

AGENDA CUYAHOGA COUNTY COUNCIL REGULAR MEETING TUESDAY, JANUARY 22, 2013 CUYAHOGA COUNTY JUSTICE CENTER COUNCIL CHAMBERS – 1ST FLOOR 5:00 PM

- 1. CALL TO ORDER
- 2. ROLL CALL
- 3. PLEDGE OF ALLEGIANCE
- 4. SILENT MEDITATION
- 5. PUBLIC COMMENT RELATED TO AGENDA
- 6. APPROVAL OF MINUTES
 - a) January 8, 2013 Regular Meeting (See Page 11)
 - b) January 15, 2013 Committee of the Whole (See Page 24)
- 7. ANNOUNCEMENTS FROM THE COUNCIL PRESIDENT
- 8. MESSAGES FROM THE COUNTY EXECUTIVE
- 9. COMMITTEE REPORT AND CONSIDERATION OF A MOTION OF COUNCIL FOR SECOND READING ADOPTION UNDER SUSPENSION OF RULES
 - a) <u>M2012-0030:</u> A Motion amending the Rules of the Cuyahoga County Council to clarify the process and powers related to the sponsorship of legislation and the Standing Committees of Council, and declaring the necessity that this Motion become immediately effective. (See Page 26)

Sponsors: Councilmembers Greenspan, Connally and Brady

Committee Assignment and Chair: Rules, Charter Review, Ethics & Council Operations – Greenspan

10. CONSIDERATION OF A MOTION OF COUNCIL FOR THIRD READING ADOPTION

a) M2012-0031: A Motion approving the expiration and retention of pending legislation in accordance with County Council Rule 11F, and declaring the necessity that this Motion become immediately effective. (See Page 29)

Sponsor: Council President Connally/Clerk of Council

11. CONSIDERATION OF A RESOLUTION OF COUNCIL FOR FIRST READING ADOPTION UNDER SUSPENSION OF RULES

a) <u>R2013-0011:</u> A Resolution proclaiming the month of January as School Boards Recognition Month in Cuyahoga County. (See Page 32)

Sponsor: Councilmember Rogers

12. CONSIDERATION OF ORDINANCES OF COUNCIL FOR FIRST READING AND REFERRAL TO COMMITTEE

a) <u>O2013-0001:</u> An Ordinance establishing general definitions for the Cuyahoga County Code and authorizing the Clerk of Council to officially publish the Cuyahoga County Code as approved by the County's Director of Law; and declaring the necessity that this Ordinance become immediately effective. (See Page 34)

Sponsors: Council President Connally/Clerk of Council and Director of Law

b) <u>O2013-0002:</u> An Ordinance amending the Public Records Policy for Cuyahoga County, and declaring the necessity that this Ordinance become immediately effective. (See Page 37)

Sponsor: Councilmember Miller

13. CONSIDERATION OF A RESOLUTION FOR FIRST READING ADOPTION UNDER SUSPENSION OF RULES

a) R2013-0012: A Resolution amending the 2012/2013 Biennial Operating Budget for 2013 by providing for additional fiscal appropriations from the General Fund and other funding sources, for appropriation transfers between budget accounts, and for cash transfers between budgetary

funds, in order to meet the budgetary needs of various County departments, offices and agencies. (See Page 40)

Sponsor: County Executive FitzGerald/Fiscal Officer/Office of Budget & Management

14. CONSIDERATION OF RESOLUTIONS FOR FIRST READING ADOPTION UNDER SUSPENSION OF RULES / EXECUTIVE SESSION

a) R2013-0003: A Resolution authorizing Settlement and Release Agreements with various plaintiffs in the total amount of \$143,257.00 in connection with *Teel et al. v. Cuyahoga County*, U.S.D.C. Case No. 11cv2582; authorizing the County Executive to execute said agreements; authorizing the appropriation of funds for payment of settlement amounts set forth herein; and declaring the necessity that this Resolution become immediately effective. (See Page 51)

Sponsor: County Executive FitzGerald

b) R2013-0013: A Resolution approving a Memorandum of Understanding between Cuyahoga County Common Pleas Court, Juvenile Court Division and Service Employees International Union (SEIU), District 1199, The Health Care and Social Service Union, in connection with a collective bargaining agreement representing approximately 111 probation, clerks and transportation staff for the period 11/1/2009 - 10/31/2012; directing that funds necessary to implement the Memorandum of Understanding be budgeted and appropriated; authorizing the County Executive to execute the Memorandum of Understanding and all other documents consistent with this Resolution; and declaring the necessity that this Resolution become immediately effective. (See Page 54)

Sponsors: Council President Connally on behalf of Juvenile Court

c) Imminent and pending litigation.

15. CONSIDERATION OF RESOLUTIONS FOR FIRST READING AND REFERRAL TO COMMITTEE

a) R2013-0014: A Resolution making an award on RQ25617 to Karvo Paving, Co. in the amount not-to-exceed \$1,547,652.00 for resurfacing Taylor Road from Euclid Avenue to East Cleveland South Corporation Line in the City of East Cleveland; authorizing the County Executive to execute the contract and all other documents consistent with said award and this Resolution; and authorizing the County Engineer, on behalf of the County Executive, to make an application for allocation from County Motor

Vehicle \$7.50 License Tax Funds in the amount of \$1,238,121.60 to fund said contract. (See Page 57)

Sponsor: County Executive FitzGerald/Department of Public Works/ Division of County Engineer

b) R2013-0015: A Resolution making an award on RQ24707 to The Guardian Life Insurance Company of America in the amount not-to-exceed \$12,300,000.00 for dental group healthcare benefits for County employees and their dependents for the period 1/1/2013 - 12/31/2015; authorizing the County Executive to execute the contract and all other documents consistent with said award and this Resolution; and declaring the necessity that this Resolution become immediately effective. (See Page 77)

Sponsor: County Executive FitzGerald/Department of Human Resources

c) R2013-0016: A Resolution amending the 2012/2013 Biennial Operating Budget for 2012 by providing for additional fiscal appropriations from the General Fund and other funding sources and for appropriation transfers between budget accounts, in order to meet the budgetary needs of various County departments, offices and agencies related to year-end close-out activities, in accordance with Resolution No. R2012-0228. (See Page 84)

Sponsor: County Executive FitzGerald/Fiscal Officer/Office of Budget & Management

- d) <u>R2013-0017:</u> A Resolution approving the appropriation of funds for Year 2013 based on the Statement of Appropriation Status dated 12/31/2012: (See Page 183)
 - 1) free balances for grants and capital projects;
 - encumbrances for grants and capital projects; and
 - 3) encumbrances for all other funds.

Sponsor: County Executive FitzGerald/Fiscal Officer/Office of Budget & Management

e) R2013-0018: A Resolution making an award on RQ22383 to MHS, Inc. in the amount not-to-exceed \$1,148,293.00 for emergency shelter services for homeless women and families, located at 2227 Payne Avenue, Cleveland, for the period 1/1/2013 - 12/31/2013; authorizing the County Executive to execute the contract and all other documents consistent with said award and this Resolution; and declaring the necessity that this Resolution become immediately effective. (See Page 187)

Sponsor: County Executive FitzGerald/Department of Health and Human Services/Division of Community Initiatives/Office of Homeless Services

f) R2013-0019: A Resolution making an award on RQ22383 to Lutheran Metropolitan Ministry in the amount of \$1,654,920.00 for emergency shelter services for homeless men, located at 2100 Lakeside Avenue, Cleveland, for the period 1/1/2013 - 12/31/2013; authorizing the County Executive to execute the contract and all other documents consistent with said award and this Resolution; and declaring the necessity that this Resolution become immediately effective. (See Page 192)

Sponsor: County Executive FitzGerald/Department of Health and Human Services/Division of Community Initiatives/Office of Homeless Services

- g) R2013-0020: A Resolution making awards on RQ25318 to various providers in the total amount not-to-exceed \$2,400,000.00 for child and family-focused services for the period 2/1/2013 1/31/2015, and authorizing the County Executive to execute the contracts and all other documents consistent with said awards and this Resolution: (See Page 196)
 - 1) Applewood Centers, Inc. in the amount not-to-exceed \$240,000.00.
 - 2) The Bair Foundation in the amount not-to-exceed \$150,000.00.
 - 3) Beech Brook in the amount not-to-exceed \$278,570.00.
 - 4) Bellefaire Jewish Children's Bureau in the amount not-to-exceed \$278,570.00.
 - 5) Catholic Charities Corporation fka Catholic Charities Services Corporation in the amount not-to-exceed \$278,570.00.
 - 6) Centers for Families and Children in the amount not-to-exceed \$278,570.00.
 - 7) The Cleveland Christian Home, Incorporated in the amount not-to-exceed \$278,570.00.
 - 8) Guidestone in the amount not-to-exceed \$278,570.00.
 - 9) Mental Health Services for Homeless Persons, Inc. in the amount not-to-exceed \$188,580.00.
 - 10) Specialized Alternatives for Families and Youth of Ohio, Inc. in the amount not-to-exceed \$150,000.00.

Sponsor: County Executive FitzGerald/Department of Health and Human Services/Division of Children and Family Services

16. COMMITTEE REPORTS AND CONSIDERATION OF RESOLUTIONS FOR SECOND READING ADOPTION UNDER SUSPENSION OF RULES

a) R2012-0240: A Resolution authorizing the County Executive and Department of Public Works to release an amount not-to-exceed \$500,000.00 from the Medical Mart Contingency Fund to the Cleveland Sports Commission when the 2013 Senior Games are held at the Convention Center, and authorizing the County Executive and Department of Public Works to negotiate and execute any necessary contract or other documents for same. (See Page 212)

Sponsor: County Executive FitzGerald/Department of Public Works

Committee Assignment and Chair: Economic Development & Planning – Schron

b) R2013-0004: A Resolution declaring that public convenience and welfare requires rehabilitation of East 49th Street Bridge No. 00.54 over abandoned railroad tracks in the City of Cleveland; total estimated construction cost \$1,940,000.00; finding that special assessments will neither be levied nor collected to pay for any part of the County's costs of said improvement; and authorizing the County Executive to enter into an agreement of cooperation with said municipality in connection with said project. (See Page 215)

Sponsor: County Executive FitzGerald/Department of Public Works/ Division of County Engineer

Committee Assignment and Chair: Public Works, Procurement & Contracting – Jones

c) R2013-0005: A Resolution approving the issuance by the Illinois Finance Authority of bonds in an aggregate principal amount not-to-exceed \$25,000,000.00, in one or more series, for the purpose of (1) refunding revenue bonds issued by Cuyahoga County, Ohio and (2) financing and refinancing facilities located in Cuyahoga County, Ohio; and declaring the necessity that this Resolution become immediately effective. (See Page 217)

Sponsors: County Executive FitzGerald/Department of Development and Councilmembers Germana and Miller

Bond Counsel: Jones Day

Committee Assignment and Chair: Finance & Budgeting – Miller

d) R2013-0006: A Resolution making an award on RQ24978 to Educational Service Center of Cuyahoga County in the amount of \$692,930.00 for fiscal agent services for the Families and Schools Together Program in connection with the Ohio Children's Trust Fund FY2013 grant program for the period 7/1/2012 - 9/30/2014; authorizing the County Executive to execute the agreement and all other documents consistent with said award and this Resolution. (See Page 221)

Sponsor: County Executive FitzGerald/Department of Health and Human Services/Division of Community Initiatives/Family and Children First Council

Committee Assignment and Chair: Health, Human Services & Aging – Brady

- e) R2013-0007: A Resolution making awards on RQ21514 to various providers in the total amount not-to-exceed \$2,700,000.00 for residential treatment services for the Youth and Family Community Partnership Program for the period 2/1/2013 1/31/2015, and authorizing the County Executive to execute a master contract and all other documents required in connection with said awards and consistent with this Resolution: (See Page 223)
 - 1) Applewood Centers, Inc.
 - 2) Beech Brook
 - 3) Bellefaire Jewish Children's Bureau
 - 4) Carrington Youth Academy LLC
 - 5) Catholic Charities Services Corporation
 - 6) The Cleveland Christian Home Incorporated
 - 7) New Directions, Inc.
 - 8) The Village Network

Sponsors: County Executive FitzGerald on behalf of Juvenile Court

Committee Assignment and Chair: Justice Affairs – Simon

- f) R2013-0008: A Resolution making awards on RQ21516 to various providers in the total amount not-to-exceed \$2,250,000.00 for residential treatment services for the period 2/1/2013 1/31/2015, and authorizing the County Executive to execute a master contract and all other documents required in connection with said awards and consistent with this Resolution: (See Page 226)
 - 1) Applewood Centers, Inc.
 - 2) Beech Brook

- 3) Bellefaire Jewish Children's Bureau
- 4) Carrington Youth Academy LLC
- 5) Catholic Charities Services Corporation
- 6) The Cleveland Christian Home Incorporated
- 7) Cornell Abraxas Group, Inc.
- 8) Guidestone, fka Berea Children's Home
- 9) Keystone Richland Center, LLC dba Foundations for Living
- 10) New Directions, Inc.
- 11) Safe House Ministries, Inc.
- 12) The Glen Mills Schools
- 13) The House of Emmanuel, Inc.
- 14) The Village Network

Sponsors: County Executive FitzGerald on behalf of Juvenile Court

Committee Assignment and Chair: Justice Affairs – Simon

g) R2013-0009: A Resolution authorizing a contract with Starting Point in the amount not-to-exceed \$1,832,782.00 for administration of the Special Needs Child Care Program for the Invest in Children Program for the period 1/1/2013 - 12/31/2013, and authorizing the County Executive to execute the contract and all other documents required consistent with this Resolution. (See Page 229)

Sponsor: County Executive FitzGerald/Department of Health and Human Services/Division of Community Initiatives/Office of Early Childhood and Councilmember Conwell

Committee Assignment and Chair: Health, Human Services & Aging – Brady

17. CONSIDERATION OF A RESOLUTION FOR THIRD READING ADOPTION

a) R2012-0260: A Resolution making an award on RQ23209 to R.W. Armstrong & Associates, Inc. in the amount of \$4,200,000.00 for general engineering services for various projects in connection with the Master Plan Update and Airport Layout Plan for the Cuyahoga County Airport, and authorizing the County Executive to execute the contract and all other documents required in connection with said award and consistent with this Resolution. (See Page 231)

Sponsor: County Executive FitzGerald/Department of Public Works

18. COMMITTEE REPORT AND CONSIDERATION OF A RESOLUTION FOR THIRD READING ADOPTION

a) R2012-0264: A Resolution making awards on RQ24534 to Geis Property Management, LLC for sale of the Ameritrust Complex and entry of a lease for a consolidated County headquarters at the site of the P & H Buildings; authorizing the County Executive to execute all documents in connection with said awards and consistent with this Resolution; authorizing the Director of Public Works to administer said project; and declaring the necessity that this Resolution become immediately effective. (See Page 233)

Sponsors: County Executive FitzGerald/Department of Public Works and Councilmember Germana

Committee Assignment and Chair: Committee of the Whole - Connally

19. CONSIDERATION OF ORDINANCES FOR FIRST READING AND REFERRAL TO COMMITTEE

a) <u>O2013-0003:</u> An Ordinance providing for modifications to and adoption of the Cuyahoga County Human Resources Personnel Policies and Procedures Manual to be applicable to all County employees, and declaring the necessity that this Ordinance become immediately effective. (See Page 238)

Sponsor: County Executive FitzGerald/Department of Human Resources

b) O2013-0004: An Ordinance amending Ordinance No. O2011-0045, as amended, which established divisions and sections within the Department of Health and Human Services, to create the Office of the Director and to delineate displacement, reinstatement and reemployment procedures; and declaring the necessity that this Ordinance become immediately effective. (See Page 392)

Sponsor: County Executive FitzGerald/Department of Health and Human Services

20. CONSIDERATION OF ORDINANCES FOR THIRD READING ADOPTION

a) <u>O2012-0030:</u> An Ordinance establishing the organizational structure of the Department of Development and creating divisions within the department, and declaring the necessity that this Ordinance become immediately effective. (See Page 395)

Sponsor: County Executive FitzGerald/Department of Development

b) O2012-0035: An Ordinance amending Ordinance No. O2011-0027 dated 7/12/2011, which established divisions and sections within the Department of Public Works, to add the operations of the Cuyahoga County Airport. (See Page 398)

Sponsors: County Executive FitzGerald and Councilmembers Germana and Conwell

- 21. MISCELLANEOUS COMMITTEE REPORTS
- **22. MISCELLANEOUS BUSINESS**
- 23. PUBLIC COMMENT UNRELATED TO AGENDA
- 24. ADJOURNMENT

NEXT MEETING

REGULAR MEETING:

TUESDAY, FEBRUARY 12, 2013 5:00 PM / COUNCIL CHAMBERS

^{*}In accordance with Ordinance No. O2011-0020, as amended, complimentary parking in the Huntington Park Garage will be available for the public on any day when the Council or any of its committees holds meetings. Please see the Clerk to obtain a parking pass.



MINUTES

CUYAHOGA COUNTY COUNCIL REGULAR MEETING
TUESDAY, JANUARY 8, 2013
CUYAHOGA COUNTY JUSTICE CENTER
COUNCIL CHAMBERS – 1ST FLOOR
5:00 PM

1. CALL TO ORDER

The meeting was called to order by Council President Connally at 5:02 p.m.

2. ROLL CALL

Council President Connally asked Clerk Schmotzer to call the roll. Councilmembers Germana, Gallagher, Schron, Conwell, Jones, Rogers, Simon, Greenspan, Miller, Brady and Connally were in attendance and a quorum was determined.

3. PLEDGE OF ALLEGIANCE

The Pledge of Allegiance was recited.

4. SILENT MEDITATION

Council President Connally requested a moment of silent meditation for personal reflection on wishes for the new year.

5. PUBLIC COMMENT RELATED TO AGENDA

No public comments were given related to the agenda.

- 6. APPROVAL OF MINUTES
 - a) December 11, 2012 Work Session
 - b) December 11, 2012 Regular Meeting
 - c) January 2, 2013 Organizational Meeting
 - d) January 2, 2013 Committee of the Whole Meeting

A motion was made by Mr. Rogers, seconded by Mr. Miller and approved by unanimous vote to approve the minutes of the December 11, 2012 work session, December 11, 2012 regular meeting, January 2, 2013 organizational meeting and January 2, 2013 Committee of the Whole meeting.

7. ANNOUNCEMENTS FROM THE COUNCIL PRESIDENT

Council President Connally thanked three retiring members of the County Executive's administration: Richard Sensenbrenner, Treasurer; Bob Reid, Sheriff; and Gerald Fuerst, Clerk of Courts, for their service to the County.

8. MESSAGES FROM THE COUNTY EXECUTIVE

County Executive FitzGerald reported the following:

- The State of the County Address will be held on Tuesday, February 19th; and
- 2) State budget hearings will begin soon; Council will be apprised of any developments.
- 9. CONSIDERATION OF A MOTION OF COUNCIL FOR FIRST READING ADOPTION UNDER SUSPENSION OF RULES

A motion was made by Mr. Gallagher, seconded by Mr. Germana and approved by unanimous vote to suspend Rules 9D and 12A and to place on final passage Motion No. M2013-0002.

a) <u>M2013-0002:</u> A Motion appointing Anne Camille Klonowski to serve on a Board of Revision for Cuyahoga County, and declaring the necessity that this Motion become immediately effective.

Sponsors: Councilmembers Conwell and Gallagher

On a motion by Ms. Conwell with a second by Mr. Rogers, Motion No. M2013-0002 was considered and approved by unanimous vote.

- 10. COMMITTEE REPORT AND CONSIDERATION OF A MOTION OF COUNCIL FOR SECOND READING
 - a) <u>M2012-0031:</u> A Motion approving the expiration and retention of pending legislation in accordance with County Council Rule 11F, and declaring the necessity that this Motion become immediately effective.

Sponsor: Council President Connally/Clerk of Council

Committee Assignment and Chair: Rules, Charter Review, Ethics & Council Operations – Greenspan

Clerk Schmotzer read Motion No. M2012-0031 into the record.

This item will move to the January 22, 2013 Council meeting agenda for consideration for third reading adoption.

11. CONSIDERATION OF A RESOLUTION OF COUNCIL FOR FIRST READING ADOPTION UNDER SUSPENSION OF RULES

A motion was made by Mr. Gallagher, seconded by Mr. Germana and approved by unanimous vote to suspend Rules 9D and 12A and to place on final passage Resolution No. R2013-0010.

a) <u>R2013-0010:</u> A Resolution adjusting compensation of Cuyahoga County Council staff, and declaring the necessity that this Resolution become immediately effective.

Sponsors: Council President Connally and Councilmember Gallagher

On a motion by Mr. Brady with a second by Mr. Rogers, Resolution No. R2013-0010 was considered and adopted by unanimous vote.

12. CONSIDERATION OF A RESOLUTION FOR FIRST READING ADOPTION UNDER SUSPENSION OF RULES

A motion was made by Mr. Gallagher, seconded by Mr. Germana and approved by unanimous vote to suspend Rules 9D and 12A and to place on final passage Resolution No. R2013-0001.

a) R2013-0001: A Resolution amending the 2012/2013 Biennial Operating Budget for 2013 by providing for additional fiscal appropriations from the General Fund and other funding sources, for appropriation transfers between budget accounts, and for cash transfers between budgetary funds, in order to meet the budgetary needs of various County departments, offices and agencies; and declaring the necessity that this Resolution become immediately effective.

Sponsor: County Executive FitzGerald/Fiscal Officer/Office of Budget & Management

Mr. Miller introduced a proposed substitute to Resolution No. R2013-0001 on the floor. Discussion ensued. A motion was then made by Mr. Miller, seconded by Mr. Brady and approved by unanimous vote to accept the proposed substitute to Resolution No. R2013-0001.

On a motion by Mr. Miller with a second by Mr. Germana, Resolution No. R2013-0001 was considered and adopted by unanimous vote, as substituted.

13. CONSIDERATION OF RESOLUTIONS FOR FIRST READING ADOPTION UNDER SUSPENSION OF RULES / EXECUTIVE SESSION

[Note: Item No. 13 was taken out of order and considered after Item No. 22 on the agenda.]

A motion was made by Mr. Gallagher, seconded by Mr. Germana and approved by unanimous roll-call vote to move to Executive Session for the purpose of discussing matters concerning collective bargaining, settlement and release agreements and imminent and pending litigation and for no other purpose whatsoever. Executive Session was then called to order by Council President Connally at 6:03 p.m. The following members were present: Councilmembers Germana, Gallagher, Schron, Conwell, Jones, Rogers, Simon, Greenspan, Miller, Brady and Connally. The following additional attendees were also present: County Executive Ed FitzGerald, Law Director Majeed Makhlouf, Deputy Law Director Nora Hurley, Judge Nancy Fuerst, Court Administrator Greg Popovich, Deputy Court Administrator Jim Ginley, Civil Division Chief David Lambert, Chief of Staff Joe Nanni, Legislative Budget Analyst Trevor McAleer and Clerk of Council Jeanne Schmotzer. At 7:21 p.m., Executive Session was adjourned, without objection, and Council President Connally then reconvened the regular meeting.

a) R2013-0002: A Resolution approving the extension of a Collective Bargaining Agreement between Cuyahoga County Court of Common Pleas and Communication Workers of America, representing approximately 166 Court Probation Officers, lab technicians and support staff for the period 1/31/2013 - 8/2/2013; directing that funds necessary to implement the contract extension terms be budgeted and appropriated; authorizing the County Executive to execute the agreement and all other documents consistent with this Resolution; and declaring the necessity that this Resolution become immediately effective.

Sponsors: County Executive FitzGerald/Department of Law on behalf of Cuyahoga County Court of Common Pleas

A motion was then made by Mr. Gallagher, seconded by Mr. Germana and approved by unanimous vote to suspend Rules 9D and 12A and to place on final passage Resolution No. R2013-0002.

On a motion by Mr. Brady with a second by Mr. Gallagher, Resolution No. R2013-0002 was considered and adopted by unanimous vote.

b) R2013-0003: A Resolution authorizing Settlement and Release Agreements with various plaintiffs in the total amount of \$143,757.00 in connection with *Teel et al. v. Cuyahoga County*, U.S.D.C. Case No. 11cv2582; authorizing the County Executive to execute said agreements; authorizing the appropriation of funds for payment of settlement amounts set forth herein; and declaring the necessity that this Resolution become immediately effective.

Sponsor: County Executive FitzGerald

Resolution No. R2013-0003 was not read into the record nor was any legislative action taken by Council.

c) Imminent and pending litigation.

14. CONSIDERATION OF RESOLUTIONS FOR FIRST READING AND REFERRAL TO COMMITTEE

a) R2013-0004: A Resolution declaring that public convenience and welfare requires rehabilitation of East 49th Street Bridge No. 00.54 over abandoned railroad tracks in the City of Cleveland; total estimated construction cost \$1,940,000.00; finding that special assessments will neither be levied nor collected to pay for any part of the County's costs of said improvement; and authorizing the County Executive to enter into an agreement of cooperation with said municipality in connection with said project.

Sponsor: County Executive FitzGerald/Department of Public Works/ Division of County Engineer

Council President Connally referred Resolution No. R2013-0004 to the Public Works, Procurement & Contracting Committee.

b) R2013-0005: A Resolution approving the issuance by the Illinois Finance Authority of bonds in an aggregate principal amount not-to-exceed \$25,000,000.00, in one or more series, for the purpose of (1) refunding revenue bonds issued by Cuyahoga County, Ohio and (2) financing and refinancing facilities located in Cuyahoga County, Ohio; and declaring the necessity that this Resolution become immediately effective.

Sponsor: County Executive FitzGerald/Department of Development

Bond Counsel: Jones Day

Council President Connally referred Resolution No. R2013-0005 to the Finance & Budgeting Committee.

c) R2013-0006: A Resolution making an award on RQ24978 to Educational Service Center of Cuyahoga County in the amount of \$692,930.00 for fiscal agent services for the Families and Schools Together Program in connection with the Ohio Children's Trust Fund FY2013 grant program for the period 7/1/2012 - 9/30/2014; authorizing the County Executive to execute the agreement and all other documents consistent with said award and this Resolution.

Sponsor: County Executive FitzGerald/Department of Health and Human Services/Division of Community Initiatives/Family and Children First Council

Council President Connally referred Resolution No. R2013-0006 to the Health, Human Services & Aging Committee.

- d) R2013-0007: A Resolution making awards on RQ21514 to various providers in the total amount not-to-exceed \$2,700,000.00 for residential treatment services for the Youth and Family Community Partnership Program for the period 2/1/2013 1/31/2015, and authorizing the County Executive to execute a master contract and all other documents required in connection with said awards and consistent with this Resolution:
 - 1) Applewood Centers, Inc.
 - 2) Beech Brook
 - 3) Bellefaire Jewish Children's Bureau
 - 4) Carrington Youth Academy LLC
 - 5) Catholic Charities Services Corporation
 - 6) The Cleveland Christian Home Incorporated
 - 7) New Directions, Inc.
 - 8) The Village Network

Sponsors: County Executive FitzGerald on behalf of Juvenile Court

Council President Connally referred Resolution No. R2013-0007 to the Justice Affairs Committee.

- e) R2013-0008: A Resolution making awards on RQ21516 to various providers in the total amount not-to-exceed \$2,250,000.00 for residential treatment services for the period 2/1/2013 1/31/2015, and authorizing the County Executive to execute a master contract and all other documents required in connection with said awards and consistent with this Resolution:
 - 1) Applewood Centers, Inc.

- 2) Beech Brook
- 3) Bellefaire Jewish Children's Bureau
- 4) Carrington Youth Academy LLC
- 5) Catholic Charities Services Corporation
- 6) The Cleveland Christian Home Incorporated
- 7) Cornell Abraxas Group, Inc.
- 8) Guidestone, fka Berea Children's Home
- 9) Keystone Richland Center, LLC dba Foundations for Living
- 10) New Directions, Inc.
- 11) Safe House Ministries, Inc.
- 12) The Glen Mills Schools
- 13) The House of Emmanuel, Inc.
- 14) The Village Network

Sponsors: County Executive FitzGerald on behalf of Juvenile Court

Council President Connally referred Resolution No. R2013-0008 to the Justice Affairs Committee.

f) R2013-0009: A Resolution authorizing a contract with Starting Point in the amount not-to-exceed \$1,832,782.00 for administration of the Special Needs Child Care Program for the Invest in Children Program for the period 1/1/2013 - 12/31/2013, and authorizing the County Executive to execute the contract and all other documents required consistent with this Resolution.

Sponsor: County Executive FitzGerald/Department of Health and Human Services/Division of Community Initiatives/Office of Early Childhood

Council President Connally referred Resolution No. R2013-0009 to the Health, Human Services & Aging Committee.

- 15. COMMITTEE REPORT AND CONSIDERATION OF A RESOLUTION FOR SECOND READING
 - a) R2012-0260: A Resolution making an award on RQ23209 to R.W. Armstrong & Associates, Inc. in the amount of \$4,200,000.00 for general engineering services for various projects in connection with the Master Plan Update and Airport Layout Plan for the Cuyahoga County Airport, and authorizing the County Executive to execute the contract and all other documents required in connection with said award and consistent with this Resolution.

Sponsor: County Executive FitzGerald/Department of Public Works

Committee Assignment and Chair: Public Works, Procurement & Contracting – Jones

Clerk Schmotzer read Resolution No. R2012-0260 into the record.

This item will move to the January 22, 2013 Council meeting agenda for consideration for third reading adoption.

16. COMMITTEE REPORTS AND CONSIDERATION OF RESOLUTIONS FOR SECOND READING ADOPTION UNDER SUSPENSION OF RULES

A motion was made by Mr. Gallagher, seconded by Mr. Germana and approved by unanimous vote to suspend Rule 9D and to place on final passage Resolution Nos. R2012-0248, R2012-0249, R2012-0259, R2012-0261, R2012-0262 and R2012-0263.

a) R2012-0248: A Resolution authorizing a Next Stage Fund Pilot Loan in the amount not-to-exceed \$700,000.00 to Sparkbase, Inc. for building and deploying loyalty solutions directly to large scale merchants; authorizing the Deputy Chief of Staff of Development or Director of Development to execute all documents required in connection with said loan and consistent with this Resolution; and declaring the necessity that this Resolution become immediately effective.

Sponsor: County Executive FitzGerald/Department of Development

Committee Assignment and Chair: Economic Development & Planning – Schron

On a motion by Mr. Schron with a second by Mr. Miller, Resolution No. R2012-0248 was considered and adopted by unanimous vote.

b) R2012-0249: A Resolution making an award on RQ23900 to Economic and Community Development Institute, Inc. in the amount of \$550,000.00 for management of the Cuyahoga County Microenterprise Revolving Loan Fund for the period 2/1/2013 - 1/31/2014; authorizing the County Executive to execute the contract and all other documents required in connection with said award and consistent with this Resolution; and declaring the necessity that this Resolution become immediately effective.

Sponsor: County Executive FitzGerald/Department of Development

Committee Assignment and Chair: Economic Development & Planning – Schron

On a motion by Mr. Schron with a second by Ms. Conwell, Resolution No. R2012-0249 was considered and adopted by unanimous vote.

c) R2012-0259: A Resolution making an award on RQ24986 to Perk Company, Inc. in the amount not-to-exceed \$3,814,000.00 for resurfacing of Rockside Road from Turney Road to Broadway Avenue in the Cities of Bedford, Garfield Heights and Maple Heights; authorizing the County Executive to execute the contract and all other documents required in connection with said award and consistent with this Resolution; and authorizing the County Engineer, on behalf of the County Executive, to make an application for allocation from County Motor Vehicle \$5.00 License Tax Funds in the amount of \$762,800.00 to fund said contract.

Sponsors: County Executive FitzGerald/Department of Public Works/ Division of County Engineer and Councilmember Jones

Committee Assignment and Chair: Public Works, Procurement & Contracting – Jones

On a motion by Mr. Jones with a second by Mr. Germana, Resolution No. R2012-0259 was considered and adopted by unanimous vote.

d) R2012-0261: A Resolution reauthorizing the HELP Loan Program in the amount not-to-exceed \$40,000,000.00 for the purpose of providing linked deposits to financial institutions making below-market home repair loans to Cuyahoga County residents; authorizing the County Executive to execute the agreements and all other documents required consistent with this Resolution; and declaring the necessity that this Resolution become immediately effective.

Sponsors: County Executive FitzGerald/Department of Development and Councilmembers Germana, Miller, Rogers, Connally, Schron and Gallagher

Committee Assignment and Chair: Economic Development & Planning – Schron

On a motion by Mr. Rogers with a second by Mr. Germana, Resolution No. R2012-0261 was considered and adopted by unanimous vote.

e) R2012-0262: A Resolution authorizing amendments to contracts with various providers for the Staff Secure Shelter Program and Placement Planning Day Report services for the period 3/1/2012 - 2/28/2014 for additional funds in the total amount not-to-exceed \$3,273,852.00, and authorizing the County Executive to execute the amendments and all other documents required consistent with this Resolution:

- 1) No. CE1200344-01 with Carrington Youth Academy LLC in the amount not-to-exceed \$2,121,340.00.
- 2) No. CE1200345-01 with The Cleveland Christian Home Incorporated in the amount not-to-exceed \$1,152,512.00.

Sponsors: County Executive FitzGerald on behalf of Juvenile Court

Committee Assignment and Chair: Justice Affairs – Simon

On a motion by Mr. Schron with a second by Ms. Simon, Resolution No. R2012-0262 was considered and adopted by unanimous vote.

f) R2012-0263: A Resolution making an award on RQ25744 to Starting Point in the amount not-to-exceed \$2,038,762.00 for administration of the Family Child Care Home Regional System for the Invest in Children Program for the period 1/1/2013 - 12/31/2013, and authorizing the County Executive to execute the contract and all other documents required in connection with said award and consistent with this Resolution.

Sponsors: County Executive FitzGerald/Department of Health and Human Services/Division of Community Initiatives/Office of Early Childhood and Councilmember Jones

Committee Assignment and Chair: Health, Human Services & Aging – Brady

On a motion by Mr. Brady with a second by Mr. Miller, Resolution No. R2012-0263 was considered and adopted by unanimous vote.

- 17. COMMITTEE REPORT AND CONSIDERATION OF A RESOLUTION FOR SECOND READING AND REFERRAL TO COMMITTEE
 - a) R2012-0264: A Resolution making awards on RQ24534 to Geis Property Management, LLC for sale of the Ameritrust Complex and entry of a lease for a consolidated County headquarters at the site of the P & H Buildings; authorizing the County Executive to execute all documents in connection with said awards and consistent with this Resolution; authorizing the Director of Public Works to administer said project; and declaring the necessity that this Resolution become immediately effective.

Sponsors: County Executive FitzGerald/Department of Public Works and Councilmember Germana

A brief presentation was given by representatives of CBRE and URS. Discussion ensued. Council President Connally then referred Resolution No. R2012-0264 to the Committee of the Whole.

- 18. COMMITTEE REPORTS AND CONSIDERATION OF ORDINANCES FOR SECOND READING
 - a) <u>O2012-0030:</u> An Ordinance establishing the organizational structure of the Department of Development and creating divisions within the department, and declaring the necessity that this Ordinance become immediately effective.

Sponsor: County Executive FitzGerald/Department of Development

Committee Assignment and Chair: Economic Development & Planning – Schron

Clerk Schmotzer read Ordinance No. O2012-0030 into the record.

This item will move to the January 22, 2013 Council meeting agenda for consideration for third reading adoption.

b) <u>O2012-0035:</u> An Ordinance amending Ordinance No. O2011-0027 dated 7/12/2011, which established divisions and sections within the Department of Public Works, to add the operations of the Cuyahoga County Airport.

Sponsors: County Executive FitzGerald and Councilmembers Germana and Conwell

Committee Assignment and Chair: Public Works, Procurement & Contracting – Jones

Clerk Schmotzer read Ordinance No. O2012-0035 into the record.

This item will move to the January 22, 2013 Council meeting agenda for consideration for third reading adoption.

- 19. CONSIDERATION OF AN ORDINANCE FOR THIRD READING ADOPTION
 - a) <u>O2012-0034:</u> An Ordinance amending the Administrative Rules for the Cuyahoga County Human Resource Commission, and declaring the necessity that this Ordinance become immediately effective.

Sponsors: County Executive FitzGerald on behalf of Human Resource Commission

Ms. Conwell introduced an amendment to Ordinance No. O2012-0034 on the floor. Discussion ensued. A motion was then made by Ms. Conwell, seconded by Mr. Schron and approved by unanimous vote to amend Section 4.06 of Ordinance No. O2012-0034 as follows:

"If a proposed Rule Change involves a proposed ordinance that requires the Commission's recommendation, the Commission may modify the time requirements for posting and acting upon such proposed ordinance, but in no case shall the proposed rule change be presented and posted less than three (3) days prior to the Commission action on such proposed ordinance."

On a motion by Ms. Conwell with a second by Mr. Schron, Ordinance No. O2012-0034 was considered and adopted by unanimous vote, as amended.

20. MISCELLANEOUS COMMITTEE REPORTS

Mr. Miller reported that the Finance & Budgeting Committee will meet on Monday, January 14, 2013 at 1:00 p.m.

Mr. Greenspan reported that the Rules, Charter Review, Ethics & Council Operations Committee will meet on Tuesday, January 15, 2013 at 4:00 p.m.

Ms. Connally reported that the Committee of the Whole will meet on Tuesday, January 15, 2013 at 5:00 p.m.

Mr. Jones reported that the Public Works, Procurement & Contracting Committee will meet on Thursday, January 10, 2013 at 11:00 a.m.

21. MISCELLANEOUS BUSINESS

Mr. Germana thanked Clerk Schmotzer for the reformatted agenda with page numbering and references.

22. PUBLIC COMMENT UNRELATED TO AGENDA

No public comments were given unrelated to the agenda.

[Note: Item No. 13 was taken out of order and considered after Item No. 22 on the agenda.]

23. ADJOURNMENT

With no further business to discuss, Council President Connally adjourned the meeting at 7:24 p.m., without objection.



MINUTES

CUYAHOGA COUNTY COMMITTEE OF THE WHOLE MEETING
TUESDAY, JANUARY 15, 2013
CUYAHOGA COUNTY JUSTICE CENTER
COUNCIL CHAMBERS – 1ST FLOOR
5:00 PM

1. CALL TO ORDER

The meeting was called to order by Council President Connally at 5:04 p.m.

2. ROLL CALL

Council President Connally asked Clerk Schmotzer to call the roll. Councilmembers Gallagher, Schron, Conwell, Jones, Rogers, Greenspan, Miller, Brady, Germana and Connally were in attendance and a quorum was determined. Councilmember Simon was absent from the meeting.

A motion was made by Mr. Miller, seconded by Ms. Conwell and approved by unanimous vote to excuse Ms. Simon from the meeting.

3. PUBLIC COMMENT RELATED TO THE AGENDA

The following citizens addressed the Committee regarding issues of concern to them relating to Resolution No. R2012-0264, a Resolution making an award to Geis Property Management, LLC for the sale of the Ameritrust Complex and entry of a lease for a consolidated County headquarters at the site of the P & H Buildings:

- a) Ronald Nicklos, representing Nicklos & Company
- b) Homer S. Taft, citizen
- c) Arne F. Goldman, Optima Ventures, LLC/Marous Brothers Construction
- d) Chaim Schochet, Optima Ventures, LLC

4. ITEM REFERRED TO COMMITTEE

Ms. Connally read a statement to the public regarding the history of the County's contracting process and oversight by various boards and the Council as it relates to

Resolution No. R2012-0264. The Clerk then read Resolution No. R2012-0264 into the record.

a) R2012-0264: A Resolution making awards on RQ24534 to Geis Property Management, LLC for sale of the Ameritrust Complex and entry of a lease for a consolidated County headquarters at the site of the P & H Buildings; authorizing the County Executive to execute all documents in connection with said awards and consistent with this Resolution; authorizing the Director of Public Works to administer said project; and declaring the necessity that this Resolution become immediately effective.

Representatives of CBRE and URS provided a brief presentation regarding the project, including proposed design and costs. Discussion ensued. Ms. Bonnie Teeuwen, Director of Public Works, addressed the Committee regarding recommendations for the project. Discussion ensued. Committee members asked questions pertaining to the item, which were answered accordingly. Committee members also asked for additional information to be provided regarding the design and costs of the project.

On a motion by Ms. Conwell with a second by Mr. Germana, Resolution No. R2012-0264 was considered and approved by unanimous vote to be referred to the full Council agenda for third reading.

5. MISCELLANEOUS BUSINESS

There was no miscellaneous business.

6. OTHER PUBLIC COMMENT

There was no public comment.

7. ADJOURNMENT

With no further business to discuss, the meeting was adjourned at 6:49 p.m., without objection.

County Council of Cuyahoga County, Ohio

Motion No. M2012-0030

Sponsored by: Councilmembers	A Motion amending the Rules of the		
Greenspan, Connally and Brady	Cuyahoga County Council to clarify the		
	process and powers related to the sponsorship		
	of legislation and the Standing Committees of		
	Council, and declaring the necessity that this		
	Motion become immediately effective.		

WHEREAS, the Cuyahoga County Charter provides in Article 2, Section 10, Subsection 2 that the Cuyahoga County Council shall adopt its own rules; and

WHEREAS, the Council's purpose and intent in adopting these rules is to develop operating procedures for the Council that are clear and understandable, promote fair and orderly conduct of the Council's business, facilitate openness and public participation, provide for easy access to public records, and encourage ethical conduct; and

WHEREAS, Council approved permanent rules on April 26, 2011 in Motion No. M2011-0007, as amended on May 8, 2012 by Motion No. M2012-0004; and

WHEREAS, Council deems it necessary to amend the permanent Rules of Council in order to provide for a more informative and accessible record of the sponsorship of legislation, to amend the process for members to withdraw legislation in Rule 11F, to provide the mechanism by which members are listed as sponsors or cosponsors of legislation, and to revise the manner by which Standing Committees are established.

NOW, THEREFORE, BE IT APPROVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. The Rules of Cuyahoga County Council, approved by Council in Motion No. M2011-0007, as amended on May 8, 2012 by Motion No. M2012-0004, are hereby amended in part to read as follows (additions are bolded and underlined; deletions are stricken out):

Rule 7G: Sponsorship

Council member(s) and/or the Executive introducing legislation for consideration by Council shall be listed as sponsors of the legislation. At any point prior to final passage of legislation in Council, any sponsor may withdraw his/her name as a sponsor by notifying the Clerk.

At any point <u>after the initial introduction of legislation for consideration by Council</u> <u>and</u> prior to <u>the</u> final passage of legislation in Council, any Member may add or withdraw his/her name as a <u>cosponsor</u> by notifying the Clerk.

Rule 11F: Expiration of Pending Legislation

- (1) Prior to the end of each year in which a General Election for Members of Council takes place, the Council shall consider a Motion of Council, providing that legislation introduced prior to July 1 of that year and not adopted shall expire. At least two weeks prior to the consideration of that Motion of Council, the Clerk shall provide each Member a list of legislation sponsored by that Member that is set to expire. The Council may adopt this Motion of Council so as to decide which items of legislation set to expire shall expire.
- (2) In addition, legislation may be withdrawn at any time by letter to the Clerk from all of the <u>initial</u> sponsors and cosponsors of the legislation, requesting that it be withdrawn. The letter may be sent by electronic mail. At each meeting of Council, the Clerk shall announce any legislation that has been withdrawn by letter, and the items withdrawn shall be listed in the minutes of the meeting.

Rule 12G: Standing Committees - The following standing committees are established for the Council:

- 1. Health, Human Services & Aging
- 2. Public Safety & Justice Affairs
- Justice Affairs
- <u>3.</u> 4. Economic Development & Planning
- 4. 5. Finance & Budgeting
- <u>5. 6.</u> Public Works, Procurement & Contracting
- 7. Intergovernmental Relations & Collaboration
- 6. 8. Human Resources, Appointments & Equity
- 7. 9. Education, Environment & Sustainability
- 8. 10. Rules, Charter Review, Ethics and Council Operations & Intergovernmental Relations

SECTION 2. Effective Date. Provided that this Motion receives the affirmative vote of at least eight (8) of the members of Council, it shall take effect immediately upon adoption by Council. It is hereby determined to be necessary that this Motion become immediately effective in order that the County Council continue to have operating rules in place without interruption, and so that the usual and daily operations of county government may continue to function.

SECTION 3. It is found and determined that all formal actions of the Council and Council committees relating to the adoption of this motion occurred in open meetings and that all deliberations of this Council and any of its committees that resulted in such formal action took place in meetings open to the public, in compliance with all legal requirements, including those of Section 121.22 of the Ohio Revised Code.

On a motion byapproved.	, seconded by, the fore	going Motion was dul
Yeas:		
Nays:		
	County Council President	Date
	Clerk of Council	Date
	d to Committee: November 13, 2012 ed: Rules, Charter Review, Ethics & Co	ouncil Operations

Journal CC009 January 22, 2013

County Council of Cuyahoga County, Ohio

Motion No. M2012-0031

Sponsored by: Council President	A Motion approving the expiration and		
Connally/Clerk of Council	retention of pending legislation in accordance		
	with County Council Rule 11F, and declaring		
	the necessity that this Motion become		
	immediately effective.		

WHEREAS, County Council Rule 11F(1) requires that, prior to the end of each year in which a General Election for Members of Council takes place, the Council shall consider a Motion of Council, providing that legislation introduced prior to July 1 of that year and not adopted shall expire, such Motion to be used to decide which items of legislation set to expire shall expire; and

WHEREAS, at least two weeks prior to the consideration of that Motion of Council, the Clerk of Council is to provide each Member a list of legislation sponsored by that Member that is set to expire, thereby allowing the Member to request retention of the legislation for further consideration; and

WHEREAS, on October 31, 2012 the Clerk of Council provided each Councilmember a list of legislation sponsored by that Councilmember that is set to expire; and

WHEREAS, Councilmembers notified the Clerk of Council requesting to retain certain legislation that is set to expire; and

WHEREAS, it is necessary that this Motion become immediately effective in order to provide for the usual, daily operation of the County Council.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the County Council hereby approves expiration of the following pending legislation:

- a) M2011-0052: A Motion amending Rule 9D of the Rules of the Cuyahoga County Council, and declaring the necessity that this Motion become immediately effective.
- b) O2011-0031: An Ordinance providing for the adoption of a campaign finance policy for Cuyahoga County.

- c) O2011-0032: An Ordinance establishing guidelines under which Council is to consider and approve subsidies to boards, commissions, agencies, corporations, and other entities or authorities over which the County Council does not have direct, immediate oversight of daily, operational management decisions.
- d) <u>O2011-0057:</u> An Ordinance establishing a General Fund Balance Reserve Replenishment Plan, and declaring the necessity that this Ordinance become immediately effective.
- e) <u>O2011-0059:</u> An Ordinance establishing a Health and Human Services Fund Balance Reserve Replenishment Plan, and declaring the necessity that this Ordinance become immediately effective.
- f) O2011-0060: An Ordinance establishing a Discretionary Funds Utilization Policy, and declaring the necessity that this Ordinance become immediately effective.

SECTION 2. That the County Council hereby approves retention of the following pending legislation at the request of the sponsor(s):

- a) R2011-0188: A Resolution providing for the submission to the Electorate of an amendment to Article 3, Section 3 of the Charter of Cuyahoga County; and declaring the necessity that this Resolution become immediately effective.
- b) R2011-0189: A Resolution providing for the submission to the Electorate of an amendment to Article 3, Section 4, Subsection 2 of the Charter of Cuyahoga County; and declaring the necessity that this Resolution become immediately effective.
- c) O2011-0048: An Ordinance requiring the County Executive to contract for and conduct a Contracting, Employment, and Workforce Training Disparity Study beginning in 2012 and, if required by law to validate the Disparity Study, every five years thereafter; and requiring annual updates of said Study, if required by law.
- d) O2012-0017: An Ordinance establishing a County Road Maintenance Program for the Department of Public Works in order to reinstate the financial responsibility of the County for design, construction, repair and road surface maintenance projects for County Roads, effective March 1, 2014.

SECTION 3. It is necessary that this Motion become immediately effective in order that the usual and daily operation of the County Council may continue. Provided that this Motion receives the affirmative vote of eight members of Council, this Motion shall become immediately effective upon approval by Council.

SECTION 4. It is found and determined that all formal actions of this Council relating to the approval of this Motion were adopted in an open meeting of the Council, and that all deliberations of this Council and any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byapproved.	, seconded by, the fo	regoing Motion was duly
Yeas:		
Nays:		
	County Council President	Date
	Clerk of Council	Date
	to Committee: November 27, 2012 l: Rules, Charter Review, Ethics & Co	ouncil Operations
Committee Report/Second	ond Reading: January 8, 2013	
Journal CC009		

January 22, 2013

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0011

Sponsored by: Councilmember	A Resolution proclaiming the month of	
Rogers	January as School Boards Recognition	
	Month in Cuyahoga County.	

WHEREAS, the service provided by the members of Cuyahoga County's city, village, local and joint vocational boards of education and educational service centers governing boards is critical to ensuring quality public education for Cuyahoga County's schoolchildren; and

WHEREAS, through partnerships with teachers, school administrators, staff and parents, school board members provide leadership and support for young people in Cuyahoga County; and

WHEREAS, service on a school board requires a devotion of time and service to carry on the mission of a school district and envision a community's education future; and

WHEREAS, the National School Boards Association and the Ohio School Boards Association have proclaimed January as School Boards Recognition Month; and

WHEREAS, the Cuyahoga County Council wishes to acknowledge the stellar service of the various school boards throughout the county; and to encourage public appreciation for such service.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. The Cuyahoga County Council proclaims the month of January as School Boards Recognition Month to promote awareness in Cuyahoga County about the dedication of school board members and the importance of school board service to the progress of our community and our children's future.

SECTION 2. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by	, the foregoing Resolution was
Yeas:		
Nays:		
	County Council President	
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date

Journal CC009 January 22, 2013

County Council of Cuyahoga County, Ohio

Ordinance No. O2013-0001

Sponsored by: Council President	An Ordinance establishing general		
Connally/Clerk of Council and	definitions for the Cuyahoga County Code		
Director of Law	and authorizing the Clerk of Council to		
	officially publish the Cuyahoga County		
	Code as approved by the County's Director		
	of Law; and declaring the necessity that		
	this Ordinance become immediately		
	effective.		

WHEREAS, the Cuyahoga County Clerk of Council and Department of Law have been working on codification of the Cuyahoga County Ordinances to officially launch the Cuyahoga County Code;

WHEREAS, the Clerk of Council and Law Department have determined that it is necessary to include a definitions chapter in the General Provisions of the Cuyahoga County Code that is applicable to all legislation adopted by the Cuyahoga County Council; and,

WHEREAS, it is necessary that this Ordinance become immediately effective in order that critical services provided by Cuyahoga County can continue and to provide for the usual, daily operation of a County entity.

NOW, THEREFORE, BE IT ENACTED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That Cuyahoga County Council hereby adopts the following Definitions Chapter to the Cuyahoga County Code:

Chapter 101: Definitions

In the construction of this Code of the County of Cuyahoga, the following definitions shall be observed unless an alternative definition is otherwise provided in the Code:

"Charter" shall mean the County of Cuyahoga Charter also known as the Charter of Cuyahoga County, effective on January 1, 2010, and shall include any amendments to such Charter.

"Code" shall mean the Cuyahoga County Code, unless the context otherwise requires.

- "County" shall mean the County of Cuyahoga, Ohio created by the Cuyahoga County Charter. As used in the Code, the term "County" shall include all County contracting authorities.
- "County Council" or "Council" shall mean the County Council of the County of Cuyahoga, Ohio.
- "C.C.C." or "CCC" shall mean the Cuyahoga County Code.
- "County Executive" shall mean the County Executive of the County of Cuyahoga, Ohio.
- "Day" shall mean a calendar day, unless otherwise expressly stated.
- "Month" shall mean a calendar month, unless otherwise expressly stated.
- "Oath" includes affirmation and "swear" includes affirm.
- "Ohio R.C.", "ORC", "Revised Code", "R.C.", or "RC" refer to the Ohio Revised Code.
- "Person" includes an individual, corporation, business trust, estate, trust, partnership, and association.
- "Premises," as applied to property, shall include land and buildings.
- "Property" shall mean real and personal property.
- "The State" or "this State" shall mean the State of Ohio.
- "Year" shall mean a calendar year unless otherwise expressly stated.
- **SECTION 2.** With the incorporation of the foregoing definitions, the Clerk of Council is hereby authorized to officially publish the Cuyahoga County Code as approved by the County's Director of Law. The Clerk of Council shall continually update the Code in an expeditious manner and as approved by the County's Director of Law.
- **SECTION 3.** It is necessary that this Ordinance become immediately effective in order that critical services provided by Cuyahoga County can continue, and to continue the usual and daily operation of a County entity. Provided that this Ordinance receives the affirmative vote of eight members of Council, this Ordinance shall become immediately effective upon the signature of the County Executive.
- **SECTION 4.** It is found and determined that all formal actions of this Council relating to the adoption of this Ordinance were adopted in an open meeting of the

On a motion byduly enacted.	, seconded by, the forego	oing Ordinance was
Yeas:		
Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date
First Reading/Referred to Committee(s) Assigned:	Committee:	
Journal, 20	_	

Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

County Council of Cuyahoga County, Ohio

Ordinance No. O2013-0002

Sponsored by: Councilmember	An Ordinance amending the Pub	lic
Miller	Records Policy for Cuyahoga County, a	nd
	declaring the necessity that this Ordinan	ice
	become immediately effective.	

WHEREAS, on January 3, 2011 the Cuyahoga County Council passed Ordinance No. O2011-0003, which adopted a Public Records Policy for Cuyahoga County; and was amended by Ordinance No. O2011-0012; and

WHEREAS, it is necessary to amend the Public Records Policy to clarify potentially ambiguous provisions therein and to ensure that the Public Records Policy accurately reflects Council's original intent in enacting the Public Records Policy; and

WHEREAS, it is necessary that this Ordinance become immediately effective in order that critical services provided by Cuyahoga County can continue and to continue the usual and daily operation of the County.

NOW, THEREFORE, BE IT ENACTED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. Section 5:5(B) of Chapter 5: Public Records, and Sections 7.1 and 7.2 of Chapter 7: Costs of Public Records, as amended by Ordinance O2011-0012, are hereby amended to read as follows (additions are bolded, deletions are stricken out):

Section 5.5: Designation of Public Records Managers

- B.) For the following offices, the person designated as public records manager shall be an employee of the office who works at the principal place at which that office does business:
- a. the County Executive, including all executive office staff
- b. the County Council
- c. the Law Department of Law
- d. the Sheriff
- e. the Medical Examiner
- f. the Clerk of Courts
- g. the Department of Economic Development, which shall include the Office of Collaboration
- h. the Information Officer Department of Information Technology
- i. the Department of Public Works
- j. the Department of Purchasing
- k. the Department of Human Resources

- the Fiscal Office
- m. the Child Support Enforcement Agency Cuyahoga Job and Family Services (CJFS)
- n. the Division of Children and Family Services
- o. the Division of Employment and Family Services
- po. the Division of Senior and Adult Services
- qp. the Director of Human Services, which shall include the Director's Office and all other offices in the Department of Health and Human Services not covered by items twelve (12) through fifteen (15) above.
- Fq. the Agency of the Inspector General
- r. the Department of Regional Collaboration
- s. the Department of Communications
- t. the Department of Public Safety and Justice
- u. the Department of Consumer Affairs
- v. the Department of Internal Auditing
- Section 7.1: County Council to Determine Copy Costs for Public Records

 Persons requesting copies of public records shall be required to pay for the cost of making copies, at a rate not to exceed the actual cost of making copies. Payment in advance may be required.

Starting in calendar year 2011, the County Council shall —biannually biennially (once every two years) determine and establish the copying costs for public records.

Section 7.2: Interim Copy Costs for Public Records

Except as otherwise provided by court order **or law**, the following copying costs shall apply until the County Council first determines and establishes copying costs for public records:

SECTION 2. It is necessary that this Ordinance become immediately effective in order that critical services provided by Cuyahoga County can continue, and to continue the usual and daily operation of a County entity. Provided that this Ordinance receives the affirmative vote of eight members of Council, this Ordinance shall become immediately effective upon the signature of the County Executive.

SECTION 3. It is found and determined that all formal actions of this Council relating to the adoption of this Ordinance were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion by _	, seconded by _	, the foregoing Ordinance was
duly enacted.		

Yeas:		
Nays:		
	County Council President	Date
	County Executive Date	e
	Clerk of Council	Date
First Reading/Referred to Committee: Committee(s) Assigned:		
Journal, 20		

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0012

Sponsored by: County Executive	A Resolution amending the 2012/2013
FitzGerald/Fiscal Officer/Office	Biennial Operating Budget for 2013 by
of Budget & Management	providing for additional fiscal
	appropriations from the General Fund and
	other funding sources, for appropriation
	transfers between budget accounts, and
	for cash transfers between budgetary
	funds, in order to meet the budgetary
	needs of various County departments,

WHEREAS, on December 11, 2012, the Cuyahoga County Council adopted the Biennial Operating Budget and Capital Improvements Program Update for 2013 (Resolution No. R2012-0232) establishing the 2013 biennial budget update for all County departments, offices and agencies; and

offices, and agencies.

WHEREAS, it is necessary to adjust the Biennial Operating Budget for 2013 to reflect budgetary funding increases, funding reductions, to transfer budget appropriations, and to transfer cash between budgetary funds, in order to accommodate the operational needs of certain County departments, offices, and agencies; and

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the 2012/2013 Biennial Operating Budget for 2013 be amended to provide for the following additional appropriation increases and decreases:

Fund Nos./Budget Accounts

Journal Nos.

A.	40A069 – Capital Project Future D	ebt Issue		BA1306003
	CC768150 – Data Center Fit Plan -	 Medical 	Examiner's Build	ling
	Personal Services	\$	65,000.00	
	Other Expenses	\$	515,000.00	
	Capital Outlay	\$	2,570,000.00	

Funding Source: Funding for the Information Technology Data Center relocation will come from a future debt issue.

B1.	40A069 – Capital Proje	ect Future Debt Issue		BA1306004
	CC766329- New HVA	C Units – County Airport		
	Other Expenses	\$	742.50	

B2.	40A069 – Capital Project Future CC767095 – Replace Runway Li		•	BA1306004
	Personal Services	\$	4,803.81	
В3.	40A069 – Capital Project Future CC767103 – 1 st & 2 nd Floors – Ja Personal Services	Debt Issue ine Edna Hunter \$	Building 69,208.29	BA1306004
B4.	40A069 – Capital Project Future CC766873 – HVAC Repair and I		/arious Build	BA1306004 ings
	Personal Services Capital Outlay	\$ \$	51,172.01 33,052.70	
B5.	40A069 – Capital Project Future			BA1306004
	CC767160 – Health & Safety Impersonal Services	provements - Ar \$	chives 14,323.37	
B6.	40A069 – Capital Project Future CC767269 – Witness Victim Ren		ngo Title	BA1306004 BA1306005
	Personal Services Capital Outlay	\$ \$	71,421.57 3,969.58	
B7.	40A069 – Capital Project Future CC767326 – Replace Security Sy		Building	BA1306005
	Personal Services Capital Outlay	\$ \$	12,315.73 145.00	
B8.	40A069 – Capital Project Future CC767483 – Replace Security Sy		Building	BA1306005
	Personal Services	\$	11,947.05	
B9.	40A069 – Capital Project Future CC767822 – 2 nd Floor Recorder's Personal Services		Administratio 70,070.58	BA1306005 in Building
	Capital Outlay	\$	33,785.76	
B10.	40A069 – Capital Project Future CC767384 – Health & Safety Im		irgrounds	BA1306006
	Capital Outlay	\$ (168,975.92)	
B11.	40A069 – Capital Project Future CC766329 – New HVAC Units -			BA1306006
	Other Expenses Capital Outlay	\$	(2,311.38) (3,363.10)	

B12.	40A069 – Capital Project Future D CC767095 – Replace Runway Ligh	hting - Air _l	•	BA1306006
	Personal Services Capital Outlay	\$ \$	(9,748.00) (18,544.44)	
N13.	40A069 – Capital Project Future D CC767103 – 1 st & 2 nd Floors – Jane Personal Services	ebt Issue e Edna Hui \$	nter Building (69,726.09)	BA1306006
	Other Expenses Capital Outlay	\$ \$	(25,000.00) (210,222.39)	
B14.	40A069 – Capital Project Future D CC766873 – HVAC Repair and Re Personal Services	eplacement \$	(51,812.00)	BA1306006 lings
	Capital Outlay	\$	(39,335.04)	
B15.	40A069 – Capital Project Future D CC767160 – Health & Safety Impr		- Archives	BA1306006
	Personal Services Capital Outlay	\$ \$	(54,421.00) (4,339.00)	
B16.	40A069 – Capital Project Future D CC767269 – Witness Victim Reno	vation – C	•	BA1306006
	Personal Services Capital Outlay	\$	(74,557.00) (46,635.16)	
B17.	40A069 – Capital Project Future D CC767236 – Replace Security Sys		ion Building	BA1306006
	Personal Services Capital Outlay	\$ \$	(32,263.66) (44,014.03)	
B18.	40A069 – Capital Project Future D CC767491 – Replace Chilled Water		- Board of Electi	BA1306006 ons
	Personal Services Capital Outlay	\$ \$	(9,000.00) (111,000.00)	
B18.	40A069 – Capital Project Future D CC767483 – Arts & Crafts Buildin		ling	BA1306007
	Personal Services Capital Outlay	\$ \$	(12,500.00) (226,823.01)	
B19.	40A069 – Capital Project Future D CC767822 – 2 nd Floor Recorder's Personal Services	Debt Issue Renovatior \$	n – Administratio (73,930.00)	BA1306007 on Building
	Capital Outlay	\$	(46,056.02)	
Funding S	ource: Funding for these capital proj	jects came	from past and fu	ture debt offerings.
C1.	40A099 – Maintenance Projects CC768101 – Countywide Painting			BA1306008
	Other Expenses	\$	1,000,000.00	

C2.	40A099 – Maintenance Pro CC768119 – Countywide C	,		BA1306008
	Other Expenses	\$	1,000,000.00	
-	ource: Funding for these two debt issuance.	maintenance proje	ects will come fro	om the General Fund
D.	40A069 – Capital Projects I CC767962 – Medical Exam Personal Services Capital Outlays			BA1306009 Out
Funding So	ource: Funding for the Crime	Lab Build-Out is	from the General	Fund.
E1.	30A910 – Brownfield Debt DS039966 – Brownfield De Other Expenses		(350,000.00)	BA1306010
E2.	30A905 – Gateway Arena DS100370 – Gateway Aren Other Expenses	a Project \$	350,000.00	BA1306011
Funding So	ource: General Fund.			
relating to Council, a resulted in	HON 2. It is found and det the adoption of this Resol and that all deliberations of such formal action were is equirements, including Sec	ution were adopt this Council and n meetings open	ted in an open maded in any of its conto to the public, in	neeting of the ommittees that a compliance with
On a motion	on by, second ted.	ed by	_, the foregoing	g Resolution was
Yeas:				
Nays:				
	Co	unty Council Pre	esident	Date
	\overline{Co}	unty Executive		Date

Clerk of Council	Date

Journal CC009 January 22, 2013







January 15, 2012

Clerk of County Council

Dear Ms. Schmotzer:

A brief summary of the fiscal items which will be presented for your consideration for adoption on first reading at the regular County Council meeting scheduled for January 22, 2013, are presented below.

<u>Additional Appropriation Summary</u> – Additional appropriations are needed when there is a new or increased revenue source, or a revision to the original appropriation level that is required to cover expenditures that exceed the original estimate. A budget review document is provided for General Fund and Health & Human Services Levy Fund impact items.

A reduction in appropriation is requested in conjunction with the close-out of a program, grant, project or decertification of an encumbrance.

^{*} Impact of fiscal item is included in the current projection and ending fund balance.

Other Operating Funds	Amount
Debt Service – An decrease in appropriation from the Brownfield Debt Service and a corresponding increase to the Gateway Arena Project to cover 2013 debt payment (\$350,000). Funding is from the General Fund.	\$0.00
TOTAL	0

Grants/Projects	Amount
Capital Projects/Data Center Fit Plan – Additional appropriation to fund the first phase of the relocation of the	
IT Data Center from the Sterling Building to the Medical Examiner's Building. Funding is from a future debt	\$3,150,000.00
issuance.	
Capital Projects/Various – Appropriation increases and decreases to close-out various capital projects that have	¢057.610.20
been completed. Funding is from past and future debt issuances.	-\$957,619.29
Capital Projects/Maintenance Projects – Additional appropriation for the Countywide Painting (\$1,000,000) and	\$2,000,000.00
Countywide Carpeting (\$1,000,000) projects. Funding is from the General Fund and a future debt issuance.	\$2,000,000.00
Capital Projects/Medical Examiner's Regional Crime Labe Build-Out – Additional appropriation to cover the cost	\$2,773,600.00
of the project. Funding is from the General Fund.	32,773,000.00
TOTAL	\$6,965,980.71

Total Additional Appropriations All Funds	\$6,965,980.71
Total Additional Appropriations - All Funds	\$0,905,980./1

The following represents the overall changes made to the 2012-2013 Biennial Appropriation Measure for 2013 since its adoption on December 11, 2011 Resolution R2011-0232. The changes reflect the Additional Appropriations, Appropriation Transfers and Cash Transfers to the original adopted appropriation resolution.

APPROPRIATION STATUS SUMMARY: Adjus						
		1/22/2013 Agenda	<u> </u>	Year to Date*		Appropriation
General Fund Impact	\$	0.00	\$	361,708,846.00	\$	362,089,445.00
HHS Levy Impact	\$	0.00	\$	222,615,901.00	\$	222,615,901.00
Other Fund Impact	\$	<u>6,965,980.71</u>	\$	919,467,552.00	\$	926,433,532.71
Total Impact	\$	6,965,980.71	\$	1,503,792,299.00	\$	1,511,138,878.71

* 2013 appropriation levels adopted by resolution R2012-0232 on December 11, 2012.

Thank you for your consideration regarding this matter.

Sincerely,

Matthew Rubino

Director, Office of Budget & Management

mrubino@cuyahogacounty.us

(216) 443-7448

Fax: (216) 443-8193

MEMORANDUM

TO: Jeanne Schmotzer, Clerk of Council **REVISED: January 4,2013**

FROM: Matthew Rubino, Director, Office of Budget & Management

DATE: January 15, 2013

RE: Agenda Items

The Office of Budget & Management requests that the following fiscal items be presented to the members of County Council for their consideration for approval on first reading at the meeting of January 22, 2013. The requested fiscal items including additional appropriations, appropriation transfers, and cash transfers meet agency budgetary needs.

Resolution: Additional Appropriations

A.	40A069 – Capital Project F	40A069 – Capital Project Future Debt Issue		
	CC768150 – Data Center Fi	t Plan – Medical	Examiner's Building	5
	Personal Services	\$	65,000.00	
	Other Expenses	\$	515,000.00	
	Capital Outlay	\$	2.570.000.00	

Appropriation is requested to fund the relocation of the IT Data Center from the Sterling Building to the Medical Examiner's Building located at 11011 Cedar Avenue. The total amount of this request (\$3,150,000.00) represents the cost of the first phase of the project. The second phase will cost \$3,300,000.00 bringing the total cost of the project to \$6,450,000.00. Appropriation for the second phase will be requested at a later date. Funding for the Information Technology Data Center relocation will come from a future debt issue.

B1.	40A069 – Capital Project Future Debt Issue CC766329– New HVAC Units – County Airport			BA1306004
	Other Expenses	\$	742.50	
B2.	40A069 – Capital Project Future Debt Issue CC767095 – Replace Runway Lighting – County Airport			BA1306004
	Personal Services	\$	4,803.81	
B3.	40A069 – Capital Project Future D			BA1306004
CC767103 – 1 st & 2 nd Floors – Jane Edna Hunter Building			Building	
	Personal Services	\$	69,208.29	

Fiscal Office Office of Budget & Management 1219 Ontario Street, Cleveland, OH 44113, (216) 443-7220, FAX (216) 443-8193 Ohio Relay Service (TTY) 711

B4.	40A069 – Capital Project Future D CC766873 – HVAC Repair and Rep	lacement – Va	_	BA1306004
	Personal Services Capital Outlay	\$ \$	51,172.01 33,052.70	
B5.	40A069 – Capital Project Future D CC767160 – Health & Safety Impro Personal Services		chives 14,323.37	BA1306004
B6.	40A069 – Capital Project Future D CC767269 – Witness Victim Renov Personal Services Capital Outlay		go Title 71,421.57 3,969.58	BA1306004 BA1306005
В7.	40A069 – Capital Project Future D CC767326 – Replace Security Syste Personal Services Capital Outlay		Building 12,315.73 145.00	BA1306005
B8.	40A069 – Capital Project Future D CC767483 – Replace Security Syste Personal Services		Building 11,947.05	BA1306005
В9.	40A069 – Capital Project Future D CC767822 – 2 nd Floor Recorder's F Personal Services Capital Outlay		dministration 70,070.58 33,785.76	BA1306005 Building
B10.	40A069 – Capital Project Future D CC767384 – Health & Safety Impro Capital Outlay	ovements - Fai	irgrounds (168,975.92)	BA1306006
B11.	40A069 – Capital Project Future D CC766329 – New HVAC Units - Air Other Expenses Capital Outlay		(2,311.38) (3,363.10)	BA1306006
B12.	40A069 – Capital Project Future D CC767095 – Replace Runway Light Personal Services Capital Outlay		(9,748.00) (18,544.44)	BA1306006
N13.	40A069 – Capital Project Future D CC767103 – 1 st & 2 nd Floors – Jane Personal Services Other Expenses Capital Outlay	Edna Hunter \$ \$	Building (69,726.09) (25,000.00) (210,222.39)	BA1306006

B14.	40A069 – Capital Project Future De CC766873 – HVAC Repair and Repla Personal Services Capital Outlay		arious Buildings (51,812.00) (39,335.04)	BA1306006
B15.	40A069 – Capital Project Future De CC767160 – Health & Safety Improv Personal Services Capital Outlay		chives (54,421.00) (4,339.00)	BA1306006
B16.	40A069 – Capital Project Future De CC767269 – Witness Victim Renova Personal Services Capital Outlay		go Title (74,557.00) (46,635.16)	BA1306006
B17.	40A069 – Capital Project Future De CC767236 – Replace Security System Personal Services Capital Outlay		Building (32,263.66) (44,014.03)	BA1306006
B18.	40A069 – Capital Project Future De CC767491 – Replace Chilled Water Personal Services Capital Outlay	System – Bo \$	ard of Elections (9,000.00) (111,000.00)	BA1306006
B18.	40A069 – Capital Project Future De CC767483 – Arts & Crafts Building F Personal Services Capital Outlay	Remodeling \$	(12,500.00) (226,823.01)	BA1306007
B19.	40A069 – Capital Project Future De CC767822 – 2 nd Floor Recorder's Re Personal Services Capital Outlay		Administration (73,930.00) (46,056.02)	BA1306007 Building

Appropriation increases and reductions are requested for various capital projects that have been completed. The purpose is to remove remaining balances by sub-object from various index codes. Funding for these capital projects came from past and future debt offerings.

C1.	40A099 – Maintenance Projects CC768101 – Countywide Painting		BA1306008
	Other Expenses	\$ 1,000,000.00	
C2.	40A099 – Maintenance Projects CC768119 – Countywide Carpeting Other Expenses	\$ 1,000,000.00	BA1306008

Additional appropriation is requested for two maintenance project accounts: Countywide Painting and Countywide Carpeting. Funding for these two maintenance projects will come from the General Fund and future debt issuance.

D.	40A069 – Capital Projects Future Debt Issuance			BA1306009
	CC767962 – Medical Exami	CC767962 – Medical Examiner's Regional Crime Lab Build-Out		
	Personal Services	\$	140,000.00	
	Capital Outlays	\$	2,633,600.00	

Additional appropriation is requested for the Medical Examiner's Regional Crime Lab Build-Out to cover the cost of the project. Funding for the Crime Lab Build-Out is from the General Fund.

E1.		30A910 – Brownfield Debt Service DS039966 – Brownfield Debt Service				
	Other Expenses	\$	(350,000.00)			
E2.	30A905 – Gateway Arena			BA1306011		
	DS100370 – Gateway Arena I	Project				
	Other Expenses	\$	350,000.00			

An appropriation decrease is requested in the Brownfield Debt Service Fund to move funds from this fund to the Gateway Arena project fund to cover the Gateway 2013 debt payment. Funding for the Brownfield and Gateway Arena Funds debt service comes from the General Fund.

S:_NewServer\Resos\Fiscal Agendas\2013\ADDJAN22.doc

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0003

Sponsored by: County Executive	A Resolution authorizing Settlement and		
FitzGerald	Release Agreements with various plaintiffs in		
	the total amount of \$143,257.00 in connection		
	with Teel et al. v. Cuyahoga County, U.S.D.C.		
	Case No. 11cv2582; authorizing the County		
	Executive to execute said agreements;		
	authorizing the appropriation of funds for		
	payment of settlement amounts set forth		
	herein; and declaring the necessity that this		
	Resolution become immediately effective.		

WHEREAS, Cuyahoga County has been named in a lawsuit filed in the United States District Court, Northern District of Ohio, in *Teel et al. v. Cuyahoga County*, U.S.D.C. Case No. 12cv2582, by current and former employees of the Cuyahoga County Coroner's Office, maintaining that their rights to overtime compensation under the federal Fair Labor Standards Act were violated; and,

WHEREAS, pursuant to Ordinance No. O2011-0033, Section 1, subsection (c), the Cuyahoga County Executive is authorized to settle or compromise any lawsuit against the County except that any settlement resulting in the County's expenditure of an amount in excess of \$100,000.00 shall require the prior approval of the Council; and,

WHEREAS, the County's legal representatives have recommended settlement of the claims of these Plaintiffs in the amounts set forth below, the total amount of which is One Hundred Forty Three Thousand, Two Hundred Fifty Seven Dollars (\$143,257.00); and,

WHEREAS, the Plaintiffs have executed Settlement and Release Agreements for specified amounts contingent upon the approval of the County Council; and,

WHEREAS, it is necessary that this Resolution become effective immediately to authorize immediate processing of the settlement of these claims and closure of the case.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the County Executive is hereby authorized to execute the Settlement and Release Agreements with the following Plaintiffs in *Teel et al. v. Cuyahoga County*, for the following amounts:

1.	Teel, Terry	\$2,188.00
2.	Gaul, Kevin	\$10,555.00
3.	McCollins, Gloria	\$9,266.00
4.	Cahal, Karen	\$10,393.00
5.	Celinski, Pamela	\$15,535.00
6.	Teel, Charles	\$15,128.00
7.	Baran, Alex	\$7,294.00
8.	Bolger, Roger	\$17,858.00
9.	Ciula, Thomas	\$10,740.00
10.	Csank, Jackelyn	\$9,379.00
11.	Dean, Mary Ann	\$3,893.00
12.	Dornback, James	\$500.00
13.	Fischer, Anjanette	\$6,833.00
14.	Hengoed, Melissa	\$500.00
15.	Kilcoyne, Blythe	\$3,714.00
16.	Kolb, Christine	\$4,737.00
17.	Smith, Lawrence	\$3,733.00
18.	Taliano, Michaelene	\$6,913.00
19.	Treece, Melanie	\$2,770.00
20.	Wallace, Paula	\$1,328.00

SECTION 2. That the County Executive is authorized to execute all other necessary documents required to provide for the payments set forth in the settlement agreements.

SECTION 3. It is necessary that this Resolution become immediately effective in order that the proceedings in *Teel et al. v. Cuyahoga County* may proceed as directed by the Court and to continue the usual and daily operations of the County Departments affected by matter. Provided that this Resolution receives the affirmative vote of eight members of the Council, this Resolution shall become immediately effective upon the signature of the County Executive.

SECTION 4. It is found and determined that all formal actions of the Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, incompliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by	, the foregoing Resolution was
Yeas:		
Nays:		

County Council President	Date
County Executive	Date
Clerk of Council	 Date

No Action Taken by Council: <u>January 8, 2013</u>

Journal CC009 January 22, 2013

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0013

Sponsored by: Council President
Connally on behalf of Juvenile
Court

A Resolution approving a Memorandum of Understanding between Cuyahoga County Common Pleas Court, Juvenile Court Division and Service Employees International Union (SEIU), District 1199, The Health Care and Social Service Union, in connection with a collective bargaining agreement representing approximately 111 probation, clerks transportation staff for the period 11/1/2009 -10/31/2012; directing that funds necessary to implement the Memorandum of Understanding be budgeted and appropriated; authorizing the County Executive to execute the Memorandum of Understanding and all other documents consistent with this Resolution; and declaring the necessity that this Resolution become immediately effective.

WHEREAS, the Cuyahoga County Common Pleas Court, Juvenile Court Division has been engaged in wage-reopener negotiations with Service Employees International Union (SEIU), District 1199, The Health Care and Social Service Union, for a Memorandum of Understanding in connection with a collective bargaining agreement representing approximately 111 Probation, Clerks and Transportation staff for the period 11/1/2009 – 10/31/2012; and,

WHEREAS, on or about August 28, 2012, the members of SEIU, District 1199, The Health Care and Social Service Union met and voted to ratify the proposed Memorandum of Understanding for the wage re-opener terms; and,

WHEREAS, O.R.C. 4117.10 (B) requires that a public employer submit a request for funds necessary to implement an agreement, and for approval of any other matter requiring the approval of the appropriate legislative body to the legislative body within thirty days of the date on which the parties finalize the agreement, unless otherwise specified or if the legislative body is not in session at the time, then within fourteen days after it convenes; and,

WHEREAS, O.R.C. 4117.10(B) further states that the legislative body must approve or reject the submission as a whole, and the submission is deemed approved if the legislative body fails to act within thirty days after the public employer submits the agreement; and,

WHEREAS, the Council President on behalf of Cuyahoga County Common Pleas Court, Juvenile Court Division are recommending that Council approve the proposed Memorandum of Understanding; and,

WHEREAS, it is necessary that this Resolution become immediately effective to ensure the efficient operation of the Cuyahoga County Common Pleas Court, Juvenile Court Division.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council hereby approves the Memorandum of Understanding between Cuyahoga County Common Pleas Court, Juvenile Court Division and Service Employees International Union (SEIU), District 1199, The Health Care and Social Service Union, in connection with a collective bargaining agreement representing approximately 111 Probation, Clerks and Transportation staff for the period 11/1/2009 - 10/31/2012.

SECTION 2. That funds necessary to implement the Memorandum of Understanding between Cuyahoga County Common Pleas Court, Juvenile Court Division and Service Employees International Union (SEIU), District 1199, The Health Care and Social Service Union shall be budgeted and appropriated.

SECTION 3. It is necessary that this Resolution become immediately effective in order that critical services provided by Cuyahoga County can continue, and to continue the usual and daily operation of a County entity. Provided that this Resolution receives the affirmative vote of eight members of Council, this Resolution shall become immediately effective upon the signature of the County Executive.

SECTION 4. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion by duly adopted.	, seconded by	, the forego	ing Resolution was
Yeas:			
Nays:			
	County Counc	il President	Date

County Executive	Date
Clerk of Council	Date

Journal CC009 January 22, 2013

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0014

Sponsored by: County Executive FitzGerald/Department of Public Works/Division of County Engineer

Resolution making award an RQ25617 to Karvo Paving, Co. in the amount not-to-exceed \$1,547,652.00 for resurfacing Taylor Road from Euclid Avenue East Cleveland South to Corporation Line in the City of East Cleveland: authorizing the County Executive to execute the contract and all other documents consistent with said award and this Resolution; and authorizing the County Engineer, on behalf of the County Executive, to make an application for allocation from County Motor Vehicle \$7.50 License Tax Funds in the amount of \$1.238.121.60 to fund said contract.

WHEREAS, the County Executive FitzGerald/Department of Public Works/ Division of County Engineer has recommended an award on RQ25617 to Karvo Paving, Co. in the amount not-to-exceed \$1,547,652.00 for resurfacing Taylor Road from Euclid Avenue to East Cleveland South Corporation Line in the City of East Cleveland; and

WHEREAS, the Council has determined that awarding RQ25617 to Karvo Paving, Co. is in the best interest of the County; and

WHEREAS, the funding for this project is as follows: (a) 80% will be paid from the County Motor Vehicle \$7.50 License Tax Fund, and (b) 20% from the Ohio Public Works Commission Issue 1 Fund.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council hereby makes an award on RQ25617 to Karvo Paving, Co. in the amount not-to-exceed \$1,547,652.00 for resurfacing Taylor Road from Euclid Avenue to East Cleveland South Corporation Line in the City of East Cleveland.

SECTION 2. That the County Executive is authorized to execute a contract in connection with said award and all documents consistent with this Resolution.

SECTION 3. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by, the forego	oing Resolution was
Yeas:		
Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date
First Reading/Referred to Co	ommittee:	
Committee(s) Assigned:		
Journal		





Item Details:

Agency/Dept.

County Engineer

Agency/Dept.Head Thomas

P. Sotak,

Name:

Name:

P.E.

Type of Request:

Award Recommendation

Request Prepared Juliann Conway

Telephone No.

216-348-3838

by:

SUMMARY OF REQUESTED ACTION:

Department of Public Works

A) SCOPE OF WORK SUMMARY

- 1. is recommending an award on RQ25617 to Karvo Paving, Co. in the amount not-toexceed \$1,547,652.00 for the resurfacing of Taylor Road from Euclid Avenue to the East Cleveland South Corporation Line in the City of East Cleveland
- 2. and enter into a contract with Karvo Paving, Co. They were determined to be the lowest qualified bidder and submitted all necessary documentation per the Bid Specifications.
- 3) N/A.

PROCUREMENT

- 1) Competitive Bid process.
- 2) The Office of Procurements & Diversity (OPD) assessed a thirty (30%) percent SBE Goal. Karvo Paving, Co. met this goal and it was approved by OPD.
- 3) The engineer's estimate was \$1,732,038.50. The bids were open on December 12, 2012.

There were twelve (12) proposals pulled from OPD and seven (7) bids were submitted for review and an award recommendation is being made to the lowest qualified bidder.

3) N/A.

C) CONTRACTOR AND PROJECT INFORMATION

Karvo Paving, Co. 330-929-9676

4524 Hudson Drive

Stow, Ohio 44224

Council District - N/A

3. Construction is in the City of East Cleveland (Council District 10).

D. PROJECT STATUS AND PLANNING

1. As per specifications construction is scheduled to begin in April of 2013 and will be completed in October of 2013.

E. FUNDING

1. 80% Cuyahoga County using funds from the \$7.50 Vehicle License Tax Fund and 20% from the Ohio Public Works Commission (Issue 1)

The contractor has completed ethics training as required

PURPOSE/OUTCOMES - PRINCIPAL OWNER(S):

Principal Owner (s): George Karvounides, President

The work within the limits of this 0.52 mile resurfacing improvement includes the removal of the existing asphalt concrete wearing course, the repair of the deteriorated rigid (brick/concrete) base, and the removal and replacement of impacted curb, sidewalk and drive aprons. The construction of a uniform three inch (3") asphalt concrete overlay; the installation of ADA compliant curb ramps; the removal and replacement of the existing water main and service connections; new traffic signals at Terrace Road intersection and other related items as shown on the plans or stipulated in the specifications, plan notes, proposal notes or elsewhere in the bid package.

Explanation for late submittal:

Contract/Agreement Information:

Procurement Method:

Competitive Bid

Explanation for Increase/Decrease in \$ Amount for current request:

Financial Information:

Funding source: Explanation:

Other See Summary of Requested Action.

Total Amount Requested:

\$1,547,652.00

Department of Public Works Bid Results

Thursday, December 13, 2012 9:35.46 AM

Letting Date 12/12/2012	Cuyahoga County Engineer's Estimate: \$1,732,038.50		RQ No.: 25617	Rep. No.: 03-13
Taylor Road; Resurfacing			*	
Euclid Avenue to the	Buclid Avenue to the East Cleveland South Corporation Line in the City of East Cleveland	City of East Cleveland		
Low Bidder Karvo Paving Company	Company	\$1,547,652.00	-10.65%	
Fabrizi Trucki	Fabrizi Trucking & Paving Co., Inc.	\$1,614,987.90	~92'9-	
C.A. Agresta (C.A. Agresta Construction Co.	\$1,615,935.25	- %02'9-	
Perk Company, Inc.	y, Inc.	\$1,661,531.50	4.07%	
Terrace Constr	Terrace Construction Company, Inc.	\$1,716,918.40	-0.87%	
Great Lakes Crushing, Ltd.	rushing, Ltd.	\$1,800,627.89	3.96%	
Vandra Brothe	Vandra Brothers Construction Co.	\$1,810,283.80	4.52%	

TAYLOR ROAD (C.R. 28) FROM EUCLID AVE. TO THE EAST CLEVELAND SOUTH CORPORATION LIMIT

Comparison Com	RE-	P DESCRIPTION	TIND	ESTIMATED	DOLLARS CTS	Karvo Paving Company,	_	. UNIT PRICE EST.	Fabrizi Trucking & Devino Co. Inc.	UNIT PRICE EST.	C.A. Agresta Construction Co
	2										
2. B. P. PARLEMENT CREATION CREAT	-	CLEARING AND GRUBBING	LUMP			4	-	-			\$ 1,000.00
4 SURVEYERS COUNTER, REPOLANCE 69 YO 200 9774 8 YO 200 8 2 X 200 8 <t< td=""><td>2</td><td></td><td>SQ YD</td><td>3418</td><td></td><td>49</td><td>_</td><td></td><td></td><td></td><td>\$ 34,180.00</td></t<>	2		SQ YD	3418		49	_				\$ 34,180.00
	ო	П	SQ YD	9724		\$		_			\$ 26,741.00
5 CANADITY REBONDED SEPT	4		SQYD	2500				_			\$ 7,500.00
	3	1	SQ FT	3274		₩.	-	_			\$ 3,274.00
7.9 PREMERIORIEGIZA FAVO LINDER FT 80 \$ <t< td=""><td>φ</td><td></td><td>Ŀ</td><td>2626</td><td></td><td>69</td><td></td><td>_</td><td></td><td></td><td>\$ 10,100.00</td></t<>	φ		Ŀ	2626		69		_			\$ 10,100.00
Decided the North Company	7		Ŀ	80		. 8		_			\$ 400.00
DECOMMENDED DECOMMENDED DECOMPRED	<u>_</u>	Г	EACH	2		s					\$ 200.00
11 ILLANGA GANDANICANICANICANICANICANICANICANICANICANIC	6		EACH	2		s	_	_			\$ 200.00
12 STATES CONDUITY CHEENER MANUMENT BOXASSEMBLY, TYPE 1 SACK STATES CONDUITY CHEENER MANUMENT BOXASSEMBLY, TYPE 2 SACK STATES CONDUITY CHEENER MANUMENT BOXASSEMBLY, TYPE 3 SACK	6	1	CY DO	94		\$				*	\$ 2,275.00
1.0	2		SQ YD	3026		· 64		_			\$ 3,026.00
1.00 CUMANICAC COUNTY CENTERLAME NOVAMENTE DOX ASSEABLY, TYPE 1 Dich 3 5 5 5 5 5 5 5 5 5	12	T	SQ YD	1310		€					\$ 3,930.00
14 CHYAMICANI COLNITY MONIUMENT BOX ELCH 1 1 1 1 1 1 1 1 1	5	_	EACH	8		es.	-				\$ 1,350.00
15 MONUMENT ECX ADLISTED TO GRADE, AS PER PLAN ELCH 10 \$ 440.00 \$ 440.00 \$ 440.00 \$ 450.00 \$	4	-	EACH	3		ø	\vdash	-			\$ 1,350.00
10 ROCKIONER ACMINIARY MONUMENT EACH 1 S 650.00 S 450.00 S 450.00 S 450.00 S 450.00 S 450.00 S 150.00 S	15	_	EACH	10							\$ 4,500.00
AN EACH 10 6 6400 8 6500 8 6500 8 1500 0 1,5000 8 5000 1 2,0000 1	5		EACH	-		s					\$ 60.00
10 CUCNOCRETE WALK, AS PER PLAN SÖFF 351 \$ 4.50 \$ 1,450.8.6.0 \$ 1,556.00 \$ 1,556.	7	-	EACH	10		\$					\$ 500.00
19 CUING RAMIP, TYPE EAI, AS PER PLAN EACH 3 \$ 6000 \$ 1,600.00 \$ 700.00 \$ 2,000.00 \$ 500.00 </td <td>18</td> <td></td> <td>SQ FT</td> <td>3313</td> <td></td> <td>\$</td> <td>_</td> <td>_</td> <td></td> <td></td> <td>\$ 15,736.75</td>	18		SQ FT	3313		\$	_	_			\$ 15,736.75
20 CURRE RAJAP TYPE RAJA SPER PLAN EACH 4 5 460.00 5 1,800.00 5 2,600.00 5 5,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.00 5 7,000.	13		EACH	. 3		\$	_		V		\$ 1,500.00
21 CLINE ROANIN, TYPE EZ, AS PER PLAN EACH 5 \$ 470.00 \$ 2,330.00 \$ 3,500.00 \$ 3,500.00 \$ 3,500.00 \$ 5,500.	8		EACH	4		49		_			\$ 2,000:00
SECTION TOTAL ROADWAY: FACH Z S 120,807.56 S 120,807.56 S 130,326.60 S 130,326.60 S 130,326.60 S 130,326.60 S 130,326.60 S 130,000 S 130,000 <td>2</td> <td></td> <td>EACH</td> <td>5</td> <td></td> <td>s</td> <td></td> <td></td> <td></td> <td></td> <td>\$ 2,500.00</td>	2		EACH	5		s					\$ 2,500.00
22 SOIL MAILYSIS TEST EROSION CONTROL 23 TOPOGNAL MAILYSIS TEST CONMERCIAL PERTILIZER 24 SEEDING MILCHING, AS PER PLAN 25 TOPOGNAL MILCHING, AS PER PLAN 26 SECONDAR CONTROL 27 SECONDAR CONTROL 28 SECONDAR CONTROL 29 SECONDAR CONTROL 20 SECONDAR CONTROL 21 SECONDAR CONTROL 22 SECONDAR CONTROL 23 SECONDAR CONTROL 24 SECONDAR CONTROL 25 SECONDAR CONTROL 26 SECONDAR CONTROL 27 SECONDAR CONTROL 28 SECONDAR CONTROL 28 SECONDAR CONTROL 29 SECONDAR CONTROL 29 SECONDAR CONTROL 20 SECONDAR C	SEC	STION TOTAL ROADWAY:				1038	99.				\$ 122,312.75
22 SOIL ANALYSIS TEST ERCOSION CONTROL EACH 2 \$ 150.00 \$ 150.00 \$ 300.00 \$ 50.00 <th< td=""><td>L.</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	L.										
22 SOLL ANALLYSIS TEST EACH 2 \$ 150.00 \$ 150.00 \$ 50.00		EROSION CONTROL									
23 TOPSOIL CU VD 73 \$ 5.00 \$ 2,555.00 \$ 36.00 \$ 2,628.00 \$ 30.00 24 SEEDING AND MULCHING, AS PER PLAN SQ YD 1,310 \$ 150.00 \$ 1,300.00 \$ 2,555.00 \$ 2,620.00 \$ 100 25 SEEDING AND MULCHING, AS PER PLAN TON 0.20 \$ 750.00 \$ 150.00 \$ 2,600.00 \$ 100 \$ 100 26 COMMERCIAL FERTILIZER ACRE 0.27 \$ 750.00 \$ 150.00 \$ 100 <td< td=""><td>23</td><td></td><td>EACH</td><td>2</td><td></td><td>s</td><td></td><td>_</td><td></td><td></td><td>\$ 100.00</td></td<>	23		EACH	2		s		_			\$ 100.00
24 SEEDING AND MULCHING, AS PER PLAN SQ YD 1,310 \$ 1,200 \$ 1,200 \$ 2,620,00 \$ 1,00 25 COMMERCIAL FERTILIZER TON 0.20 \$ 750,00 \$ 1,600 \$ <td< td=""><td>g</td><td></td><td>CU YD</td><td>73</td><td></td><td>€9</td><td>\rightarrow</td><td></td><td></td><td></td><td>\$ 219.00</td></td<>	g		CU YD	73		€9	\rightarrow				\$ 219.00
25 COMMERCIAL PERTILIZER TON 0.20 \$ 750.00 \$ 150.00 \$ 160.00 \$ 160.00 \$ 500.00 <td>24</td> <td></td> <td>SQ YD</td> <td>1,310</td> <td></td> <td>69</td> <td>_</td> <td></td> <td></td> <td></td> <td>\$ 1,310.00</td>	24		SQ YD	1,310		69	_				\$ 1,310.00
20 LIME ACRE 0.27 \$ 275.00 \$ 74.26 \$ 800.00 \$ 216.00 \$ 10.00 27 WATER M GAL 7 \$ 1.00 \$ 7.50 \$ 7.50 \$ 1.00 \$	25	$\overline{}$	TON	0.20		69					\$ 100.00
27 WATER TOWATER TOWAT	8		ACRE	0.27		s					\$ 2.70
28 EROSION CONTROL EACH 7,500 \$ 7,500.00 \$ 7,500.00 \$ 7,500.00 \$ 7,500.00 \$ 1,00 \$ 7,500.00 \$ 1,000.00	_		MGAL	7		47					\$ 7.00
SECTION TOTAL EROSION CONTROL: SECTION TOTAL EROSION CONTROL: \$ 11,896.25 \$ 11,896.25 \$ 13,494.00 28 © CONDUIT, TYPE F FOR UNDERDRAIN OUTLETS, AS PER PLAN FT 440 \$ 20.00 \$ 2,800.00 \$ 1,200.00 \$ 1,200.00 \$ 1,200.00 \$ 5000.00 \$ 5000.00 \$ 5000.00 \$ 5000.00 \$ 5000.00 \$ 2,400.00	_		EACH	7,500		₩	_	_			\$ 7,500.00
28 6"CONDUIT, TYPE F FOR UNDERDRAIN OUTLETS, AS PER PLAN FT 40 \$ 20.00 \$ 2,800.00 \$ 1,200.00 \$ 1,200.00 \$ 5,00	_	STION TOTAL EROSION CONTROL:		٠			.25				\$ 9,238.70
23 6° CONDUIT, TYPE F FOR UNDERDRAIN OUTLETS, AS PER PLAN FT 40 \$ 20.00 \$ 2,800.00 \$ 1,120.00 \$ 9.00 30 12° CONDUIT, TYPE B, AS PER PLAN FT 40 \$ 30.00 \$ 1,200.00 \$ 5,000.00 \$ \$ 5,000.00 \$ 5,000.00 \$ 5,000.00 \$ 5,000.00 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>											
29 6" CONDUIT, TYPE F FOR UNDERDRAIN OUTLETS, AS PER PLAN FT 40 \$ 20.00 \$ 2,800.00 \$ 1,120.00 \$ 9.00 30 12" CONDUIT, TYPE R, AS PER PLAN FT 40 \$ 30.00 \$ 1,200.00 \$ 5,000.00 \$ <t< td=""><td>لِـا</td><td>DRAINAGE</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	لِـا	DRAINAGE									
30 12° CONDUIT, TYPE B, AS PER PLAN FT 40 \$ 30,00 \$ 1,200.00 \$ 1,250.00 \$ 5,000.00 \$ 5,000.00 \$ 5,000.00 \$ 5,000.00 \$ 3,000.0			F	\$	v	\$	_				\$ 1,260.00
31 12° CONDUIT, TYPE C, AS PER PLAN FT 40 \$ 30.00 \$ 1,200.00 \$ 95.00 \$ 3,800.00 \$ 3,600.00 \$ 3,600.00 \$ 3,600.00 \$ 2,950.00 \$ 2,950.00 \$ 2,400.00 </td <td></td> <td></td> <td>Ħ</td> <td>\$</td> <td></td> <td>s</td> <td>→</td> <td></td> <td></td> <td></td> <td></td>			Ħ	\$		s	→				
32 CATCH BASIN, AS PER PLAN EACH 2 \$ 1,800.00 \$ 2,950.00 \$ 5,900.00 \$ 2,400.00 33 CATCH BASIN ADLUSTED TO GRADE. AS PER PLAN EACH 10 \$ 655.00 \$ 6,550.00 \$ 6,350.00 \$ 750.00	a a		Ħ	4		s		_			\$ 1,200.00
33 CATCH BASIN ADJUSTED TO GRADE. AS PER PLAN 655.00 \$ 6.550.00 \$ 6.550.00 \$ 6.350.00 \$ 750.00			EACH	2	-	49		_			\$ 4,800.00
	ı	3 CATCH BASIN ADJUSTED TO GRADE, AS PER PLAN	EACH	10		€9	_				\$ 7,500.00

TAYLOR ROAD (C.R. 26) FROM EUCLID AVE. TO THE EAST CLEVELAND SOUTH CORPORATION LIMIT

1	A S	DESCRIPTION	UNIT	ESTIMATED	DOLLARS I CT	S	Karvo Paving Company, Inc.	UNIT PRICE EST DOLLARS I C	EST.	Fabrizi Trucking & Paving Co., Inc.	UNIT PRICE EST.	C.A. Agresta Construction Co.	gresta ctlon Co.
SMANCHELL MONTHER LOCK AND TEACH AND CONTROL STATE AND CONTROL S	ļ.,		EACH	4	69	8	3,600.00	6	_			S	3,200.00
	es	7	EACH	2	69	-	2,800.00		-				4,800.00
	<u>س</u>	1	EACH	23	တ		9,200.00	€9					17,250.00
Section Sect	<u>س</u>	$\overline{}$	EACH	8	49	_	4,800.00	s				_	6,400,00
SECTION CONTINUE BANKED NAME PANAMENT NAME NAME NAME NAME NAME NAME NAME NAME	۳,	_	ㅂ	2,554	€	_	17,878.00		_			cs.	22,986.00
Patricular Comparison Patricular Comparison Patricular Control Comparison Patricular C	60		POUND	30,000	64	_	30,000.00	sý	_			_	3,000.00
PAMERINT	S	CTION TOTAL DRAINAGE:				47	83,628.00						74,396.00
PATRICA DEPTRICATION PROPRENTY REPARK AND PATRICATION AND PLOTAGE REPLACEMENTARY AND PATRICATION AND PATRICA	L												
Application Depth Application Teachers (Table And Pacification Teachers (Table And Table		PAVEMENT									•		
A CHAIL CREPHY HOUGH DAYBERT TRANSVALLAND PLENIER FEBLACENERY, A SQT 0 1750 S 0 1500 S 0 15	4	1	SQ YD	415	\$		12,450.00	\$				63	20,335.00
4.2 FILT DOWNSTER ENDARGE NORMONE FT 7,889 6 2,00 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 6 15,000 8 15,	4	$\overline{}$	SQ YD	1,250	s		56,250.00	69				s	82,500.00
44 ACCINCIPATE BASE, AS PERP LAAN CUVTO 220 5 14000 5 155000 5	4	_	ㅂ	7,668	ક		15,338.00	s	-			\$	19,170.00
	4	1	CU YD	280	49	_	39,200.00	(A)				4	39,200.00
4. 1. 1. 1. 1. 1. 1. 1.	4		CU YD	. 425	\$		27,200.00	\$				49	19,550.00
FUNCIONATION CONTINUED NET COUNTINUE NATION CONTINUE NATION CONTINUE NATION COUNTINUE NATION COUNTINUE NATION COUNTINUE NATION COUNTINUE NATION COUNTINUE NATION COUNTINUE NATION NATI	4	-	SQ YD	2,874	\$		169,566.00	#				\$	218,424.00
	4	_	GALLON	1,471	69		2,942.00	€9	_				4,413.00
ASPHALT CONCRETE INTERNALIDAY TE COUNSEL, TYPE 2, FORAS 2, A PERR ASPHALT CONCRETE INTERNALIDAY TE COUNSEL, TYPE 2, FORAS 2, 165.00 5 6, 626.00 5 6, 41.62.00 5 6, 4	4	_	GALLON	610	67		1,220.00	69	-			_	1,830.00
49 AMATTER NORMETTE SOURCETTE SURFACE COURSE, TYPE 1, POTD ZAM, AS ENR PLAN CUYD 424 \$ 15500 \$ 22,00 \$ 5,050.00 \$ 5,015.00 \$ 2000 \$ 5,015.00 \$ 2000 \$ 5,015.00 \$ 2000 \$ 5,000.00 </td <td>4</td> <td>1</td> <td>CU YD</td> <td>593</td> <td>es.</td> <td>_</td> <td>80,055.00</td> <td>€9</td> <td>_</td> <td></td> <td></td> <td>s</td> <td>78,276.00</td>	4	1	CU YD	593	es.	_	80,055.00	€9	_			s	78,276.00
90 97 FERINY CACED CONCYRETE PAVEMENT USING CLASS MS CONCYRETE, AS EACY D 118 \$ 8.0.0 \$ 6.138.00	4	_	CU YD	424	G	20	82,680.00	G	_			s	84,800.00
51 CHANGATERING COLONGETICE AND ALGENING CLASS MS CONONGETICE, SON ON ACTION CONONGETICE AND ALGENING CLASS MS CONONGETIC AND ALGENING CLASS MS CONONG	Γ,	_	SQ YD	118	es		6,136.00	છ		100		_	6,844.00
COUNCEPTE PAVEMBENT CLEANED AND FILLED FT 2,500 S 1,00 S 3,030,00 S 45,450,00 S 1,00 S 2,00		$\overline{}$	SQ YD	426	ь		15,782.00	49				\$	21,300.00
SS CONCIDENTE PAYEMENT JOINT CLEANED AND FILLED FT 7500 \$ 2,500.00 \$ 5,000.00 \$ 0.00 \$ <td>l_{rQ}</td> <td>$\overline{}$</td> <td>Ħ</td> <td>2,525</td> <td>8</td> <td></td> <td>30,300.00</td> <td>49</td> <td></td> <td></td> <td></td> <td>49</td> <td>31,562.50</td>	l _{rQ}	$\overline{}$	Ħ	2,525	8		30,300.00	49				49	31,562.50
SECTION TOTAL PAVEMENT CRACK CLEAVED AND FILLED FT 750 \$ 6.02.05 \$ 6.00.00 \$ 1.00 \$ 6.00.00 \$ 7.50.00 \$ 9.00	3		E	2,500	s	_	2,500.00	\$	_				2,000.00
WATERWORK FT 2780 \$ 642,197.00 \$ 64	Ω.,	4 CONCRETE PAVEMENT CRACK CLEANED AND FILLED	Ħ	750	49	_	900.009	\$	_				60D.0D
State Stat	SE	CTION TOTAL PAVEMENT:				45	542,197.00		*	200			30,804.50
Fig. Pipe RemoveD, 24" AND UNDER, Pipe RemoveD, 24" AND													
56 PIPE REMOVED, 24° AND UNDER FT 2780 \$ 3.50 \$ 1.00 \$ 2.780.00 \$ 0.30 0.30 \$ 0.30 \$ <td></td> <td>WATERWORK</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>-</td> <td></td> <td></td> <td></td> <td></td>		WATERWORK							-				
54 VALUE BOX, ADJUSTED TO GRADE FACH 5 \$ <	3	1	Æ	2780	69		9,730.00	\$					834.00
57 EVANTER WARD CONTILE GONINGE CONNICCTION STATEMENT AND BETAINED MEDIATION STATEMENT AND BETAINED MEDIATION STAND BETAINED MEDIATION STAND LAND EACH CONNICCTION STAND LAND EACH CONNICCTION STAND LAND EACH CONNICCTION STAND LAND EACH CONNICCTION STAND LAND LAND EACH CONNICCTION STAND LAND LAND EACH CONNICCTION STAND LAND LAND LAND EACH CONNICCTION STAND LAND LAND LAND LAND LAND LAND LAND L	(C)	$\overline{}$	EACH	5	69		250.00	s,	_				1,875.00
69 Power Remains Indicate Remains Number Lemman Decreased Remains Number Lemman Remains Number Remains Remman Remains Remains Remains Remains Remains Remains Remains Rema	<u>س</u>		. 14	25	S		5,250.00	s		3			4,561.25
59 8 GATE VALVE WITH VALVE BOX, COMPLETE EACH 8 1,100.00 \$ 1,200.00 \$ 7,200.00 \$ 1,200.00 \$ 7,200.00	_		FT	2755	s		256,215.00	. \$				64	244,506.25
60 G** CUTTINGE.IN VALVE BOX, COMPLETE EACH 3 4,590.00 \$ 15,900.00 <	_	$\overline{}$	EACH	9	4	_	6,600.00		_				7,500.00
61 FURNISHING AND SETTING STATING STANDANT ASSEMBLY, COMPLETE EACH 8 \$ 5,000.00 \$ \$ 40,000.00 \$ \$ 3,650.00 \$ \$ 4,600.00 \$ \$ 40,000.00 \$ \$ 40,000.00 \$ \$ 4,000.00 \$ \$ 4,000.00 \$ \$ 4,000.00 \$ \$ 1.00 \$ \$ 2.0			EACH	ဇ	€9	_	15,900.00		_				3,600.00
62 MINTERNANCE OF WATER SERVICE POLIND 2000 \$ 4,000.00 \$ 1.50 \$ 1.00 \$ 1.00 \$ 2.0 63 MAINTENANCE OF WATER SERVICE LUMP 1 \$ 64,600.00 \$ 64,600.00 \$ 64,600.00 \$ 67,000.00 \$ 7,4750.00 \$ 7,4750.00	_	т	EACH	8	8		40,000.00		-			s	34,752.00
63 MAINTENANCE OF WATER SERVICE LUMP 1 \$ 64,600.00 \$ 64,600.00 \$ 64,600.00 \$ 64,600.00 \$ 64,600.00 \$ 64,600.00 \$ 64,600.00 \$ 600.00 \$ 600.00 \$ 600.00 \$ 7,000.	_	$\overline{}$	POUND	2000	69		4,000.00	69					2,000.00
64 SHEETING LEFTING LE	\perp	$\overline{}$	LUMP	-		-	54,600.00						57,960.00
65 WATER WORK, MISC.: WATER SERVICE CONNECTION 2" AND UNDER, LONG EACH 13 \$ 2,100.00 \$ 27,300.00 \$ 35,100.00 \$ 35,100.00 \$ 1,575.00 \$ 35,100.00 \$ 1,500.00 \$ 35,100.00 \$ 1,575.00 \$ 35,100.00 \$ 1,575.00 \$ 35,175.00 \$ 1,575.00 \$ 35,175.00 \$	L	_	MBF	2	89	_	1,000.00	s					20.00
66 WATER WORK, MISC.: WATER SERVICE CONNECTION 3" AND UNDER, SHOR EACH 10 \$ 4,200.00 \$ 42,000.00 \$ 5,475.00 \$ 54,750.00 \$ 54,750.00 \$ 58,750.00 \$ 5095.00 \$			EACH	5	s	-	27,300.00					G	23,205.00
67 WATER WORK, MISC.: WATER SERVICE CONNECTION 3" AND LARGER, LONG EACH. 10 \$ 4,200.00 \$ 42,000.00 \$ 5,475.00 \$ 54,750.00 \$ 6,095.00 \$		$\overline{}$	EACH	38	s	_	46,800.00						64,701.00
			EACH	0	4		42,000.00		_	54,750.00			60,950.00

TAYLOR ROAD (C.R. 28) FROM EUCLID AVE. TO THE EAST CLEVELAND SOUTH CORPORATION LIMIT

A ON	DESCRIPTION	TIND	GUANTITY	DOLLARS I CTS		Karvo Paving Company, Inc.	UNIT PRICE EST. DOLLARS I CTS	Fabrizi Trucking & Paving Co., Inc.		UNIT PRICE EST. DOLLARS I CTS	Constru	C.A. Agrosta Construction Co.
8	WATER WORK, MISC.: WATER SERVICE CONNECTION 3" AND LARGER, SHOP	EACH		\$ 2,600.00	\$ 00	2,600.00	\$ 3,250.00	s	8	4,241.00	4	4,241.00.
8	$\overline{}$	EACH	8	\$ 70.00	ø 8	1,400.00	\$ 50.00	49	1,000.00 \$	21.00		420.00
8		EACH	99	\$ 40.00	\$	2,640.00	\$ 25.00	67	1,650.00 \$	8.50	₩	561.00
7	$\overline{}$	EACH	8	\$ 850,00	\$ 00	6,800.00	\$ 150.00	s	1,200.00 \$	21.00	\$	168.00
SE	SECTION TOTAL WATERWORK:				49	523,085.00		\$ 481,700.00	00.00		\$	511,854.50
Ш												
	TRAFFIC CONTROL			2								
72	CONDUIT, 2", 725.04	FT	69	\$ 12.50	\$ 09	862.50	\$ 13.40	€>	924.60 \$	12.50	69	862.50
2	CONDUIT, 3", 725.04	E	22	\$ 12.50	\$ 09	675.00	\$ 13.40	\$	723.60 \$	12.50	s	675.00
72	CONDUIT, JACKED OR DRILLED, 725.04, 3*	ㅌ	164	\$ 45.00	\$ 00	7,380.00	\$ 48,00		7,872.00 \$	45.00		7,380.00
75	TRENCH	L	123	\$ 7.00	\$ 00	861.00	\$ 7.50	76 \$	922.50 \$	7.00	\$	861.00
92	. PULLBOX, 725.06, 17%30"	EACH	4	\$ 750.00	\$ 00	3,000.00	\$ 805.00	s	3,220.00 \$	750.00	\$	3,000.00
4	PULLBOX, 725.06, 24'x36"	EACH		00'096 \$	\$ 00	950.00	\$ 1,050.00	€9	1,050.00 \$	950.00	\$	950.00
82	_	EACH	6	\$ 125.00	\$ 00	1,125.00	\$ 140.00	. \$	1,260.00 \$	125.00	69	1,125.00
182	PLASTIC CAUTION TAPE	E	123	\$ 0.75	75 \$	92.26	\$ 0.80	s	98.40 \$	97.D	69	92.25
8		EACH	. 4	\$ 295.00	\$ Q	1,180.00	\$ 315.00	\$ 1,26	1,260.00 \$	295.00	₩	1,180.00
20	SIGN, FLAT SHEET, AS PER PLAN	SQ FT	63	30,00	\$ 00	1,890.00	\$ 32.00	87	2,016.00 \$	30.00	69	1,890.00
8	VEHICULAR SIGNAL HEAD, 3-SECTION, 12" LENS, 1-WAY, AS PER PLAN	EACH	8	\$ 600.00	\$ 00	4,800.00	\$ 645.00	\$	5,160.00 \$	600.00	s	4,800.00
8	PEDESTRIAN SIGNAL HEAD, COUNTDOWN, TYPE D2, AS PER PLAN	EACH .	80	\$ 475.00	\$ 00	3,800.00	\$ 510.00	€9	4,080.00 \$	475.00	8	3,800.00
84	COVERING OF VEHICULAR SIGNAL HEAD	EACH	8	\$ 40.00	\$ 00	320.00	\$ 43.00	ક્ર	344.00 \$	40.00	9	320.00
85	COVERING OF PEDESTRIAN SIGNAL HEAD	EACH	8	\$. 40.00	\$ 00	320.00	\$ 43.00	69	344.00 \$	40.00	8	320.00
88	PEDESTRIAN PUSH BUTTON, AS PER PLAN	EACH	8	\$ 225.00	\$ 00	1,800.00	\$., 245.00	65	1,980.00 \$	225.00	₩.	1,800.00
87	DETECTOR LOOP, AS PER PLAN	EACH	ιο	\$ 800.00	\$	4,000.00	\$ 910.00	s	4,550.00 \$	850.00	8	4,250.00
8	SIGNAL CABLE, 5 CONDUCTOR, NO. 14 AWG	ㅂ	923	\$	1.75 \$	1,615.25	\$. 2.00	\$	1,846.00 \$	1.75	67	1,615.25
88	SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG	FT	732	\$ 2.3	2.25 \$	1,647.00	\$ 2.45	69	1,793.40 \$	2.25	47	1,647.00
8	SIGNAL SUPPORT FOUNDATION	EACH	က	\$ 3,200.00	\$ 00	00:009'6	\$ 3,425.00	4	10,275.00 \$	3,200.00	s	9,600.00
92	PEDESTAL FOUNDATION	EACH	9	00'009 \$	\$ 00	3,000.00	\$ 645.00	€	3,225.00 \$	00'009	8	3,000.00
88	LOOP DETECTOR LEAD-IN CABLE	Ħ	1288	\$ 1.	1.40 \$	1,803.20	\$ 1.46	\$	1,867.60 \$	1.35	S	1,738.80
ន	POWER CABLE, 3 CONDUCTOR, NO. 8, AWG	FT	49	\$ 4.	4.00	196.00	\$ 4,50		\$ 05.022	4.00	4	196.00
8	POWER SERVICE, AS PER PLAN	EACH	1	\$ 1,800.00	\$ 00	1,800.00	\$ 1,925.00	63	1,925.00 \$	1,800.00	4	1,800.00
ß	SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 4 POLE, WITH MAST ARMS TC-81.2	EACH	1 .	\$ 7,100.00	\$ 00	7,100.00	\$ 7,600.00	\$ 7,60	7,600.00 \$	7,100.00	\$	7,100.00
8	SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 1, AS PER PLAN	EACH	1	\$ 3,900.00	\$ 00	3,900.00	\$ 4,175.00	မာ	4,175.00 \$	3,900.00	€9	3,900.00
97	SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN	EACH	-	\$ 4,100.00	\$ 00	4,100.00	\$ 4,385.00	4	4,385.00 \$	4,100.00		4,100.00
86	PEDESTAL, 8', TRANSFORMER BASE, AS PER PLAN	EACH	5	\$ 575.00	\$ 00	2,875.00	\$ 615.00	\$	3,076.00 \$	675.00	49	2,875.00
8	REMOVAL OF TRAFFIC SIGNAL INSTALLATION, AS PER PLAN	EACH	1	\$ 1,500.00	\$ 00	1,500.00	\$ 1,600.00	\$	1,600.000 \$	1,500.00	s	1,500.00
9	D SIGNALIZATION, MISC.: FOUNDATION TEST HOLE	EACH	2	\$ 400.00	ه 8	800.00	\$ 430.00	€>	\$ 00.098	400.00	s	800.00
ē	1 CONTROLLER UNIT, TYPE TS2/A2, WITH CABINET, TYPE TS2, AS PER PLAN	EACH	1	\$ 14,400.00	\$ 00	14,400.00	\$ 15,400.00	65	15,400.00 \$	14,400.00	G	14,400.00
102	2 CABINET RISER	EACH	-	\$ 725.00	\$ 00	725.00	\$ 775.00	65	\$ 00.577	725.00	\$	725.00
103	3 CABINET FOUNDATION	EACH	1	\$ 2,100.00	\$ 00	2,100.00	\$ 2,250.00	69	2,250.00 \$	2,100.00	69.	2,100.00
Š	SECTION TOTAL TRAFFIC SIGNALS:				49	90,217.20		\$.97,057.60	7.60		S	90,402.80
L												

TAYLOR ROAD (C.R. 26) FROM EUCLID AVE. TO THE EAST CLEVELAND SOUTH CORPORATION LIMIT

P S	DESCRIPTION	TINO	QUANTITY	DOLLARS I CTS	-	raivo reving company, fnc.	DOLLARS LCTS	-	Paving Co., Inc.	DOLLARS I CTS	Constr	Construction Co.
								+				
					_				00 110			00 000
104	GROUND MOUNTED SUPPORT, NO. 3 POST	Ē	20	w		350.00	0	_	3/3.00		100	0,000
105 S	SIGN SUPPORT ASSEMBLY, POLE MOUNTED	EACH	89	· •	\$ 00.5	440.00		8	480.00	\$ 25.00		440.00
_	SIGN, FLAT SHEET	SQFT	62	S	12.00 \$	624.00	s	13.00 \$	676.00	\$ 12.00	s	624.00
$\overline{}$	REMOVAL OF GROUND MOUNTED SIGN AND DISPOSAL	EACH	2	es	15.00 \$	75.00	\$. 16.00	\$ 00	80.00	\$ 15.00	ss.	75.00
\neg	REMOVAL OF GROUND MOUNTED SIGN AND REERECTION	EACH	8	ဟ	\$ 00'99	165.00	\$ 60.00	\$ 00	180.00	\$ 55.00	s	165.00
$\overline{}$	REMOVAL OF GROUND MOUNTED POST SUPPORT AND DISPOSAL	· EACH	9	s	30.00 \$	80.00	\$ 33.00	\$ 00	99.00	\$ 30.00	S	90.00
	REMOVAL OF POLE MOUNTED SIGN AND DISPOSAL	EACH	80	s	15.00 \$	120.00	\$	16.00 \$	128.00	\$ 15.00	S	120.00
111	ENI dollar	Ŀ	140	\$	6.00 \$	840.00	s	6.50 \$	910.00	\$ 6.00	ca.	840.00
_	CROSSWALKLINE	E	999	49	3.00 \$	1,980.00	\$ 37	3.25 \$	2,145.00	\$ 3.00	4	1,980.00
	LANE ARROW	EACH	11	47	125.00 \$	1,375.00	\$ 135.00	\$ 00	1,485.00	\$. 125.00	4	1,375.00
	WORD ON PAVEMENT 72"	EACH	2	49	165.00 \$	310.00	\$ 165.00	\$ 00	330.00	\$ 155.00	\$	310,00
	ANE INF 4"	MILE	0.65		1,800.00	1,170.00	\$ 1,925.00	₽	1,251.25	\$ 1,800.00	.43	1,170.00
		MILE	09.0	\$	4,800.00 \$	2,880.00	\$ 5,135.00	\$ 00	3,081.00	\$ 4,800.00	\$	2,880.00
	CHANNELIZING LINE 8"	ᇤ	270	· ss	1.80 \$	486.00	\$ 2.00	\$ 00	540.00	\$ 1.80	\$	486.00
	WORK ZONE LANE LINE, CLASS III	MILE	0.65		\$ 00.009	390.00	. \$ 645.00	\$ 00	419.25	\$ 600.00	s	390.00
	WORK ZONE CENTERLINE, CLASS III	MILE	09.0	s	\$ 00.008	480.00	\$ 865.00	\$ 00	519.00	\$ 800.00	\$	480.00
8	WORK ZONE CHANNELIZING LINE CLASS III	F	270	S	0.80	216.00	s	\$ 06'0	243.00	\$ 0.80	•	216.00
	WORK ZONE STOP LINE, CLASS III	Ŀ	140	s	3.00	420.00		3.25. \$	455.00	\$ 3.00	&	420.00
	WORK ZONE CROSSWALK LINE, CLASS III	ㅂ	099	s	1.50 \$	990.00	\$	1.65 \$	1,089.00	\$ 1.50	49	990.00
	123 WORK ZONE ARROW, CLASS III	EACH	. 11.	s	\$ 00.09	550.00	\$ 55.00	s 00	605.00	\$ 50.00	64	. 550.00
124	124 WORK ZONE WORD ON PAVEMENT, 72", CLASS III	EACH	. 2	s	\$ 00.03	100.00	\$. 55.00	\$ 00	110.00	\$ 50.00	49	100.00
CTIC	SECTION TOTAL TRAFFIC CONTROL:				63	14,051.00		•	15,200.50		•	14,051.00
-								-				
<u> </u>	MAINTENANCE OF TRAFFIC							_				
125 1	TRAFFIC COMPACTED SURFACE, TYPE A OR B , AS PER PLAN	· CU YD	510	s	9.00 \$	4,590.00	49	44.00	22,440.00			12,750.00
126 2	2" ASPHALT CONCRETE WALK	SQFT	1600	69	1.00 \$	1,600.00	↔	2.00 \$	3,200.00	1.00	8	1,600.00
127 L	LAW ENFORCEMENT OFFICER WITH PATROL CAR FOR ASSISTANCE	HOUR	80	\$	\$ 00'59	5,200.00	65	\rightarrow	4,000.00			4,400.00
	DETOUR SIGNING, AS PER PLAN	LUMP	1	\$	3,500.00 \$	3,500.00	\$ 4,500.00	89 89	4,500.00	\$ 2,800.00	s	2,800.00
83	129 REPLACEMENT SIGN	EACH	15	89	\$ 00.57	1,125.00	S	\$ 00.38	1,275.00	\$ 75.00	ķ	1,125.00
8	130 ASPHALT CONCRETE FOR MAINTAINING TRAFFIC	CUYD	10	s	150.00 \$	1,500.00	s	8.	5,000.00	\$ 250.00		2,500.00
33	131 PORTABLE CHANGEABLE MESSAGE SIGN	SIGN MONTH	24	s	200.00	4,800.00	S	\$	20,520.00	\$ 800.00	ès	19,200.00
132	ROADS FOR MAINTAINING TRAFFIC, AS PER PLAN	LUMP	-	\$ 10	10,000.00	10,000.00	\$ 3,375.00	90.	3,375.00	\$ 1,000.00	s	1,000.00
	133 WATER	MGAL	50	\$	1.00 \$	50.00		10.00 \$	500.00	\$ 5.00	\$	250.00
134	CALCIUM CHLORIDE	NOT	2	8	1.00 \$	5.00	\$ 250.00	.00	1,250.00	\$ 10:00	s C	20.00
ECTIC	SECTION TOTAL MAINTENANCE OF TRAFFIC:				4	32,370.00		s	66,060.00		45	45,675.00
	C. Marine .							-			ŀ	
	MISCELLANEOUS							1	000000	OC COL OF		20 200 00
-		•	•	-	50 50 50 50	E2 500 00	,	_				3

TAYLOR ROAD (C.R. 26) FROM EUCLID AVE. TO THE EAST CLEVELAND SOUTH CORPORATION LIMIT

E C	DESCRIPTION	UNIT	ESTIMATED	DOLLARS LOTS	Karvo Paving Company.	OOLI ARS LOTS	Fabrizi Trucking & Paving Co. Inc.	DOLLARS I CTS	Construction Co.
38	136 FIELD OFFICE, TYPE B, AS PER PLAN	MONTH	9	\$ 1,600.00			000000		100
137	137 CONSTRUCTION LAYOUT STAKES, AS PER PLAN	LUMP	-	\$ 1,000.00	\$ 1,000.00	\$ 8,000.00	\$ 8,000.00	\$ 4,000.00	\$ 4,000.00
88	138 MOBILIZATION	- LUMP	-	\$ 65,000.00	\$ 65,000.00	\$ 15,000.00	\$ 15,000.00	\$ 30,000,00	\$ 30,000.00
8	139 PRECONSTRUCTION PHOTOGRAPHIC PROJECT DOCUMENTATION (SEE PRO	LUMP	-	\$ 1,500.00	\$ 1,500.00 \$	\$ 1,500.00	\$ 1,500.00	\$ 1,500.00	\$ 1,500.00
5	SECTION TOTAL MISCELLANEOUS:				\$ 129,600.00		\$ 52,400.00		\$ 117,200.00
	GRAN	ID TOTAL	GRAND TOTAL ESTIMATED		\$ 1,547,652.00		\$ 1,614,987.90		\$ 1,615,935.25

TAYLOR ROAD (C.R. 26) FROM EUCLID AVE. TO THE EAST CLEVELAND SOUTH CORFORATION LIMIT

REF NO.	DESCRIPTION	UNIT	QUANTITY	DOLLARS I CTS	CIS.	Perk Company, Inc.	DOLLARS I CTS	CIS	8	Company, Inc.
	ROADWAY						٠			
-	CLEARING AND GRUBBING	LUMP	1	5	50.00	50.00	s	_	\$	1,259.60
7	PAVEMENT REMOVED	SQ YD	3418		12.50 \$				4	31,103.80
60	WEARING COURSE REMOVED	SQ YD	9724	s	2.25 \$		s ₂		4	28,199.60
4	BRICK BASE REMOVED, AS PER PLAN	SQ YD	2500	S	8.50 \$	21,250.00		3.30	8	8,250.00
2	WALK REMOVED	SQFT	3274	s	1.50 \$	4,911.00	₩	0.85	49	2,782.90
8	CURB REMOVED	ե	2525	G	2.50 \$	6,312.50	s	1.00	\$	2,525.00
1	PIPE REMOVED, 24" AND UNDER	ᆫ	08	49	20.00 \$	1,600.00	s	5.05	5	404.00
8	MANHOLE REMOVED	EACH	2	69	350.00 \$	00:002	\$	201.55	\$	403,10
6	CATCH BASIN REMOVED	EACH	2		350.00 \$	700.00		201.55	s	403.10
5	EXCAVATION	CU YD	9	€9	35.00 \$	3,185.00	\$	49.95	. 8	4,545.45
=	SUBGRADE COMPACTION	SQYD	3026	8	1.50 \$	4,539.00	s	0.50	s	1,513.00
12	LINEAR GRADING, AS PER PLAN	SQ YD	1310		6.75	8,842.50	s	9.10	s	11,921.00
5	CUYAHOGA COUNTY CENTERLINE MONUMENT BOX ASSEMBLY, TYPE 1	EACH	т		\$ 00.039	1,650.00	es.	683,45	es.	2,050.35
4	CUYAHOGA COUNTY MONUMENT BOX	EACH	က	₩ ₩	\$ 00.035		\$	513.10	s	1,539.30
5	MONUMENT BOX ADJUSTED TO GRADE, AS PER PLAN	EACH	10	S es	\$ 00.055	5,500.00	49	209.90	S	2,099.00
16	RIGHT OF WAY MONUMENT	EACH	-	€.	\$ [00.038	550.00	\$	314.90	ca.	314.90
17	MONUMENT REFERENCING	EACH	٠ 1		225.00 \$	2,250.00	67	314.90	47	3,149.00
48	4" CONCRETE WALK, AS PER PLAN	SOFT	3313	s	5.50 \$		G	4.30	49	14,245.90
9	CURB RAMP, TYPE A1, AS PER PLAN	EACH	3	88	\$ 00.038		S		. 69	4,877.65
ន	CURB RAMP, TYPE A2, AS PER PLAN	EACH	4	. \$	\$ 00.058	3,400.00	ıs		65	5,525.80
2	CURB RAMP, TYPE B2, AS PER PLAN	EACH	.5	88	\$50.00 \$		69	1,932.85	S	9,664.25
ECT	SECTION TOTAL ROADWAY:				57	156,715.50				136,776.60
					1			·		
	EROSION CONTROL									
8	SOIL ANALYSIS TEST	EACH	2	s			4		S	251.90
ន	TOPSOIL	CU YD	22		35.00 \$			700	S	4,599.00
24	SEEDING AND MULCHING, AS PER PLAN	SQYD	1,310	49	1.50 \$	1	47	_	s	2,489.00
32	COMMERCIAL FERTILIZER	TON	0.20	σ ••	\$ 00.038				63	377.87
8	ГІМЕ	ACRE	0.27	\$	450.00 \$		\$		49	153.04
27	WATER	MGAL	. 7		30.00		s	-	4	441.00
88	EROSION CONTROL	EACH	7,500	s	1.00 \$	7,500.00	B	8.		7,500.00
Ĕ	SECTION TOTAL EROSION CONTROL:				44	12,641.50			40	15,811.81
67	DRAINAGE									
8	6" CONDUIT, TYPE F FOR UNDERDRAIN OUTLETS, AS PER PLAN	FT	140	ક	9.50	1,330.00	¢\$		67	1,183.00
ន of	12" CONDUIT, TYPE B, AS PER PLAN	F	40	s 1	125.00 \$		s	72.95	69	2,918.00
3	12" CONDUIT, TYPE C, AS PER PLAN	E	. 40	s	95.00		so.	_	69	1,558.00
8 03	CATCH BASIN, AS PER PLAN	EACH	2	\$ 2,7	2,725.00 \$		\$ 2		S	4,472.80
8	CATCH BASIN AD II ISTED TO GRADE AS PER PLAN	EACH	9	5	550.00	5,500.00		368 55	v:	3,685,50

Š	DESCRIPTION	LIND	QUANTITY	DOLLAR	DOLLARS I CTS	Perk Company, Inc.		DOLLARS I CTS	ĕ	Company, Inc.
34	CATCH BASIN RECONSTRUCTED TO GRADE, AS PER PLAN	EACH	4	69	1,100.00	\$ 4,400.00		824.40	673	3,297.60
38		EACH	2	69	4,300.00	\$ 8,600.00	\$ 0	5,425.55	69	10,851.10
98	MANHOLE ADJUSTED TO GRADE, AS PER PLAN	EACH	23	49	550.00	\$ 12,650.00	\$ Q	318.10	G	7,316.30
37	MANHOLE RECONSTRUCTED TO GRADE, AS PER PLAN	EACH	8	69	1,100.00	\$ 8,800.00	\$ 0.	597.05	es-	4,776.40
38	6" BASE PIPE UNDERDRAINS WITH FABRIC WRAP, AS PER PLAN	Ħ	2,554	69	09'6	\$ 24,263.00	\$ 0	8,50	69	21,709.00
စ္တ	MISCELLANEOUS METAL	POUND	30,000	S	1.00	\$ 30,000.00	\$ 0	1.05	64	31,500.00
S	SECTION TOTAL DRAINAGE:					\$ 109,793.00	0		40>	93,267.70
	DANGARINE									
1		1			3		-	1		
₹ ;	PARTIAL DEPTH PAVEMENT REPAIR, AS PER PLAN	2 8	415	69 G	49.00		-	62.70	69 (26,020.50
₹	סרט סיי איי	SQ YD	1,250	<i>a</i>	98.00		-	60.95	0	/6,187.50
42	\neg	ᇤ	2,668	49	2.65	\$ 20,320.20	Ф	2.90	s	22,237.20
43	ASPHALT CONCRETE BASE, PG64-22, AS PER PLAN	CUYD	280	63	140.00	\$ 39,200.00	\$	179.20	s	50,176.00
44	AGGREGATE BASE, AS PER PLAN	CU YD	425	69	37.00	\$ 15,725.00	. \$ 0	54.05	49	22,971.25
45	14" CONCRETE BASE, AS PER PLAN	SQ YD	2,874	69	62.50	\$ 179,625.00	\$ 0	75.75	69	217,705.50
46	TACK COAT	GALLON	1,471	· ·	3.00	\$ 4,413.00	\$ 0	3.85	s	5,663.35
47		GALLON	610	G	3.00	\$ 1,830.00	. \$ 0	3.85	4	2,348.50
8		CU YID	593	s	140.00	\$ 83,020.00	\$ 0	179.20	47	106,265.60
\$		CU YD	424	4	208.00	\$ 88,192.00	\$ 0	266.25	47	112,890.00
ន	8" REINFORCED CONCRETE PAVEMENT USIN	SQ YD	118	49	61.00	\$ 7,198.00	\$ 0	67.35	₩	7,947.30
5	6" NON-REINFORCED CONCRETE PAVEMENT USING CLASS MS CONCRETE,	SQ YD	426	4	53.00	\$. 22,578.00	\$ 0	45.50	€9	19,383.00
-25	CURB, TYPE 6 USING CLASS MS CONCRETE, AS PER PLAN	FF	2,525	49	19.40	\$ 48,985,00	\$ 0	14.00		35,350.00
53	CONCRETE PAVEMENT JOINT CLEANED AND FILLED	E	2,500	69	3.00	\$ 7,500.00	\$ 0	2.50	69	6,250.00
54	CONCRETE PAVEMENT CRACK CLEANED AND FILLED	FT	750	好	3.00	\$ 2,250.00	\$ 0	2.50	s	1,875.00
SEC	SECTION TOTAL PAVEMENT:					\$ 623,671.20	,		47	713,270.70
iii										
	WATERWORK					*		10 10 10 10 10 10 10 10 10 10 10 10 10 1		
25		ь	2780	v,	1.75	\$ 4,865.00	s o	0.60	4	1,668.00
56	VALVE BOX, ADJUSTED TO GRADE	EACH	ഗ	G	550,00	\$ 2,750.00	\$ 0	235.60	49	1,178.00
57	G WATER MAIN DUCTILE IRON PIPE WITH PUSH-ON JOINTS AND RETAINED MECHANICAL JOINT FITTINGS, ANDI CLASS 62	F	52	↔	75.00	\$ 1,875.00	#	187.40	€9	4,685.00
58	G"WATER MAN DUCTLE IRON PPEWITH PUSH-ON JOINTS AND RETAINED MECHANICAL JOINT FITTINGS, ANSI CLASS 52	FT	2755	•	71.50	\$ 196,982.50	\$ 0	30.05	€\$	248,087.75
29		EACH	8	s	1,125.00	\$ 6,750.00	. \$ 0	1,137.60	69	6,825.60
8	6" CUTTING-IN VALVE WITH VALVE BOX, COMPLETE	EACH	3	\$	3,350.00	\$ 10,050.00	\$ 0	1,133.50	63	3,400.50
64	FURNISHING AND SETTING 6" HYDRANT ASSEMBLY, COMPLETE	EACH	8	4	4,200.00	\$ 33,600.00	\$ 0	4,462.90	S	35,703.20
62	MISCELLANEOUS METAL WORK	POUND	2000	67	1.25	\$ 2,500.00	\$ 0	2.50	s	5,000.00
8	MAINTENANCE OF WATER SERVICE	LUMP	-	. 69	46,300.00	\$ 46,300.00	s	46,752.95	es.	46,752.95
O .	SHEETING LEFT IN PLACE	MBF	2	69	1,100.00	\$ 2,200.00	\$ 0	10.00	49	20.00
65	WATER WORK, MISC .: WATER SERVICE CONNECTION 2" AND UNDER, LONG	EACH	13	49	_	\$ 33,800.00	\$	-	4	23,838.75
8	WATER WORK, MISC .: WATER SERVICE CONNECTION 2" AND UNDER, SHOR	EACH	38	49	1,625.00	\$ 63,375.00	8	1,283.00	69	50,037.00
67	PINO E DECOR - CINO SE NOTIFICIAL CONTROL CONTROL SE VICTOR CONTROL CO	TOUL	4		200000	W W 63	•	10000	•	

REPORT NO. 03-13

2	NO. DESCRIPTION	LIND	QUANTITY	DOLLARS I CTS	. 92	Perk Company, Inc.	DOLLARS I CTS		Company, Inc.
8	8 WATER WORK, MISC.: WATER SERVICE CONNECTION 3" AND LARGER, SHOP	EACH	-	\$ 4,100.00	\$ 00	4,100.00	2 3	40	7,439.40
68	9 WATER WORK, MISC.: REMOVE EXISTING VALVE BOX	EACH	20	\$ 75	75.00, \$	1,500.00	\$ 21.	21.25 \$	425,00
2	WATER WORK, MISC.: REMOVE EXISTING	EACH	99	\$ 30	30.00	1,980.00	89	8.50 \$	561.00
۲	WATER WORK, MISC .: REMOVE HYDRANT ASSEMBLY INCLUDING TEE AND B	EACH	8	\$ 175	175.00 \$	1,400,00	\$ 21	21.25 \$	170.00
SE	SECTION TOTAL WATERWORK:				*	467,027.50		S	529,220.65
	Control of Control							+	
								_	
22	\neg	ᄩ	89			959.10	8	_	941.85
2	3 CONDUIT, 3", 725,04	ቴ	54	\$.22	.22.00 \$	1,188.00	\$ 21.	21.90 \$	1,182.60
74	CONDUIT, JACKED OR DRILLED, 725.04, 3"	ы	164	\$ 45.	45.00 \$	7,380.00	\$ 45.95	95 \$	7,535.80
75	5 TRENCH	E	123	2 \$	\$ 00.7	861.00	s	6.10 \$	750.30
92	S PULLBOX, 725.08, 17'x30"	EACH	7	\$ 600.00	\$ 00:	2,400.00	\$ 553.05	\$ 90	2,212.20
12	7 PULLBOX, 725.06, 24"x36"	EACH	1 .	\$ 1,050.00	\$ 00	1,050.00	\$ 1,017.00	\$ 00	1,017.00
78	B GROUND ROD	EACH	65	\$ 120.00	\$ 00:	1,080.00	\$ 116.30	\$ 08	1,046.70
62	9 PLASTIC CAUTION TAPE	Ħ	123	· O	\$ 96.0	43.05	\$	0.35 \$	43.05
ន	SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN	EACH	4	\$ 295.00	\$ 00	1,180.00	\$ 371.60	\$ 09	1,486.40
8	SIGN, FLAT SHEET, AS PER PLAN	SQFT	8	0E \$	30.00	1,890.00	\$ 37.	37.80 \$	2,381.40
82	VEHICULAR SIGNAL HEAD, 3-SECTION, 12" LENS, 1-WAY, AS PER PLAN	EACH	8	\$ 775.00	\$ 00:	6,200.00	\$ 758.60	\$ 09	6,068.80
83	PEDESTRIAN SIGNAL HEAD, COUNTDOWN, TYPE D2, AS PER PLAN	EACH	ω.	\$ 390.00	\$ 00	3,120.00	\$ 381.95	95 \$	3,055.60
84	COVERING OF VEHICULAR SIGNAL HEAD	EACH	8	\$ 24.	24.00 \$	192.00	\$ 22.85	\$ 3	182.80
88	5 COVERING OF PEDESTRIAN SIGNAL HEAD	EACH	80	\$ 16.	16.00 \$	120.00	\$ 19.70	20	157.60
88	S PEDESTRIAN PUSH BUTTON, AS PER PLAN	EACH	8	\$. 200.00	\$ 00	1,600.00	\$ 188.45	45	1,507.60
87	DETECTOR LOOP, AS PER PLAN	EACH	5	\$ 800.00	\$ 00	4,000.00	\$ 782.20	\$ 02	3,911.00
88		FT	923	\$. 1.	1.65 \$	1,430.65	4	1.50 \$	1,384.50
88	9 SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG	Ħ	732	\$.1	1.90 \$	1,390.80	\$ 1.1	1.85	1,354.20
8	SIGNAL SUPPORT FOUNDATION	EACH	3	\$ 2,600.00	\$ 00.	7,800.00	\$ 2,513.60	\$	7,540.80
9	PEDESTAL FOUNDATION .	EACH	. 5	\$ \$00	\$ 00.003	2,500.00	\$ 493.50	\$ 09	2,467.50
85	LOOP DETECTOR LEAD-IN CABLE	FT	1288	\$ 1	1.40 \$	1,803.20	\$ 1.	1.35 \$	1,738.80
8	POWER CABLE, 3 CONDUCTOR, NO. 6, AWG	FT	49	\$ 10	10.00	490.00	\$ 11.15	15 \$	546.35
8	POWER SERVICE, AS PER PLAN	EACH	1	\$ 1,085.00	\$	1,085.00	\$ 1,064.80	80 \$	1,064.80
8	5 SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 4 POLE, WITH MAST ARMS TC-81.2	EACH	-	\$ 7,860.00	8	7,860.00	\$ 7,711.00	\$ 00	7,711.00
96 Pa	SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 1, AS PER PLAN	EACH	1	\$ 4,300.00	\$ 00.	4,300.00	\$ 4,217.10	10 \$	4,217.10
	SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN	EACH	١ ، ١	\$ 4,725.00	.00 \$	4,725.00	\$ 4,626.50	\$ 05	4,626.50
8	B PEDESTAL, 8', TRANSFORMER BASE, AS PER PLAN	· EACH	5	\$ 465	465.00 \$	2,325.00	\$ 452.00	\$ 00	2,280.00
ន 69	REMOVAL OF TRAFFIC SIGNAL INSTALLATION, AS PER PLAN	EACH	1	\$ 1,200.00	\$ 00:	1,200.00	\$ 1,260.45	45 \$	1,260.45
5	IO SIGNALIZATION, MISC.: FOUNDATION TEST HOLE	EACH	2	\$ 450.00	8. 8	900.00	\$ 441.56	S5 S	883.10
of ₽	11 CONTROLLER UNIT, TYPE TS2/A2, WITH CABINET, TYPE TS2, AS PER PLAN	EACH	-	\$ 15,780.00	₩	15,780.00	\$ 16	\$ 09	16,058.60
102	2 CABINET RISER	EACH	-	\$ 755	\$ 00.557	756.00	64	۶۶ ده	740.70
ë 0 :	103 CABINET FOUNDATION	EACH	7	\$ 1,150.00	\$ 00·	1,150.00	\$ 1,039.85	₩	1,039.85
ij	RECTION TOTAL TRAFFIC SIGNALS:		•		43	88,757.80		49	88.374.95

STATE STAT	NO. DESCRIPTION	TIND	QUANTITY	DOLLARS / CTS	CTS	Perk Company, Inc.	DOLLARS CTS		Company, Inc.
NAME Park									
OST FT 60 \$ TO \$ <td></td> <td></td> <td></td> <td></td> <td>_</td> <td></td> <td></td> <td></td> <td></td>					_				
NTED BACH B S 650 \$ 4400 \$ 693 \$ AND DISPOSAL EACH 62 5 15.00 5 16.10	4 GROUND MOUNTED SUPPORT, NO. 3 POST	Ħ	90	€9					440.00
SOFT SO SO SO SO SO SO SO S		EACH	80	€9			\$		554.40
AMD DISPOSALL EACH 6 9 1500 8 7500 8 1500<	6 SIGN, FLAT SHEET	SQFT	29	S			\$		785.20
Table Determendation Earch 3 5 56.00 5 165.00 5 165.00 5 175.	7 REMOVAL OF GROUND MOUNTED SIGN AND DISPOSAL	EACH	တ	S			4		94.50
TSUPPORTAND DISPOSAL	B REMOVAL OF GROUND MOUNTED SIGN AND REERECTION	FACH	ဇ	4			s		207.90
PT		EACH	€.	4			s		113.40
FT 140 \$ 0.00	REMOVAL OF POLE MOUNTED SIGN AND DISPOSAL	EACH	8	G			. 69		151.20
FT 860 \$ 1,080 \$ 1,080 \$ 3.80 \$ 3.80 \$ 8	1 STOP LINE	E.	140	69			\$		1,057.00
SSIII		ㅂ	099	1/2	-	-	s	-	2,508.00
SACH 2 5 155.00 5 310.00 5 3		ЕАСН	11	65			€9		1,731.95
MILE 0.65 \$ 1,100.00 \$ 2,280.00 \$ 2,000.00 \$	WORD ON PAVEMENT, 72"	. EACH	. 2				\$	-	390.50
MILE 0.60 \$ 4,800.00 \$ 2,880.00 \$ 5,046.00 \$	CANELINE, 4"	MILE	0,65		-		s	_	1,473.71
FT 270	CENTERLINE	MILE	09'0		-		*		3,627.60
SSIII FT 270 \$ 600.00 \$ 756.76 \$ 775.76 \$ 850.00 \$ 140.00 \$ 775.76 \$ 850.00 \$ 140.00 \$ 1,007.65 \$ 850.00 \$ 1,007.65 \$ 850.00 \$ 1,007.65 \$ 850.00 \$ 1,007.65 \$ 850.00 \$ 1,007.65 \$ 850.00 \$ 1,007.65 \$ 850.00 \$ 1,007.65 \$ 850.00 \$ 1,007.65 \$ 850.00 \$ 1,007.65 \$ 1,007.	CHANNELIZING LINE, 8"	F	270	69			4	_	607.50
SSIII FT 270 \$ 800.00 \$ 480.00 \$ 1,007.65 \$ 8.83 SSIII FT 270 \$ 800.00 \$ 216.00 \$ 1,007.65 \$ 8.83 SSIII FT 270 \$ 8 00.00 \$ 216.00 \$ 1.00 \$ \$ 1.00 \$ \$ 8.83 SSIII FT 140 \$ 1.00 \$ 1.00 \$ 1.00 \$ 1.00 \$ 1.00 \$ 1.00		· MILE ·	0.65				\$		491.24
SSIII FT 140 \$ 0.80 \$ 216.00 \$ 1.00 \$ 2.00 \$ 5 1	WORK ZONE CENTERLINE, CLASS III	MILE	0.60				69		604.59
FIT 140 \$ 3.00 \$ 420.00 \$ 3.80 \$ SIII FT 660 S 1.50 \$ 50.00 S 1.90 \$ 1.90		Н.	. 0/2	\$	-	X	\$	_	270.00
SIII FT 660 S 1.50 S 590.00 S 1.90 S 1.10		F	140	63			₩		532.00
FACH 11 \$ 50.00 \$ 550.00 \$ 63.		Ħ	099	s	_		s		1,254.00
CALASS III EACH 2 \$ 50.00 \$ 14061.00 \$ 68.00 \$ 14061.00 E OF TRAFFIC CUYD \$10 \$ 14,061.00 \$ 17,860.00 \$ 47710 \$ 1 A OR B, AS PER PLAN CUYD \$10 \$ 15,860.00 \$ 2.65	WORK ZONE ARROW, CLASS III	EACH	11	9			. ·	-	693.00
EOF TRAFFIC A.ORB, AS PERPLAN C.UYD SQ.FT LUMP CLUMP	WORK ZONE WORD ON PAVEMENT, 72", CLAS	EACH	. 2	49	-		· ·	-	126.00
E OF TRAFFIC ATROL CAR FOR ASSISTANCE HOUR CU YD SQ FT (600 \$ \$ 17,850.00 \$ \$ 47.10 \$ \$ 5.50 \$ \$ 17,850.00 \$ \$ 47.10 \$ \$ 5 5 5 \$ \$ 5 5 5 \$ \$ 5 5 5 \$ \$ 5 5 5 \$ \$ 5 5 5 \$ \$ 5 5 5 \$ \$ 5 5 5 5 \$ \$ 5 5 5 5 \$ \$ 5 5 5 5 \$ \$ 5 5 5 5 5 \$ \$ 5	TION TOTAL TRAFFIC CONTROL:							cs.	17,713.69
E OF TRAFFIC AOR B, AS PER PLAN CU YD SQ FT (600 \$ 35.00 \$ 17,850.00 \$ 47.10 \$ 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5									
ATROL CAR FOR ASSISTANCE HOUR SQ FT 1600 \$ 55.00 \$ 17,850.00 \$ 47.10 \$ 5 75.00 \$ 17,850.00 \$ 47.10 \$ 5 75.00 \$ 1 7,850.00 \$ 1 75.00 \$ 2.65 \$ 1 75.00 \$ 1 75.00 \$ 2.65 \$ 1 75.00 \$ 2.800.00	MAINTENANCE OF TRAFFIC								
ATROL CAR FOR ASSISTANCE HOUR 80 \$ 55.00 \$ 160.00 \$ 2.65 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	TRAFFIC COMPACTED SURFACE, TYPE A OR B, AS PER PLAN	CU YD	510	4		T	₩		24,021.00
ATROL CAR FOR ASSISTANCE HOUR 80 \$ 4,400.00 \$ 75.60 \$ 75.00 \$ 75.00 \$ 75.00 \$ 75.00 \$ 75.00 \$ 75.00 \$ 75.00 \$ 75.00 \$ 700.00 \$ 1,05.00 \$ 1,05.00 \$ <t< td=""><td></td><td>SQFT</td><td>1600</td><td>67</td><td>_</td><td></td><td>69</td><td></td><td>4,240.00</td></t<>		SQFT	1600	67	_		69		4,240.00
LUMP 1 5 2,800.00 5 4,030.65 5 5 EACH 15 5 75.00 5 1,125.00 5 1,25.95 5 GTRAFFIC CUYD 10 5 350.00 5 1,25.95 5 5 GN MONTH 24 5 800.00 5 1,25.05 5 1,060.80 5 PER PLAN LUMP 1 5 50.00 5 1,060.80 5 1,168.75 5 FR PLAN MGAL 50 5 10,00 5 1,168.75 5 TON 5 5 150.00 5 750.00 5 1,060.80 5 ANEOUS NEOUS NEO	LAW ENFORCEMENT OFFICER WITH PATROL		80	69	-		s	_	6,048.00
GTRAFFIC CU YD 15 \$ 75.00 \$ 1,125.00 \$ 125.95 \$ GN CU YD 10 \$ 350.00 \$ 1,99.86 \$ \$ 1,99.86 \$ \$ \$ 1,99.86 \$ \$ \$ \$ 1,99.86 \$ </td <td>DETOUR SIGNING, AS PER PLAN</td> <td>LUMP</td> <td>-</td> <td></td> <td>-</td> <td></td> <td>s</td> <td></td> <td>4,030.65</td>	DETOUR SIGNING, AS PER PLAN	LUMP	-		-		s		4,030.65
GTRAFFIC CU YD 10 \$ 35,00.00 \$ 199.86 \$ GN SIGN MONTH 24 \$ 800.00 \$ 1,060.80 \$ 1,060.80 \$ 2 1,060.80 \$ 1,060.80 \$ 1,060.80 \$ 1,168.75 \$ \$ 1,060.80 \$ 1,168.75 \$ \$ \$ 1,060.80 \$ 1,168.75 \$ \$ \$ \$ 1,168.75 \$	REPLACEMENT SIGN	EACH	15	s			. \$		1,889.25
GN SIGN MONTH 24 \$ 800.00 \$ 19,200.00 \$ 1,080.80 \$ 2 PER PLAN LUMP 1 \$ 50.00 \$ 1,148.75 \$ \$ MGAL 50 \$ 10,00 \$ 600.00 \$ 1,148.75 \$ TON 5 \$ 160.00 \$ 750.00 \$ 610.15 \$ ANEOUS ANEOUS 8 160,335.00 \$ 610.15 \$ 7	ASPHALT CONCRETE FOR MAINTAINING TRAFFIC	CU YD	10	s,			43		1,998.50
PER PLAN LLMP 1 \$ 50.00 \$ 50.00 \$ 1,188.75 \$ 1,188.75 \$ 23.25 \$ 23.25 \$ 23.25 \$ 20.00 \$ 23.25 \$ 20.00	PORTABLE CHANGEABLE MESSAGE SIGN	SIGN MONTH	24	s			69	$\overline{}$	25,459.20
MGAL 50 \$ 10.00 \$ 500.00 \$ 23.25 \$ \$ 10.00 \$ 10.00 \$ 23.25 \$ \$ 10.00 \$ 10.00 \$ 23.25 \$ \$ 10.00	ROADS FOR MAINTAINING TRAFFIC, AS PER P	LUMP	,	¢3			\$	_	1,168.75
ANEOUS TON 5 \$ 150.00 \$ 750.00 \$ 610.15 \$ 7	WATER	MGAL	20	ક			4		1,162.50
ANEOUS \$ 60,335.00 \$	CALCIUM CHLORIDE	NOT	ĸ	64			69	_	3,050.75
	TION TOTAL MAINTENANCE OF TRAFFIC:							₩	73,068.60
					_			_	

REPORT NO. 03-13

	Ħ
	ĕ
	₹
	ğ
	SA
	Ĭ
	8
	AND SOUTH
	Š
	핒
	5
	젌
	TO THE EA
	뿓
26)	9
ď	Ę
0	§
8	븠
RR	ĭ
5	S
¥	FR

REF DESCRIPTION	TINO	COANTITY	DOLLARS I CTS	Perk Company, Inc.	DOLLARS I CTS	Temaco Construction · Company, Inc.
136 FIELD OFFICE, TYPE B, AS PER PLAN	MONTH	ဖ	\$ 1,900.00	\$ 11,400.00	\$ 2,394,35	\$ 14,366.10
137 CONSTRUCTION LAYOUT STAKES, AS PER PLAN	LUMP	-	\$ 00.005,9	\$ 00.002,000.00	\$ 2,141.30	\$ 2,141.30
138 MOBILIZATION	LUMP	-	\$ 36,000.00	\$ 36,000.00	\$ 3,784.55	\$ 3,764,55
139 PRECONSTRUCTION PHOTOGRAPHIC PROJECT DOCUMENTATION (SEE PRO	LUMP	1	\$ 1,500.00	\$ 1,500.00	\$ 801.10	\$ 801.10
SECTION TOTAL MISCELLANEOUS:				\$. 138,539.00	•	\$ 49,413.70
GRANE	TOTAL E	GRAND TOTAL ESTIMATED		\$ 1,661,531.50		\$ 1,716,918.40

_	Š	DESCRIPTION	UNIT	QUANTITY	DOLLARS I CTS	. Ltd.	DOLLARS I CTS	CTS	Construction Co.
		ROADWAY							
_	-	CLEARING AND GRUBBING	LUMP	-	\$ 11,500.00	00.005,11,500.00	\$ 00.00	5,000.00 \$	5,000.00
	2	PAVEMENT REMOVED	SQ YD	3418	\$ 10.00	\$	\$ 00.00	14.00 \$	
_	60	WEARING COURSE REMOVED	SQ YD	9724	\$ 3.50	34,034.00	4.00 \$	3.00 \$	
-	4	BRICK BASE REMOVED, AS PER PLAN	SQ YD	2500	\$ 5.50	30 \$ 13,750.00	\$ 00.00	8 00'9	15,000.00
	Ð	WALK REMOVED	SOF	3274	\$ 0.90	မ	2,946.60 \$	2.00 \$	6,548.00
_	9	CURB REMOVED	ᇤ	2525	\$ 2.45	\$	6,186.25 \$	3.00 \$	
_	^	PIPE REMOVED, 24" AND UNDER	F	08	\$ 20.00	\$	1,600.00	15.00 \$	1,200.00
-	۵	MANHOLE REMOVED	EACH	2	\$ 450.00	4	\$ 00.006	500.00	1,000:00
_	6	CATCH BASIN REMOVED	EACH	2	\$ 400.00	\$	\$ 00.008	\$ 00.003	1,000.00
	9	EXCAVATION	CUYD	9	\$ 20.00	69	1,820.00 \$	27.00 \$	2,457.00
-	F		SQYD	3028	\$ 2.15	€	6,505.90 \$	1.00 \$	3,026.00
	12	-	SQ YD	1310	00'9 . \$	69	\$ 00:098'2	2,00 \$	2,620.00
_	13		EACH	က	\$ 650.00	. +5	1,950.00 \$	\$ 00.009	1,800.00
_	4	CUYAHOGA COUNTY MONUMENT BOX	EACH		\$ 650.00	₩.	1,950.00 \$	\$ 00:009	1,800.00
	15	MONUMENT BOX ADJUSTED TO GRADE, AS PER PLAN	EACH	10	\$ 225.00	S	2,250.00 \$	450.00 \$	4,500:00
J	16	RIGHT OF WAY MONUMENT	EACH	1	\$ 500,000	s	\$ 00.003	\$ 00.005	600.00
	17	MONUMENT REFERENCING	EACH	04	\$ 300.00	G	3,000.00 \$. 100.00	
-	18	4" CONCRETE WALK, AS PER PLAN	SQFT	3313	\$ 5.50	49	1.50 \$	6.00	
	19	CURB RAMP, TYPE A1, AS PER PLAN	EACH	8	\$ 600.00	49	1,800.00 \$	200.007	
_	8	CURB RAMP, TYPE A2, AS PER PLAN	EACH	4	\$ 600.00	€9	2,400.00 \$	700.00	
_	21	CURB RAMP, TYPE B2, AS PER PLAN	EACH	2	\$ 600.00	69	3,000.00 \$	200.00	
	SECT	SECTION TOTAL ROADWAY:				\$ 157,154.25	4.25		\$ 160,328.00
-		EROSION CONTROL .							
_	ន	SOIL ANALYSIS TEST	EACH	2	\$. 80.00	s		_	
_	83	TOPSOIL	CUYD	73	\$ 25.00	4	1,825.00 \$	40.00	
	54	SEEDING AND MULCHING, AS PER PLAN	SQYD	1,310	\$ 1.50	4	\$ 00.596,	3.50	4,585.00
	52	COMMERCIAL FERTILIZER	NOT	0.20	\$. 800.00	€9	160.00 \$	1,000.00	
_	92	UME	ACRE	0.27	\$ 350.00	s	94.50 \$	200.00	54.00
	27	WATER	M GAL	7	\$ 100.00	U)		\$ 00.09	350.00
D-	88	EROSION CONTROL	EACH	7,500	\$ 1.00	2 \$	\$ 00.003,	1,00 \$	7,500.00
	SECT	SECTION TOTAL EROSION CONTROL:				\$ 12,404.50	4.50		\$ 15,809.00
- 7		DRAINAGE							
7	29	6" CONDUIT, TYPE F FOR UNDERDRAIN OU	ᇤ	140	\$ 19.00	69	2,660.00 \$	12.00 \$	1,680.00
~1	8	_	ᄩ	40	\$ 57.00	\$	2,280.00 \$	135.00 \$	No.
	31	12" CONDUIT, TYPE C, AS PER PLAN	Ħ	40	\$ 57.00	&	_	-	
<u>\</u>	32	CATCH BASIN, AS PER PLAN	EACH	2	.\$ 1,250.00			3,200.00 \$	
•	83	CATCH BASIN ADJUSTED TO GRADE, AS PER PLAN	EACH	9	\$ 325.00	64	3,250.00 \$	\$ 00.008	9,000.00

REPORT NO. 03-13

. =

TAYLOR ROAD (C.R. 26) FROM EUCLID AVE, TO THE EAST CLEVELAND SOUTH CORPORATION LIMIT

NO.	DESCRIPTION	UNIT	QUANTITY	DOLLARS I CT	I CTS	Groat Lakos Grusning, Ltd.		DOLLARS I CTS	CTS	Constr	Construction Co.
34	CATCH BASIN RECONSTRUCTED TO GRADE, AS PER PLAN	EACH	4	49	850.00	\$ 3,	3,400.00	\$ 1,	0		4,400.00
32	MANHOLE, NO. 3, AS PER PLAN	EACH	2	€9	3,000.00	\$ 6,	6,000.00	\$ 4,2	4,250.00 \$		8,500.00
36	MANHOLE ADJUSTED TO GRADE, AS PER PLAN	EACH	23	s	325.00	2 2	7,475.00	\$	475.00 \$		10,925.00
37	MANHOLE RECONSTRUCTED TO GRADE, AS PER PLAN	EACH	8	G	950.00	5 7,	7,600.00	·	750.00 \$		6,000.00
38	6" BASE PIPE UNDERDRAINS WITH FABRIC WRAP, AS PER PLAN	FF	2,554	49	7.00	\$ 17,		\$	13.00 \$		33,202.00
39	MISCELLANEOUS METAL	POUND	30,000	•	1.20	\$ 36,	36,000.00	. \$	1.25 \$		37,500.00
SECT	SECTION TOTAL DRAINAGE:					\$ 91,	91,323.00		"	45	127,207.00
					7			•	1		
	PAVEMENT										
40	PARTIAL DEPTH PAVEMENT REPAIR, AS PER PLAN	SQ YD	415	69	50.00	\$ 20,	20,750.00	\$	40.00 \$		16,600.00
41	FULL DEPTH RIGID PAVEMENT REMOVAL AND FLEXIBLE REPLACEMENT, AS	SQ YD	1,250	ક	82.00	\$ 102,	02,500.00	53	\$ 00.53		68,750.00
42	FULL DEPTH PAVEMENT SAWING	ㅂ	7,668	\$	2.25	\$. 17,	17,253.00	\$	2.50 \$		19,170.00
43	ASPHALT CONCRETE BASE, PG64-22, AS PER PLAN	CU YD	280	49	180.00	\$ 50,	50,400.00		170.00 \$		47,600.00
4	AGGREGATE BASE, AS PER PLAN	CUYD	425	69	45.00	\$ 19,	19,125.00	မှ	38.00 \$		16,150.00
\$	14" CONCRETE BASE, AS PER PLAN	SQ YD	2,874	49	70,00	\$. 201,	201,180.00	4	\$ 00.08		258,660.00
46	TACK COAT	GALLON	1,471	69	4.00	\$ 5,	5,884.00	\$	\$ 00%		4,413.00
47	TACK COAT FOR INTERMEDIATE COURSE	GALLON	610 .	S	6.00	\$ 3,	3,660.00	\$	3.00 \$		1,830.00
48	ASPHALT CONCRETE INTERMEDIATE COURSE, TYPE 2, PG64-22, AS PER	CU YD	593	s	132.00	\$ 78.	78,276.00	4	140.00 \$	L. U.S.	83,020.00
48	ASPHALT CONCRETE SURFACE COURSE, TYPE 1, PG70-22M, AS PER PLAN	CU YD	424	•	225.00	\$ 95,	95,400.00	S. 2	215.00 \$		91,160.00
80	8" REINFORCED CONCRETE PAVEMENT USING CLASS MS CONCRETE, AS	SQ YD	118	ક	95.00	\$ 11,	11,210.00	\$	\$ 00.08		9,440.00
51	6" NON-REINFORCED CONCRETE PAVEMENT USING CLASS MS CONCRETE,	SQ YD	. 426	S	65.00	\$ 27,	27,680.00	\$	68.00 \$		28,968,00
25	CURB, TYPE & USING CLASS MS CONCRETE, AS PER PLAN	FT	2,525	4	14.00	\$ 35,	35,350.00	49	19.00 \$		47,975.00
83	CONCRETE PAVEMENT JOINT CLEANED AND FILLED	FT	2,500	67	. 6.50	\$ 16,	16,250.00	\$	2.00 \$		5,000.00
54	CONCRETE PAVEMENT CRACK CLEANED AND FILLED	Ħ	750	\$	2.75	\$ 2,	2,062.50	\$	3.00 \$	Vaga et al	2,250.00
SECT	SECTION TOTAL PAVEMENT:					\$ 686,5	686,990.50		*		700,986.00
	WATERWORK			W				A 1888			
. 55	PIPE REMOVED, 24" AND UNDER	Ŀ	2780	s	10.00	\$ 27,		es.	2.00 \$		5,560.00
99	VALVE BOX, ADJUSTED TO GRADE	EACH	τO	cs.	225.00	5 1,	1,125.00	. 8	450.00 \$	22.8	2,250.00
67 .	O WATER MAIN OUCTLE IRON PIPE WITH PUSH-ON JOINTS AND RETAINED MECHANICAL JOINT FITTINGS, ANSI CLASS 32	Ŀ	52	49	125.00	ۍ ع	3,125.00	5	\$ 00.69		1,725.00
28	BY WATER MAIN DUCTLE IRON PIPE WITH PUSH-CH JOHTS AND RETAINED MECHANICAL JOHT FITTINGS, ANSI CLASS 52	E	2755	49	94.00	\$ 258,	258,970.00	*	78.00 \$		214,890.00
59	8" GATE VALVE WITH VALVE BOX, COMPLETE	EACH	9	\$	1,600.00	. \$	9,600.00	\$ 1,2	1,250.00 \$		7,500.00
09	6" CUTTING-IN VALVE WITH VALVE BOX, COMPLETE	EACH	. 3	€9	3,200,00	\$. 9,	9,600.00	9'8 \$	3,600.00 \$	630	10,800.00
61	FURNISHING AND SETTING 6" HYDRANT ASSEMBLY, COMPLETE	EACH	.80	s	4,300.00	\$ 34,	34,400.00	\$ 4,1	4,125.00 \$		33,000.00
62	MISCELLANEOUS METAL WORK	POUND	2000	s	1.20	\$ 2,	2,400.00	S	2.50 \$		5,000.00
63	MAINTENANCE OF WATER SERVICE	LUMP	+	4	30,000.00	\$ 30'	30,000,00	\$ 62,0	62,000.00 \$		62,000.00
8	SHEETING LEFT IN PLACE	MBF	2	4	_	\$ 2,	_	. 	550.00		1,100.00
	WATER WORK, MISC.: WATER SERVICE CONNECTION 2" AND UNDER, LONG	EACH	. 51		-		_		2,850.00 \$		37,050.00
8	WATER WORK, MISC.: WATER SERVICE CONNECTION 2" AND UNDER, SHOR	EACH	æ	49			70,200.00		1,675.00 \$		65,325.00
10		A CONTRACTOR AND			The contract of the state of th						

CUYAHOGA COUNTY DEPARTMENT OF PUBLIC WORKS ITEMIZED ESTIMATE

TAYLOR ROAD (C.R. 26) FROM EUCLID AVE. TO THE EAST CLEVELAND SOUTH CORPORATION LIMIT

MATTER WORK, MISC: WATTER SERVICE CONNECTION 9" AND LARGER, SHOF EACH 1		╎┈╎┈╎┈┤╸┤╸╎╶┤╶╏╸ ┤		8, 4, 11, 9, 72, 8, 11, 11, 12, 4, 4, 14, 14, 14, 14, 14, 14, 14, 14,		\$ 3,350.00 \$ 5.00 \$ \$ 50.00 \$ \$ \$ 50.00 \$ \$ \$ \$ 50.00 \$ \$ \$ \$ \$ 50.00 \$ \$ \$ \$ \$ 50.00 \$ \$ \$ \$ \$ \$ 50.00 \$ \$ \$ \$ \$ \$ \$ 50.00 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	3,350,00 1,000.00 1,650,00 1,200,00 609,660,00 7,380,00 861,00 861,00 3,000,00 950,00 1,125,00 1,180,00 1,180,00 3,000,00 3,000,00 3,000,00 3,000,00 3,000,00 3,000,00
WATER WORK, MISC.: REMOVE EXISTING VALVE BOX		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	175.00 1,200.00 1,200.00 13.00 1,000.00 1,000.00 1,000.00 310.00 830.00 830.00 830.00 842.00	* * * * * * * * * * * * * * * * * * *	 		
WATER WORK, MISC. REMOVE ENDING CIECAND EACH			175.00 1,200.00 13.00 13.00 47.00 8.00 787.00 1,000.00 130.00 310.00 830.00 830.00 842.00	27. 8. 8. 8. 8. 8. 8. 8. 8. 8. 8. 8. 8. 8.	 		
WATTER WORK, MISC.: REMOVE EXISTING CURES AND 1-DTF VALVE BOAN WATTER WORK, MISC.: REMOVE HYDRANT ASSEMBLY INCLUDING TEE AND 6 ECONDUIT. 2. 725.04 FT 69 5		, w w w w w w w w w w w w w	1,200.00 13.00 13.00 47.00 8.00 787.00 1,000.00 130.00 310.00 831.00 831.00 830.00 830.00 42.00	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	 		8
Treeling of Pedestrain Signal, Targetic Control		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	13.00 13.00 13.00 1,000.00 1,000.00 130.00 310.00 530.00 520.00	2 17 8 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			8
TRAFFIC CONTROL		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	13.00 13.00 47.00 8.00 787.00 1,000.00 130.00 310.00 310.00 830.00 530.00 500.00				
72 CONDUIT, 2°, 725.04 FT 69 \$ 73 CONDUIT, 2°, 725.04 FT 69 \$ 74 CONDUIT, 2°, 725.04 FT 64 \$ 75 CONDUIT, 3°, 725.04 FT 164 \$ 76 TRENCH FT 123 \$ 77 PULLBOX, 725.06, 175.00 FC 4 \$ 77 PULLBOX, 725.06, 175.00 EACH 4 \$ 78 PULLBOX, 725.06, 175.00 EACH 4 \$ 79 PLASTIC CAUTION TAPE EACH 4 \$ 80 SIGNY HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 8 \$ 81 SIGNY FLAT SHEET, AS PER PLAN EACH 8 \$ 82 VEHICULAR SIGNAL HEAD EACH 8 \$ 83 FEDESTRAND VISITAN SIGNAL HEAD EACH 8 \$ 84 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 85 SIGNAL CABLE, GONDUCTOR, NO. 14 AWIG		00 00 00 00 00 00 00 00 00 00 00 00 00	13.00 13.00 47.00 8.00 787.00 1,000.00 130.00 310.00 830.00 630.00 42.00	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8			
TRAFFIC CONTROL FT 69 \$ 72 CONDUIT, 2", 725.04 FT 69 \$ 73 CONDUIT, 3", 725.04 FT 164 \$ 74 CONDUIT, 3", 725.04 FT 164 \$ 75 TRENCH FT 173 \$ 77 PULLBOX, 725.06, 24"736" EACH 1 \$ 78 GROUNDIN TADE EACH 1 \$ 80 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 4 \$ 81 GROUND ROD FACH 4 \$ 82 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 8 \$ 83 SIGN FLAT SHEET, AS PER PLAN EACH 8 \$ 84 SIGNAL HEAD EACH 8 \$ 85 SIGNAL CABLE, GONDUCTOR, AS PER PLAN EACH 8 \$ 86 FEDESTRIAN SIGNAL HEAD EACH 8 \$ 87 SIGNAL CABLE, CONDUCTOR, NO. 14 AWG FT 1		v v v v v v v v v v v v v v v v v v	13.00 13.00 47.00 8.00 787.00 1,000.00 130.00 310.00 830.00 630.00 42.00	w w w w w w w w w w w w w w w w w w w			
72 CONDUIT, 2", 725.04 FT 64 \$ 73 CONDUIT, 3", 725.04 FT 64 \$ 74 CONDUIT, 3", 725.04 FT 143 \$ 75 TRENCH FT 143 \$ 75 TRENCH FT 143 \$ 76 PULLBOX, 725.06, 24X36" FC 14 \$ 78 PULLBOX, 725.06, 24X36" FC 123 \$ 78 PULLBOX, 725.06, 24X36" FC 13 \$ 79 PLASTIC CAUTION TAPE EACH 4 \$ 80 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 4 \$ 81 SIGNAL HEAD, 3-SECTION, 12" LENS, 1-WAY, AS PER PLAN EACH 8 \$ 82 VEHICULAR SIGNAL HEAD EACH 8 \$ 83 VONERING OF PEDESTRANS SIGNAL HEAD EACH 8 \$ 84 PEDESTRANS PRIVAN EACH 8 \$ 85 SIGNAL CABLE, 5 CONDUCTOR, NO. 14 AWG		w w w w w w w w w w w w w	13.00 13.00 47.00 8.00 787.00 1,000.00 130.00 310.00 830.00 630.00 42.00	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8			
73 CONDUIT, 3°, 725.04 FT 64 \$ 74 CONDUIT, 3°, 725.04 FT 164 \$ 75 TRENCH FT 123 \$ 76 PULLBOX, 725.06, 17%30° FACH 4 \$ 77 PULLBOX, 725.06, 24%36° EACH 4 \$ 78 PULLBOX, 725.06, 24%36° EACH 4 \$ 79 PULLBOX, 725.06, 24%36° EACH 4 \$ 80 SCOUNDING ROAD EACH 4 \$ 80 SIGN HANDERGE ASSERBLY, MAST ARM, AS PER PLAN EACH 8 \$ 81 SIGN HANGER ASSERBLY, MAST ARM, AS PER PLAN EACH 8 \$ 82 VEHCULAR SIGNAL HEAD EACH 8 \$ 83 VEHCULAR SIGNAL HEAD EACH 8 \$ 84 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 85 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 86 FEDESTRIAN PUSH PUAN<		0	13.00 47.00 8.00 787.00 1,000.00 130.00 310.00 831.00 830.00 620.00 42.00	ω ω ω ω ω ω ω ω ω ω ω ω ω ω ω ω ω ω ω			
74 CONDUIT., JACKED OR DRILLED, 725.04, 3" FT 164 \$ 75 TREDICH. 7 123 \$ 76 PULLBOX, 725.06, 24*736" 6 4 \$ 77 PULLBOX, 725.06, 24*736" 7 123 \$ 78 BULLBOX, 725.06, 24*736" 6 4 \$ 79 PLASTIC CAUTION TAPE EACH 4 \$ 80 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 4 \$ 81 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 8 \$ 82 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 8 \$ 83 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 8 \$ 84 SIGNAL HEAD, SGIGNAL HEAD EACH 8 \$ 85 SOWERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 86 PEDESTRAN SIGNAL HEAD EACH 8 \$ 86 SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG FT 7 \$		w, w, w, w, w, w, w, w, w, w	47.00 8.00 787.00 1,000.00 1,000.00 310.00 310.00 831.00 830.00 500.00 42.00	w w w w w w w w w w			
77 CONCERNIA SIGNAL HEAD FT 123 \$ 78 PULL BOX, 725.06, 24*26° 4 \$ 79 PULL BOX, 725.06, 24*26° 6 4 \$ 79 PULL BOX, 725.06, 24*26° 6 5 \$ 79 PLASTIC CAUTION TAPE EACH 1 \$ \$ 80 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 4 \$ \$ 81 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 8 \$ \$ \$ \$ 82 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 8 \$ <td< td=""><td></td><td>w, w, w, w, w, w, w, w, w</td><td>8.00 787.00 1,000.00 130.00 6.90 310.00 830.00 500.00 42.00</td><td>w w w w w w w w w</td><td></td><td></td><td></td></td<>		w, w, w, w, w, w, w, w, w	8.00 787.00 1,000.00 130.00 6.90 310.00 830.00 500.00 42.00	w w w w w w w w w			
POWER SERVICE, ACASE OF, 17730** EACH 4 \$ 77 PULL BOX, 725.06, 24736** 1 \$ 78 GROUND ROO. EACH 1 \$ 79 PLUL BOX, 725.06, 24736** 1 \$ \$ 79 PLASTIC CAUTION TAPE FT 123 \$ 80 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 4 \$ 81 SIGN, FLAT SHEET, AS PER PLAN EACH 8 \$ \$ 82 VEHICULAR SIGNAL HEAD, SECTION, 12° LENS, 1-WAY, AS PER PLAN EACH 8 \$ \$ 83 VEHICULAR SIGNAL HEAD EACH 8 \$ \$ \$ \$ 84 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$		44 44 45 45 45 47 47 47 V7 V7	787.00 1,000.00 130.00 6.30 310.00 310.00 530.00 500.00 42.00	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			
77 PULLEDOX, 725.06, 24*X36" 7 \$ 78 GROUND KOO 78 \$ 79 PLASTIC CAUTION TAPE FT 123 \$ 80 SIGN HANGER ASSEMBLY, MAST ARM, AS PER PLAN EACH 4 \$ 81 SIGN FLAT SHEET, AS PER PLAN EACH 6 \$ 82 VEHICULAR SIGNAL HEAD, COUNTDOWN, TYPE D2, AS PER PLAN EACH 8 \$ 83 VEHICULAR SIGNAL HEAD EACH 8 \$ 84 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 85 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 86 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 87 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 88 COVERING OF PEDESTRIAN SIGNAL HEAD EACH \$ \$ 89 SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG FT 732 \$ 80 SIGNAL SUPPORT FOUNDATION EACH FT 49 \$		w w w w w w w w w	1,000.00 130.00 0.90 310.00 310.00 630.00 500.00 42.00	w w w w w w w			
PLASTIC CAUTION TAPE FT 123 \$		w w w w w w w	130.00 0.90 310.00 31.00 630.00 590.00 42.00	0 0 0 0 0 0 0 0 0			
SIGNAL CAUTION TAPE		w w w w w w	310.00 310.00 31.00 500.00 42.00	w w w w w w			
80 SIGNA HANGER ASSEMBLY, MAST ARIM, AS PER PLAN EACH 4 \$ 81 SIGNA HANGER ASSEMBLY, MAST ARIM, AS PER PLAN EACH 8 \$ 82 VEHICULAR SIGNAL HEAD, COUNTDOWN, TYPE DZ, AS PER PLAN EACH 8 \$ 83 VEHICULAR SIGNAL HEAD EACH 8 \$ 84 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 85 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 86 PEDESTRIAN PUSH BUTTON, AS PER PLAN EACH 8 \$ 87 OVERING OF PEDESTRIAN PUSH BUTTON, AS PER PLAN EACH 5 \$ 88 SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG FT 732 \$ 89 SIGNAL CABLE, 7 CONDUCTOR, NO. 6, AWG EACH 5 \$ 91 PEDESTRAL FOUNDATION EACH 5 \$ 92 LOOP DETECTOR LEAD-IN CABLE FT 49 \$ 93 SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 1, PAS PER PLAN EACH 1 \$ 94 <		w w w w w	310.00 31.00 630.00 500.00 42.00	w w w w	\square		
80 SIGNAT CABLE, TARRIAN SIGNAT EACH ES \$ 81 SIGNAL HEAD, SECTION, 12" LENS, 1-WAY, AS PER PLAN EACH 8 \$ 82 VEHICULAR SIGNAL HEAD EACH 8 \$ 83 PEDESTRIAN SIGNAL HEAD EACH 8 \$ 84 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 85 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 86 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 87 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 88 PEDESTRIAN SIGNAL HEAD EACH \$ \$ 89 SIGNAL CABLE, GONDUCTOR, NO. 14 AWG FT 732 \$ 89 SIGNAL SUPPORT FOUNDATION EACH 5 \$ 90 SIGNAL SUPPORT FOUNDATION EACH 7 \$ 91 PEDESTAL FOUNDATION FT 49 \$ 92 LOOP DETECTOR LEAD-IN CABLE FT 49 <t< td=""><td></td><td>w, w, w w w</td><td>31.00 630.00 500.00 42.00</td><td>N N N N</td><td>\square</td><td></td><td>1,890.0 4,800.0 3,800.0</td></t<>		w, w, w w w	31.00 630.00 500.00 42.00	N N N N	\square		1,890.0 4,800.0 3,800.0
8 SECTION, 12" LENS, 1-WAY, AS PER PLAN EACH 8 \$ 82 VEHICULAR SIGNAL HEAD, COUNTDOWN, 17PE D2, AS PER PLAN EACH 8 \$ 84 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 85 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 86 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 87 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 88 PEDESTRIAN PUSH BUTTON, AS PER PLAN EACH 5 \$ 89 SIGNAL CABLE, FOUNDATION FT 732 \$ 89 SIGNAL SUPPORT FOUNDATION EACH 5 \$ 90 SIGNAL SUPPORT FOUNDATION EACH 5 \$ 91 PEDESTAL FOUNDATION FT 1238 \$ 92 LOOP DETECTOR LEAD-IN CABLE FT 49 \$ 93 POWER CABLE, 3 CONDUCTOR, NO. 6, AWG FT 49 \$ 94 POWER SERVICE, AS PER PLAN EACH 1		69 69 69 69	500.00	N N N	\vdash		3,800.0
82 VEHICULAR SIGNAL HEAD, SAPENTIAN, 12 LLING, 1707, 170		w w w	500.00	ω ω	-		3,800.0
84 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 85 COVERING OF VEHICULAR SIGNAL HEAD EACH 8 \$ 86 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 87 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 88 PEDESTRIAN PUSH BUTTON, AS PER PLAN FT 923 \$ 89 SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG FT 732 \$ 89 SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG FT 732 \$ 91 PEDESTAL FOUNDATION EACH 5 \$ 92 LOOP DETECTOR LEAD-IN CABLE FT 1288 \$ 93 POWER CABLE, 3 CONDUCTOR, NO. 6, AWG FT 49 \$ 94 POWER SERVICE, AS PER PLAN EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 1, AS PER PLAN EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1 \$		w w	42.00	4	L	-	, 000
84 COVERNING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 85 COVERING OF PEDESTRIAN SIGNAL HEAD EACH 8 \$ 87 DETECTOR LOOP, AS PER PLAN EACH 5 \$ 88 SIGNAL CABLE, FOUNDUCTOR, NO. 14 AWG FT 732 \$ 89 SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG FT 732 \$ 90 SIGNAL SUPPORT FOUNDATION EACH 5 \$ 91 PEDESTAL FOUNDATION EACH 5 \$ 92 LOOP DETECTOR LEAD-IN CABLE FT 1288 \$ 93 POWER CABLE, 3 CONDUCTOR, NO. 6, AWG FT 49 \$ 94 POWER SERVICE, AS PER PLAN EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 4 POLE, WITH MAST ARMS TC-81.2 EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 2, AS PER PLAN EACH 1 \$ 96 SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1 \$		S			336.00 \$		320.0
85 COVERING OF PEDES TRIAN SIGNAL HEAD CACH 8 \$ 86 PEDESTRIAN PUSH BUTTON, AS PER PLAN EACH 8 \$ 87 DETECTOR LOOP, AS PER PLAN EACH 5 \$ 88 SIGNAL CABLE, F CONDUCTOR, NO. 14 AWG FT 732 \$ 89 SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG EACH 3 \$ 91 PEDESTAL FOUNDATION EACH 5 \$ 92 LOOP DETECTOR LEAD-IN CABLE FT 1288 \$ 93 POWER CABLE, 3 CONDUCTOR, NO. 6, AWG FT 49 \$ 94 POWER SERVICE, AS PER PLAN EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 4 POLE, WITH MAST ARMS TC-81.2 EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1 \$ 96 SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1 \$		•	42.00	49		40.00	320.00
86 PEDESTRIAN POSH BULLON, AS PER PLAN PER PLAN S \$ 87 DETECTOR LOOP, AS PER PLAN EACH 5 \$ 88 SIGNAL CABLE, 6 CONDUCTOR, NO. 14 AWG FT 732 \$ 89 SIGNAL SUPPORT FOUNDATION EACH 5 \$ 91 PEDESTAL FOUNDATION EACH 5 \$ 92 LOOP DETECTOR LEAD-IN CABLE FT 1288 \$ 93 POWER CABLE, 3 CONDUCTOR, NO. 6, AWG FT 49 \$ 94 POWER SERVICE, AS PER PLAN EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-8121 DESIGN 4 POLE, WITH MAST ARMS TC-812 EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-8121, DESIGN 2, AS PER PLAN EACH 1 \$ 96 SIGNAL SUPPORT, TYPE TC-8121, DESIGN 2, AS PER PLAN EACH 1 \$		· vs	236.00	69	_	225.00 \$	1,800.00
87 DEFECTOR LOCK, AS PER PLAN 88 SIGNAL CABLE, 6 CONDUCTOR, NO. 14 AWG FT 923 \$ 89 SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG FT 732 \$ 90 SIGNAL SUPPORT FOUNDATION EACH 5 \$ 91 PEDESTAL FOUNDATION FT 1288 \$ 92 LOOP DETECTOR LEAD-IN CABLE FT 1288 \$ 93 POWER CABLE, 3 CONDUCTOR, NO. 6, AWG FT 49 \$ 94 POWER SERVICE, AS PER PLAN EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-8121 DESIGN 4 POLE, WITH MAST ARMS TC-812 EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-8121, DESIGN 2, AS PER PLAN EACH 1 \$ 96 SIGNAL SUPPORT, TYPE TC-8121, DESIGN 2, AS PER PLAN EACH 1 \$		co	890.00	69	_	\$ 00.058	4,250.00
88 SIGNAL CABLE, 6 CONDUCTOR, NO. 14 AWG FT 732 \$ 89 SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG FT 732 \$ 90 SIGNAL SUPPORT FOUNDATION EACH 5 \$ 92 LOOP DETECTOR LEAD-IN CABLE FT 1288 \$ 93 POWER CABLE, 3 CONDUCTOR, NO. 8, AWG FT 49 \$ 94 POWER SERVICE, AS PER PLAN EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-8121 DESIGN 4 POLE, WITH MAST ARMS TC-812 EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-8121, DESIGN 2, AS PER PLAN EACH 1 \$ 96 SIGNAL SUPPORT, TYPE TC-8121, DESIGN 2, AS PER PLAN EACH 1 \$		v	1,85		-	1.75 \$	
SIGNAL CABLE, 7 CONDUCTOR, NO. 14 AWG FI 732 5 SIGNAL SUPPORT FOUNDATION EACH 5 \$ PEDESTAL FOUNDATION EACH 5 \$ LOOP DETECTOR LEAD-IN CABLE FT 1288 \$ POWER CABLE, 3 CONDUCTOR, NO. 6, AWG FT 49 \$ POWER SERVICE, AS PER PLAN EACH 1 \$ SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 4 POLE, WITH MAST ARMS TC-81.2 EACH 1 \$ SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 1, AS PER PLAN EACH 1 \$ SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1 \$	+	2	39.6			_	ŀ
90 SIGNAL SUPPORT FOUNDATION 5 3 3 3 3 3 3 3 3 3 3 3 3 3 3 4 5 \$ 4 5 \$ 4 \$ 4 \$ \$ 4 \$ \$ 4 \$ </td <td></td> <td>0 6</td> <td>25.5</td> <td>÷ 6</td> <td>-</td> <td>+</td> <td></td>		0 6	25.5	÷ 6	-	+	
91 PEDESTAL FOUNDATION 92 LOOP DETECTOR LEAD-IN CABLE FT 1288 \$ 92 LOOP DETECTOR LEAD-IN CABLE FT 49 \$ 94 POWER SERVICE, AS PER PLAN EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 4 POLE, WITH MAST ARMS TC-81.2 EACH 1 \$ 95 SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 1, AS PER PLAN EACH 1 \$ 97 SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1 \$	-	9 4	630.00	9 69	-	_	
92 LOOP DELECTOR LEGION LEAD-IN CABLE 93 POWER CABLE, 3 CONDUCTOR, NO. 6, AWG F7 49 \$ 94 POWER SERVICE, AS PER PLAN 1 \$ 95 SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 4 POLE, WITH MAST ARMS TC-81.2 EACH 1 \$ 96 SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 1, AS PER PLAN EACH 1 \$ 97 SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1 \$			1.38	69	0.000		1,738.80
94 POWER CABLE, 3 CONDOCTOR, NO. 9, AND EACH 1 \$ 94 POWER SERVICE, AS PER PLAN 1 \$ 95 SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 4 POLE, WITH MAST ARMS TC-81.2 EACH 1 \$ 97 SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1 \$			4.20	69	-	4.00 \$	198.00
94 POWER SERVICE, AS FOR PLAN 95 SIGNAL SUPPORT, TYPE TC-81.21 DESIGN 4 POLE, WITH MAST ARMS TC-81.2 EACH 1 \$ 97 SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1 \$		69	1,890.00	65	1,890.00 \$	1,800.00	1,800.00
SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 1, AS PER PLAN EACH 1 \$ SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1 \$		S	7,455.00	65	7,455.00 \$	7,100.00 \$	7,100.00
SIGNAL SUPPORT, TYPE TC-81.21, DESIGN 2, AS PER PLAN EACH 1		69	4,100.00	69	4,100.00 \$	3,900.00 \$	3,900.00
9/ SIGNAL SUPPORT, LIPE 10-8121, DESIGNA, AS TENT DAY		69	4,300.00	8	4,300.00 \$	4,100.00 \$	4,100.00
AN DEPOTOTAL OF TOANISCHOOLINGS AS DED DIAN	ļ.	· va	603.00	€9		\$ 575.00	2,875.00
SO PEDESTAL OF DEVISION DESCRIPTION AS DED DIAN	-	69	1,575.00	€	1,575.00 \$	1,500.00	1,500.00
SE REMOVAL OF INAT PLOUDE TO SECURITION TEST HOLE		€9	420.00	49	840.00 \$	400.00	800.00
AND CONTROL EP INIT TYPE TSOLA WITH CABINET TYPE TS2. AS PER PLAN EACH 1 \$	ACH 1	49	15,120.00	\$	15,120.00 \$	14,400.00 \$	14,400.00
100 CABINET RISER	ACH 1	es	760.00	67	\$ 00.087	725.00 \$	
5	ACH 1	49	2,205.00	63	2,205.00 \$	2,100.00 \$	
10				\$ 94,863.69	3.69	\$	90,402.80

TAYLOR ROAD (C.R. 28) FROM EUCLID AVE. TO THE EAST CLEVELAND SOUTH CORPORATION LIMIT

REF DE	DESCRIPTION	UNIT	GUANTITY	UNIT PRICE EST.	EST.	Great Lakes Crushing,		DOLLARS I CTS	S	Construction Ca.
-										
104 GR	GROUND MOUNTED SUPPORT, NO. 3 POST	E .	89	\$	7.35	\$ 367.50	\$	7.00	43	350.00
	SIGN SUPPORT ASSEMBLY, POLE MOUNTED	EACH	8	\$	57.00	\$ 456,00	\$ 00	. 55.00	55	440.00
	SIGN. FLAT SHEET	SQ FT	25	67	12,60	\$ 655.20	. \$ 0	12.00	43	624.00
	REMOVAL OF GROUND MOUNTED SIGN AND DISPOSAL	EACH	5.	49	15.75	\$ 78.75	\$ 5.	15.00	ဖ	75.00
_		EACH	8	47	58.00	\$ 174.00	\$ 00	55.00	\$	165.00
		EACH	m	69	31.50	\$ 84.50	\$ 00	30.00	છ	90.00
	REMOVAL OF POLE MOUNTED SIGN AND DISPOSAL	EACH	8	47	15.75	\$· 126.00	\$ 00	15.00	s	120.00
	INI ACTS	t	140	w	6.30	\$ 882.00	\$ 00	900'9		840.00
_	CROSSWAIKINE	E	099	ss	3.20	\$. 2,112.00	\$ 00	3.00	s	1,980.00
	LANE ARROW	EACH	11	cs.	130.00	\$ 1,430.00	\$ 00	125.00		1,375.00
	WORD ON PAVEMENT, 72"	EACH	2		163.00	\$ 326.00	\$ 00	155.00	4	310.00
115	LANE LINE 4"	MILE	0.65		1,890.00	\$ 1,228.50	\$ 09	1,800.00	49	1,170.00
	CENTERLINE	MILE	09.0	\$	5,040.00	\$ 3,024.00	\$ 00	4,800.00		2,880.00
42	CHANNELIZING LINE 8"	F	270	40	2.00	\$ 540.00	\$ 00	1.80	4	486.00
118 W	WORK ZONE LANE LINE, CLASS III	MILE	0.65		630,00	\$ 409.50	\$ 05	600.00	€9	390.00
W W	119 WORK ZONE CENTERLINE, CLASS III	MILE	09'0	\$	840.00	\$ 504.00	\$ 00	800.00	49	480.00
20 🕅	120 WORK ZONE CHANNELIZING LINE, CLASS III	Ŀ	270	€	0.85	\$ 229.50	8	08:0	€9	216.00
21 W	121 WORK ZONE STOP LINE, CLASS III	FT	140	€	3.15	\$ 441.00	<u>چ</u>	3.00	€9-	420.00
122 W	WORK ZONE CROSSWALK LINE, CLASS #	Ħ	999	€	1.60	\$ 1,056.00	پ 8	1.50	69	00'066
123 W	WORK ZONE ARROW, CLASS III	EACH	11	€>	52.50	\$ 577.50	_	20.00	69	550.00
124 W	124 WORK ZONE WORD ON PAVEMENT, 72", CLASS III	EACH.	7	69	52.50	\$ 105.00	<u>پ</u>	50.00	49	100.00
ECTION	SECTION TOTAL TRAFFIC CONTROL:					\$ 14,816.95	9		: 69	14,051.00
F										
-	MAINTENANCE OF TRAFFIC									
125 TF	TRAFFIC COMPACTED SURFACE, TYPE A OR B , AS PER PLAN	CU YD	910	c/a	43.00	\$ 21,930.00	\$ 00	10.00	s	5,100.00
	2* ASPHALT CONCRETE WALK	SQFT	. 1600	49	2.25	\$ 3,600.00	\$ 00	1.50	s	2,400.00
_	LAW ENFORCEMENT OFFICER WITH PATROL CAR FOR ASSISTANCE	HOUR	80	65	60.00	\$ 4,800.00	\$	00.00	(A)	4,800.00
128	DETOUR SIGNING, AS PER PLAN	LUMP	1	€	3,400,00	\$ 3,400.00		2,800.00	44	2,800.00
129 RE	REPLACEMENT SIGN	EACH	15	€5	105.00	\$ 1,575.00		75.00		1,125.00
130 AS	ASPHALT CONCRETE FOR MAINTAINING TRAFFIC	CU YD	10	69	150.00	\$ 1,500.00	\$	300,00	_	3,000.00
	PORTABLE CHANGEABLE MESSAGE SIGN	SIGN MONTH	24	\$	500.00	\$ 12,000.00	\$ 00	800.00	63	19,200.00
	ROADS FOR MAINTAINING TRAFFIC, AS PER PLAN	LUMP	-	\$ 15	15,000.00	\$ 15,000.00	\$ 00	1,000.00	63	1,000.00
133 W	WATER	MGAL	50	€9	20.00	\$ 1,000.00	_	90.09		2,500.00
134 C/	CALCIUM CHLORIDE	TON	υ	\$	200.00	\$ 1,000.00	8	125.00	100	625.00
ECTION	SECTION TOTAL MAINTENANCE OF TRAFFIC:					\$ 66,805.00	9		w	42,550.00
							-			
	MISCELLANEOUS									
135 M	MAINTAINING TRAFFIC	LUMP		35	35,000,00	\$ 35,000.00	8	75,000,00	59	75,000.00

4

	₹
	ORPORATION
	EVELAND SOUTH CORPORA
	STC
ROAD (C.R. 26)	ICLID AVE. TO THE EA
TAYLOR	FROM EU

	-	ESTIMATED	UNIT PRICE EST.	Great Lakes Crushing.	UNIT PRICE EST.	Vandra Brothers
ESCRIPTION	LINO	QUANTIFY	DOLLARS I CTS	ĹĦ.	DOLLARS I CTS	
IELD OFFICE, TYPE B, AS PER PLAN	MONTH	9	\$: 1,200,00	\$ 7,200.00	00:008'1	\$ 10,800.00
ONSTRUCTION LAYOUT STAKES, AS PER PLAN	LUMP	1	\$ 18,000.00	\$ 18,000.00	\$ 2,500.00	\$ 2,500.00
IOBILIZATION	LUMP	1	\$ 40,000.00	\$ 40,000.00	00'000'09 \$	\$ 60,000,00
RECONSTRUCTION PHOTOGRAPHIC PROJECT DOCUMENTATION (SEE PRO	LUMP	1	\$ 2,000.00	\$ 2,000.00	1,000,00	1,000.00
N TOTAL MISCELLANEOUS:				\$ 102,200.00	•	\$ 149,300.00
GRAND	D TOTAL E	GRAND TOTAL ESTIMATED		\$ 1,800,627.89		\$ 1,810,283.80

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0015

Sponsored by: County Executive	A Resolution making an award on RQ24707
FitzGerald/Department of Human	to The Guardian Life Insurance Company of
Resources	America in the amount not-to-exceed
	\$12,300,000.00 for dental group healthcare
	benefits for County employees and their
	dependents for the period 1/1/2013 -
	12/31/2015; authorizing the County
	Executive to execute the contract and all
	other documents consistent with said award
	and this Resolution; and declaring the
	necessity that this Resolution become
	immediately effective.

WHEREAS, the County Executive/Department of Human Resources has recommended an award on RQ24707 to The Guardian Life Insurance Company of America in the amount not-to-exceed \$12,300,000.00 for dental group healthcare benefits for County employees and their dependents for the period 1/1/2013 – 12/31/2015; and

WHEREAS, this project will be funded 100% from the General Fund and the schedule of payments is monthly by premium report.

WHEREAS, it is necessary that this Resolution become immediately effective because the projects term began January 1, 2013. The project was delayed because the vendor needed time to pull together all required forms and documentation.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council hereby makes an award on RQ24707 to The Guardian Life Insurance Company of America in the amount not-to-exceed \$12,300,000.00 for dental group healthcare benefits for County employees and their dependents for the period 1/1/2013 - 12/31/2015.

SECTION 2. That the County Executive is authorized to execute a contract in connection with said award and all documents consistent with this Resolution.

SECTION 3. It is necessary that this Resolution become immediately effective in order that critical services provided by the Cuyahoga County Department of Human Resources can continue without interruption, and to provide for the usual, daily operation of the County. Provided that this Resolution receives the affirmative vote of at least eight

members of County Council, this Resolution shall become immediately effective upon the signature of the County Executive.

SECTION 4. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by, the foreg	oing Resolution was
Yeas:		
Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date
First Reading/Referred to Committee(s) Assigned:	Committee:	
Journal		



3

Item Details:

Agency/Dept.

Office of Human Agency/Dept.Head Elise Hara

Name:

Resources

Name:

Type of Request:

Contract/Amendment

Request Prepared Michael Brown

Telephone No.

216-698-6436

by:

SUMMARY OF REQUESTED ACTION:

The Office of Human Resources is recommending an award on RQ24707 and would like to enter into a contract with Guardian Life Insurance Company of America in the amount not-to-exceed \$12,300,000.00 for Dental Insurance coverage for the period of January 1, 2013 - December 31, 2015. (Resolution No. DC2012-182 - authority to seek proposals.)

A. Scope of Work SummaryThe Office of Human Resources is recommending an award on RQ24707 and would like to enter into a contract with Guardian Life Insurance Company of America in the amount not-to-exceed \$12,300,000.00 for Dental Insurance coverage for the period of January 1, 2013 - December 31, 2015. (Resolution No. DC2012-182 - authority to seek proposals.) **B. Procurement**1. The procurement method for this project was RQ24707. 2. The RFQ was closed on September 10, 2012. There is an SBE participation/goal of 5 % 3. There were 9 bids/proposals/applications pulled from OPD, 9 bids/proposals/ applications submitted for review, 1 bid/proposal/application approved. **C. Contractor and Project Information**1. Guardian Life Insurance Company of America

7 Hanover Square

New York, NY 10004-2616

- 2. The President for the contractor/vendor is Deanna M. Mulligan
- 3. The project is located in Council District 7**D. Project Status and Planning**1. The project is on-going due to the nature of Dental Insurance coverage. 2. The project's term will begin January 1, 2013. The reason there was a delay in this request is the vendor needed time to pull together all required forms and documentation. **E. Funding**1. The

project is funded 100% by the premium report.	he General Fund 2. The sche	dule of payments is monthly by
PURPOSE/OUTCOMES - PR	RINCIPAL OWNER(S):	
Explanation for late submi	ittal:	
Contract/Agreement Inforn	nation:	
Procurement Method:		
Explanation for Increase/	Decrease in \$ Amount for o	current request:
Financial Information:		
Funding source: Explai General Fund	nation:	
Total Amount Requested:		
ATTACHMENTS:		
Click to download		
Annual Report		
Auditor's Findings		
 ☐ Bidder Compliance Form ☐ Certificate of Incumbancy 		
Certificate of Non-Discrimination		
☐ Contract Cover		
Insurance Certificate		
Department Acknowledgement Form	<u>n</u>	
☐ Dental Insurance Contract		
☐ History Non-Collusion		
☐ SBE Plan		
Tabulation Sheet		
□ <u>W-9</u>		
Workers' Comp Certificate		
Award Letter		
☐ Evaluation ☐ SOS		
History	VA.1	Amproval
Time	Who Office of Procurement & Diversity	Approval



TABULATION OF PROPOSALS RECEIVED

DEPARTMENT NAME:	Human Resources	sonuces		PROP	OSAL D	UE DAT	PROPOSAL DUE DATE: September 10, 2012	
REP TITLE	COBRA Ad	ministration, \	COBRA Administration, Various Instrance	RF	RFP # 24707	707	SBE 5%	
TO BE COMPLETED BY OPD	3Y OPD		TO BE COMPLETED BY SBE CONTRACT COMPLIANCE OFFICER	3Y SBE CA	ONTRAC	T COMP	LIANCE OFFICER	USER DEPT.
VENDOR NAME & ADDRESS	ESS	IG/ETHICS REGISTRATION COMPLETE	SBE SUBCONTRACTOR NAME AMOUNT & PERCENTAGE	SBE PRIME%	TOTAL SBE %	COMPLY	COMMENTS & INSTITUTES	AWARD
Aena Life insurance Company 151 Parmington Avenuë Harford, CT 06156		IGC# MF NCA - yes BCF - yes ASC 9(1) N2	WF WHOO Pruiting BF 1 & Choice Staffing Irz		%2 \(\times \)	XXXX	Ferrendoge All 3/12. Total Grand All 3/12. Also requestingfull waiver. Also requestingfull waiver.	
Anthem BlueCross BlueShield 4361 Irwin Simpson Road Mason, OH 45040		- yes	WF Factor Limited #49,000 20%		202	20% Samo	Did not specify Total And glistis And glistis	
Ceridian Corporation 3201 34 th Street South St. Petersburg, FL 33711		IGC# NCA-yes BCF-yes ACF-yes ACF-yes			2	8-7	Mon collet Serviced One collet Serviced And Alas 2 1/1/2 #	极
CIGNA HealthCare 3 Summit Park Drive, Suite 250 Independence, OH 44131	,	8 8 g	GC# WF South End Ruiting \$3,750 1,25% NCA-yes HF 25 Promotions BCF-yes HF 25 Promotions bCF-yes 41,250 3.75%	~	53	R -	Did wet specify literation Bruts. And 9/13/10 rm 9/12/12. Specify &	
Delta Dental Plan of Ohio, Inc. Penton-Media Building, Suite 1703 1300 East 9th Street Cleveland, OH 44114	50	IGC# HF NCA - yes BCF - yes LE - 9 N 2	HF Tugiol Rechect about 5 ps Promotions \$ 10,630 5%		5/3	- James	and Inched redention water	

DEPARTMENT NAME:	Human Resources	PROPOSAL DU	PROPOSAL DUE DATE: September 10, 2012
RFP TITLE:	COBRA Administration, Various Insurance	RFP #: 24707	77 SBE 5%
Onardian Life Insurance Company of America 6480 Rockside Woods Blvd. South, Suite 220 Independence, OH 4413;	ica IGC# NM Progradure Buefts 20 15. 4 11 2 22.747 5 8 BCF - yes (*) * Not complete	569	Did not specify retention
Rumana 6050 Oak Tree Bonlevard, Suite 100 Cleveland, OR 44131	IGC# NCA - yes BCF - yes ACF - yes	100 N	29 * Mary Did word Sequent Wowler 29 * Mary Did word Supers our SEC And 9/13/19
Mimesota Life Insurance Company Securian Life Insurance Company 400 Robert Street North St. Paul, MN 55101-2098	IGC# NCA - yes BCF - yes	Se Zee	29 North partern all octuities - Will partern all octuities without 5055.
United Healthcare 9200 Worthington Road Westerville, OH 43082	IGC# 86 Bell # Luckson 1.2. 9 1.1.2. \$37,962 22 23 NCA - yes (*) BCF - yes * Not Complete	N N N	And Phille Did wet Stope SEZ Form or F. Hout the required information 3/10
	NCA = Non-Collusion Affidavit BCF = Bidder Compiliance Form		

Date

Department Director Signature of Approval

Department Director Name 4prab - Updated 12/8/2011

CONTRACT/AGREEMENT EVALUATION FORM

(To be completed in its entirety by user department for all contract/agreement renewals or amendments.)

Contract/Agreement No.: CE1000475	Time Period: 1-1-10 - 12-31-12
Service Description: Provide group dental and their dependents.	insurance benefits for County employees
Original Contract/Agreement Amount: \$ 1	1,405,666.04
Prior Amendment(s) Amount(s): None	
Performance Indicators: Accuracy and sponeeds, concerns, processing claims and pay	eed of response in replying to employee ment of claims.
Actual performance versus performance in processing are within the industry parame	ndicators (include statistics): Payments and ters.
ating of Overall Performance of Contractor	(Check One):
☐ Superior	
X Above Average	
☐ Average	
☐ Below Average ☐ Poor	
□ 1 001	
ustification of Rating: We have not received	d many complaints about Guardian and

Human Resources / Eleanor Haugh, Benefits Manager User Department

1/10/2013 Date

s: evaluation

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0016

Sponsored by: County Executive
FitzGerald/Fiscal Officer/Office
of Budget & Management

A Resolution amending the 2012/2013 Biennial Operating Budget for 2012 by additional providing for fiscal appropriations from the General Fund and other funding sources and appropriation transfers between budget accounts, in order to meet the budgetary needs of various County departments, offices and agencies related to year-end close-out activities, in accordance with Resolution No. R2012-0228.

WHEREAS, on December 13, 2011, the Cuyahoga County Council adopted the 2012/2013 Biennial Operating Budget and Capital Improvements Program (Resolution No. R2011-0291) establishing the 2012/2013 biennial budgets for all County departments, offices and agencies;

WHEREAS, it is necessary to adjust the Biennial Operating Budget for 2012 to reflect budgetary funding increases, funding reductions, and to transfer budget appropriations in order to accommodate the operational needs of certain County departments, offices, and agencies; and

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the 2012/2013 Biennial Operating Budget for 2012 be amended to provide for the following additional appropriation increases and decreases, to be determined by the Office of Budget & Management (OBM), for the purpose of year-end reconciliation of accounts, provided that the maximum amount of any one item shall not exceed \$200,000.00, unless the additional appropriation increase is offset by a corresponding decrease of the same amount, except that any one item identified to fund pending service charge(s) shall not exceed \$700,000.00.

Fund Nos./Budget Accounts

Journal Nos.

ITEMS SUBMITTED THROUGH DECEMBER 11, 2012

Resolutions: Additional Appropriations

A1. 24A601 – Senior and Adult Services

BA1201035

SA138321 – Administrative Services-SAS Other Expenses \$ (72,281.00)

A2. 24A601 – Senior and Adult Services
SA138305 – Community Social Services Programs
Other Expenses \$ (9,009.00)

Appropriation decreases are requested to reduce excess appropriations in various index codes in the contracts and client services lines for Senior and Adult Services. The funding source is the Health and Human Services Levy Fund, the Public Assistance allocations, and fees for direct services provided to clients.

ITEMS SUBMITTED THROUGH DECEMBER 18, 2012

B1. 21A421 – Sex Offender Treatment FY12-13 BA1201050

JC753558 – Youth Services Subsidy – FDCC -2106

Other Expenses \$ (15,778.80)

B2. 21A421 –Sex Offender Treatment FY12-13
JC753541 – Substance Abuse Treatment FY12-13 -2107
Other Expenses \$ (13,636.85)

B3. 21A421 –Sex Offender Treatment FY12-13 JC753533 – Clinical Assessment & SA FY12 -2217 Other Expenses \$ (33,392.98)

This request removes the excess appropriation from the grant budget. Funding derives from RECLAIM Ohio grants from the Ohio Department of Youth Services for sex offender, substance abuse, and clinical assessment services.

C1. 54A901 – Sanitary Engineer/Note Retirement ST540625 – Sanitary Engineer – Note Retirement Other Expenses \$ (17.00)

C2. 54P900 – Sanitary Engineer Bond Retirement DV752006 – Bond Retirement Chagrin Highlands 543A Other Expenses \$ 17.00

An appropriation decrease/increase is requested to pay the principal payment on the Chagrin Highlands Sanitary Sewer bonds. The original budget was incorrect by \$17. Funding for the Chagrin Highlands Bond Retirement comes from the proceeds of its Bond Offering.

D. 01A001 - General Fund BA1201044
AE511451 - Board and Care of Prisoners
Other Expenses \$ 150,000.00

Request to increase appropriation to the Board and Care budget to cover anticipated payments to municipal jails for the housing of County prisoners in response to over-crowding in the County Jail. This budget is exclusively supported by the County's General Fund, though it

should be noted that there are Federal and State prisoners in the County Jail that the County is housing for a fee: this revenue offsets the cost of sending County prisoners to outside jails.

ITEMS SUBMITTED THROUGH DECEMBER 26, 2012

E. 20A331 – Indigent Guardianship
PC404665 – Indigent Guardianship
Other Expenses \$ 25,000.00

Request to increase appropriation to Probate Court's Indigent Guardianship Fund to support a transfer of cash from the Court to the County's Department of Senior and Adult Services in support of the Adult Guardianship Program managed by Lutheran Metropolitan Ministries (via a contract with DSAS). The Indigent Guardianship Fund is supported by filing fees in the Court. This request does not impact the County's General Fund budget.

F. 01A001 – General Fund BA1201049
AE511451 – Board and Care of Prisoners
Other Expenses \$ 50,000.00

Request to increase appropriation to the Board and Care budget to cover anticipated payments to municipal jails for the housing of County prisoners in response to over-crowding in the County Jail. This budget is exclusively supported by the County's General Fund, though it should be noted that there are Federal and State prisoners in the County Jail that the County is housing for a fee: this revenue offsets the cost of sending County prisoners to outside jails.

G. 20A812 – Common Pleas Special Project I
SH456483 – Sheriff's Dept Special Project I
Personal Services \$ 5,060.00

Provide additional appropriations in the Sheriff's Department Special Project I from the Common Pleas Court Special Project Fund operated for foreclosure activities. Sufficient funds exist in the fund for the additional appropriation. Funding covers the period January 1, 2012 through December 31, 2012.

H. 54A100 – Sanitary Engineer BA1201112 ST540583 – Sanitary Engineer – Debt Services Other Expenses \$ 200,000.00

Appropriation is requested to provide for loan repayments for sanitary sewer development projects. Funding is generated from sewer district assessments.

I. 22A046 – Community Development Block Grant 2011 **BA1201113**DV713362 – Project Plan 2011
Other Expenses \$ (192,197.05)

A decrease is requested in the 2011 CDBG Project Plan grant to move appropriation to the 2012 Project Plan grant. The source of funding is the U.S. Dept. of Housing and Urban Development.

J. 61A607 – Centralized Custodial Services BA1201090

CT577395 – Buildings and Grounds – Trades Services
Personal Services \$ 169,795.61

Appropriation is requested to cover workers' compensation and other year end payroll expenses. Funding is generated from charges from user agencies for space maintenance.

K. 24A435 –Cuy Tapestry System of Care

CF135004 – DCFS-Cuy Tapestry System of Care

Personal Services \$ (3,000.00)

Other Expenses \$ (103,000.00)

The Department of Children and Family Services has requested appropriation decreases to move appropriations from the Tapestry (CTSOC) fund to cover projected shortfalls in the Public Assistance funds within the department for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund. (See related additional appropriation, item L1-L5, below.)

L1.	24A301 – Children and Family Services CF135483 – Training			BA1200869
	Personal Services	\$	500.00	
L2.	24A301 – Children and Family Services CF135491 – Information Services	vices		
	Personal Services	\$	27,500.00	
L3.	24A301 – Children and Family Serv CF135525 – Supportive Services	vices		
	Personal Services	\$	6,000.00	
L4.	24A301 – Children and Family Serv CF135582 – Permanent Custody Ad			
	Personal Services	\$	18,000.00	
L5.	24A301 – Children and Family Serv			
	CF135616 – CFS Foster Homes/Re	source Mgt		
	Personal Services	\$	54,000.00	

The Department of Children and Family Services has requested appropriation increases to move appropriations from the Tapestry (CTSOC) fund to cover projected shortfalls in the Public Assistance funds within the department for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund. (See related decrease appropriation, item K, on page 3.)

ITEMS SUBMITTED THROUGH JANUARY 2, 2013

M. 20A192 – TASC HHS BA1201062 CO456533 – TASC HHS Other Expenses \$ 99,500.00 Request to increase appropriation to the Treatment Alternative to Street Crimes (TASC) program in the Court of Common Pleas to enable the space maintenance chargeback to post. This program is supported by a combination of grants and the County's Health and Human Services levy; the chargebacks are solely supported by the HHS Levy Fund.

N. 20R320 – Board of Developmental Disabilities BA1201094 MR845024 – County Bd of Developmental Disabilities Other Expenses \$ (1,052,577.00)

Appropriation decreases are requested to reduce excess appropriations in the commodities, contracts, client services, and other operating lines for the Board of Developmental Disabilities. The Cuyahoga County Board of Developmental Disabilities is supported through its own levy and the Ohio Rehabilitation Services Commission. (See related additional appropriation, item T, on page 7.)

O. 24A510 – Work & Training Admin BA1201095 WT137315 – Work First Services Other Expenses \$ (318,041.66)

Request to decrease appropriation within Job and Family Services. Appropriation is being decreased because corresponding contracts have been decertified and funding is no longer available. The corresponding contracts were for services prior to 2011. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

P1. 20A303 – Children Services Fund CF134031 – CFS Foster Care Other Expenses \$ (235,000.00)

P2. 20A303 – Children Services Fund CF134049 – Purchased Congregate & Foster Care Other Expenses \$ (1,200,000.00)

The Department of Children and Family Services has requested appropriation decreases to move appropriation surpluses from the Children Services Fund to cover projected shortfalls in the Public Assistance Funds for controlled services charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

Q. 24A301 – Children and Family Services
CF135467 – Administrative Services
Other Expenses \$ 1,435,000.00

The Department of Children and Family Services has requested appropriation increase to offset the move of appropriation surpluses from the Children Services fund to cover projected shortfalls in the Public Assistance funds in Controlled Services charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

R1. 22A956 – Community Development Block Grant 2005 BA1201115 DV709824 – Project Plan 2005 Other Expenses \$ (11,680.00)

R2.	22A970 – Community Development Block Grant 2006 DV712224 – Project Plan 2006						
	Other Expenses	\$	(33,848.67)				
R3.	22A975 – Community Developmen DV712430 – Project Plan 2007		ant 2007				
	Other Expenses	\$	(0.87)				
R4.	22A984 – Community Developmen DV712711 – Project Plan 2008	t Block Gra	ant 2008				
	Other Expenses	\$	(5,780.18)				
R5.	22A795 – Community Development DV713032 – Project Plan 2009						
	Other Expenses	\$	(115,843.57)				
R6.	22A795 – Community Developmen DV713016 – Project Plan 2009		ant 2009				
	Other Expenses	\$	(1,647.27)				
R7.	22A795 – Community Developmen DV713008 – Administrative Operat		ant 2009				
	Other Expenses	\$	(6,857.87)				
R8.	22A613 – Community Developmen DV713255 – Project Plan 2010		ant 2010				
	Other Expenses	\$	(88,804.74)				
R9.	22A613 – Community Development Block Grant 2010 DV713263 – Administrative Operations 2010						
	Personal Services	\$	(27,821.65)				
	Other Expenses	\$	(70,321.80)				
	Capital Outlays	\$	(4,035.75)				
R10.	22A613 – Community Development DV713271 – Rehabilitation Operation		ant 2010				
	Personal Services	\$	(29,632.12)				
	Other Expenses	\$	(35,779.11)				
	Capital Outlays	\$	(2,968.75)				
R11.	22A613 – Community Development DV713289 – Program Operations 20		ant 2010	BA1201116			
	Personal Services	\$	(1,326.49)				
	Other Expenses	\$	(30,940.46)				
	Capital Outlays	\$	(308.75)				
R12.	22A046 – Community Development Block Grant 2011 DV713362 – Project Plan 2011						
	Other Expenses	\$ ((4,073,480.12)				
R13.	22A046 – Community Developmen	t Block Gra	ant 2011				

DV713370 – Administration Operations 2011

Other Expenses \$ (5,000.00)

R14. 22A685 – Community Development Block Grant 2012

DV713917 – Project Plan 2012

Other Expenses \$ 4,546,078.17

Appropriation decreases with corresponding appropriation increases are requested from 2005-2011 CDBG grants to the new 2012 grant. The source of funding is the U.S. Dept. of Housing and Urban Development.

S1. 22A685 – Community Development Block Grant 2012 **BA1201117**

DV713883 – Administration Operations 2012

Other Expenses \$ 101,309.27

S2. 22A685 – Community Development Block Grant 2012

DV713909 – Rehabilitation Operations 2012

Other Expenses \$ 58,320.60

S3. 22A685 – Community Development Block Grant 2012

DV713891 - Program Operations 2012

Other Expenses \$ 32,567.18

Increases are requested for the 2012 CDBG grant. A decrease of appropriation in a corresponding amount (BA1201113) provides funding for these increases. The source of funding is the U.S. Dept. of Housing and Urban Development. (Appropriation decrease was approved on the December 11, 2012, R2012-0257.)

ITEMS SUBMITTED THROUGH JANUARY 8, 2013

T. 20R320 – Board of Developmental Disabilities BA1201291 MR845024 – County Bd of Developmental Disabilities

Other Expenses \$ 1,052,577.00

Appropriation increases are requested to restore appropriations in the commodities, contracts, client services, and other operating lines for the Board of Developmental Disabilities. The Cuyahoga County Board of Developmental Disabilities is supported through its own levy and the Ohio Rehabilitation Services Commission. (See related decrease appropriation, item N, on page 4.)

U. 40A524 – County Engineer BA1201260 CE785170 – County Engineer – West Bridge Street Bridge #256

Capital Outlays \$ 1,896.00

Appropriation is requested to reclassify an expense that was a capital outlay for contract CE785170. Funding is generated from State Issue 1.

V. 20D447 – Western Reserve Fund **BA1201261**

DV520676 – Community Development

Other Expenses \$ (19,732,407.60)

BA1201114

Appropriation decrease is requested to remove unused appropriation. The source of funding for the fund will be proceeds from County bonds.

W. 62A603 – County Garage BA1201261

CT575001 – Maintenance Garage

Personal Services \$ (32,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

X. 62A603 - County Garage BA1201261

CT575001 – Maintenance Garage

Personal Services (16,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

Y. 62A603 – County Garage BA1201261

CT575001 – Maintenance Garage

\$ Other Expenses (94,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

Z. 62A603 – County Garage BA1201261

CT575001 – Maintenance Garage

Other Expenses (4,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

BA1201261 AA. 62A603 – County Garage

CT575001 - Maintenance Garage

Other Expenses (409,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

AB. 62A603 - County Garage BA1201261

CT575001 – Maintenance Garage

Other Expenses (2,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

AC. 63A100 – Information Services Center

> IS821009 - Cuyahoga County Information Services Center Other Expenses 243,000.00 \$

BA1201253

An increase in appropriation is requested to cover 2012 expenses. Of this increase, \$180,000.00 would go to the controlled services line to cover indirect expense chargebacks. The other \$63,000.00 would cover 2012 expenses associated with the County's wide area network. Funding comes from charges assessed to user agencies for data processing and telephone use.

AD. 64A601 – Office Supplies BA1201288
FS109751 – County Office Supply Contract
Other Expenses \$ (36,200.00)

A decrease would remove surplus appropriation from the 2012 budget. Funding comes from charges to agencies for the purchase of office supplies.

AE. 68A100 – Hospitalization – Self Insurance Fund CC499004 – Hospitalization – Self Insurance Other Expenses \$ (324,000.00)

This request would decrease appropriation from the 2012 budget. Revenues are generated from charges for co-payments to the employees and the County.

AF. 68A200 – Hospitalization – Regular Insurance Fund CC499012 – Hospitalization – Regular Insurance Other Expenses \$ (1,400,000.00)

A decrease would remove surplus appropriation from the 2012 budget. Funding for this account comes from contributions made from the County as well as payroll deductions from employees.

AG. 67A011 – Workers' Compensation Retrospective 2011 BA1201288
CC498881- Workers' Compensation Retrospective 2011
Other Expenses \$ (420,000.00)

This request would decrease unneeded appropriation from the 2012 retrospective account budget. Funding comes from charges to departments to reimburse for the costs of the premium and claims based on injury experience and the size of each department to calculate a risk level.

AH. 63A100 – Information Services Center IS690107 – Information Services Center – Overhead Other Expenses \$ 118,000.00

Additional appropriation would cover 2012 expenses for the wide area network and would cover data processing expenses that will be charged back to user agencies once substantiated. Funding comes from charges assessed to user agencies for data processing and telephone use.

AI. 01A001 – General Fund BE473058 – General Election
Personal Services \$ (170,000.00)
Other Expenses \$ (1,230,000.00)

A decrease in appropriation would remove surplus appropriation from the 2012 budget. Funding comes from the General Fund. (See related additional appropriation, item AJ, below.)

AJ. 01A001 – General Fund BA1201220 JC372052 – Juvenile Court – Judges Other Expenses \$ 1,400,000.00

The space maintenance charges increased substantially for the Juvenile Court in its first full year of occupancy in the new Juvenile Justice Center. The 2012 appropriation did not accommodate this increase, therefore an increase in appropriation is requested for this purpose. Funding comes from the General Fund. (See related decrease appropriation, item AI, above.)

AK. 20A811 – Juvenile Court Detention & Probation Services

JC107516 – Juvenile Court – Detention Services

Other Expenses \$ 65,000.00

The Juvenile Court budget did not have sufficient appropriation for the increased costs of space maintenance at the new Juvenile Justice Center. This increase would resolve this deficit. An equal decrease in appropriation is requested from the Juvenile Court Youth and Family Community Partnership fund in BA1201222. This budget receives funding from the Health and Human Services Levy Fund. (See related appropriation decrease, item AL, below.)

AL. 20A823 – Youth & Family Community Partnership

JC108092 – JC Youth & Family Community Partnership

Other Expenses \$ (65,000.00)

This budget decrease would balance an increase of \$65,000 in the Juvenile Court Detention & Probation Services fund requested in BA1201221 to cover increased space maintenance costs at the Juvenile Justice Center. Funding comes from the Health and Human Services Levy Fund. (See related additional appropriation, item AK, above.)

AM1. 01A001 – General Fund **BA1201168**

SH350272 – Law Enforcement-Sheriff

Other Expenses \$ 200,000.00

AM2. 01A001 – General Fund

SH350470 – Jail Operations-Sheriff

Other Expenses \$ 750,000.00

Requesting additional appropriations in the Sheriff Department for vehicle chargebacks in the Law Enforcement Division and space maintenance charges in Jail. Funding is from General Fund covering the period January 1, 2012 through December 31, 2012. Other General Fund accounts have been reduced providing for the increase. (See related decrease appropriation, item AN, below.)

AN. 01A001 – General Fund **BA1201167**

BE473058 – General Elections

Personal Services \$ (950,000.00)

Provide reduce appropriations in the Board of Elections, General Elections account. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012. (See related additional appropriation, item AM1 and AM2, above.)

AO.	20A809 – Witness Victim HHS	BA1201163
	JA107425 – Witness Victim HHS	
	Personal Services	\$ (375,311.61)
	Other Expenses	\$ (85,596.39)
	Capital Outlays	\$ (23,115.00)

Reducing appropriations in the Department of Public Safety and Justice Services Witness Victim HHS account in accordance to Generally Accepted Accounting Principles (GAAP). Funding is from the Health and Human Service Levies covering the period January 1, 2012 through December 31, 2012.

AP.	24A878 – Public Assistance Funds		BA1201164	
	HS749069 – HHS-Office of	ReEntry		
	Personal Services	\$	(25,100.00)	
	Other Expenses	\$	(672,007.00)	
	Capital Outlays	\$	(533.00)	

Reducing appropriations in the Office of ReEntry HHS account in accordance to Generally Accepted Accounting Principles (GAAP). Funding is from the Health and Human Service Levies covering the period January 1, 2012 through December 31, 2012.

AQ.	20A891 – Common Pleas H	HHS Subsidy	BA1201211
	CO456517 – Common Pleas HHS Subsidy		
	Other Expenses	\$	(113,300.00)

Reducing appropriations in the Common Pleas HHS Subsidy account for year-end expenses in another HHS account in Common Pleas (see corresponding increase to CO456533, document BA1201212 fiscal item AV, on page 12). Funding is from the Health and Human Services Levy Fund covering the period January 1, 2012 through December 31, 2012.

20A192 – TASC HHS		BA1201212
CO456533 – TASC HHS		
Personal Services	\$	3,300.00
Other Expenses	\$	110,000.00
	CO456533 – TASC HHS Personal Services	CO456533 – TASC HHS Personal Services \$

Requesting additional appropriations in the Common Pleas TASC HHS account for year-end expenses, specifically workers compensation charges and space maintenance. See reduction in another Common Pleas HHS account for the same amount (see corresponding decrease to CO456517, document BA1201211 fiscal item AU, on page 11). Funding is from the Health and Human Services Levies covering the period January 1, 2012 through December 31, 2012.

AS.	20A192 – TASC HHS		BA1201176
	CO456533 – TASC HHS		
	Other Expenses	\$ 114,000.00	

Request to increase appropriation to the Treatment Alternative to Street Crimes budget in the Court of Common Pleas to support 2012 space maintenance chargebacks. This program is

supported by grants and the County's Health and Human Services Levy, the latter of which exclusively covers space charges. This request is offset by a corresponding reduction in the Court's Probation/Psychiatric division, which is also subsidized by the County's Health and Human Services Fund. (See related decrease appropriation, item AT, below.)

AT. 20A891 – Common Pleas HHS Subsidy
CO456517 – Common Pleas HHS Subsidy
Other Expenses \$ (114,000.00)

Request to reduce surplus appropriation to the Court of Common Pleas' Probation/Psychiatric division to support an increase in the Treatment Alternative to Street Crimes' budget, both of which are supported by the County's Health and Human Services Fund. (See related additional appropriation, item AS, above.)

AU1.	52A100 – County Airport DV520031 – County Airport	•	271 020 71	BA1201129
	Other Expenses	\$	271,028.71	
AU2.	65A604 – Postage CT577353 – County Mailroom			BA1201128
	Other Expenses	\$	35,000.00	
AU3.	01A001 – General Fund CT577106 – Risk and Property Ma	nagement		
	Personal Services	\$	12,500.00	
AU4.	01A100 – General Fund	N		BA1201119
	AE210005 – Soldiers' and Sailors' Other Expenses	Monument \$	11,000.00	
	Office Expenses	Ψ	11,000.00	

Appropriation is requested to cover year end payroll, space maintenance, data processing, and indirect cost charges. The source of funding for the Airport and County Mailroom is user fees and a General Fund subsidy for the Airport.

AV. 22A046 – Community Development Block Grant Proj.

FY 2011

BA1201144

DV713362 – CDBG Year 37 2011

Other Expenses \$ 192,197.05

This item reverses an earlier decrease on the year end resolution (BA1201113) that was already approved by Council on the December 11, 2012 agenda, R2012-0257. (See related decrease appropriation, item I, on page 3.)

AW. 22A685 – Community Development Block Grant 2012 **BA1201145**DV713917 – Project Plan 2012
Other Expenses \$ (4,546,078.17)

This item reverses an earlier increase on the year end resolution (BA1201114) that was already approved by Council on the December 11, 2012 agenda, R2012-0257. (See related additional appropriation, item R14, on page 6.)

AX1.	20A600 - General Fund			BA1201239
	SE496000 – Support Enforcem	nent Agency		
	Personal Services	\$	(740,023.00)	
	Other Expenses	\$	(1,937,270.00)	
	Capital Outlay	\$	(85,000.00)	
AX2.	24A430 - General Fund			BA1201239
	HS157289 – Executive Office	for H&HS		
	Personal Services	\$	(55,958.00)	
	Other Expenses	\$	(136,200.00)	
	Capital Outlay	\$	(2,000.00)	

A request to reduce surplus appropriation. Both agencies are supported by the Health and Human Services Levy Fund.

AY.	20A810 – Criminal Justice Inte	ervention HHS	BA1201215
	JA107433 – Criminal Justice In	ntervention HHS	
	Personal Services	\$	(147,141.72)

Reducing appropriations in the Department of Public Safety and Justice Services Criminal Justice Intervention HHS account in accordance to Generally Accepted Accounting Principles (GAAP). Funding is from the Health and Human Service Levies covering the period January 1, 2012 through December 31, 2012.

AZ.	20A809 – Witness Victim HHS	BA1201216
	JA107425 – Witness Victim HHS	
	Other Expenses	\$ 3,890.00

Requesting to increase appropriations in the Department of Public Safety and Justice Services Witness Victim HHS account. This increase is a correction to previous reduction seen on BA1201163 (fiscal item AR, on page 11). Funding is from the Health and Human Service Levy Fund covering the period January 1, 2012 through December 31, 2012.

BA1.	30A905 – Gateway Arena			BA1200957
	DS100370 - Gateway Arena Pro	ject		
	Other Expenses	\$	(29.54)	
BA2.	30A910 – Brownfield Debt Serv DS039966 – Brownfield Debt Se			
	Other Expenses	\$	(2,610.57)	
BA3.	30A913 – Community Redevelo	pment Deb	t Service	
	DS040121 – Community Redeve	elopment D	ebt Service	
	Other Expenses	\$	(8,646.90)	

Appropriation reductions are requested for three debt service accounts: Gateway Arena Project, Brownfield Debt Service and Community Redevelopment Debt Service. The appropriation reductions will bring the actual debt service and the budgeted debt service into balance and thereby zero out the remaining free balance. Funding for the debt service accounts comes from the General Fund.

BB1.	24A510 – Work & Training Admin WT137109 – Admin Services-Gene	BA1201194					
	Other Expenses	\$ \$	(1,686,500.00)				
	Capital Outlay	\$ \$	(3,500.00)				
	Capital Outray	φ	(3,300.00)				
BB2.	24A510 – Work & Training Admin WT137141 – Client Support Services						
	Other Expenses	\$	(249,800.00)				
	Capital Outlay	\$	(9,800.00)				
BB3.	24A510 – Work & Training Admin WT137315 – Work First Services						
	Personal Services	\$	(6,000.00)				
	Other Expenses	\$	(703,000.00)				
BB4.	24A510 – Work & Training Admin WT137414 – Southgate NFSC	\$	(16,100.00)				
	Other Expenses Capital Outlay	\$ \$	(6,000.00)				
	Capital Outlay	Φ	(0,000.00)				
BB5.	24A510 – Work & Training Admin WT137430 – Ohio City NFSC						
	Other Expenses	\$	(767,000.00)				
	Capital Outlay	\$	(6,400.00)				
BB6.	24A510 – Work & Training Admin WT137455 – Quincy Place NFSC Other Expenses	\$	(128,500.00)				
BB7.	24A510 – Work & Training Admin WT137463 – VEB Building NFSC						
	Other Expenses	\$	(1,090,000.00)				
BB8.	24A510 – Work & Training Admin WT137471 – Mount Pleasant NFSC	2					
	Other Expenses	\$	(197,000.00)				
	Capital Outlay	\$	(4,400.00)				
BB9.	24A510 – Work & Training Admin WT137539 – West Shore NFSC			BA1201195			
	Other Expenses	\$	(129,500.00)				
	Capital Outlay	\$	(7,000.00)				
BB10.	24A530 – Children With Medical H WT137935 – Children With Medica Other Expenses	_	ap (341,000.00)				
BB12.	24A510 – Work & Training Admin WT137943 – Information Services						
	Other Expenses	\$	(276,830.00)				

The Department of Jobs and Family Services has requested to decrease excess appropriation. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

BC1.	24A601 – Senior and Adult Services SA138321 – Administrative Services-SAS		BA1201195			
	Other Expenses	\$	(172,700.00)			
BC2.	24A601 – Senior and Adult Services					
	SA138354 – Management Services					
	Personal Services	\$	(337,800.00)			
	Other Expenses	\$ \$	(10,000.00)			
	Capital Outlay	\$	(14,000.00)			
BC3.	24A601 – Senior and Adult Services					
	SA138388 – Home Care Skilled Se	rvices				
	Personal Services	\$	(31,900.00)			
	Other Expenses	\$	(22,000.00)			
BC4.	24A601 – Senior and Adult Services					
	SA138420 – Home Support					
	Personal Services	\$	(17,000.00)			
	Other Expenses	\$	(90,300.00)			
BC5.	24A601 – Senior and Adult Services					
	SA138479 – Protective Services					
	Personal Services	\$	(47,500.00)			
	Other Expenses	\$	(20,800.00)			
	Capital Outlay	\$	(4,500.00)			

The Department of Senior and Adult Services has requested to decrease excess appropriation. The funding source is the Health and Human Services Levy Fund, the Public Assistance allocations, and fees for direct services provided to clients.

BD1.	20D446 – Brownfield Revolving Lo DV520726 – Brownfield Revolving Other Expenses		(99,880.77)	BA1201914
BD2.	64A601 – Supplies CC577700 – County Supplies Other Expenses	\$	(21,854.39)	
BD3.	64A601 – Supplies FS109751 – Fiscal – County Supplie Other Expenses	es \$	(5,100.00)	
BD4.	64A606 – Fast Copier CT577551 – Fast Copy Other Expenses	\$	(886,253.53)	

Reductions are requested to remove surplus appropriation. Funding for the Brownfield Revolving Fund is from loan repayments and the Supplies and Fast Copier fund generate revenues from charges for supplies procurement and print shop and fast copier services.

BE1.	20A303 – Children Services Fund CF134049 – Purchased Congregate Care			BA1200273
	Other Expenses	\$	(616,000.00)	
BE2.	20A303 – Children Services Fund CF134031 – CFS Foster Care			
	Other Expenses	\$	(164,000.00)	
BE3.	20A303 – Children Services Fund CF134023 – Adoption Services			
	Other Expenses	\$	(2,000.00)	
BE4.	20A303 – Children Services Fund CF134015 – Client Supportive Serv	rices		
	Other Expenses	\$	(61,000.00)	

Reduction of unused appropriation for Children Services Fund. Funding is from the Federal and State revenues, as well as the Health & Human Services Levy Fund

BF1.	24A301 - Children & Family Serv CF135467-CFS Administration	ices		BA1200883		
	Personal Services	\$	(60,000.00)			
	Other Expenses	\$	(759,000.00)			
BF2.	24A301 - Children & Family Serv CF135483-Training	ices				
	Other Expenses	\$	(74,000.00)			
BF3.	24A301 - Children & Family Serv	ices				
	CF135491-Information Services					
	Personal Services	\$	(14,000.00)			
	Other Expenses	\$	(264,000.00)			
	Capital Outlays	\$	(4,300.00)			
BF4.	24A301 - Children & Family Services					
	CF135509-Direct Services					
	Personal Services	\$	(261,700.00)			
	Other Expenses	\$	(146,500.00)			
BF5.	24A301 - Children & Family Services					
	CF135525-Supportive Services					
	Personal Services	\$	(12,000.00)			
	Other Expenses	\$	(1,813,500.00)			
BF6.	24A301 - Children & Family Services CF135541-Visitation Center					
	Other Expenses	\$	(94,000.00)			

BF7. 24A301 - Children & Family Services
CF135582-Permanent Custody Adoption
Other Expenses \$ (15,000.00)

Appropriation decreases are requested to reduce excess appropriations in various index codes in the salaries lines in Children and Family Services. The funding source is primarily the Health and Human Services levies.

BG1.	24A301 - Children & Family Services CF135582-Permanent Custody Adoption			BA1200884		
	Other Expenses	\$	(9,000.00)			
BG2.	24A301 - Children & Family Services CF135608-Contracted Placements					
	Other Expenses	\$	(149,000.00)			
BG3.	24A301 - Children & Family Servi CF135616-CFS Foster Homes	ces				
	Personal Services	\$	(15,000.00)			
	Other Expenses	\$	(27,500.00)			
BG4.	24A435 - Cuyahoga Tapestry System of Care (CTSOC) CF135004 – DCFS- Cuyahoga Tapestry System of Care					
	Other Expenses	\$	(2,009,000.00)			
BG5.	24A640 - FCFC Public Assistance FC451492 – Family and Children First Council PA					
	Personal Services		(15,000.00)			
		\$				
	Other Expenses	\$	(558,000.00)			
BG6.	24A635 - EC- Invest In Children PA EC451484 – Early Childhood Admin Services					
	· · · · · · · · · · · · · · · · · · ·					
	Other Expenses	\$	(101,000.00)			
BG7.	24A635 - EC- Invest In Children PA					
	EC451450 – Quality Child Care	ф	(27.500.00)			
	Other Expenses	\$	(27,500.00)			
BG8.	24A641 - PA-Homeless Services HS158097 – PA-Homeless Services					
		\$ \$	(26,000,00)			
	Other Expenses	Φ	(26,000.00)			

Appropriation decreases are requested to reduce excess appropriations in several departments' index codes within Health and Human Services budget lines. The funding source is primarily the Health and Human Services levies.

BH1.	01A001 – General Fund		BA1201928
	PD140053 – Public Defender		
	Other Expenses	\$ 30,000.00	

BH2. 01A001 – General Fund

PR191056 - Prosecutor-General Office

Other Expenses \$ 144,000.00

BH3. 01A001 – General Fund

DR391052 - Domestic Relations Court

Other Expenses \$ 35,000.00

BH4. 01A001 – General Fund

CO380121 - Common Pleas-Judicial/General

Other Expenses \$ 175,000.00

BH5 20A600 – Cuyahoga Support Enforcement Agency

SE496000 – Child Support Enforcement Agency

Other Expenses \$ 15,000.00

BH7. 20A809 – Witness Victim HHS

JA107425 – Witness Victim HHS

Other Expenses \$ 15,000.00

Provide increased appropriations in the various departments for November 2012 data processing Funding is from the various funding sources including General Fund and Health and Human Service Levies covering the period January 1, 2012 through December 31, 2012.

BI. 24A635 - EC- Invest In Children PA

BA1200885

EC451484 – Early Childhood Admin Services

Personal Services \$ (7,000.00)

An appropriation decrease is requested to offset corresponding increase in appropriations for the Family and Children First Council (FCFC) budget line in order to cover data center charges in controlled services budget line. The funding source is primarily the Health and Human Services levies.

BJ. 24A640 - FCFC Public Assistance

FC451492 – Family and Children First Council PA BA1200886

Other Expenses \$ 7,000.00

An appropriation increase is requested to offset corresponding decrease in appropriations in Early Childhood for unused appropriation and to cover data center charges in the Family and Children First Council controlled services budget line. The funding source is primarily the Health and Human Services levies.

SECTION 2. That the 2012/2013 Biennial Operating Budget for 2012 be amended to provide for the following appropriation transfers, to be determined by OBM, for the purpose of year-end reconciliation of budget accounts and line items, provided that transfer items adhere to the currently established financial policies for processing transfers:

Fund Nos. /Budget Accounts

Journal Nos.

ITEMS SUBMITTED THROUGH DECEMBER 11, 2012

Resolutions: Appropriation Transfers

A. FROM: 24A601 – Senior and Adult Services BA1201008

SA138479 – Protective Services

Other Expenses \$ 8,000.00

TO: 24A601 – Senior and Adult Services

SA138602 – Home Based Services

Other Expenses \$ 8,000.00

The Department of Senior and Adult Services has requested an appropriation transfer to realign appropriations within index codes to cover projected travel reimbursement expenses for the remainder of the year. The funding source is the Health and Human Services Levy Fund, the Public Assistance allocations, and fees for direct services provided to clients.

B. FROM: 61A608 – Central Custodial Services BA1200936

SH352005 – Building Security Services-OPBA-Officers Other Expenses \$ 23,000.00

TO: 61A608 – Central Custodial Services

SH352005 – Building Security Services-OPBA-Officers Capital Outlay \$ 23,000.00

The appropriation transfer is to fund replacement guns in the Sheriff's Protective Services Division. Funding is from the internal service fund which derives revenues from charges to county departments which utilize the services of security for the buildings. The internal service fund covers the period January 1, 2012 through December 31, 2012.

C. FROM: 21A837 – State Homeland Security (SHSP) BA1200993

JA763425 – State Homeland Security (SHSP) 2009/2012 Capital Outlays \$ 583.43

TO: 21A837 – State Homeland Security (SHSP)

JA763425 – State Homeland Security (SHSP) 2009/2012 Personal Services \$ 583.43

To provide sufficient appropriations in personal services for a pending expense adjustment by year's end in the State Homeland Security Grant. Funding is from the United States Department of Homeland Security passed through the Ohio Emergency Management Agency covering the period August 1, 2010 through September 30, 2012.

D. FROM: 21A579 – VAWA Administration Grant BA1201013

JA752907 – FY2010 VAWA Administration Fund CY2011

Other Expenses \$ 2,814.91

TO: 21A579 – VAWA Administration Grant

JA752907 - FY2010 VAWA Administration Fund CY2011 \$

Personal Services

2.814.91

To provide sufficient appropriations in personal services for a pending expense adjustment for closure of the grant. Funding is from the State of Ohio, Office of Criminal Justice Services covering the period January 1, 2011 through December 31, 2011.

ITEMS SUBMITTED THROUGH DECEMBER 18, 2012

E. FROM: 01A001 - General Fund BA1201051

HC019018 - Human Resource Commission

Other Expenses \$ 16,000.00

TO: 01A001 – General Fund

HC019018 – Human Resource Commission

Personal Services 16,000.00

An appropriation transfer of surplus budget from contractual services to cover a projected payroll budget shortfall. Funding comes from the General Fund.

F. FROM: 20A301 – Real Estate Assessment BA1201052

FS109702 – Tax Assessments

37,000.00 Other Expenses

TO: 20A301 – Real Estate Assessment

BR420067 - Board of Revision

Other Expenses 37,000.00

A transfer is requested to purchase software that provides scanning and electronic file capability in preparation of anticipated appeals to the sexennial reappraisal. The source of funding is a percentage of real property tax receipts.

G. FROM: 01A001 - General Fund BA1201042

CO380121 - Judicial Administration

Other Expenses 200,000.00

TO: 01A001 - General Fund

CO380410 – Probation Psychiatric

Personal Services 200,000.00

Request to transfer appropriation within the General Fund of the Court of Common Pleas to cover fringe benefit expenses in the Probation division through year-end.

H. FROM: 01A001 - General Fund BA1201041

CO380121 – Judicial Administration

Other Expenses 217,000.00

01A001 - General Fund

CO380220 - Court Services

Capital Outlays 11,000.00 01A001 - General Fund

CO380410 – Probation Psychiatric

Capital Outlays \$ 15,000.00

TO: 01A001 – General Fund

CO380196 – Magistrates

Personal Services \$ 89,000.00 Other Expenses \$ 9,000.00

01A001 - General Fund

CO380121 – Judicial Administration

Personal Services \$ 119,000.00 Capital Outlays \$ 26,000.00

Request to transfer appropriation within the General Fund budget in the various divisions of the Court of Common Pleas to cover anticipated expenses through the rest of the year.

I. FROM: 01A001 – General Fund **BA1201043**

MI512657 - Miscellaneous

Other Expenses \$ 9,000.00

TO: 01A001 – General Fund

MI512657 - Miscellaneous

Personal Services \$ 9,000.00

Request to transfer appropriation within the Miscellaneous Obligations budget to cover projected legacy expenses related to the Early Retirement Incentive Program implemented in 2009. This budget is supported by the County's General Fund.

ITEMS SUBMITTED THROUGH DECEMBER 26, 2012

J. FROM: 01A001 – General Fund **BA1201048**

PC400051 – Probate Court

Other Expenses \$ 55,000.00

TO: 01A001 – General Fund

PC400051 – Probate Court

Personal Services \$ 55,000.00

Request to transfer available appropriation within Probate Court's General Fund budget to post the final payroll of the year.

K. FROM: 01A001 – General Fund **BA1201058**

CO380196 – Magistrates

Personal Services \$ 91,000.00

01A001 – General Fund

CO380220 – Court Services

Personal Services \$ 222,000.00

TO: 01A001 – General Fund

Judicial Administration

Personal Services \$ 113,000.00 Other Expenses \$ 200,000.00

Request to transfer appropriation within the Court of Common Pleas' General Fund budget to post remaining year expenses.

L. FROM: 01A001 – General Fund **BA1201089**

JC372060 - Juvenile Court - Legal

Personal Services \$ 60,000.00

01A001 - General Fund

JC372052 – Juvenile Court – Judges

Personal Services \$ 15,000.00

01A001 - General Fund

JC370056 - Juvenile Court - Detention Home

Personal Services \$ 130,000.00

TO: 01A001 – General Fund

JC372060 – Juvenile Court - Legal

Other Expenses \$ 30,000.00

01A001 - General Fund

JC372052 – Juvenile Court – Judges

Other Expenses \$ 15,000.00

01A001 – General Fund

JC370056 – Juvenile Court – Detention Home

Other Expenses \$ 160,000.00

The requested transfers would clear year-end expenditures in the above budget accounts. Funding comes from the General Fund.

M. FROM: 21A762 – Port Security Grant **BA1201018**

JA767855 – Port Security Grant FFY09-2009/2012 Other Expenses \$ 1,598.22

TO: 21A762 – Port Security Grant

JA767855 – Port Security Grant FFY09-2009/2012 Personal Services \$ 1,598.22

The appropriations transfer is requested to prepare the grant for closure. Funding is from the United States Department of Homeland Security passed through the Ohio Emergency Management Agency covering the period August 1, 2010 through May 31, 2013.

N. FROM: 61A608 – Central Security Services-Sheriff **BA1201025**

SH352005 – Building Security Services-OPBA-Officers Personal Services \$ 81,900.00 TO: 61A608 – Central Security Services-Sheriff

SH352013 – Building Security Services-OPBA-Sergeants Personal Services \$ 77,000.00

TO: 61A608 – Central Security Services-Sheriff

SH352021 – Building Security Services-Non-Bargaining Personnel

Personal Services \$ 4,900.00

The appropriations transfer is to realign appropriations within the Sheriff's Protective Services Division grouped by bargaining and non-bargaining units. Funding is from the internal service fund which derives revenues from charges to county departments which utilize the services of security for the buildings. The internal service fund covers the period January 1, 2012 through December 31, 2012.

O. FROM: 21A500 – Urban Area Security Initiative (URSI) **BA1201029**

JA741199 – Urban Area Security Initiative (URSI) Personal Services \$ 766.06

TO: 21A500 – Urban Area Security Initiative (URSI)

JA741199 – Urban Area Security Initiative (URSI) Capital Outlays \$ 766.06

The appropriations transfer is requested to prepare the grant for closure. Funding is from the United States Department of Homeland Security passed through the Ohio Emergency Management Agency covering the period August 1, 2010 through September 30, 2012.

P. FROM: 61A608 – Central Security Services-Sheriff **BA1201032**

SH352005 – Building Security Services-OPBA-Officers Personal Services \$ 22,050.00

TO: 61A608 – Central Security Services-Sheriff

SH352013 – Building Security Services-OPBA-Sergeants Personal Services \$ 16,250.00

TO: 61A608 – Central Security Services-Sheriff

SH352021 – Building Security Services-Non-Bargaining Personnel

Personal Services \$ 5,800.00

The appropriations transfer is to realign appropriations within the Sheriff's Protective Services Division grouped by bargaining and non-bargaining units. Funding is from the internal service fund which derives revenues from charges to county departments which utilize the services of security for the buildings. The internal service fund covers the period January 1, 2012 through December 31, 2012.

Q. FROM: 01A001 – General Fund **BA1201069**

PR194720 – General Office-Prosecutor

Other Expenses \$ 176,000.00

TO: 01A001 – General Fund

PR194720 – General Office-Prosecutor

Personal Services \$ 176,000.00

The appropriations transfer is to provide sufficient appropriations with the Prosecutor's General Office account for last pay in December. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

R. FROM: 24A301 – Children and Family Services **BA1200867**

CF135467 – Administrative Services

Other Expenses \$ 5,200.00

TO: 24A301– Children and Family Services

CF135467- Administrative Services

Capital Outlays \$ 5,200.00

The Department of Children and Family Services requests an appropriations transfer from one budget line to another within the same index code in order to provide appropriations to the cover approved computer hardware expenses. The Department of Children and Family Services is funded both by Federal and State revenue as well as the HHS Levy Fund.

S. FROM: 24A301 – Children and Family Services **BA1200863**

CF135442 – Caregiver Parent Recruitment

Personal Expenses \$ 1,000.00

24A301 – Children and Family Services CF135467 – Administrative Services

Personal Expenses \$ 7,000.00

24A301 – Children and Family Services

CF135541 – Multi Systemic Therapy (MST) Unit Personal Expenses \$ 36,200.00

24A301 – Children and Family Services CF135608 – Contracted Placements

Personal Expenses \$ 1,500.00

TO: 24A301– Children and Family Services

CF135483 – Administrative Services

Personal Expenses \$ 200.00

24A301 – Children and Family Services

CF135509 - Direct Services

Personal Expenses \$ 41,000.00

24A301 – Children and Family Services

CF135525 – Supportive Services

Personal Expenses \$ 4,500.00

The Department of Children and Family Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

ITEMS SUBMITTED THROUGH JANUARY 2, 2013

T. FROM: 01A001 – General Fund **BA1201076**

CR180026 - Medical Examiner-Operations

Personal Services \$ 9,000.00

TO: 01A001 – General Fund

CR180026 – Medical Examiner-Operations

Other Expenses \$ 9,000.00

The appropriations transfer with the Medical Examiners Operations account for year-end expenses. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

U. FROM: 20A390 – Emergency Management **BA1201078**

JA100123 – Emergency Management

Personal Services \$ 15,400.00

TO: 20A390 – Central Security Services-Sheriff

JA100123 – Emergency Management

Other Expenses \$ 15,400.00

The appropriations transfer is to realign appropriations within Public Safety and Justice Services Emergency Management Division for controlled costs such as telephone, data processing and indirect costs. Funding is from grants and services along with a General Fund subsidy covering the period January 1, 2012 through December 31, 2012.

V. FROM: 01A001 – General Fund **BA1201079**

SH350470 – Jail Operations-Sheriff

Other Expenses \$ 650,920.00

TO: 01A001 – General Fund

SH350579 – Sheriff Operations

Other Expenses \$ 650,920.00

The appropriations transfer is to realign appropriations between the Sheriff's Jail Operations and Sheriff Operations for controlled costs such as telephone, data processing and space maintenance. Funding is from General Fund covering the period January 1, 2012 through December 31, 2012.

W. FROM: 01A001 – Law Enforcement-Sheriff BA1201081

SH350272 - Law Enforcement-Sheriff

Personal Services \$ 18,300.00

TO: 01A001 – General Fund

SH350579 – Sheriff Operations

Personal Services \$ 18,300.00

The appropriations transfer is to realign appropriations between Law Enforcement and Operations for fringe benefits. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

X. FROM: 01A001 – General Fund **BA1201082**

SH350579 – Sheriff Operations

Other Expenses \$ 1,000.00 Capital Outlays \$ 6,000.00

TO: 01A001 – General Fund

SH350579 – Sheriff Operations

Personal Services \$ 7,000.00

The appropriations transfer is to provide sufficient appropriations within the Sheriff's Operations account for last pay in December. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

Y. FROM: 01A001 – General Fund **BA1201083**

SH350470 – Jail Operations-Sheriff

Other Expenses \$ 127,000.00

TO: 01A001 – General Fund

SH350470 – Jail Operations-Sheriff

Personal Services \$ 127,000.00

The appropriations transfer is to provide sufficient appropriations within the Sheriff's Jail Operations account for fringe benefits in December. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

Z. FROM: 01A001 – General Fund **BA1201110**

SH351080 – Impact Unit/Community Policing Capital Outlays \$ 5,400.00

TO: 01A001 – General Fund

SH351080 – Impact Unit/Community Policing Personal Services \$ 5,400.00

The appropriations transfer is to provide sufficient appropriations within the Sheriff's Impact Unit/Community Policing account for the last pay in December. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AA.FROM: 21A837 – State Homeland Security (SHSG) **BA1201161**

JA763441 – State Homeland Security (SHSP) 2010/2013 Capital Outlays \$ 143,000.00

TO: 21A837 – State Homeland Security (SHSG)

JA763441 – State Homeland Security (SHSP) 2010/2013 Other Expenses \$ 143,000.00

The appropriations transfer is to provide sufficient appropriations within the State Homeland Security Program (SHSP) 2010/2013 for pending request for proposal to purchase a system for

the fire service discipline. Funding is from the United States Department of Homeland Security passed through the Ohio Emergency Management Agency covering the period August 1, 2010 through March 31, 2013.

AB.FROM: 24A430 - Executive Office of HHS **BA1201175**

HS157289 - Executive Office of H&HS

Personal Services \$ 102,000.00

TO: Executive Office of HHS

HS157289 - Executive Office of H&HS

Other Expenses \$ 102,000.00

Request to transfer available appropriation within the budget for the Office of Health and Human Services to enable the 2012 indirect chargeback to post. This budget is supported by the County's Public Assistance allocations and the Health and Human Services Levy Fund.

AC1.FROM: 24A601 – Senior and Adult Services BA1201093

SA138354 – Management Services

Personal Services \$ 1,000.00

TO: 24A601 – Senior and Adult Services

SA138321 – Administrative Services-SAS

Personal Services \$ 1,000.00

AC2.FROM: 24A601 – Senior and Adult Services **BA1201105**

SA138321 - Administrative Services-SAS

Other Expenses \$ 5,300.00

TO: 24A601 – Senior and Adult Services

SA138602 – Home Based Services

Other Expenses \$ 5,300.00

The Department of Senior and Adult Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary/fringe and commodities charges for the remainder of the year. The funding source is the Health and Human Services Levy Fund, the Public Assistance allocations, and fees for direct services provided to clients.

AD.FROM: 24A510 – Work & Training Admin BA1201011

WT137109 – Admin Services-General Manager Personal Services \$ 53,800.00

TO: 24A510 – Work & Training Admin

WT137141 – Client Support Services

Personal Services \$ 14,400.00

24A510 – Work & Training Admin WT137430 – Ohio City NFSC

Personal Services \$ 1,700.00

24A510 – Work & Training Admin WT137943 – Information Services

Personal Services \$ 28,000.00

24A510 – Work & Training Admin WT137455 – Quincy Place NFSC

Personal Services \$ 9,700.00

The Department of Jobs and Family Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

AE.FROM: 24A510 – Work & Training Admin BA1201102

WT137463 – VEB Building NFSC

Personal Services \$ 10,000.00

24A510 – Work & Training Admin WT137539 – West Shore NFSC

Personal Services \$ 13,000.00

24A510 – Work & Training Admin WT137471 – Mount Pleasant NFSC

Other Expenses \$ 81,033.00

TO: 24A510 – Work & Training Admin

WT137414 - Southgate NFSC

Personal Services \$ 104,033.00

The Department of Jobs and Family Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

AF. FROM: 24A510 – Work & Training Admin **BA1201103**

WT137471 - Mount Pleasant NFSC

Personal Services \$ 53,800.00

24A510 – Work & Training Admin WT137315 – Work First Services

Personal Services \$ 7,900.00

TO: 24A510 – Work & Training Admin

WT137455 – Quincy Place NFSC

Personal Services \$ 61,700.00

The Department of Jobs and Family Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

AG.FROM: 24A641 –PA Homeless Services BA1200870

HS158097 – PA Homeless Services

Other Expenses \$ 1,596.00

TO: 24A641 –PA Homeless Services

HS158097 – PA Homeless Services

Personal Expenses \$ 1,596.00

The Department of Homeless Services has requested appropriation transfers to realign appropriations between index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

AH1.FROM: 01A001 – General Fund **BA1201099**

FS109637 – Financial Reporting

Other Expenses \$ 155,000.00

TO: 01A001 – General Fund

FS109611 – Fiscal Office Administration

Other Expenses \$ 40,000.00

01A001 – General Fund

FS109652 - Fiscal Operations – Contractual Services Other Expenses \$ 110,000.00

01A001 - General Fund

FS109686 - Fiscal Operations – Property Valuations Other Expenses \$ 5,000.00

AH2.FROM: 01A001 – General Fund

HR018010 - Human Resources

Personal Services \$ 185,000.00

TO: 01A001 – General Fund

HR018010 – Human Resources

Other Expenses \$ 185,000.00

This appropriation transfer would provide sufficient appropriation for controlled expenses including data processing and space maintenance chargebacks. This includes reallocation of \$155,000 among Fiscal Office divisions and a transfer of \$185,000 from Human Resources salary surplus to cover Human Resources data processing expenses. Funding comes from the General Fund.

AI. FROM: 20A301 – Real Estate Assessment **BA1201100**

FS109702 – Fiscal Operations - Tax Assessments Other Expenses \$ 405,000.00

TO: 20A301 – Real Estate Assessment

BR420067 – Board of Revision

Other Expenses \$ 405,000.00

This request would transfer surplus appropriation from the consulting contracts for the sexennial reappraisal and surplus controlled services from the Fiscal Office to the Board of

Revision to cover space maintenance charges. The space maintenance charge is anticipated to double from the 2011 Board of Revision expenses because it has increased its square footage to accommodate its caseload.

ITEMS SUBMITTED THROUGH JANUARY 8, 2013

AJ. FROM: 24A510 – Work & Training Admin **BA1201293**

WT137109 – Admin Services-General Manager Personal Services \$ 1,500.00

TO: 24A510 – Work & Training Admin

WT137539 - West Shore NFSC

Personal Services \$ 1,500.00

The Department of Jobs and Family Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

AK.FROM: 54A100 – Sanitary Engineer BA1201232

ST540252 – Sanitary Engineer - Administration Capital Outlays \$ 290,000.00

TO: 54A100 – Sanitary Engineer

ST540252 – Sanitary Engineer-Administration Other Expenses \$ 290,000.00

A transfer is requested to cover the certification of a new contract (CE1200145) that has been awarded for general engineering services. This item will also cover indirect cost charges. Revenues come from charges for services.

AL.FROM: 01A001 – General Fund **BA1201233**

FS109637 – Financial Reporting

Other Expenses \$ 24,000.00

01A001 - General Fund

HR018010 – Human Resources

Personal Services \$ 30,000.00

TO: 01A001 – General Fund

FS109611 – Fiscal Office Administration

Other Expenses \$ 24,000.00

01A001 – General Fund

HR018010 - Human Resources

Other Expenses \$ 30,000.00

The budget adjustments would cover space maintenance expenses. Funding comes from the General Fund.

AM.FROM: 01A001 – General Fund **BA1201234**

EX016006 - Executive Office

Personal Services \$ 110,500.00

TO: 01A001 – General Fund

EX016006 - Executive Office

Other Expenses \$ 110,500.00

A budget adjustment would use vacancy savings from salary to cover data processing expenses. Funding comes from the General Fund.

AN.FROM: 63A100 – Information Services Center BA1201252

IS821009 - Cuyahoga County Information Services Center

Personal Services \$ 396,400.00 Capital Outlay \$ 96,000.00

TO: 63A100 – Information Services Center

IS821009 - Cuyahoga County Information Services Center

Other Expenses \$ 492,400.00

A budget adjustment would use surplus appropriation to cover indirect charges within the controlled budget category. Funding comes from charges to user agencies for computer and telephone services.

AO.FROM: 01A001 – General Fund **BA1201254**

FS109629 – Office of Budget & Management

Personal Services \$ 21,000.00

01A001 – General Fund

FS109686 – Fiscal Operations – Property Valuation Personal Services \$ 14,100.00

TO: 01A001 – General Fund

FS109652 – Fiscal Operations – Contractual Services Personal Services \$ 9,000.00

01A001 – General Fund

FS109660 - Treasury Management

Personal Services \$ 26,100.00

This request would reallocate personal services appropriation within the Fiscal Office General Fund budget to cover year-end payroll expenses. Funding comes from the General Fund.

AP. FROM: 01A001 – General Fund **BA1201273**

FS109611 – Fiscal Office Administration

Personal Services \$ 56,740.00

01A001 - General Fund

FS109629 - Office of Budget & Management

Other Expenses \$ 11,293.06

01A001 - General Fund

FS109637 – Financial Reporting

Other Expenses \$ 5,500.00

01A001 – General Fund

FS109645 – Fiscal Operations – Records & Licenses Capital Outlay \$ 4,300.00

01A001 – General Fund

HR018010 - Human Resources

Personal Services \$ 338,115.00

TO: 01A001 – General Fund

EX016006 - Executive Office

Other Expenses \$ 24,000.00

01A001 - General Fund

FS109611 – Fiscal Office Administration

Other Expenses \$ 149,750.00

01A001 - General Fund

FS109629 – Office of Budget & Management

Personal Services \$ 5,100.00

01A001 – General Fund

FS109645 – Fiscal Operations – Records & Licenses Personal Services \$ 24,548.06

01A001 – General Fund

FS109660 - Treasury Management

Personal Services \$ 5,050.00

01A001 - General Fund

HR018010 – Human Resources

Other Expenses \$ 207,500.00

This request would transfer funds within General Fund accounts to cover year-end expenses. Funding for all transfers come from the General Fund.

AQ.FROM: 01A001 – General Fund **BA1201192**

JC372052 – JC Judges

Capital Outlay \$ 22,850.00

20A811 – JC Detention & Probation Services

JC107532 – JC Legal Services

Other Expenses \$ 33,000.00

TO: 01A001 – General Fund

JC375055 – JC Child Support

Personal Services \$ 22,850.00

20A811 – JC Detention & Probation Services

JC107516 – JC Probation Services

Personal Services \$ 33,000.00

This transfer would utilize surplus appropriation to cover 2012 Workers' Compensation charges including fringe benefits and controlled charges. Funding comes from the General Fund and the Health and Human Services Levy Fund.

AR.FROM:	01A001 – General Fund			BA1201274			
	IA018002 – Internal Audit						
	Personal Services	\$	239,975.00				
	Other Expenses	\$	166,500.00				
	01A001 – General Fund						
	HC019018 – Human Resou	rce Commi	ission				
	Other Expenses	\$	181,333.00				
	Capital Outlay	\$	1,020.00				
	01A001 – General Fund						
	FS109611 - Fiscal Office A	dministrati	on				
	Personal Services	\$	52.00				
	Other Expenses	\$	46.00				
	Capital Outlay	\$ \$	1,373.00				
	Capital Outlay	φ	1,373.00				
	01A001 – General Fund						
	FS109629 – Office of Budget & Management						
	Personal Services	\$	665.00				
	Other Expenses	\$	43,859.00				
	Capital Outlay	\$	1,850.00				
	01A001 – General Fund						
	FS109637 – Financial Reporting						
	Personal Services	\$	271,460.00				
	Other Expenses	\$	380,785.00				
	Capital Outlay	\$	1,850.00				
	Capital Outlay	φ	1,050.00				
TO:	01A001 – General Fund						
	HR018010 – Human Resou	irces					
	Other Expenses	\$	4,700.00				
	01A001 – General Fund						
	JC372052 – Juvenile Court	Judges					
	Other Expenses	\$	1,291,398.00				

This budget adjustment would transfer unspent appropriation in various General Fund agency budgets primarily to help cover the 2012 space maintenance expenses associated with the costs at the new Juvenile Justice Center as well as rent payments for Human Resources space. Funding is from the General Fund.

AS. FROM: 01A001 – General Fund BA1201280 AE511501 – Bureau of Inspection

	Other Expenses	\$	11,075.00
	01A001 – General Fund AU150011 – General Acco Other Expenses	unting \$	80,356.00
	-	Ψ	00,330.00
	01A001 – General Fund BR420059 - Board of Revis	sion	
	Other Expenses	\$	12,061.00
	Capital Outlay	\$	6,267.00
	01A001 – General Fund CC012088 – Office of Proc	ouramant & I	Divarcity
		\$	22,830.00
	Other Expenses	φ	22,830.00
	01A001 – General Fund		
	EX016006 -Executive Office		202.00
	Other Expenses	\$	393.00
	01A001 – General Fund	agords & Li	aansas
	FS109645 – Fiscal Ops – R Personal Services		
	Other Expenses	\$ \$	2,105.00 176,485.00
	Capital Outlay	\$ \$	78.00
	Capital Guilay	Ψ	70.00
	01A001 – General Fund CT577601 – Archives		
	Other Expenses	\$	400,000.00
	Other Expenses	Ψ	100,000.00
	01A001 – General Fund		
	DV014100 – Economic De	velopment	
	Other Expenses	\$	600,000.00
TO:	01A001 – General Fund		
	JC370056 – Juvenile Court		
	Personal Services	\$	64,600.00
	Other Expenses	\$	701,400.00
	01A001 – General Fund		
	JC372052 – Juvenile Court	Judges	
	Personal Services	\$	12,550.00
	Other Expenses	\$	533,100.00

This request would transfer appropriation among various General Fund accounts to cover the 2012 space maintenance costs at the Juvenile Court Justice Center and Juvenile Court Workers' Compensation charges.

AT.FROM:	01A001 – General Fun	d		BA1201218
	JC107524 – JC - Deten	tion Services		
	Personal Services	\$	1,231.00	
	Other Expenses	\$	5,288.00	

	Capital Outlay	\$ 875.00
	01A001 – General Fur JC107516 – JC Probat	
	Capital Outlay	\$ 29.00
	01A001 – General Fur JC107532 - JC - Legal	
	Personal Services	\$ 11,979.00
	Other Expenses	\$ 6,468.00
	Capital Outlay	\$ 378.00
TO:	01A001 – General Fur JC107516 – JC – Prob	 ;
	Personal Services	\$ 12,967.00
	Other Expenses	\$ 13,281.00

This budget adjustment would transfer HHS Levy funds within Juvenile Court accounts to cover year expenses for space maintenance and workers' compensation charges.

AU.FROM:	01A001 – General Fund DR391052 – Domestic Re Personal Services	elations Court \$	14,126.00	BA1201166
FROM:	01A001 – General Fund DR495515 – Domestic Re Other Expenses	lations Child S	Support 1,154.00	
TO:	01A001 – General Fund DR391052 – Domestic Re Other Expenses Capital Outlays	lations Court \$ \$	776.00 1,190.00	

01A001 – General Fund DR495515 – Domestic Relations Child Support

TO:

Personal Services \$ 6,066.00 Capital Outlays \$ 7,248.00

The appropriations transfer within the Domestic Relations Court for year-end expenses. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AV.FROM:	01A001 – General Fur SH350579 – Sheriff O		BA1201169
	Other Expenses	\$ 177,000.00	
TO:	01A001 – General Fur SH350470 – Jail Oper		
	Other Expenses	\$ 177,000.00	

The appropriations transfer within the Sheriff's Department for space maintenance charges. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AW.FROM: 01A001 – General Fund **BA1201172**

JA302224 – Public Safety Grants Administration
Personal Services \$ 207,300.00
Other Expenses \$ 80,000.00

TO: 01A001 – General Fund

JA050088 - Justice Affairs Administration

Other Expenses \$ 287,300.00

The appropriations transfer within Public Safety and Justice Service general fund accounts for space maintenance charges. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AX.FROM: 01A001 – General Fund **BA1201283**

JA050088 – Justice Affairs Administration

Personal Services \$ 59,800.00

TO: 01A001 – General Fund

JA050088 – Justice Affairs Administration

Other Expenses \$ 59,800.00

The appropriations transfer within Public Safety and Justice Service, Justice Services Administration account for space maintenance charges. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AY.FROM: 01A001 – General Fund **BA1201182**

PR200071 - Prosecutor Child Support

Personal Services \$ 140,200.00

TO: 01A001 – General Fund

PR191056 - Prosecutor General Office

Personal Services \$ 140,200.00

The appropriations transfer within Prosecutor Office accounts for year-end expenses. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AZ.FROM: 61A608 – Central Security Services-Sheriff **BA1201184**

SH352005 – Building Security Services-OPBA-Officers Personal Services \$ 186,600.00

TO: 61A608 – Central Security Services-Sheriff

SH352005 – Building Security Services-OPBA-Officers Other Expenses \$ 162,000.00 Capital Outlays \$ 24,600.00

The appropriations transfer within the Sheriff's Department, Protective Services Division for year-end expenses. Funding is from an internal service fund that charges applicable

departments with the County for security services covering the period January 1, 2012 through December 31, 2012.

BA.FROM: 21SA029 – ARRA-Early Case Disposition 2009/2013 **BA1201185**

PR736942 – ARRA-Early Case Disposition 2009/2013 Other Expenses \$ 25,571.78

TO: 21SA029 – ARRA-Early Case Disposition 2009/2013

PR736942 – ARRA-Early Case Disposition 2009/2013 Personal Services \$ 25,571.78

The appropriations transfer to be in compliance with the approved budget adjustment and to prepare the grant for closeout in the ARRA-Early Disposition Grant awarded to the Prosecutor's Office through the Sheriff's Office from the City of Cleveland funded by the American Recovery and Reinvestment Act from the United States Department of Justice, Bureau of Assistance Edward Byrne Memorial Justice Assistance Grant Program (JAG) covering the period March 1, 2009 through February 28, 2013.

BB.FROM: 61A608 – Central Security Services-Sheriff **BA1201207**

SH352005 – Building Security Services-OPBA-Officers Personal Services \$ 110,000.00

TO: 61A608 – Central Security Services-Sheriff

SH352005 – Building Security Services-OPBA-Officers Capital Outlays \$ 110,000.00

The appropriations transfer within the Sheriff's Department, Protective Services Division for year-end expenses. Funding is from an internal service fund that charges applicable departments with the County for security services covering the period January 1, 2012 through December 31, 2012.

BC.FROM: 01A001 – General Fund **BA1201214**

SH350470 – Jail Operations

Personal Services \$ 77,000.00 Capital Outlays \$ 11,000.00

01A001 - General Fund

SH350579 – Sheriff Operations

Personal Services \$ 76,000.00

TO: 01A001 – General Fund

SH350470 – Jail Operations

Other Expenses \$ 164,000.00

The appropriations transfer within the Sheriff's Department for year-end food expenses in Jail Operations Division. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

BD.FROM: 01A001 – General Fund **BA1201237**

PD140053 - Public Defender

Personal Services \$ 185,302.00

TO: 01A001 – General Fund

PD140053 – Public Defender

Other Expenses \$ 185,302.00

Request to transfer appropriation within Public Defender's budget to support space maintenance charge-backs. The Public Defender is supported by the General Fund.

BE. FROM: 01A001 – General Fund **BA1201277**

CO380121 - Judicial Administration

Other Expenses \$ 416,000.00

TO: 01A001 – General Fund

DR391052 - Domestic Relations

Other Expenses \$ 244,000.00

01A001 - General Fund

DR495515 – Bureau of Support

Other Expenses \$ 172,000.00

Request to transfer appropriation from the Court of Common Pleas to Domestic Relations Court to post space maintenance chargebacks. Both Courts are supported by the County's General Fund.

BF. FROM: 01A001 – General Fund **BA1201279**

CO380220 - Court Services

Personal Services \$ 18,000.00 Other Expenses \$ 295,000.00

TO: 01A001 – General Fund

CO380410 – Probation/Psychiatric

Personal Services \$ 295,000.00 Other Expenses \$ 10,000.00

01A001 - General Fund

CO380121 – Judicial Administration

Other Expenses \$ 8,000.00

Request to transfer appropriation within the Court of Common Pleas' General Fund budget to post year-end expenses.

BG.FROM: 01A001 – General Fund **BA1201135**

CO380196 - Magistrates

Other Expenses \$ 4,000.00

01A001- General Fund

CO380121 – Judicial Administration

Other Expenses \$ 25,000.00

TO: 01A001 – General Fund

CO380196 – Magistrates

Personal Services \$ 4,000.00

01A001 - General Fund

CO380121 – Judicial Administration

Personal Services \$ 25,000.00

Request to transfer appropriation within the Court of Common Pleas' General Fund budget to post 2012 workers' compensation charges.

BH.FROM: 01A001 – General Fund **BA1200136**

CO380196 - Magistrates

Other Expenses \$ 8,000.00

01A001 – General Fund

CO380121 – Judicial Administration

Other Expenses \$ 16,000.00

01A001 – General Fund CO380220 – Court Services

Personal Services \$ 15,000.00

TO: 01A001 – General Fund

CO380410 – Probation/Psychiatric

Personal Services \$ 39,000.00

Request to transfer appropriation within the Court of Common Pleas General Fund budget to post 2012 workers' compensation expenses.

BI.FROM: 20A812 – Common Pleas Special Project I BA1201137

CO456475 – Common Pleas Special Project I Personal Services \$ 3,000.00

TO: 20A812 – Common Pleas Special Project I

SH456483 – Sheriff Special Project

Personal Services \$ 3,000.00

Request to transfer appropriation from the Court of Common Pleas to the Sheriff's Office — within the same fund — to support benefits expenses in the division of the Sheriff's Office that specifically deals with foreclosures. This Fund, established by the Court of Common Pleas, is supported by a \$200 fee on foreclosure filings and covers related expenses in the Court of Common Pleas, the Clerk of Courts Office, and the Sheriff's Office. This request does not impact the County's General Fund.

BJ.FROM: 20A099 – TASC Medicaid **BA1201140**

CO456525 – TASC Medicaid (CO)

Other Expenses \$ 523.55

TO: 20A099 – TASC Medicaid

CO456525 – TASC Medicaid (CO)

Personal Services \$ 523.55

Request to transfer appropriation within the Medicaid portion of the Treatment Alternative to Street Crimes' budget to post expenses at the subobject level. This budget captures reimbursement for expenses from Federal Medicaid. This request does not impact the County's General Fund.

BK.FROM: 24A301 – Children and Family Services BA1200878

CF135467 - Admin Services-CFS

Other Expenses \$ 26,000.00

TO: 24A301 – Children and Family Services

CF135467 - Admin Services-CFS

Capital Outlays \$ 26,000.00

The Department of Jobs and Family Services has requested appropriation transfers to realign appropriations within the same index code in order to cover pre-encumbrances and expenses for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

BL1.FROM: 65A604 - Postage BA1201126

CT577353 – County Mailroom

Personal Services \$ 49,353.06

TO: 65A604 - Postage

CT577353 - County Mailroom

Other Expenses \$ 49,353.06

BL2.FROM: 61A607 – Centralized Custodial Services BA1201125

CT571000 – Buildings and Grounds - Administration Personal Services \$ 482,489.26 Other Expenses \$ 60,863.00

FROM: 61A607 – Centralized Custodial Services

CT577379 – Buildings and Grounds – Custodial Services Personal Services \$ 344,664.43

TO: 61A607 – Centralized Custodial Services

CT577411 – Buildings and Grounds – Other Services Other Expenses \$ 888,016.69

BL3.FROM: 26A601 – General Gas and License Fees BA1201124

CE835249 – Maintenance Engineer Road & Bridge Personal Services \$ 85,000.00

TO: 26A601 – General Gas and License Fees

CE835025 – Road & Bridge Administration

Personal Services \$ 85.000.00

Transfers are requested to cover year end payroll, space maintenance, data processing, and indirect cost charges. The source of funding for the Custodial Fund and County Mailroom is user fees. The Road & Bridge account receives funding from license and gas taxes.

BM.FROM: 61A607 – Centralized Custodial Services BA1201133

CT577395 – Buildings and Grounds – Trades Services Personal Services \$ 92,000.00

TO: 61A607 – Centralized Custodial Services

CT577411 – Buildings and Grounds – Other Services Other Expenses \$ 92,000.00

A transfer is requested to cover outstanding utilities expenses for the building maintenance fund. Revenues come from charges to user agencies for space maintenance.

BN1.FROM: 61A607 – Centralized Custodial Services BA1201146

CT577395 – Buildings and Grounds – Trades Services Personal Services \$ 42,000.00

TO: 61A607 – Centralized Custodial Services

CT571000 – Buildings and Grounds – Administration Personal Services \$ 27,000.00 Other Expenses \$ 15,000.00

BN2.FROM: 61A607 – Centralized Custodial Services BA1201147

CT577379 – Buildings and Grounds – Custodial Services Personal Services \$ 56,000.00

TO: 61A607 – Centralized Custodial Services

CT577411 – Buildings and Grounds – Other Services Other Expenses \$ 56,000.00

BN3.FROM: 65A604 – Postage BA1201148

CT577353 - County Mailroom

Other Expenses \$ 6,400.00

TO: 65A604 – Postage

CT577353 - County Mailroom

Personal Services \$ 6,400.00

Transfers are requested to cover outstanding workers' compensation and telephone charges for the building maintenance and mailroom funds. Revenues come from charges to user agencies for space maintenance and mailroom services.

BO.FROM: 20N306 – Soil and Water Conservation BA1201151

SW500058 - Soil and Water Conservation

Other Expenses \$ 1,464.66

TO: 20N306 – Soil and Water Conservation

SW500058 – Soil and Water Conservation

Personal Services \$ 1.464.66

A transfer is requested to cover outstanding workers' compensation charges for Soil and Water Conservation District. Revenues come from municipalities, the State, and a County subsidy for pollution prevention technical services.

BN.FROM: 01A001 – General Fund **BA1201243**

PC400051 – Probate Court

Other Expenses \$ 8,650.00

TO: 01A001 – General Fund

PC400051 – Probate Court

Personal Services \$ 8.650.00

Request to transfer appropriation within the Probate Court's budget to cover worker's compensation charges. The Probate Court is supported by the General Fund.

BO.FROM: 01A001 – General Fund **BA1201243**

LL440008 – Law Library

Other Expenses \$ 160.00

TO: 01A001 – General Fund

LL440008 – Law Library

Personal Services \$ 160.00

Request to transfer appropriation within the Law Library's budget to cover worker's compensation charges. The Law Library is supported by the General Fund.

BP. FROM: 01A001 – General Fund **BA1201238**

CO380121 – Judicial Administration

Other Expenses \$ 140,000.00

TO: 01A001 – General Fund

PD140053 – Public Defender

Other Expenses \$ 140,000.00

Request to transfer appropriation from the Court of Common Pleas budget to the Public Defender's budget to support space maintenance charge-backs. The Public Defender is supported by the General Fund.

BQ.FROM: 01A001 – General Fund **BA1201275**

DR391052 - Domestic Relations

Other Expenses \$ 50,000.00

TO: 01A001 – General Fund

PD140053 - Public Defender

Other Expenses \$ 50,000.00

Request to transfer appropriation from Domestic Relations Court's budget to the Public Defender's budget to support space maintenance charge-backs. The Public Defender is supported by the General Fund.

BR.FROM: 01A001 – General Fund BA1201108

JC370056 – Juv. Court – Detention Home

Personal Services \$ 154,500.00 Other Expenses \$ 537,700.00 01A001 - General Fund

JC372052 – Juv. Court – Judges

Personal Services 62,900.00

01A001 - General Fund

JC372060 - Juv. Court - Legal

Personal Services 272,300.00

01A001 - General Fund

JC375055 – Juv. Court – Child Support

Personal Services 92,100.00 \$ Other Expenses 20,489.32 \$ Capital Outlay 2,750.00

20A811 – Juv. Court Detention & Probation Services

JC107516 – Juv. Court – Probation Services

Personal Services \$ 171,500.00 \$ Capital Outlay 9,200.00

20A811 – Juv. Court Detention & Probation Services JC107524 – Juvenile Court – Detention Services

Other Expenses \$ 36,300.00

TO: 01A001 – General Fund

JC372060 – Juv. Court – Legal

Other Expenses 1,142,739.32

20A811 – Juv. Court Detention & Probation Services

JC107516 - Juv. Court - Probation Services

Other Expenses 198,000.00

20A811 – Juv. Court Detention & Probation Services

JC107524 – Juv. Court – Detention Services

Personal Services 19,000.00

The budget adjustments would adjust within General Fund and within HHS Levy accounts to cover year-end expenditures space maintenance charges.

BS. FROM: 01A001 - General Fund BA1201223

FS109652 – Fiscal Operations – Tax Assessment

Other Expenses 15,000.00

01A001 - General Fund

IG030411 – Inspector General

Personal Services 235,000.00 \$ Other Expenses 35,465.00

01A001 – General Fund

CT577106 – Risk & Property Management

Other Expenses 239,000.00 TO: 01A001 – General Fund

JC370056 – Juv. Court – Detention Home

Other Expenses \$ 2,215.00

01A001 - General Fund

JC372052 – Juv. Court – Judges

Other Expenses \$ 200,000.00

01A001 - General Fund

JC372060 - Juv. Court - Legal

Other Expenses \$ 320,000.00

01A001 - General Fund

FS109637 – Financial Reporting

Other Expenses \$ 1,500.00

01A001 - General Fund

HR018010 - Human Resources

Other Expenses \$ 750.00

Appropriation transfers are requested to cover the 2012 space maintenance expenses at the new Juvenile Justice Center as well as to cover year-end expenditures. Funding comes from the General Fund.

BT.FROM: 01A001 – General Fund **BA1201913**

EX016006 - County Executive

Capital Outlays \$ 5,000.00

TO: 01A001 – General Fund

EX016006 - County Executive

Other Expenses \$ 5,000.00

A transfer is requested to cover outstanding space maintenance expenses.

BU.FROM: 01A001 – General Fund **BA1201916**

AE210005 – Soldiers' and Sailors' Monument Capital Outlays \$ 1,000.00

TO: 01A001 – General Fund

AE210005 – Soldiers' and Sailors' Monument

Other Expenses \$ 1,000.00

A transfer is requested to post year end data processing and telephone charges.

BV.FROM: 01A001 – General Fund **BA1201932**

JC372052 – JC – Judges

Personal Services \$ 12.000.00

01A001 - General Fund

FS109645 – Fiscal Ops – Records & Licenses

	Other Expenses	\$	23,000.00		
	01A001 – General Fund FS109652 – Fiscal Ops – Other Expenses	Contractual So	ervices 12,000.00		
	01A001 – General Fund FS109660 – Treasury Ma Other Expenses	nagement \$	24,000.00		
TO:	01A001 – General Fund JC370056 – JC – Detention Other Expenses	on Home \$	24,000.00		
	01A001 – General Fund FS109629 – Office of Budget & Management Other Expenses \$ 23,000.00				
	01A001 – General Fund FS109678 – Office of Pro Other Expenses	ocurement & D	Diversity 24,000.00		

Appropriation transfers are requested to cover year end data processing, phone, and supplies expenses. Funding comes from the General Fund.

BW.FROM:	01A001 – General Fund JC370056 – JC – Detention	Ното		BA1201924		
	Personal Services	\$	62,700.00			
	Capital Outlay	\$	5,000.00			
	01A001 – General Fund JC370056 – JC – Judges Capital Outlay	\$	4,400.00			
	01A001 – General Fund JC372060 – JC – Legal Capital Outlay	\$	1,400.00			
	01A001 – General Fund FS109660 – Treasury Mana	ngement				
	Other Expenses	\$	63,000.00			
	01A001 – General Fund FS109686 – Fiscal Operation Personal Services	ons – Property \$	Valuation 18,000.00			
	01A001 – General Fund					
	HC019018 – Human Resou	rce Commissi	ion			
	Other Expenses	\$	9,350.00			
	Capital Outlay	\$	4,000.00			

01A001 – General Fund IA018002 – Internal Audit

Other Expenses \$ 35,000.00

01A001 - General Fund

IG030411 – Inspector General

Personal Services \$ 18,000.00

01A001 – General Fund

FS109678 – Office of Procurement & Diversity Other Expenses \$ 41,000.00

TO: 01A001 – General Fund

JC370056 – JC – Detention Home

Other Expenses \$ 2,500.00

01A001 - General Fund

FS109611 – Fiscal Office Administration

Other Expenses \$ 30,000.00

01A001 - General Fund

FS109637 – Financial Reporting

Other Expenses \$ 89,950.00

01A001 – General Fund

FS109645 – Fiscal Ops – Records & Licenses

Other Expenses \$ 47,000.00

01A001 - General Fund

HR018010 - Human Resources

Other Expenses \$ 92,400.00

Appropriation transfers are requested to cover year end data processing, phone, and supplies expenses. Funding comes from the General Fund.

BX.FROM: 01A001 – General Fund **BA1201929**

PR200071 - Prosecutor Child Support

Capital Outlays \$ 2,000.00

TO: 01A001 – General Fund

PR200071 – Prosecutor Child Support

Other Expenses \$ 2,000.00

The appropriations transfer within Prosecutor Office Child Support account for data processing charges. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

BY.FROM: 01A001 – General Fund **BA1201930**

CO380204 – Common Pleas-Foreclosure

Other Expenses \$ 160,000.00

01A001 – General Fund

CO380428 - Common Pleas-Psy. Clinic

Other Expenses \$ 115,000.00

TO: 01A001 – General Fund

CO380121 – Common Pleas-Judicial/General

Other Expenses \$ 275,000.00

The appropriations transfer within the Common Pleas Court for data processing charges. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

BZ. FROM: 20A301 – Real Estate Assessment BA1201935

BR420067 – Board of Revision

Personal Services \$ 11,945.00 Capital Outlay \$ 11,725.00

TO: 20A301 – Real Estate Assessment

BR420067 - Board of Revision

Other Expenses \$ 23,670.00

A budget adjustment would use surplus appropriation to cover year-end data processing chargebacks. Funding comes from a fee collected on tax duplicates other than estate taxes.

CA.FROM: 24A601 – Senior and Adult Services BA1201295

SA138503 –Information and Outreach

Personal Services \$ 21,295.00

TO: 24A601 – Senior and Adult Services

SA138321 – Administrative Services - SAS

Other Expenses \$ 21,090.00

24A601 – Senior and Adult Services SA138354 – Management Services

Other Expenses \$ 205.00

The Department of Senior and Adult Services has requested appropriation transfers to realign appropriations within index codes to cover projected controlled and other expenses charges for the remainder of the year. The funding source is the Health and Human Services Levy Fund, the Public Assistance allocations, and fees for direct services provided to clients.

SECTION 3. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion by $_$, seconded by	, the foregoing Resolution was
duly adopted.		

Yeas:		
Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date
First Reading/Referred Committee(s) Assigned		
Journal, 2013		

MEMORANDUM

TO: Jeanne Schmotzer **Year-End Fiscal Agenda**

Year-End RESOLUTION R2012-0228

FROM: Matthew Rubino

DATE: January 22, 2013

RE: Agenda Items

The Office of Budget & Management is submitting the following 2012 year-end fiscal items processed under Resolution R2012-0228.

ITEMS SUBMITTED THROUGH DECEMBER 11, 2012

Resolutions: Additional Appropriations

A1. 24A601 – Senior and Adult Services BA1201035

SA138321 – Administrative Services-SAS

Other Expenses (72,281.00)\$

A2. 24A601 - Senior and Adult Services

SA138305 - Community Social Services Programs

Other Expenses (9,009.00)

Appropriation decreases are requested to reduce excess appropriations in various index codes in the contracts and client services lines for Senior and Adult Services. The funding source is the Health and Human Services Levy Fund, the Public Assistance allocations, and fees for direct services provided to clients.

ITEMS SUBMITTED THROUGH DECEMBER 18, 2012

B1. 21A421 –Sex Offender Treatment FY12-13 BA1201050

JC753558 - Youth Services Subsidy - FDCC -2106

Other Expenses (15,778.80)

B2. 21A421 – Sex Offender Treatment FY12-13

> JC753541 - Substance Abuse Treatment FY12-13 -2107 Other Expenses \$ (13,636.85)

B3. 21A421 – Sex Offender Treatment FY12-13

JC753533 – Clinical Assessment & SA FY12 -2217

Other Expenses (33,392.98)

This request removes the excess appropriation from the grant budget. Funding derives from RECLAIM Ohio grants from the Ohio Department of Youth Services for sex offender, substance abuse, and clinical assessment services.

C1. 54A901 – Sanitary Engineer/Note Retirement BA1200955

ST540625 – Sanitary Engineer – Note Retirement

Other Expenses \$ (17.00)

C2. 54P900 – Sanitary Engineer Bond Retirement BA1200956

DV752006 – Bond Retirement Chagrin Highlands 543A Other Expenses \$ 17.00

An appropriation decrease/increase is requested to pay the principal payment on the Chagrin Highlands Sanitary Sewer bonds. The original budget was incorrect by \$17. Funding for the Chagrin Highlands Bond Retirement comes from the proceeds of its Bond Offering.

D. 01A001 - General Fund **BA1201044**

AE511451 – Board and Care of Prisoners

Other Expenses \$ 150,000.00

Request to increase appropriation to the Board and Care budget to cover anticipated payments to municipal jails for the housing of County prisoners in response to over-crowding in the County Jail. This budget is exclusively supported by the County's General Fund, though it should be noted that there are Federal and State prisoners in the County Jail that the County is housing for a fee: this revenue offsets the cost of sending County prisoners to outside jails.

ITEMS SUBMITTED THROUGH DECEMBER 26, 2012

E. 20A331 – Indigent Guardianship BA1201045

PC404665 – Indigent Guardianship

Other Expenses \$ 25,000.00

Request to increase appropriation to Probate Court's Indigent Guardianship Fund to support a transfer of cash from the Court to the County's Department of Senior and Adult Services in support of the Adult Guardianship Program managed by Lutheran Metropolitan Ministries (via a contract with DSAS). The Indigent Guardianship Fund is supported by filing fees in the Court. This request does not impact the County's General Fund budget.

F. 01A001 – General Fund **BA1201049**

AE511451 - Board and Care of Prisoners

Other Expenses \$ 50,000.00

Request to increase appropriation to the Board and Care budget to cover anticipated payments to municipal jails for the housing of County prisoners in response to over-crowding in the County Jail. This budget is exclusively supported by the County's General Fund, though it should be noted that there are Federal and State prisoners in the County Jail that the County is housing for a fee: this revenue offsets the cost of sending County prisoners to outside jails. The total expenditures for 2012 were \$ XXXXXXXX

G. 20A812 – Common Pleas Special Project I BA1201073

SH456483 – Sheriff's Dept Special Project I

Personal Services \$ 5,060.00

Provide additional appropriations in the Sheriff's Department Special Project I from the Common Pleas Court Special Project Fund operated for foreclosure activities. Sufficient funds exist in the fund for the additional appropriation. Funding covers the period January 1, 2012 through December 31, 2012.

H. 54A100 – Sanitary Engineer BA1201112

ST540583 – Sanitary Engineer – Debt Services
Other Expenses \$ 200,000.00

Appropriation is requested to provide for loan repayments for sanitary sewer development projects. Funding is generated from sewer district assessments.

I. 22A046 – Community Development Block Grant 2011 **BA1201113**DV713362 – Project Plan 2011
Other Expenses \$ (192,197.05)

A decrease is requested in the 2011 CDBG Project Plan grant to move appropriation to the 2012 Project Plan grant. The source of funding is the U.S. Dept. of Housing and Urban Development.

J. 61A607 – Centralized Custodial Services
 CT577395 – Buildings and Grounds – Trades Services
 Personal Services
 \$ 169,795.61

Appropriation is requested to cover workers' compensation and other year end payroll expenses. Funding is generated from charges from user agencies for space maintenance.

K. 24A435 – Cuy Tapestry System of Care CF135004 – DCFS-Cuy Tapestry System of Care Personal Services \$ (3,000.00)
Other Expenses \$ (103,000.00)

The Department of Children and Family Services has requested appropriation decreases to move appropriations from the Tapestry (CTSOC) fund to cover projected shortfalls in the Public Assistance funds within the department for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund. (See related additional appropriation, item L1 – L 5, below.)

L1.	24A301 – Children and Fan CF135483 – Training	24A301 – Children and Family Services CF135483 – Training		
	Personal Services	\$	500.00	
L2.	24A301 – Children and Fan CF135491 – Information Se	•		
	Personal Services	\$	27,500.00	

L3. 24A301 – Children and Family Services

CF135525 - Supportive Services

Personal Services \$ 6,000.00

L4. 24A301 – Children and Family Services

CF135582 - Permanent Custody Adoptions

Personal Services \$ 18,000.00

L5. 24A301 – Children and Family Services

CF135616 – CFS Foster Homes/Resource Mgt

Personal Services \$ 54,000.00

The Department of Children and Family Services has requested appropriation increases to move appropriations from the Tapestry (CTSOC) fund to cover projected shortfalls in the Public Assistance funds within the department for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund. (See related decrease appropriation, item K, on page 3.)

ITEMS SUBMITTED THROUGH January 2, 2013

M. 20A192 – TASC HHS BA1201062

CO456533 - TASC HHS

Other Expenses \$ 99,500.00

Request to increase appropriation to the Treatment Alternative to Street Crimes (TASC) program in the Court of Common Pleas to enable the space maintenance chargeback to post. This program is supported by a combination of grants and the County's Health and Human Services levy; the chargebacks are solely supported by the HHS Levy Fund.

N. 20R320 – Board of Developmental Disabilities **BA1201094**

MR845024 - County Bd of Developmental Disabilities

Other Expenses \$ (1,052,577.00)

Appropriation decreases are requested to reduce excess appropriations in the commodities, contracts, client services, and other operating lines for the Board of Developmental Disabilities. The Cuyahoga County Board of Developmental Disabilities is supported through its own levy and the Ohio Rehabilitation Services Commission. (See related additional appropriation, item T, on page 7.)

O. 24A510 – Work & Training Admin **BA1201095**

WT137315 - Work First Services

Other Expenses \$ (318,041.66)

Request to decrease appropriation within Job and Family Services. Appropriation is being decreased because corresponding contracts have been decertified and funding is no longer available. The corresponding contracts were for services prior to 2011. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

P1. 20A303 – Children Services Fund BA1200872 CF134031 – CFS Foster Care

Other Expenses \$ (235,000.00)

P2. 20A303 – Children Services Fund

CF134049 – Purchased Congregate & Foster Care

Other Expenses \$ (1,200,000.00)

The Department of Children and Family Services has requested appropriation decreases to move appropriation surpluses from the Children Services Fund to cover projected shortfalls in the Public Assistance Funds for controlled services charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

Q. 24A301 – Children and Family Services BA1200873

CF135467 – Administrative Services

Other Expenses \$ 1,435,000.00

The Department of Children and Family Services has requested appropriation increase to offset the move of appropriation surpluses from the Children Services fund to cover projected shortfalls in the Public Assistance funds in Controlled Services charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

R1. 22A956 – Community Development Block Grant 2005 BA1201115
DV709824 – Project Plan 2005
Other Expenses \$ (11,680.00)

R2. 22A970 – Community Development Block Grant 2006

DV712224 – Project Plan 2006
Other Expenses \$ (33,848.67)

R3. 22A975 – Community Development Block Grant 2007

DV712430 - Project Plan 2007

Other Expenses \$ (0.87)

R4. 22A984 – Community Development Block Grant 2008

DV712711 – Project Plan 2008

Other Expenses \$ (5,780.18)

R5. 22A795 – Community Development Block Grant 2009

DV713032 - Project Plan 2009

Other Expenses \$ (115,843.57)

R6. 22A795 – Community Development Block Grant 2009

DV713016 - Project Plan 2009

Other Expenses \$ (1,647.27)

R7.	22A795 – Community Development Block Grant 2009 DV713008 – Administrative Operations 2009			
	Other Expenses	\$	(6,857.87)	
R8.	22A613 – Community Developmer DV713255 – Project Plan 2010	nt Block Gr	ant 2010	
	Other Expenses	\$	(88,804.74)	
R9.	22A613 – Community Developmer DV713263 – Administrative Operate			
	Personal Services	\$	(27,821.65)	
	Other Expenses	\$ \$ \$	(70,321.80)	
	Capital Outlays	\$	(4,035.75)	
R10.	22A613 – Community Developmer		ant 2010	
	DV713271 – Rehabilitation Operat			
	Personal Services	\$ \$ \$	(29,632.12)	
	Other Expenses	Ş	(35,779.11)	
	Capital Outlays	\$	(2,968.75)	
R11.	22A613 – Community Developmer DV713289 – Program Operations 2		ant 2010	BA1201116
	Personal Services	\$	(1,326.49)	
	Other Expenses	\$ \$	(30,940.46)	
	Capital Outlays	\$	(308.75)	
R12.	22A046 – Community Developmer DV713362 – Project Plan 2011	nt Block Gr	ant 2011	
	Other Expenses	\$	(4,073,480.12)	
R13.	22A046 – Community Developmer DV713370 – Administration Opera			
	Other Expenses	\$	(5,000.00)	
R14.	22A685 – Community Developmer DV713917 – Project Plan 2012	nt Block Gr	ant 2012	BA1201114
	Other Expenses	\$	4,546,078.17	

Appropriation decreases with corresponding appropriation increases are requested from 2005-2011 CDBG grants to the new 2012 grant. The source of funding is the U.S. Dept. of Housing and Urban Development.

S1.	22A685 – Community Dev	elopment Block G	Grant 2012	BA1201117
	DV713883 – Administratio	on Operations 201	.2	
	Other Expenses	\$	101,309.27	

S2. 22A685 – Community Development Block Grant 2012

DV713909 – Rehabilitation Operations 2012

Other Expenses \$ 58,320.60

S3. 22A685 – Community Development Block Grant 2012

DV713891 – Program Operations 2012

Other Expenses \$ 32,567.18

Increases are requested for the 2012 CDBG grant. A decrease of appropriation in a corresponding amount (BA1201113) provides funding for these increases. The source of funding is the U.S. Dept. of Housing and Urban Development. (Appropriation decrease was approved on the December 11, 2012, R2012-0257.)

ITEMS SUBMITTED THROUGH January 8, 2013

T. 20R320 – Board of Developmental Disabilities BA1201291

MR845024 – County Bd of Developmental Disabilities

Other Expenses \$ 1,052,577.00

Appropriation increases are requested to restore appropriations in the commodities, contracts, client services, and other operating lines for the Board of Developmental Disabilities. The Cuyahoga County Board of Developmental Disabilities is supported through its own levy and the Ohio Rehabilitation Services Commission. (See related decrease appropriation, item N, on page 4.)

U. 40A524 – County Engineer **BA1201260**

CE785170 – County Engineer – West Bridge Street Bridge #256 Capital Outlays \$ 1,896.00

Appropriation is requested to reclassify an expense that was a capital outlay for contract CE785170. Funding is generated from State Issue 1.

V. 20D447 – Western Reserve Fund BA1201261

DV520676 – Community Development

Other Expenses \$ (19,732,407.60)

Appropriation decrease is requested to remove unused appropriation. The source of funding for the fund will be proceeds from County bonds.

W. 62A603 – County Garage **BA1201261**

CT575001 – Maintenance Garage

Personal Services \$ (32,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

X. 62A603 – County Garage **BA1201261**

CT575001 – Maintenance Garage

Personal Services \$ (16,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

Y. 62A603 – County Garage **BA1201261**

CT575001 – Maintenance Garage

Other Expenses \$ (94,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

Z. 62A603 – County Garage **BA1201261**

CT575001 - Maintenance Garage

Other Expenses \$ (4,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

AA. 62A603 – County Garage **BA1201261**

CT575001 – Maintenance Garage

Other Expenses \$ (409,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

AB. 62A603 – County Garage **BA1201261**

CT575001 - Maintenance Garage

Other Expenses \$ (2,000.00)

Appropriation decrease is requested to remove excess appropriation. The source of funding is charges for services to other operating funds.

AC. 63A100 – Information Services Center BA1201253

IS821009 – Cuyahoga County Information Services Center
Other Expenses \$ 243,000.00

An increase in appropriation is requested to cover 2012 expenses. Of this increase, \$180,000.00 would go to the controlled services line to cover indirect expense chargebacks. The other \$63,000.00 would cover 2012 expenses associated with the County's wide area network. Funding comes from charges assessed to user agencies for data processing and telephone use.

AD. 64A601 – Office Supplies **BA1201288**

FS109751 – County Office Supply Contract

Other Expenses \$ (36,200.00)

A decrease would remove surplus appropriation from the 2012 budget. Funding comes from charges to agencies for the purchase of office supplies.

AE. 68A100 – Hospitalization – Self Insurance Fund BA1201288

CC499004 – Hospitalization – Self Insurance

Other Expenses \$ (324,000.00)

This request would decrease appropriation from the 2012 budget. Revenues are generated from charges for co-payments to the employees and the County.

AF. 68A200 – Hospitalization – Regular Insurance Fund **BA1201288**

CC499012 – Hospitalization – Regular Insurance

Other Expenses \$ (1,400,000.00)

A decrease would remove surplus appropriation from the 2012 budget. Funding for this account comes from contributions made from the County as well as payroll deductions from employees.

AG. 67A011 – Workers' Compensation Retrospective 2011 BA1201288

CC498881- Workers' Compensation Retrospective 2011
Other Expenses \$ (420,000.00)

This request would decrease unneeded appropriation from the 2012 retrospective account budget. Funding comes from charges to departments to reimburse for the costs of the premium and claims based on injury experience and the size of each department to calculate a risk level.

AH. 63A100 – Information Services Center BA1201178

IS690107 - Information Services Center - Overhead

Other Expenses \$ 118,000.00

Additional appropriation would cover 2012 expenses for the wide area network and would cover data processing expenses that will be charged back to user agencies once substantiated. Funding comes from charges assessed to user agencies for data processing and telephone use.

Al. 01A001 – General Fund **BA1201219**

BE473058 – General Election

Personal Services \$ (170,000.00) Other Expenses \$ (1,230,000.00)

A decrease in appropriation would remove surplus appropriation from the 2012 budget. Funding comes from the General Fund. (See related additional appropriation, item AJ, below.)

AJ. 01A001 – General Fund **BA1201220**

JC372052 – Juvenile Court – Judges

Other Expenses \$ 1,400,000.00

The space maintenance charges increased substantially for the Juvenile Court in its first full year of occupancy in the new Juvenile Justice Center. The 2012 appropriation did not accommodate this increase, therefore an increase in appropriation is requested for this purpose. Funding comes from the General Fund. (See related decrease appropriation, item AI, above.)

AK. 20A811 – Juvenile Court Detention & Probation Services BA1201221

JC107516 - Juvenile Court - Detention Services

Other Expenses \$ 65,000.00

The Juvenile Court budget did not have sufficient appropriation for the increased costs of space maintenance at the new Juvenile Justice Center. This increase would resolve this deficit. An equal decrease in appropriation is requested from the Juvenile Court Youth and Family Community Partnership fund in BA1201222. This budget receives funding from the Health and Human Services Levy Fund. (See related appropriation decrease, item AL, below.)

AL. 20A823 – Youth & Family Community Partnership BA1201222

JC108092 – JC Youth & Family Community Partnership

Other Expenses \$ (65,000.00)

This budget decrease would balance an increase of \$65,000 in the Juvenile Court Detention & Probation Services fund requested in BA1201221 to cover increased space maintenance costs at the Juvenile Justice Center. Funding comes from the Health and Human Services Levy Fund. (See related additional appropriation, item AK, above.)

AM1. 01A001 – General Fund BA1201168

SH350272 - Law Enforcement-Sheriff

Other Expenses \$ 200,000.00

AM2. 01A001 – General Fund

SH350470 - Jail Operations-Sheriff

Other Expenses \$ 750,000.00

Requesting additional appropriations in the Sheriff Department for vehicle chargebacks in the Law Enforcement Division and space maintenance charges in Jail. Funding is from General Fund covering the period January 1, 2012 through December 31, 2012. Other General Fund accounts have been reduced providing for the increase. (See related decrease appropriation, item AN, below.)

AN. 01A001 – General Fund **BA1201167**

BE473058 - General Elections

Personal Services \$ (950,000.00)

Provide reduce appropriations in the Board of Elections, General Elections account. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012. (See related additional appropriation, item AM1 and AM2, above.)

AO.	20A809 – Witness Victim HH	S	BA1201163
	JA107425 – Witness Victim F	IHS	
	Personal Services	\$	(375,311.61)
	Other Expenses	\$	(85,596.39)
	Capital Outlays	\$	(23,115.00)

Reducing appropriations in the Department of Public Safety and Justice Services Witness Victim HHS account in accordance to Generally Accepted Accounting Principles (GAAP). Funding is from the Health and Human Service Levies covering the period January 1, 2012 through December 31, 2012.

AP.	24A878 – Public Assistance F	unds	BA1201164
	HS749069 – HHS-Office of Re	eEntry	
	Personal Services	\$	(25,100.00)
	Other Expenses	\$	(672,007.00)
	Capital Outlays	\$	(533.00)

Reducing appropriations in the Office of ReEntry HHS account in accordance to Generally Accepted Accounting Principles (GAAP). Funding is from the Health and Human Service Levies covering the period January 1, 2012 through December 31, 2012.

AQ.	20A891 – Common Pleas HF	IS Subsidy	BA1201211
	CO456517 – Common Pleas	HHS Subsidy	
	Other Expenses	\$	(113,300.00)

Reducing appropriations in the Common Pleas HHS Subsidy account for year-end expenses in another HHS account in Common Pleas (see corresponding increase to CO456533, document BA1201212 fiscal item AV, on page 12). Funding is from the Health and Human Services Levy Fund covering the period January 1, 2012 through December 31, 2012.

AR.	20A192 – TASC HHS	BA1201212
	CO456533 – TASC HHS	
	Personal Services	\$ 3,300.00
	Other Expenses	\$ 110,000.00

Requesting additional appropriations in the Common Pleas TASC HHS account for year-end expenses, specifically workers compensation charges and space maintenance. See reduction in another Common Pleas HHS account for the same amount (see corresponding decrease to CO456517, document BA1201211 fiscal item AU, on page 11). Funding is from the Health and Human Services Levies covering the period January 1, 2012 through December 31, 2012.

AS.	20A192 – TASC HHS		BA1201176
	CO456533 – TASC HHS		
	Other Expenses	\$ 114,000,00	

Request to increase appropriation to the Treatment Alternative to Street Crimes budget in the Court of Common Pleas to support 2012 space maintenance chargebacks. This program is supported by grants and the County's Health and Human Services Levy, the latter of which

exclusively covers space charges. This request is offset by a corresponding reduction in the Court's Probation/Psychiatric division, which is also subsidized by the County's Health and Human Services Fund. (See related decrease appropriation, item AT, below.)

AT. 20A891 – Common Pleas HHS Subsidy CO456517 – Common Pleas HHS Subsidy Other Expenses \$ (114,000.00)

Request to reduce surplus appropriation to the Court of Common Pleas' Probation/Psychiatric division to support an increase in the Treatment Alternative to Street Crimes' budget, both of which are supported by the County's Health and Human Services Fund. (See related additional appropriation, item AS, above.)

AU1.	52A100 – County Airport DV520031 – County Airport			BA1201129
	Other Expenses	\$	271,028.71	
AU2.	65A604 – Postage CT577353 – County Mailroom			BA1201128
	Other Expenses	\$	35,000.00	
AU3.	01A001 – General Fund CT577106 – Risk and Property Mar	nagement		
	Personal Services	\$	12,500.00	
AU4.	01A100 – General Fund AE210005 – Soldiers' and Sailors' N	Лonument		BA1201119
	Other Expenses	\$	11,000.00	

Appropriation is requested to cover year end payroll, space maintenance, data processing, and indirect cost charges. The source of funding for the Airport and County Mailroom is user fees and a General Fund subsidy for the Airport.

AV.	22A046 – Community Development	t Block Grant	Proj. FY 2011	BA1201144
	DV713362 – CDBG Year 37 2011			
	Other Expenses	\$	192,197.05	

This item reverses an earlier decrease on the year end resolution (BA1201113) that was already approved by Council on the December 11, 2012 agenda, R2012-0257. (See related decrease appropriation, item I, on page 3.)

AW.	22A685 – Community Development Block Grant 2012			BA1201145
	DV713917 – Project Plan 2012			
	Other Expenses	\$	(4,546,078.17)	

This item reverses an earlier increase on the year end resolution (BA1201114) that was already approved by Council on the December 11, 2012 agenda, R2012-0257. (See related additional appropriation, item R14, on page 6.)

AX1.	20A600 - General Fund			BA1201239
	SE496000 – Support Enforceme	nt Agency		
	Personal Services	\$	(740,023.00)	
	Other Expenses	\$	(1,937,270.00)	
	Capital Outlay	\$	(85,000.00)	
AX2.	24A430 - General Fund			BA1201239
	HS157289 – Executive Office fo	r H&HS		
	Personal Services	\$	(55,958.00)	
	Other Expenses	\$	(136,200.00)	
	Capital Outlay	\$	(2,000.00)	

A request to reduce surplus appropriation. Both agencies are supported by the Health and Human Services Levy Fund.

AY.	20A810 – Criminal Justice Ir	tervention HHS	BA1201215
	JA107433 – Criminal Justice	Intervention HHS	
	Personal Services	\$	(147,141.72)

Reducing appropriations in the Department of Public Safety and Justice Services Criminal Justice Intervention HHS account in accordance to Generally Accepted Accounting Principles (GAAP). Funding is from the Health and Human Service Levies covering the period January 1, 2012 through December 31, 2012.

AZ.	20A809 – Witness Victim HHS	BA1201216
	JA107425 – Witness Victim HHS	
	Other Expenses	\$ 3,890.00

Requesting to increase appropriations in the Department of Public Safety and Justice Services Witness Victim HHS account. This increase is a correction to previous reduction seen on BA1201163 (fiscal item AR, on page 11). Funding is from the Health and Human Service Levy Fund covering the period January 1, 2012 through December 31, 2012.

BA1.	30A905 – Gateway Arena DS100370 – Gateway Arena Project			BA1200957
	Other Expenses	\$	(29.54)	
BA2.	30A910 – Brownfield Debt Service DS039966 – Brownfield Debt Servic	e		
	Other Expenses	\$	(2,610.57)	
BA3.	30A913 – Community Redevelopment Debt Service			
	DS040121 – Community Redevelopment Debt Service			
	Other Expenses	\$	(8,646.90)	

Appropriation reductions are requested for three debt service accounts: Gateway Arena Project, Brownfield Debt Service and Community Redevelopment Debt Service. The

appropriation reductions will bring the actual debt service and the budgeted debt service into balance and thereby zero out the remaining free balance. Funding for the debt service accounts comes from the General Fund.

BB1.	24A510 – Work & Training Admin WT137109 – Admin Services-General Manager			BA1201194
	Other Expenses	-	(1,686,500.00)	
	•	\$ \$		
	Capital Outlay	Ş	(3,500.00)	
BB2.	24A510 – Work & Training Admin WT137141 – Client Support Service	S		
	Other Expenses	\$ \$	(249,800.00)	
	Capital Outlay	\$	(9,800.00)	
BB3.	24A510 – Work & Training Admin WT137315 – Work First Services Personal Services Other Expenses	\$ \$	(6,000.00) (703,000.00)	
BB4.	24A510 – Work & Training Admin WT137414 – Southgate NFSC Other Expenses	\$	(16,100.00)	
	Capital Outlay	\$ \$	(6,000.00)	
BB5.	24A510 – Work & Training Admin WT137430 – Ohio City NFSC Other Expenses Capital Outlay	\$ \$	(767,000.00) (6,400.00)	
BB6.	24A510 – Work & Training Admin WT137455 – Quincy Place NFSC Other Expenses	\$	(128,500.00)	
BB7.	24A510 – Work & Training Admin WT137463 – VEB Building NFSC Other Expenses	\$	(1,090,000.00)	
BB8.	24A510 – Work & Training Admin WT137471 – Mount Pleasant NFSC Other Expenses Capital Outlay	\$ \$	(197,000.00) (4,400.00)	
BB9.	24A510 – Work & Training Admin WT137539 – West Shore NFSC Other Expenses Capital Outlay	\$ \$	(129,500.00) (7,000.00)	BA1201195

BB10.	24A530 – Children With Medical Handicap				
	WT137935 - Children With Medic	WT137935 – Children With Medical Handicap			
	Other Expenses	\$	(341,000.00)		
BB12.	24A510 – Work & Training Admin WT137943 – Information Services				
	Other Expenses	\$	(276,830.00)		

The Department of Jobs and Family Services has requested to decrease excess appropriation. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

BC1. 24A601 – Senior and Adult Services				BA1201195			
	SA138321 – Administrative Service						
	Other Expenses	\$	(172,700.00)				
BC2.	24A601 – Senior and Adult Service	S					
	SA138354 – Management Services						
	Personal Services	\$	(337,800.00)				
	Other Expenses	\$ \$	(10,000.00)				
	Capital Outlay	\$	(14,000.00)				
BC3.	24A601 – Senior and Adult Service	S					
	SA138388 – Home Care Skilled Services						
	Personal Services	\$	(31,900.00)				
	Other Expenses	\$	(22,000.00)				
BC4.	24A601 – Senior and Adult Service	S					
	SA138420 – Home Support						
	Personal Services	\$	(17,000.00)				
	Other Expenses	\$	(90,300.00)				
BC5.	24A601 – Senior and Adult Service	S					
	SA138479 – Protective Services						
	Personal Services	\$	(47,500.00)				
	Other Expenses	\$	(20,800.00)				
	Capital Outlay	\$	(4,500.00)				

The Department of Senior and Adult Services has requested to decrease excess appropriation. The funding source is the Health and Human Services Levy Fund, the Public Assistance allocations, and fees for direct services provided to clients.

BD1.	20D446 – Brownfield Revo	20D446 – Brownfield Revolving Loan Fund		
	DV520726 – Brownfield Re	DV520726 – Brownfield Revolving Loan Fund		
	Other Expenses	\$	(99,880.77)	

BD2.	64A601 – Supplies CC577700 – County Supplies Other Expenses	\$	(21,854.39)
BD3.	64A601 – Supplies FS109751 – Fiscal – County Sup	plies	
	Other Expenses	\$	(5,100.00)
BD4.	64A606 – Fast Copier CT577551 – Fast Copy		

Other Expenses

Reductions are requested to remove surplus appropriation. Funding for the Brownfield Revolving Fund is from loan repayments and the Supplies and Fast Copier fund generate revenues from charges for supplies procurement and print shop and fast copier services.

\$

(886,253.53)

BE1.	20A303 – Children Services Fund			BA1200273
	CF134049 – Purchased Congregate	Care		
	Other Expenses	\$	(616,000.00)	
BE2.	20A303 – Children Services Fund CF134031 – CFS Foster Care			
	Other Expenses	\$	(164,000.00)	
BE3.	20A303 – Children Services Fund CF134023 – Adoption Services			
	Other Expenses	\$	(2,000.00)	
BE4.	20A303 – Children Services Fund CF134015 – Client Supportive Serv	ices		
	Other Expenses	\$	(61,000.00)	

Reduction of unused appropriation for Children Services Fund. Funding is from the Federal and State revenues, as well as the Health & Human Services Levy Fund

BF1. 24A301		Children & Family Servi	BA1200883		
		CF135467-CFS Administ	tration		
		Personal Services	\$	(60,000.00)	
		Other Expenses	\$	(759,000.00)	
BF2.	24A301	Children & Family Servi CF135483-Training	ces		
		Other Expenses	\$	(74,000.00)	

BF3.	24A301	Children & Family Services CF135491-Information Serv	rices	
		Personal Services	\$	(14,000.00)
		Other Expenses	\$	(264,000.00)
		Capital Outlays	\$	(4,300.00)
BF4.	24A301	Children & Family Services		
		CF135509-Direct Services		
		Personal Services	\$	(261,700.00)
		Other Expenses	\$	(146,500.00)
BF5.	24A301	Children & Family Services		
		CF135525-Supportive Servi	ces	
		Personal Services	\$	(12,000.00)
		Other Expenses	\$	(1,813,500.00)
BF6.	24A301	Children & Family Services		
		CF135541-Visitation Center		
		Other Expenses	\$	(94,000.00)
BF7.	24A301	Children & Family Services		
		CF135582-Permanent Custo	ody Adop	tion
		Other Expenses	\$	(15,000.00)

Appropriation decreases are requested to reduce excess appropriations in various index codes in the salaries lines in Children and Family Services. The funding source is primarily the Health and Human Services levies.

BG1.	24A301	Children & Family Services			BA1200884
		CF135582-Permanent Cust	ody Adop	otion	
		Other Expenses	\$	(9,000.00)	
BG2.	24A301	Children & Family Services CF135608-Contracted Plac	ements		
		Other Expenses	\$	(149,000.00)	
BG3.	24A301	Children & Family Services			
		CF135616-CFS Foster Hom	es		
		Personal Services	\$	(15,000.00)	
		Other Expenses	\$	(27,500.00)	
BG4.	24A435	Cuyahoga Tapestry System CF135004 – DCFS- Cuyahog Other Expenses	•	•	
		p	τ'	(=,000,000.00)	

BG5.	24A640	FCFC Public Assistance FC451492 – Family and Chi	ldren First (Council PA
		Personal Services	\$	(15,000.00)
		Other Expenses	\$	(558,000.00)
BG6.	24A635	EC- Invest In Children PA		
		EC451484 – Early Childhoo	d Admin Se	rvices
		Other Expenses	\$	(101,000.00)
BG7.	24A635	EC- Invest In Children PA		
		EC451450 –Quality Child C	are	
		Other Expenses	\$	(27,500.00)
BG8.	24A641	PA-Homeless Services HS158097 – PA-Homeless S	Services	
		Other Expenses	\$	(26,000.00)

Appropriation decreases are requested to reduce excess appropriations in several departments' index codes within Health and Human Services budget lines. The funding source is primarily the Health and Human Services levies.

BH1.	01A001 – General Fund PD140053 – Public Defend	er		BA1201928		
	Other Expenses	\$	30,000.00			
BH2.	01A001 – General Fund	on and Office				
	PR191056 – Prosecutor-Ge		44400000			
	Other Expenses	\$	144,000.00			
BH3.	01A001 – General Fund					
	DR391052 – Domestic Rela	tions Court				
	Other Expenses	\$	35,000.00			
BH4.	01A001 – General Fund					
	CO380121 – Common Plea	s-Judicial/Ge	neral			
	Other Expenses	\$	175,000.00			
BH5	20A600 – Cuyahoga Suppo	rt Enforceme	ent Agency			
	SE496000 – Child Support Enforcement Agency					
	Other Expenses	\$	15,000.00			
BH7.	20A809 – Witness Victim H	IHS				
	JA107425 – Witness Victim	n HHS				
	Other Expenses	\$	15,000.00			

Provide increased appropriations in the various departments for November 2012 data processing Funding is from the various funding sources including General Fund and Health and Human Service Levies covering the period January 1, 2012 through December 31, 2012.

BI. 24A635 EC-Invest In Children PA BA1200885

EC451484 - Early Childhood Admin Services

Personal Services \$ (7,000.00)

An appropriation decrease is requested to offset corresponding increase in appropriations for the Family and Children First Council (FCFC) budget line in order to cover data center charges in controlled services budget line. The funding source is primarily the Health and Human Services levies.

BJ. 24A640 FCFC Public Assistance

FC451492 – Family and Children First Council PA BA1200886

Other Expenses \$ 7,000.00

An appropriation increase is requested to offset corresponding decrease in appropriations in Early Childhood for unused appropriation and to cover data center charges in the Family and Children First Council controlled services budget line. The funding source is primarily the Health and Human Services levies.

ITEMS SUBMITTED THROUGH DECEMBER 11, 2012

Resolutions: Appropriation Transfers

A. FROM: 24A601 – Senior and Adult Services **BA1201008**

SA138479 – Protective Services

Other Expenses \$ 8,000.00

TO: 24A601 – Senior and Adult Services

SA138602 – Home Based Services

Other Expenses \$ 8,000.00

The Department of Senior and Adult Services has requested an appropriation transfer to realign appropriations within index codes to cover projected travel reimbursement expenses for the remainder of the year. The funding source is the Health and Human Services Levy Fund, the Public Assistance allocations, and fees for direct services provided to clients.

B. FROM: 61A608 – Central Custodial Services BA1200936

SH352005 – Building Security Services-OPBA-Officers Other Expenses \$ 23,000.00

TO: 61A608 – Central Custodial Services

SH352005 – Building Security Services-OPBA-Officers
Capital Outlay \$ 23,000.00

The appropriation transfer is to fund replacement guns in the Sheriff's Protective Services Division. Funding is from the internal service fund which derives revenues from charges to county departments which utilize the services of security for the buildings. The internal service fund covers the period January 1, 2012 through December 31, 2012.

C. FROM: 21A837 – State Homeland Security (SHSP) BA1200993

JA763425 – State Homeland Security (SHSP) 2009/2012 Capital Outlays \$ 583.43

TO: 21A837 – State Homeland Security (SHSP)

JA763425 – State Homeland Security (SHSP) 2009/2012 Personal Services \$ 583.43

To provide sufficient appropriations in personal services for a pending expense adjustment by year's end in the State Homeland Security Grant. Funding is from the United States Department of Homeland Security passed through the Ohio Emergency Management Agency covering the period August 1, 2010 through September 30, 2012.

D. FROM: 21A579 – VAWA Administration Grant BA1201013

JA752907 – FY2010 VAWA Administration Fund CY2011 Other Expenses \$ 2,814.91 TO: 21A579 – VAWA Administration Grant

JA752907 – FY2010 VAWA Administration Fund CY2011 Personal Services \$ 2,814.91

To provide sufficient appropriations in personal services for a pending expense adjustment for closure of the grant. Funding is from the State of Ohio, Office of Criminal Justice Services covering the period January 1, 2011 through December 31, 2011.

ITEMS SUBMITTED THROUGH DECEMBER 18, 2012

E. FROM: 01A001 – General Fund **BA1201051**

HC019018 – Human Resource Commission

Other Expenses \$ 16,000.00

TO: 01A001 – General Fund

HC019018 – Human Resource Commission

Personal Services \$ 16,000.00

An appropriation transfer of surplus budget from contractual services to cover a projected payroll budget shortfall. Funding comes from the General Fund.

F. FROM: 20A301 – Real Estate Assessment BA1201052

FS109702 - Tax Assessments

Other Expenses \$ 37,000.00

TO: 20A301 – Real Estate Assessment

BR420067 – Board of Revision

Other Expenses \$ 37,000.00

A transfer is requested to purchase software that provides scanning and electronic file capability in preparation of anticipated appeals to the sexennial reappraisal. The source of funding is a percentage of real property tax receipts.

G. FROM: 01A001 – General Fund BA1201042

CO380121 - Judicial Administration

Other Expenses \$ 200,000.00

TO: 01A001 – General Fund

CO380410 - Probation Psychiatric

Personal Services \$ 200,000.00

Request to transfer appropriation within the General Fund of the Court of Common Pleas to cover fringe benefit expenses in the Probation division through year-end.

H. FROM: 01A001 – General Fund **BA1201041**

CO380121 – Judicial Administration

Other Expenses \$ 217,000.00

01A001 – General Fund CO380220 – Court Services

Capital Outlays \$ 11,000.00

01A001 - General Fund

CO380410 – Probation Psychiatric

Capital Outlays \$ 15,000.00

TO: 01A001 – General Fund

CO380196 - Magistrates

Personal Services \$ 89,000.00 Other Expenses \$ 9,000.00

01A001 - General Fund

CO380121 - Judicial Administration

Personal Services \$ 119,000.00 Capital Outlays \$ 26,000.00

Request to transfer appropriation within the General Fund budget in the various divisions of the Court of Common Pleas to cover anticipated expenses through the rest of the year.

I. FROM: 01A001 – General Fund **BA1201043**

MI512657 - Miscellaneous

Other Expenses \$ 9,000.00

TO: 01A001 – General Fund

MI512657 - Miscellaneous

Personal Services \$ 9,000.00

Request to transfer appropriation within the Miscellaneous Obligations budget to cover projected legacy expenses related to the Early Retirement Incentive Program implemented in 2009. This budget is supported by the County's General Fund.

ITEMS SUBMITTED THROUGH DECEMBER 26, 2012

J. FROM: 01A001 – General Fund **BA1201048**

PC400051 - Probate Court

Other Expenses \$ 55,000.00

TO: 01A001 – General Fund

PC400051 - Probate Court

Personal Services \$ 55,000.00

Request to transfer available appropriation within Probate Court's General Fund budget to post the final payroll of the year.

K. FROM: 01A001 – General Fund **BA1201058**

CO380196 - Magistrates

Personal Services \$ 91,000.00

01A001 – General Fund CO380220 – Court Services

Personal Services \$ 222,000.00

TO: 01A001 – General Fund

Judicial Administration

Personal Services \$ 113,000.00 Other Expenses \$ 200,000.00

Request to transfer appropriation within the Court of Common Pleas' General Fund budget to post remaining year expenses.

L. FROM: 01A001 – General Fund **BA1201089**

JC372060 - Juvenile Court - Legal

Personal Services \$ 60,000.00

01A001 - General Fund

JC372052 – Juvenile Court – Judges

Personal Services \$ 15,000.00

01A001 - General Fund

JC370056 - Juvenile Court - Detention Home

Personal Services \$ 130,000.00

TO: 01A001 – General Fund

JC372060 - Juvenile Court - Legal

Other Expenses \$ 30,000.00

01A001 - General Fund

JC372052 - Juvenile Court - Judges

Other Expenses \$ 15,000.00

01A001 - General Fund

JC370056 - Juvenile Court - Detention Home

Other Expenses \$ 160,000.00

The requested transfers would clear year-end expenditures in the above budget accounts. Funding comes from the General Fund.

M. FROM: 21A762 – Port Security Grant BA1201018

JA767855 – Port Security Grant FFY09-2009/2012 Other Expenses \$ 1,598.22

TO: 21A762 – Port Security Grant

JA767855 – Port Security Grant FFY09-2009/2012 Personal Services \$ 1,598.22

The appropriations transfer is requested to prepare the grant for closure. Funding is from the United States Department of Homeland Security passed through the Ohio Emergency Management Agency covering the period August 1, 2010 through May 31, 2013.

N. FROM: 61A608 – Central Security Services-Sheriff BA1201025

SH352005 – Building Security Services-OPBA-Officers Personal Services \$ 81,900.00

TO: 61A608 – Central Security Services-Sheriff

SH352013 – Building Security Services-OPBA-Sergeants Personal Services \$ 77,000.00

TO: 61A608 – Central Security Services-Sheriff

SH352021 – Building Security Services-Non-Bargaining Personnel

Personal Services \$ 4,900.00

The appropriations transfer is to realign appropriations within the Sheriff's Protective Services Division grouped by bargaining and non-bargaining units. Funding is from the internal service fund which derives revenues from charges to county departments which utilize the services of security for the buildings. The internal service fund covers the period January 1, 2012 through December 31, 2012.

O. FROM: 21A500 – Urban Area Security Initiative (URSI) BA1201029

JA741199 – Urban Area Security Initiative (URSI)
Personal Services \$ 766.06

TO: 21A500 – Urban Area Security Initiative (URSI)

JA741199 – Urban Area Security Initiative (URSI)
Capital Outlays \$ 766.06

The appropriations transfer is requested to prepare the grant for closure. Funding is from the United States Department of Homeland Security passed through the Ohio Emergency Management Agency covering the period August 1, 2010 through September 30, 2012.

P. FROM: 61A608 – Central Security Services-Sheriff BA1201032

SH352005 – Building Security Services-OPBA-Officers Personal Services \$ 22,050.00 TO: 61A608 – Central Security Services-Sheriff

SH352013 – Building Security Services-OPBA-Sergeants Personal Services \$ 16,250.00

TO: 61A608 – Central Security Services-Sheriff

SH352021 – Building Security Services-Non-Bargaining Personnel

Personal Services \$ 5,800.00

The appropriations transfer is to realign appropriations within the Sheriff's Protective Services Division grouped by bargaining and non-bargaining units. Funding is from the internal service fund which derives revenues from charges to county departments which utilize the services of security for the buildings. The internal service fund covers the period January 1, 2012 through December 31, 2012.

Q. FROM: 01A001 – General Fund **BA1201069**

PR194720 - General Office-Prosecutor

Other Expenses \$ 176,000.00

TO: 01A001 – General Fund

PR194720 - General Office-Prosecutor

Personal Services \$ 176,000.00

The appropriations transfer is to provide sufficient appropriations with the Prosecutor's General Office account for last pay in December. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

R. FROM: 24A301 – Children and Family Services BA1200867

CF135467 – Administrative Services

Other Expenses \$ 5,200.00

TO: 24A301– Children and Family Services

CF135467 – Administrative Services

Capital Outlays \$ 5,200.00

The Department of Children and Family Services requests an appropriations transfer from one budget line to another within the same index code in order to provide appropriations to the cover approved computer hardware expenses. The Department of Children and Family Services is funded both by Federal and State revenue as well as the HHS Levy Fund.

S. FROM: 24A301 – Children and Family Services BA1200863

CF135442 – Caregiver Parent Recruitment

Personal Expenses \$ 1,000.00

24A301 – Children and Family Services CF135467 – Administrative Services

Personal Expenses \$ 7,000.00

24A301 – Children and Family Services

CF135541 – Multi Systemic Therapy (MST) Unit Personal Expenses \$ 36,200.00

24A301 – Children and Family Services CF135608 – Contracted Placements

Personal Expenses \$ 1,500.00

TO: 24A301– Children and Family Services

CF135483 – Administrative Services

Personal Expenses \$ 200.00

24A301 - Children and Family Services

CF135509 - Direct Services

Personal Expenses \$ 41,000.00

24A301 – Children and Family Services

CF135525 - Supportive Services

Personal Expenses \$ 4,500.00

The Department of Children and Family Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

ITEMS SUBMITTED THROUGH January 2, 2013

T. FROM: 01A001 – General Fund **BA1201076**

CR180026 - Medical Examiner-Operations

Personal Services \$ 9,000.00

TO: 01A001 – General Fund

CR180026 - Medical Examiner-Operations

Other Expenses \$ 9,000.00

The appropriations transfer with the Medical Examiners Operations account for year-end expenses. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

U. FROM: 20A390 – Emergency Management BA1201078

JA100123 – Emergency Management

Personal Services \$ 15,400.00

TO: 20A390 – Central Security Services-Sheriff

JA100123 – Emergency Management

Other Expenses \$ 15,400.00

The appropriations transfer is to realign appropriations within Public Safety and Justice Services Emergency Management Division for controlled costs such as telephone, data processing and indirect costs. Funding is from grants and services along with a General Fund subsidy covering the period January 1, 2012 through December 31, 2012.

V. FROM: 01A001 – General Fund **BA1201079**

SH350470 - Jail Operations-Sheriff

Other Expenses \$ 650,920.00

TO: 01A001 – General Fund

SH350579 – Sheriff Operations

Other Expenses \$ 650,920.00

The appropriations transfer is to realign appropriations between the Sheriff's Jail Operations and Sheriff Operations for controlled costs such as telephone, data processing and space maintenance. Funding is from General Fund covering the period January 1, 2012 through December 31, 2012.

W. FROM: 01A001 – Law Enforcement-Sheriff BA1201081

SH350272 – Law Enforcement-Sheriff

Personal Services \$ 18,300.00

TO: 01A001 – General Fund

SH350579 - Sheriff Operations

Personal Services \$ 18,300.00

The appropriations transfer is to realign appropriations between Law Enforcement and Operations for fringe benefits. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

X. FROM: 01A001 – General Fund **BA1201082**

SH350579 – Sheriff Operations

Other Expenses \$ 1,000.00 Capital Outlays \$ 6,000.00

TO: 01A001 – General Fund

SH350579 – Sheriff Operations

Personal Services \$ 7,000.00

The appropriations transfer is to provide sufficient appropriations within the Sheriff's Operations account for last pay in December. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

Y. FROM: 01A001 – General Fund **BA1201083**

SH350470 – Jail Operations-Sheriff

Other Expenses \$ 127,000.00

TO: 01A001 – General Fund

SH350470 – Jail Operations-Sheriff

Personal Services \$ 127,000.00

The appropriations transfer is to provide sufficient appropriations within the Sheriff's Jail Operations account for fringe benefits in December. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

Z. FROM: 01A001 – General Fund **BA1201110**

SH351080 – Impact Unit/Community Policing
Capital Outlays \$ 5,400.00

TO: 01A001 – General Fund

SH351080 - Impact Unit/Community Policing

Personal Services \$ 5,400.00

The appropriations transfer is to provide sufficient appropriations within the Sheriff's Impact Unit/Community Policing account for the last pay in December. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AA. FROM: 21A837 – State Homeland Security (SHSG) BA1201161

JA763441 – State Homeland Security (SHSP) 2010/2013 Capital Outlays \$ 143,000.00

TO: 21A837 – State Homeland Security (SHSG)

JA763441 – State Homeland Security (SHSP) 2010/2013 Other Expenses \$ 143,000.00

The appropriations transfer is to provide sufficient appropriations within the State Homeland Security Program (SHSP) 2010/2013 for pending request for proposal to purchase a system for the fire service discipline. Funding is from the United States Department of Homeland Security passed through the Ohio Emergency Management Agency covering the period August 1, 2010 through March 31, 2013.

AB. FROM: 24A430 - Executive Office of HHS **BA1201175**

HS157289 - Executive Office of H&HS

Personal Services \$ 102,000.00

TO: Executive Office of HHS

HS157289 - Executive Office of H&HS

Other Expenses \$ 102,000.00

Request to transfer available appropriation within the budget for the Office of Health and Human Services to enable the 2012 indirect chargeback to post. This budget is supported by the County's Public Assistance allocations and the Health and Human Services Levy Fund.

AC1. FROM: 24A601 – Senior and Adult Services BA1201093

SA138354 – Management Services

Personal Services \$ 1,000.00

TO: 24A601 – Senior and Adult Services

SA138321 - Administrative Services-SAS

Personal Services \$ 1,000.00

AC2. FROM: 24A601 – Senior and Adult Services BA1201105

SA138321 - Administrative Services-SAS

Other Expenses \$ 5,300.00

TO: 24A601 – Senior and Adult Services

SA138602 - Home Based Services

Other Expenses \$ 5,300.00

The Department of Senior and Adult Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary/fringe and commodities charges for the remainder of the year. The funding source is the Health and Human Services Levy Fund, the Public Assistance allocations, and fees for direct services provided to clients.

AD. FROM: 24A510 – Work & Training Admin **BA1201011**

WT137109 – Admin Services-General Manager Personal Services \$ 53,800.00

TO: 24A510 – Work & Training Admin

WT137141 - Client Support Services

Personal Services \$ 14,400.00

24A510 – Work & Training Admin

WT137430 - Ohio City NFSC

Personal Services \$ 1,700.00

24A510 – Work & Training Admin WT137943 – Information Services

Personal Services \$ 28,000.00

24A510 – Work & Training Admin WT137455 – Quincy Place NFSC

Personal Services \$ 9,700.00

The Department of Jobs and Family Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

AE. FROM: 24A510 – Work & Training Admin **BA1201102**

WT137463 - VEB Building NFSC

Personal Services \$ 10,000.00

24A510 – Work & Training Admin WT137539 – West Shore NFSC

Personal Services \$ 13,000.00

24A510 – Work & Training Admin WT137471 – Mount Pleasant NFSC

Other Expenses \$ 81,033.00

TO: 24A510 – Work & Training Admin

WT137414 – Southgate NFSC

Personal Services \$ 104,033.00

The Department of Jobs and Family Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

AF. FROM: 24A510 – Work & Training Admin **BA1201103**

WT137471 - Mount Pleasant NFSC

Personal Services \$ 53,800.00

24A510 – Work & Training Admin WT137315 – Work First Services

Personal Services \$ 7,900.00

TO: 24A510 – Work & Training Admin

WT137455 - Quincy Place NFSC

Personal Services \$ 61,700.00

The Department of Jobs and Family Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

AG. FROM: 24A641 –PA Homeless Services BA1200870

HS158097 - PA Homeless Services

Other Expenses \$ 1,596.00

TO: 24A641 –PA Homeless Services

HS158097 – PA Homeless Services

Personal Expenses \$ 1,596.00

The Department of Homeless Services has requested appropriation transfers to realign appropriations between index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

AH1. FROM: 01A001 – General Fund **BA1201099**

FS109637 - Financial Reporting

Other Expenses \$ 155,000.00

TO: 01A001 – General Fund

FS109611 – Fiscal Office Administration

Other Expenses \$ 40,000.00

01A001 - General Fund

FS109652 - Fiscal Operations – Contractual Services Other Expenses \$ 110,000.00

01A001 - General Fund

FS109686 - Fiscal Operations — Property Valuations Other Expenses \$ 5,000.00

AH2. FROM: 01A001 – General Fund

HR018010 - Human Resources

Personal Services \$ 185,000.00

TO: 01A001 – General Fund

HR018010 - Human Resources

Other Expenses \$ 185,000.00

This appropriation transfer would provide sufficient appropriation for controlled expenses including data processing and space maintenance chargebacks. This includes reallocation of \$155,000 among Fiscal Office divisions and a transfer of \$185,000 from Human Resources salary surplus to cover Human Resources data processing expenses. Funding comes from the General Fund.

AI. FROM: 20A301 – Real Estate Assessment BA1201100

FS109702 – Fiscal Operations - Tax Assessments
Other Expenses \$ 405,000.00

TO: 20A301 – Real Estate Assessment

BR420067 – Board of Revision

Other Expenses \$ 405,000.00

This request would transfer surplus appropriation from the consulting contracts for the sexennial reappraisal and surplus controlled services from the Fiscal Office to the Board of Revision to cover space maintenance charges. The space maintenance charge is anticipated to double from the 2011 Board of Revision expenses because it has increased its square footage to accommodate its caseload.

ITEMS SUBMITTED THROUGH January 8, 2013

AJ. FROM: 24A510 – Work & Training Admin BA1201293

WT137109 – Admin Services-General Manager
Personal Services \$ 1,500.00

TO: 24A510 – Work & Training Admin

WT137539 - West Shore NFSC

Personal Services \$ 1,500.00

The Department of Jobs and Family Services has requested appropriation transfers to realign appropriations within index codes to cover projected salary and fringe charges for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

AK. FROM: 54A100 – Sanitary Engineer BA1201232

ST540252 – Sanitary Engineer - Administration
Capital Outlays \$ 290,000.00

TO: 54A100 – Sanitary Engineer

ST540252 – Sanitary Engineer-Administration
Other Expenses \$ 290,000.00

A transfer is requested to cover the certification of a new contract (CE1200145) that has been awarded for general engineering services. This item will also cover indirect cost charges. Revenues come from charges for services.

AL. FROM: 01A001 – General Fund **BA1201233**

FS109637 - Financial Reporting

Other Expenses \$ 24,000.00

01A001 - General Fund

HR018010 – Human Resources

Personal Services \$ 30,000.00

TO: 01A001 – General Fund

FS109611 - Fiscal Office Administration

Other Expenses \$ 24,000.00

01A001 - General Fund

HR018010 - Human Resources

Other Expenses \$ 30,000.00

The budget adjustments would cover space maintenance expenses. Funding comes from the General Fund.

AM. FROM: 01A001 – General Fund **BA1201234**

EX016006 – Executive Office

Personal Services \$ 110,500.00

TO: 01A001 – General Fund

EX016006 - Executive Office

Other Expenses \$ 110,500.00

A budget adjustment would use vacancy savings from salary to cover data processing expenses. Funding comes from the General Fund.

AN. FROM: 63A100 – Information Services Center BA1201252

IS821009 - Cuyahoga County Information Services Center

Personal Services \$ 396,400.00 Capital Outlay \$ 96,000.00

TO: 63A100 – Information Services Center

IS821009 - Cuyahoga County Information Services Center

Other Expenses \$ 492,400.00

A budget adjustment would use surplus appropriation to cover indirect charges within the controlled budget category. Funding comes from charges to user agencies for computer and telephone services.

AO. FROM: 01A001 – General Fund BA1201254

FS109629 – Office of Budget & Management

Personal Services \$ 21,000.00

01A001 - General Fund

FS109686 – Fiscal Operations – Property Valuation Personal Services \$ 14,100.00

TO: 01A001 – General Fund

FS109652 – Fiscal Operations – Contractual Services Personal Services \$ 9,000.00

01A001 - General Fund

FS109660 – Treasury Management

Personal Services \$ 26,100.00

This request would reallocate personal services appropriation within the Fiscal Office General Fund budget to cover year-end payroll expenses. Funding comes from the General Fund.

AP. FROM: 01A001 – General Fund **BA1201273**

FS109611 – Fiscal Office Administration

Personal Services \$ 56,740.00

01A001 - General Fund

FS109629 – Office of Budget & Management

Other Expenses \$ 11,293.06

01A001 - General Fund

FS109637 - Financial Reporting

Other Expenses \$ 5,500.00

01A001 - General Fund

FS109645 – Fiscal Operations – Records & Licenses
Capital Outlay \$ 4,300.00

01A001 - General Fund

HR018010 - Human Resources

Personal Services \$ 338,115.00

TO: 01A001 – General Fund

EX016006 – Executive Office

Other Expenses \$ 24,000.00

01A001 - General Fund

FS109611 - Fiscal Office Administration

Other Expenses \$ 149,750.00

01A001 - General Fund

FS109629 – Office of Budget & Management

Personal Services \$ 5,100.00

01A001 - General Fund

FS109645 – Fiscal Operations – Records & Licenses Personal Services \$ 24,548.06

01A001 – General Fund

FS109660 - Treasury Management

Personal Services \$ 5,050.00

01A001 - General Fund

HR018010 – Human Resources

Other Expenses \$ 207,500.00

This request would transfer funds within General Fund accounts to cover year-end expenses. Funding for all transfers come from the General Fund.

AQ. FROM: 01A001 – General Fund **BA1201192**

JC372052 – JC Judges

Capital Outlay \$ 22,850.00

20A811 – JC Detention & Probation Services

JC107532 – JC Legal Services

Other Expenses \$ 33,000.00

TO: 01A001 – General Fund

JC375055 – JC Child Support

Personal Services \$ 22,850.00

20A811 – JC Detention & Probation Services

JC107516 – JC Probation Services

Personal Services \$ 33,000.00

This transfer would utilize surplus appropriation to cover 2012 Workers' Compensation charges including fringe benefits and controlled charges. Funding comes from the General Fund and the Health and Human Services Levy Fund.

AR.	FROM:	01A001 – General Fund IA018002 – Internal Audit			BA1201274		
		Personal Services	\$	239,975.00			
		Other Expenses	, \$	166,500.00			
		·		·			
		01A001 – General Fund					
		HC019018 – Human Resou	rce Commis	sion			
		Other Expenses	\$	181,333.00			
		Capital Outlay	\$	1,020.00			
		01A001 – General Fund					
		FS109611 - Fiscal Office Ad	ministration	ı			
		Personal Services	\$	52.00			
		Other Expenses	\$ \$ \$	46.00			
		Capital Outlay	\$	1,373.00			
		01A001 – General Fund					
		FS109629 – Office of Budget & Management					
		Personal Services	\$	665.00			
		Other Expenses	\$ \$	43,859.00			
		Capital Outlay	\$	1,850.00			
		01A001 – General Fund					
		FS109637 – Financial Repo	rting				
		Personal Services	\$	271,460.00			
		Other Expenses	\$ \$	380,785.00			
		Capital Outlay	\$	1,850.00			
	TO:	01A001 – General Fund HR018010 – Human Resou	rces				
		Other Expenses	\$	4,700.00			
		Care Experiees	T	.,,, 00.00			

01A001 – General Fund JC372052 – Juvenile Court – Judges

Other Expenses \$ 1,291,398.00

This budget adjustment would transfer unspent appropriation in various General Fund agency budgets primarily to help cover the 2012 space maintenance expenses associated with the costs at the new Juvenile Justice Center as well as rent payments for Human Resources space. Funding is from the General Fund.

AS.	FROM:	01A001 – General Fund			BA1201280
		AE511501 – Bureau of Insp		44.075.00	
		Other Expenses	\$	11,075.00	
		01A001 – General Fund AU150011 – General Accor	ınting		
		Other Expenses	\$	80,356.00	
		Other Expenses	Ų	80,330.00	
		01A001 – General Fund BR420059 - Board of Revis	ion		
		Other Expenses	\$	12,061.00	
		Capital Outlay	\$	6,267.00	
		01A001 – General Fund CC012088 – Office of Proci	ırement & Γ)iversity	
		Other Expenses	\$	22,830.00	
		Other Expenses	Y	22,030.00	
		01A001 – General Fund EX016006 -Executive Office	e		
		Other Expenses	\$	393.00	
		01A001 – General Fund FS109645 – Fiscal Ops – Re	ecords & Lice	enses	
		Personal Services	\$	2,105.00	
		Other Expenses	\$	176,485.00	
		Capital Outlay	\$	78.00	
		01A001 – General Fund CT577601 – Archives			
		Other Expenses	\$	400,000.00	
		01A001 – General Fund			
		DV014100 – Economic Dev		600 000 00	
		Other Expenses	\$	600,000.00	
	TO:	01A001 – General Fund JC370056 – Juvenile Court	– Detention	Home	
		Personal Services	\$	64,600.00	
		Other Expenses	\$	701,400.00	
			•	,	

01A001 - General Fund

JC372052 – Juvenile Court – Judges

Personal Services \$ 12,550.00 Other Expenses \$ 533,100.00

This request would transfer appropriation among various General Fund accounts to cover the 2012 space maintenance costs at the Juvenile Court Justice Center and Juvenile Court Workers' Compensation charges.

AT.	FROM:	01A001 – General Fund	l		BA1201218
		JC107524 – JC - Detenti	on Services		
		Personal Services	\$	1,231.00	
		Other Expenses	\$	5,288.00	
		Capital Outlay	\$	875.00	
		01A001 – General Fund	l		
		JC107516 – JC Probatio	n Services		
		Capital Outlay	\$	29.00	
		01A001 – General Fund	I		
		JC107532 - JC - Legal Se	ervices		
		Personal Services	\$	11,979.00	
		Other Expenses	\$	6,468.00	
		Capital Outlay	\$	378.00	
	TO:	01A001 – General Fund	l		
		JC107516 – JC – Probat	ion Services		
		Personal Services	\$	12,967.00	
		Other Expenses	\$	13,281.00	

This budget adjustment would transfer HHS Levy funds within Juvenile Court accounts to cover year expenses for space maintenance and workers' compensation charges.

AU.	FROM:	01A001 – General Fund DR391052 – Domestic Rel	ations Court		BA1201166
		Personal Services	\$	14,126.00	
	FROM:	01A001 – General Fund DR495515 – Domestic Rel	ations Child S	Support	
		Other Expenses	\$	1,154.00	
	TO:	01A001 – General Fund			
		DR391052 – Domestic Rel	ations Court		
		Other Expenses	\$	776.00	
		Capital Outlays	\$	1,190.00	

TO: 01A001 – General Fund

DR495515 – Domestic Relations Child Support
Personal Services \$ 6,066.00
Capital Outlays \$ 7,248.00

The appropriations transfer within the Domestic Relations Court for year-end expenses. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AV. FROM: 01A001 – General Fund BA1201169

SH350579 – Sheriff Operations

Other Expenses \$ 177,000.00

TO: 01A001 – General Fund

SH350470 - Jail Operations

Other Expenses \$ 177,000.00

The appropriations transfer within the Sheriff's Department for space maintenance charges. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AW. FROM: 01A001 – General Fund **BA1201172**

JA302224 – Public Safety Grants Administration
Personal Services \$ 207,300.00
Other Expenses \$ 80,000.00

TO: 01A001 – General Fund

JA050088 - Justice Affairs Administration

Other Expenses \$ 287,300.00

The appropriations transfer within Public Safety and Justice Service general fund accounts for space maintenance charges. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AX. FROM: 01A001 – General Fund **BA1201283**

JA050088 – Justice Affairs Administration

Personal Services \$ 59,800.00

TO: 01A001 – General Fund

JA050088 – Justice Affairs Administration

Other Expenses \$ 59,800.00

The appropriations transfer within Public Safety and Justice Service, Justice Services Administration account for space maintenance charges. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AY. FROM: 01A001 – General Fund **BA1201182**

PR200071 – Prosecutor Child Support

Personal Services \$ 140,200.00

TO: 01A001 – General Fund

PR191056 – Prosecutor General Office

Personal Services \$ 140,200.00

The appropriations transfer within Prosecutor Office accounts for year-end expenses. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

AZ. FROM: 61A608 – Central Security Services-Sheriff BA1201184

SH352005 – Building Security Services-OPBA-Officers Personal Services \$ 186,600.00

TO: 61A608 – Central Security Services-Sheriff

SH352005 – Building Security Services-OPBA-Officers
Other Expenses \$ 162,000.00
Capital Outlays \$ 24,600.00

The appropriations transfer within the Sheriff's Department, Protective Services Division for year-end expenses. Funding is from an internal service fund that charges applicable departments with the County for security services covering the period January 1, 2012 through December 31, 2012.

BA. FROM: 21SA029 – ARRA-Early Case Disposition 2009/2013 **BA1201185**

PR736942 – ARRA-Early Case Disposition 2009/2013 Other Expenses \$ 25,571.78

TO: 21SA029 – ARRA-Early Case Disposition 2009/2013

PR736942 – ARRA-Early Case Disposition 2009/2013 Personal Services \$ 25,571.78

The appropriations transfer to be in compliance with the approved budget adjustment and to prepare the grant for closeout in the ARRA-Early Disposition Grant awarded to the Prosecutor's Office through the Sheriff's Office from the City of Cleveland funded by the American Recovery and Reinvestment Act from the United States Department of Justice, Bureau of Assistance Edward Byrne Memorial Justice Assistance Grant Program (JAG) covering the period March 1, 2009 through February 28, 2013.

BB. FROM: 61A608 – Central Security Services-Sheriff BA1201207

SH352005 – Building Security Services-OPBA-Officers Personal Services \$ 110,000.00

TO: 61A608 – Central Security Services-Sheriff

SH352005 – Building Security Services-OPBA-Officers
Capital Outlays \$ 110,000.00

The appropriations transfer within the Sheriff's Department, Protective Services Division for year-end expenses. Funding is from an internal service fund that charges applicable departments with the County for security services covering the period January 1, 2012 through December 31, 2012.

BC.	FROM:	01A001 – General Fund	d		BA1201214
		SH350470 – Jail Opera	tions		
		Personal Services	\$	77.000.00	

Personal Services \$ 77,000.00 Capital Outlays \$ 11,000.00

01A001 - General Fund

SH350579 – Sheriff Operations

Personal Services \$ 76,000.00

TO: 01A001 – General Fund

SH350470 - Jail Operations

Other Expenses \$ 164,000.00

The appropriations transfer within the Sheriff's Department for year-end food expenses in Jail Operations Division. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

BD.	FROM:	01A001 – General Fund	BA1201237
-----	-------	-----------------------	-----------

PD140053 – Public Defender

Personal Services \$ 185,302.00

TO: 01A001 – General Fund

PD140053 - Public Defender

Other Expenses \$ 185,302.00

Request to transfer appropriation within Public Defender's budget to support space maintenance charge-backs. The Public Defender is supported by the General Fund.

BE.	FROM:	01A001 – General Fund	BA1201277
-----	-------	-----------------------	-----------

CO380121 – Judicial Administration

Other Expenses \$ 416,000.00

TO: 01A001 – General Fund

DR391052 - Domestic Relations

Other Expenses \$ 244,000.00

01A001 - General Fund

DR495515 - Bureau of Support

Other Expenses \$ 172,000.00

Request to transfer appropriation from the Court of Common Pleas to Domestic Relations Court to post space maintenance chargebacks. Both Courts are supported by the County's General Fund.

BF. FROM: 01A001 – General Fund **BA1201279**

CO380220 - Court Services

Personal Services \$ 18,000.00 Other Expenses \$ 295,000.00

TO: 01A001 – General Fund

CO380410 - Probation/Psychiatric

Personal Services \$ 295,000.00 Other Expenses \$ 10,000.00

01A001 - General Fund

CO380121 - Judicial Administration

Other Expenses \$ 8,000.00

Request to transfer appropriation within the Court of Common Pleas' General Fund budget to post year-end expenses.

BG. FROM: 01A001 – General Fund BA1201135

CO380196 - Magistrates

Other Expenses \$ 4,000.00

01A001- General Fund

CO380121 – Judicial Administration

Other Expenses \$ 25,000.00

TO: 01A001 – General Fund

CO380196 - Magistrates

Personal Services \$ 4,000.00

01A001 - General Fund

CO380121 - Judicial Administration

Personal Services \$ 25,000.00

Request to transfer appropriation within the Court of Common Pleas' General Fund budget to post 2012 workers' compensation charges.

BH. FROM: 01A001 – General Fund **BA1200136**

CO380196 - Magistrates

Other Expenses \$ 8,000.00

01A001 - General Fund

CO380121 - Judicial Administration

Other Expenses \$ 16,000.00

01A001 – General Fund

CO380220 – Court Services

Personal Services \$ 15,000.00

TO: 01A001 – General Fund

CO380410 – Probation/Psychiatric

Personal Services \$ 39,000.00

Request to transfer appropriation within the Court of Common Pleas General Fund budget to post 2012 workers' compensation expenses.

BI. FROM: 20A812 – Common Pleas Special Project I BA1201137

CO456475 - Common Pleas Special Project I

Personal Services \$ 3,000.00

TO: 20A812 – Common Pleas Special Project I

SH456483 – Sheriff Special Project

Personal Services \$ 3,000.00

Request to transfer appropriation from the Court of Common Pleas to the Sheriff's Office — within the same fund — to support benefits expenses in the division of the Sheriff's Office that specifically deals with foreclosures. This Fund, established by the Court of Common Pleas, is supported by a \$200 fee on foreclosure filings and covers related expenses in the Court of Common Pleas, the Clerk of Courts Office, and the Sheriff's Office. This request does not impact the County's General Fund.

BJ. FROM: 20A099 – TASC Medicaid **BA1201140**

CO456525 - TASC Medicaid (CO)

Other Expenses \$ 523.55

TO: 20A099 – TASC Medicaid

CO456525 – TASC Medicaid (CO)

Personal Services \$ 523.55

Request to transfer appropriation within the Medicaid portion of the Treatment Alternative to Street Crimes' budget to post expenses at the subobject level. This budget captures reimbursement for expenses from Federal Medicaid. This request does not impact the County's General Fund.

BK. FROM: 24A301 – Children and Family Services BA1200878

CF135467 - Admin Services-CFS

Other Expenses \$ 26,000.00

TO: 24A301 – Children and Family Services

CF135467 - Admin Services-CFS

Capital Outlays \$ 26,000.00

The Department of Jobs and Family Services has requested appropriation transfers to realign appropriations within the same index code in order to cover pre-encumbrances and expenses for the remainder of the year. The funding source is primarily Federal/State as well as the Health and Human Services Levy Fund.

BL1. FROM: 65A604 - Postage BA1201126

CT577353 - County Mailroom

Personal Services \$ 49,353.06

TO: 65A604 - Postage

CT577353 - County Mailroom

Other Expenses \$ 49,353.06

BL2. FROM: 61A607 – Centralized Custodial Services BA1201125

CT571000 – Buildings and Grounds - Administration Personal Services \$ 482,489.26 Other Expenses \$ 60,863.00

FROM: 61A607 – Centralized Custodial Services

CT577379 – Buildings and Grounds – Custodial Services Personal Services \$ 344,664.43

TO: 61A607 – Centralized Custodial Services

CT577411 – Buildings and Grounds – Other Services Other Expenses \$ 888,016.69

BL3. FROM: 26A601 – General Gas and License Fees BA1201124

CE835249 – Maintenance Engineer Road & Bridge Personal Services \$ 85,000.00

TO: 26A601 – General Gas and License Fees

CE835025 - Road & Bridge Administration

Personal Services \$ 85,000.00

Transfers are requested to cover year end payroll, space maintenance, data processing, and indirect cost charges. The source of funding for the Custodial Fund and County Mailroom is user fees. The Road & Bridge account receives funding from license and gas taxes.

BM. FROM: 61A607 – Centralized Custodial Services BA1201133

CT577395 – Buildings and Grounds – Trades Services Personal Services \$ 92,000.00

TO: 61A607 – Centralized Custodial Services

CT577411 – Buildings and Grounds – Other Services Other Expenses \$ 92,000.00

A transfer is requested to cover outstanding utilities expenses for the building maintenance fund. Revenues come from charges to user agencies for space maintenance.

BN1. FROM: 61A607 – Centralized Custodial Services BA1201146

CT577395 – Buildings and Grounds – Trades Services Personal Services \$ 42,000.00

TO: 61A607 – Centralized Custodial Services

CT571000 – Buildings and Grounds – Administration Personal Services \$ 27,000.00 Other Expenses \$ 15,000.00

BN2. FROM: 61A607 – Centralized Custodial Services BA1201147

CT577379 – Buildings and Grounds – Custodial Services
Personal Services \$ 56,000.00

TO: 61A607 – Centralized Custodial Services

CT577411 – Buildings and Grounds – Other Services Other Expenses \$ 56,000.00

BN3. FROM: 65A604 – Postage BA1201148

CT577353 - County Mailroom

Other Expenses \$ 6,400.00

TO: 65A604 – Postage

CT577353 - County Mailroom

Personal Services \$ 6,400.00

Transfers are requested to cover outstanding workers' compensation and telephone charges for the building maintenance and mailroom funds. Revenues come from charges to user agencies for space maintenance and mailroom services.

BO. FROM: 20N306 – Soil and Water Conservation BA1201151

SW500058 – Soil and Water Conservation

Other Expenses \$ 1,464.66

TO: 20N306 – Soil and Water Conservation

SW500058 - Soil and Water Conservation

Personal Services \$ 1,464.66

A transfer is requested to cover outstanding workers' compensation charges for Soil and Water Conservation District. Revenues come from municipalities, the State, and a County subsidy for pollution prevention technical services.

BN. FROM: 01A001 – General Fund BA1201243

PC400051 - Probate Court

Other Expenses \$ 8,650.00

TO: 01A001 – General Fund

PC400051 – Probate Court

Personal Services \$ 8,650.00

Request to transfer appropriation within the Probate Court's budget to cover worker's compensation charges. The Probate Court is supported by the General Fund.

BO. FROM: 01A001 – General Fund BA1201243

LL440008 – Law Library

Other Expenses \$ 160.00

TO: 01A001 – General Fund

LL440008 – Law Library

Personal Services \$ 160.00

Request to transfer appropriation within the Law Library's budget to cover worker's compensation charges. The Law Library is supported by the General Fund.

BP. FROM: 01A001 – General Fund BA1201238

CO380121 - Judicial Administration

Other Expenses \$ 140,000.00

TO: 01A001 – General Fund

PD140053 - Public Defender

Other Expenses \$ 140,000.00

Request to transfer appropriation from the Court of Common Pleas budget to the Public Defender's budget to support space maintenance charge-backs. The Public Defender is supported by the General Fund.

BQ. FROM: 01A001 – General Fund **BA1201275**

DR391052 - Domestic Relations

Other Expenses \$ 50,000.00

TO: 01A001 – General Fund

PD140053 - Public Defender

Other Expenses \$ 50,000.00

Request to transfer appropriation from Domestic Relations Court's budget to the Public Defender's budget to support space maintenance charge-backs. The Public Defender is supported by the General Fund.

BR. FROM: 01A001 – General Fund BA1201108

JC370056 - Juv. Court - Detention Home

Personal Services \$ 154,500.00 Other Expenses \$ 537,700.00

01A001 – General Fund

JC372052 - Juv. Court - Judges

Personal Services \$ 62,900.00

01A001 - General Fund

JC372060 - Juv. Court - Legal

Personal Services \$ 272,300.00

01A001 - General Fund

JC375055 - Juv. Court - Child Support

Personal Services \$ 92,100.00 Other Expenses \$ 20,489.32 Capital Outlay \$ 2,750.00

20A811 - Juv. Court Detention & Probation Services

JC107516 - Juv. Court - Probation Services

Personal Services \$ 171,500.00 Capital Outlay \$ 9,200.00

20A811 – Juv. Court Detention & Probation Services
JC107524 – Juvenile Court – Detention Services
Other Expenses \$ 36,300.00

TO: 01A001 – General Fund

JC372060 - Juv. Court - Legal

Other Expenses \$ 1,142,739.32

20A811 - Juv. Court Detention & Probation Services

JC107516 - Juv. Court - Probation Services

Other Expenses \$ 198,000.00

20A811 - Juv. Court Detention & Probation Services

JC107524 - Juv. Court - Detention Services

Personal Services \$ 19,000.00

The budget adjustments would adjust within General Fund and within HHS Levy accounts to cover year-end expenditures space maintenance charges.

BS. FROM: 01A001 – General Fund **BA1201223**

FS109652 – Fiscal Operations – Tax Assessment Other Expenses \$ 15,000.00

01A001 - General Fund

IG030411 – Inspector General

Personal Services \$ 235,000.00 Other Expenses \$ 35,465.00

01A001 - General Fund

CT577106 – Risk & Property Management

Other Expenses \$ 239,000.00

TO: 01A001 – General Fund

JC370056 – Juv. Court – Detention Home

Other Expenses \$ 2,215.00

01A001 - General Fund

JC372052 - Juv. Court - Judges

Other Expenses \$ 200,000.00

01A001 - General Fund

JC372060 - Juv. Court - Legal

Other Expenses \$ 320,000.00

01A001 - General Fund

FS109637 - Financial Reporting

Other Expenses \$ 1,500.00

01A001 - General Fund

HR018010 - Human Resources

Other Expenses \$ 750.00

Appropriation transfers are requested to cover the 2012 space maintenance expenses at the new Juvenile Justice Center as well as to cover year-end expenditures. Funding comes from the General Fund.

BT. FROM: 01A001 – General Fund **BA1201913**

EX016006 - County Executive

Capital Outlays \$ 5,000.00

TO: 01A001 – General Fund

EX016006 - County Executive

Other Expenses \$ 5,000.00

A transfer is requested to cover outstanding space maintenance expenses.

BU. FROM: 01A001 – General Fund **BA1201916**

AE210005 – Soldiers' and Sailors' Monument

Capital Outlays \$ 1,000.00

TO: 01A001 – General Fund

AE210005 - Soldiers' and Sailors' Monument

Other Expenses \$ 1,000.00

A transfer is requested to post year end data processing and telephone charges.

BV. FROM: 01A001 – General Fund BA1201932

JC372052 – JC – Judges

Personal Services \$ 12,000.00

01A001 - General Fund

FS109645 – Fiscal Ops – Records & Licenses

Other Expenses \$ 23,000.00

01A001 - General Fund

FS109652 – Fiscal Ops – Contractual Services

Other Expenses \$ 12,000.00

01A001 - General Fund

FS109660 – Treasury Management

Other Expenses \$ 24,000.00

TO: 01A001 – General Fund

JC370056 – JC – Detention Home

Other Expenses \$ 24,000.00

01A001 – General Fund

FS109629 - Office of Budget & Management

Other Expenses \$ 23,000.00

01A001 – General Fund

FS109678 – Office of Procurement & Diversity

Other Expenses \$ 24,000.00

Appropriation transfers are requested to cover year end data processing, phone, and supplies expenses. Funding comes from the General Fund.

BW.	FROM:	01A001 – General Fund JC370056 – JC – Detention	Home		BA1201924
		Personal Services	\$	62,700.00	
		Capital Outlay	\$	5,000.00	
		01A001 – General Fund JC370056 – JC – Judges			
		Capital Outlay	\$	4,400.00	
		01A001 – General Fund JC372060 – JC – Legal			
		Capital Outlay	\$	1,400.00	
		01A001 – General Fund FS109660 – Treasury Mana	agement		
		Other Expenses	\$	63,000.00	
		01A001 – General Fund FS109686 – Fiscal Operatio			
		Personal Services	\$	18,000.00	

01A001 - General Fund

HC019018 - Human Resource Commission

Other Expenses \$ 9,350.00 Capital Outlay \$ 4,000.00

01A001 – General Fund IA018002 – Internal Audit

Other Expenses \$ 35,000.00

01A001 - General Fund

IG030411 – Inspector General

Personal Services \$ 18,000.00

01A001 - General Fund

FS109678 - Office of Procurement & Diversity

Other Expenses \$ 41,000.00

TO: 01A001 – General Fund

JC370056 – JC – Detention Home

Other Expenses \$ 2,500.00

01A001 – General Fund

FS109611 - Fiscal Office Administration

Other Expenses \$ 30,000.00

01A001 - General Fund

FS109637 - Financial Reporting

Other Expenses \$ 89,950.00

01A001 - General Fund

FS109645 - Fiscal Ops - Records & Licenses

Other Expenses \$ 47,000.00

01A001 - General Fund

HR018010 – Human Resources

Other Expenses \$ 92,400.00

Appropriation transfers are requested to cover year end data processing, phone, and supplies expenses. Funding comes from the General Fund.

BX.	FROM:	01A001 – General Fund	BA1201929		
		PR200071 – Prosecutor Child Support Capital Outlays	\$	2,000.00	
	TO:	01A001 – General Fund PR200071 – Prosecutor Child Support Other Expenses	\$	2,000.00	

The appropriations transfer within Prosecutor Office Child Support account for data processing charges. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

BY.	FROM:	01A001 – General Fund CO380204 – Common Pleas-Foreclosure		BA1201930
		Other Expenses	\$	160,000.00
		01A001 – General Fund CO380428 – Common Pleas-Psy. Clinic		
		Other Expenses	\$	115,000.00
	TO:	01A001 – General Fund CO380121 – Common Pleas-Judicial/General		
		Other Expenses	\$	275,000.00

The appropriations transfer within the Common Pleas Court for data processing charges. Funding is from the General Fund covering the period January 1, 2012 through December 31, 2012.

BZ.	FROM:	20A301 – Real Estate Assessment BR420067 – Board of Revision		BA1201935	
		Personal Services	\$	11,945.00	
		Capital Outlay	\$	11,725.00	
	TO:	20A301 – Real Estate Assessment BR420067 – Board of Revision			
		Other Expenses	\$	23,670.00	

A budget adjustment would use surplus appropriation to cover year-end data processing chargebacks. Funding comes from a fee collected on tax duplicates other than estate taxes.

CA.	FROM:		24A601 – Senior and Adult Services SA138503 –Information and Outreach		
		Personal Services	\$	21,295.00	
	TO:	24A601 – Senior and A SA138321 – Administr Other Expenses			
		24A601 – Senior and A SA138354 – Managem Other Expenses		205.00	

The Department of Senior and Adult Services has requested appropriation transfers to realign appropriations within index codes to cover projected controlled and other expenses charges for the remainder of the year. The funding source is the Health and Human Services Levy Fund, the Public Assistance allocations, and fees for direct services provided to clients.

Revised: December 18, 2012 December 26, 2012 January 2, 2013 January 8, 2013

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0017

Sponsored by: County Executive	A Resolution approving the appropriation
FitzGerald/Fiscal Officer/Office of	of funds for Year 2013 based on the
Budget & Management	Statement of Appropriation Status dated
	12/31/2012.

WHEREAS, the County Executive/Fiscal Officer/Office of Budget & Management, recommends the appropriation of funds for Year 2013 based on the attached Statement of Appropriation Status dated 12/31/2012 for: a) free balances for grants and capital projects, b) encumbrances for grants and capital projects, and c) encumbrances for all other funds; and,

WHEREAS, pursuant to R.C. 5705.38, each year the County must appropriate funds for each of its non-trust and agency accounts; and,

WHEREAS, the County appropriates the full amount of a grant or capital project when it is first identified; and,

WHEREAS, the free balance (the unexpended and unencumbered appropriation) is then appropriated each following year until the grant or capital project is completed; and,

WHEREAS, encumbrances for capital projects, grant funds and all other funds must also be appropriated and these encumbrances represent payments in process or contract in place at the end of the year that span more than one (1) calendar year; and,

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. The County Council hereby approves the appropriation of funds for Year 2013 based on the attached Statement of Appropriation Status dated 12/31/2012 for: a) free balances for grants and capital projects, b) encumbrances for grants and capital projects, and c) encumbrances for all other funds.

SECTION 2. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

Resolution was duly adopted.	_, seconded by	_, the foregoing
Yeas:		
Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date
First Reading/Referred to Committee(s) Assigned:	mittee:	
Journal, 2013		





Item Details:

Agency/Dept. Name:

Office of Budget and Agency/Dept.Head Matthew Rubino

Management

Name:

Type of Request:

Request Prepared Millie Jones

Telephone No.

443-7220

by:

SUMMARY OF REQUESTED ACTION:

Office of Budget & Management, recommending the appropriation of funds for the Year 2013 based on Financial Reporting's Statement of Appropriation Status dated 12/31/12:

- a) free balances for grants and capital projects;
- b) encumbrances for grants and capital projects; and
- c) encumbrances for all other funds.

Each year the County must appropriate funds for each of its non-trust and agency accounts per ORC 5705.38. The County appropriates the full amount of a grant or capital project when it is first identified. The free balance (the unexpended and unencumbered appropriation) is then appropriated each following year until the grant or capital project is completed. Encumbrances for capital projects, grant funds, and all other funds must also be appropriated. These encumbrances represent payments in process or contracts in place at the end of the year that span more than one calendar year. Report 11 will be provided once it is available.

PURPOSE/OUTCOMES - PRINCIPAL OWNER(S):

Additional reporting will be provided when available in January 2013.

Explanation for late submittal:

Contract/Agreement Information:

Procurement Method:

Explanation for Increase/Decrease in \$ Amount for current request:

Financial Information:

Funding source: Explanation:

Total Amount Requested:

\$

ATTACHMENTS:

Click to download

No Attachments Available

History

Time

Who

Approval

Clerk of the Board

Novusolutions Copyright 2001-2009

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0018

Sponsored by: County Executive	A Resolution making an award on
FitzGerald/Division of	RQ22383 to MHS, Inc. in the amount not-
Community Initiatives/Office of	to-exceed \$1,148,293.00 for emergency
Homeless Services	shelter services for homeless women and
	families, located at 2227 Payne Avenue,
	Cleveland, for the period 1/1/2013 -
	12/31/2013; authorizing the County
	Executive to execute the contract and all
	other documents consistent with said award
	and this Resolution; and declaring the
	necessity that this Resolution become

WHEREAS, the County Executive FitzGerald/Division of Community Initiatives/Office of Homeless Services has recommended an award on RQ22383 to MHS, Inc. in the amount not-to-exceed \$1,148,293.00 for emergency shelter services for homeless women and families, at Norma Herr Women's Center located at 2227 Payne Avenue, Cleveland, for the period 1/1/2013 - 12/31/2013; and

immediately effective.

WHEREAS, the primary objectives of this project are to: (a) safely shelter homeless women, and through coordinated intake, assess and place homeless families; (b) reduce the shelter length of stay; and (c) assist clients to access permanent housing; and

WHEREAS, this contract is funded 100% from the General Fund, Health & Human Services Levy and the provider will be reimbursed on a monthly basis, and

WHEREAS, it is necessary that this Resolution become immediately effective because the project has begun. The submission of this contract was delayed due to the provider's delay in submitting a FY2013 budget, and the OHS efforts to meet other deadlines and responsibilities.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council hereby makes an award on RQ22383 to MHS, Inc. in the amount not-to-exceed \$1,148,293.00 for emergency shelter services for homeless women and families, at Norma Herr Women's Center located at 2227 Payne Avenue, Cleveland, for the period 1/1/2013 - 12/31/2013.

SECTION 2. That the County Executive is authorized to execute a contract and all documents consistent with this Resolution.

SECTION 3. It is necessary that this Resolution become immediately effective in order that critical services provided by Cuyahoga County can continue, and to continue the usual and daily operation of a County entity. Provided that this Resolution receives the affirmative vote of eight members of Council, this Resolution shall become immediately effective upon the signature of the County Executive.

SECTION 4. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by, the forego	oing Resolution was
Yeas:		
Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date
First Reading/Referred to Committee(s) Assigned:	Committee:	
Journal		





Item Details:

Agency/Dept. Administrator's Agency/Dept.Head Ruth Gillett

Name: Office/Office of Name:

Homeless Services

Type of Request: Contract/Amendment

Request Prepared Ruth Gillett **Telephone No.** 420-6844

by:

SUMMARY OF REQUESTED ACTION:

Submitting a contract with MHS, Inc. in the amount of \$1,148,293.00 for emergency shelter services for women and famimiles located at Norma Herr Women's Center, for the period 1/01/2013 - 12/31/2013.

A. Scope of Work Summary

- 1. Office of Homeless Services requesting approval of a contract with Mental Health Services, Inc. in the amount of \$1,148,293 to manage the Emergency Shelter for Women, the Norma Herr Women's Center, for the period January 1, 2013 through December 31, 2013.
- 2. The primary objectives of the contract are to:
- (a) safely shelter homeless women, and through coordinated intake, assess and place homeless families;
- (b) reduce the shelter length of stay;
- (c) assist clients to access permanent housing.
- 3. N/A

B. Procurement

- 1. The Procurment method was through an RFP, 22383; CON2012-42. The award recommendation identified that the RFP was for a three year period. This contract covers the second year of the three year RFP time frame.
- 2. N/A
- 3. N/A

C. Contract and Project Management

1. Mental Health Services, Inc.

1744 Payne Avenue

Cleveland, OH 44114

Council District # 7

- 2. Susan Neth is the Executive Director of MHS, Inc. MHS is a non-profit organization.
- 3. The Emergency Shelter for Women is located at:

2227 Payne Avenue

Cleveland, OH 44114

Council District #7

D. Project Status and Planning

- 1. The County has supported 24/7, 365 days/year shelter for single women since FY 2002. In FY 2004, the Community Women's Shelter moved from the Bishop Cosgrove Center to its current location at 2227 Payne Avenue. For 8 months in FY 2010, the shelter relocated to 1701 Payne so that the current site could be renovated and improved. The current site reopened in November, 2010 and was re-named the "Norma Herr Women's Center" in honor of one of the former clients. The City of Cleveland also provides funding to Mental Health Services to manage the shelter.
- 2. N/A
- 3. N/A
- 4. The contract start date is 1/01/2013. The contract was delayed in being submitted due to provider delays in submitting a FY 2013 budget, and the OHS efforts to meet other deadlines and responsibilities.

E. Funding

- 1. The contract is funded 100% through the General Fund, Health & Human Services Levy.
- 2. The provider is reimbursed on a monthly basis.
- 3. N/A

PURPOSE/OUTCOMES - PRINCIPAL OWNER(S):

See Scope of Work Summary

Explanation for late submittal:

See Scope of Work D. #4

Contract/Agreement Information:

Procurement Method:

RFP (Request for Proposal)

Explanation for Increase/Decrease in \$ Amount for current request:

Current contract is for a 12 month period vs the 7 month term of the prior contract.

Financial Information:

Funding source: Explanation:

General Fund Health & Human Services Levy

Total Amount Requested:

\$1,148,293

ATTACHMENTS:

Click to download Blueback Contract History Evaluation Contract Contract Budget Business Filing W-9 Workers Comp Auditor's Findings signature Authority Valid Certificate of Liability Insurance

History

Time Who Approval

Office of Procurement & Diversity

Novusolutions Copyright 2001-2009

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0019

Sponsored by: County Executive FitzGerald/Department of Health and Human Services/Division of Community Initiatives/Office of Homeless Services

A Resolution making an award on RQ22383 to Lutheran Metropolitan Ministry in the amount of \$1,654,920.00 for emergency shelter services homeless men, located at 2100 Lakeside Avenue, Cleveland, for the period 1/1/2013 - 12/31/2013; authorizing the County Executive to execute the contract and all other documents consistent with said award and this Resolution; and declaring the necessity that this Resolution become immediately effective.

WHEREAS, the County Executive FitzGerald/Administrator's Office/Office of Homeless Services has recommended an award on RQ22383 to Lutheran Metropolitan Ministry in the amount of \$1,654,920.00 for emergency shelter services for homeless men located at 2100 Lakeside Avenue, Cleveland, for the period 1/1/2013 - 12/31/2013; and

WHEREAS, the primary objectives of this contract are to: (a) provide clean and safe shelter on a daily basis for 365 men; (b) provide dinner and breakfast for the clients; and (c) to implement the Continuum of Care goal of focusing on assisting clients to attain permanent housing as quickly as possible; and

WHEREAS, this contract is funded 100% from the General Fund, Health & Human Services Levy and the provider will be reimbursed on a monthly basis, and

WHEREAS, it is necessary that this Resolution become immediately effective because the project term has begun. The submission of this contract was delayed due to the provider's delay in submitting a FY2013 budget, and OHS efforts to meet other deadlines and responsibilities.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council hereby makes an award on RQ22383 to Lutheran Metropolitan Ministry in the amount of \$1,654,920.00 for emergency shelter services for homeless men located at 2100 Lakeside Avenue, Cleveland, for the period 1/1/2013 - 12/31/2013.

SECTION 2. That the County Executive is authorized to execute a contract and all documents consistent with this Resolution.

SECTION 3. It is necessary that this Resolution become immediately effective in order that critical services provided by Cuyahoga County can continue, and to continue the usual and daily operation of a County entity. Provided that this Resolution receives the affirmative vote of eight members of Council, this Resolution shall become immediately effective upon the signature of the County Executive.

SECTION 4. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by, the forego	oing Resolution was
Yeas:		
Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date
First Reading/Referred to Committee(s) Assigned:	Committee:	
Journal		
20		



Item Details:

Agency/Dept. Administrator's Agency/Dept.Head Ruth Gillett

Name: Office/Office of Name:

Homeless Services

Type of Request: Contract/Amendment

Request Prepared Ruth Gillett **Telephone No.** 420-6844

by:

SUMMARY OF REQUESTED ACTION:

Submitting a contract with Lutheran Metropolitan Ministry in the amount of \$1,654,920 for managing the Emergency Shelter for Men located at 2100 Lakeside Avenue for the period 1/01/2013 - 12/31/2013.

A. Scope of Work Summary

- 1. Office of Homeless Services requesting approval of a contract with Lutheran Metropolitan Ministry (LMM) in the amount of \$1,654,920 to manage the Emergency Shelter for Men located at 2100 Lakeside Avenue.
- 2. The primary objectives of the contract are to:
- (a) provide a clean and safe shelter on a daily basis for 365 men; (b) provide dinner and breakfast for clients, and (c) to implement the Continuum of Care goal of focusing on assisting clients to attain permanent housing as quickly as possible.
- 3. N/A

B. Procurement

- 1. This provider was awarded the contract for these services through an RFP process: RFP 22383, (CON 2012-43). The Award recomendation identified that the RFP term was for a three year period. This contract is the second year of that three year award period.
- 2. N/A
- 3. N/A

C. Contractor and Project Information

1. Lutheran Metropolitan Ministry 1468 W. 25th Street Cleveland, OH 44113

Council District #3

- 2. Carol Fredrich is the ED of LMM. LMM is a non-profit organization.
- 3. The location of the shelter is:

2100 Lakeside Avenue

Cleveland, OH 44114

Council District #7

The property is owned by Cuyahoga County.

D. Project Status and Planning

- 1. Cuyahoga County has supported 24/7, 365 days a year, publicly funded shelter for single men since FY 2000. In FY 2005, the City of Cleveland and Cuyahoga County split the cost of purchasing the building located at 2100 Lakeside, with the title to the building be assumed by Cuyahoga County. The City shares in the annual operating costs of the emergency shelter for single men.
- 2. N/A
- 3. N/A
- 4. The contract start date is 1/01/2013. The contract was delayed in being submitted due to provider delays in submitting the budget for FY 2013 and OHS efforts to meet other deadlines related to funding, grant agreements, and contracts.

E. Funding

- 1. The contract is funded 100% by the General Fund, Health and Human Services Levy.
- 2. The provider is reimbursed on a monthly basis.
- 3. N/A

PURPOSE/OUTCOMES - PRINCIPAL OWNER(S):

See Scope of Work Summary

Explanation for late submittal:

See Scope of Work D. #4

Contract/Agreement Information:

Procurement Method:

RFP (Request for Proposal)

Explanation for Increase/Decrease in \$ Amount for current request:

Increase reflects 12 month budget period vs nine month term.

Financial Information:

Funding source: Explanation: General Fund HHS Levy funds

Total Amount Requested:

\$1,654,920

ATTACHMENTS:

Click to download

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0020

Sponsored by: County Executive
FitzGerald/Department of Health
and Human Services/Division of
Children and Family Services

A Resolution making awards on RQ25318 to various providers in the total amount not-to-exceed \$2,400,000.00 for child and family-focused services for the period 2/1/2013 - 1/31/2015, and authorizing the County Executive to execute the contracts and all other documents consistent with said awards and this Resolution.

WHEREAS, the County Executive/Department of Health and Human Services/Division of Children and Family Services has recommended awards on RQ25318 to various providers in the total amount not-to-exceed \$2,400,000.00 for child and family-focused services for the period 2/1/2013 - 1/31/2015 with the following:

- 1) Applewood Centers, Inc. in the amount not-to-exceed \$240,000.00.
- 2) The Bair Foundation in the amount not-to-exceed \$150,000.00.
- 3) Beech Brook in the amount not-to-exceed \$278,570.00.
- 4) Bellefaire Jewish Children's Bureau in the amount not-to-exceed \$278,570.00.
- 5) Catholic Charities Corporation fka Catholic Charities Services Corporation in the amount not-to-exceed \$278,570.00.
- 6) Centers for Families and Children in the amount not-to-exceed \$278,570.00.
- 7) The Cleveland Christian Home, Incorporated in the amount not-to-exceed \$278,570.00.
- 8) Guidestone in the amount not-to-exceed \$278,570.00.
- 9) Mental Health Services for Homeless Persons, Inc. in the amount not-to-exceed \$188,580.00.
- 10) Specialized Alternatives for Families and Youth of Ohio, Inc. in the amount not-to-exceed \$150,000.00; and,

WHEREAS, child and family focused services provide family preservation services to different types of families to help them address major challenges, stabilize the family, and enhance family functioning; and,

WHEREAS, this project is funded 100% by federal funding.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council hereby makes awards on RQ25318 to various providers in the total amount not-to-exceed \$2,400,000.00 for child and family-focused services for the period 2/1/2013 - 1/31/2015 with the following:

- 1) Applewood Centers, Inc. in the amount not-to-exceed \$240,000.00.
- 2) The Bair Foundation in the amount not-to-exceed \$150,000.00.
- 3) Beech Brook in the amount not-to-exceed \$278,570.00.
- 4) Bellefaire Jewish Children's Bureau in the amount not-to-exceed \$278,570.00.
- 5) Catholic Charities Corporation fka Catholic Charities Services Corporation in the amount not-to-exceed \$278,570.00.
- 6) Centers for Families and Children in the amount not-to-exceed \$278,570.00.
- 7) The Cleveland Christian Home, Incorporated in the amount not-to-exceed \$278,570.00.
- 8) Guidestone in the amount not-to-exceed \$278,570.00.
- 9) Mental Health Services for Homeless Persons, Inc. in the amount not-to-exceed \$188,580.00.
- 10) Specialized Alternatives for Families and Youth of Ohio, Inc. in the amount not-to-exceed \$150,000.00.

SECTION 2. That the County Executive is authorized to execute contracts in connection with said awards and all documents consistent with this Resolution.

SECTION 3. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by, the f	foregoing Resolution was
Yeas:		
Nays:		
	County Council President	Date
	County Executive	 Date

	Clerk of Council	Date
First Reading/Referred to Committee(s) Assigned:	Committee:	
Journal		



11 a-5

Item Details:

Agency/Dept.

Department

of Agency/Dept.Head Patricia Rideout

Name:

Children and Family Name:

Services

Type of Request:

Contract/Amendment

Request Prepared Freda Houchins

Telephone No.

(216) 881-3109

by:

SUMMARY OF REQUESTED ACTION:

Title: DCFS 2013-2015 Child and Family Focused Services Contract RQ#25318

A. Scope of Work Summary 1. The Division of Children and Family Services (DCFS) is requesting approval of ten (10)Child and Family-Focused contracts with various vendors at an amount not to exceed \$2,400,000.00. The time period will be from February 1, 2013 to January 31, 2015.

- 1. Guidestone- \$278,570.00
- 2. Beech Brook- \$278,570.00
- 3. The Cleveland Christian Home, Incorporated- \$278,570.00
- 4. Catholic Charities Services Corporation- \$278,570.00
- 5. Bellefaire Jewish Children's Bureau- \$278,570.00
- 6. Specialized Alternatives for Families and Youth of Ohio- \$150,000.00
- 7. The Bair Foundation- \$150,000.00
- 8. Applewood Centers, Inc.- \$240,000.00
- 9. Centers for Families and Children- \$278,570.00
- 10. Mental Health Services for Homeless Persons, Inc.- \$188,580.00
- 2. The Child and Family Focused Services (CFFS) program provides family preservation services to different types of families--- birth or biological families, resource families (including kinship, foster and adoptive families)--- to help them address major challenges, stabilize the family, and enhance family functioning. 3. N/A **B. Procurement** 1. Services were procured through an RFP (RQ25318) process. 2. The (above procurement method) closed on October 22, 2012.

3. In 2012 DCFS solicited proposals for Child and Family Focused Services (CFFS) in RFP #25318. After carefully evaluation of each vendor's proposal(s), the review panel recommended ten (10) vendors to provide said services: Applewood, The Bair Foundation, Beech Brook, Bellefaire, Catholic Charities (Parmadale), Centers for Families and Children, Cleveland Christian Home, Guidestone, Mental Health Services and Specialized Alternatives for Families and Youth (SAFY). C. Contractor and Project Information 1. The address of vendor and/or contractor is:

See attached - The 2013 Council District List.

- **D. Project Status and Planning** 1. The project reoccurs in two-year intervals. **E. Funding** 1. The project is 100% Federallly funded
- 2. The schedule of payments is by monthly invoice.

PURPOSE/OUTCOMES - PRINCIPAL OWNER(S):

The Child and Family Focused Services (CFFS) program provides family preservation services to different types of families--- birth or biological families, resource families (including kinship, foster and adoptive families)--- to help them address major challenges, stabilize the family, and enhance family functioning.

Explanation for late submittal:

Contract/Agreement Information:

Procurement Method:

RFP (Request for Proposal)

Explanation for Increase/Decrease in \$ Amount for current request:

Financial Information:

Funding source:

Explanation:

Federal

100% Federal Funding

Total Amount Requested:

\$2,400,000.00

ATTACHMENTS:

Click to download

- DepartmentAcknowledgementForm
- 2013-2015 Guidestone SD Contract
- 2013-2015 Guidestone W-9
- 2013-2015 CFFS Guidestone Secretary of State Certificate
- 2013-2015 Guidestone Workers Comp Certificate.exp 022813
- 2013-2015 Guidestone Certificate of Liability.exp 042013
- 2013-2015 Guidestone. Auditor's Findings
- 2013-2015 Guidestone Notarized Authorization Letter
- 2013 15 CFFS Guidestone Contract Cover TAB.
- 2013-2015 CFFS Cleveland Christian Home SD Contract
- 2013-2015 CFFS The Cleveland Christian Home W-9

(To be completed in its entirety by user department for all contract/agreement renewals or amendments.)

Contractor: Applewood Centers. – Child and Family-Focused Services				
Contract/Agreement No.: CE1000868-01	Time Period: 12/01/2010 – 11/30/2012			
Service Description: In-home therapeutic serve to eliminate the recidivism of abuse and neglect parents/caregivers, focusing on preserving the and linkage to community supports.	ct of children. Services are provided to			
Original Contract/Agreement Amount: \$240	0.000.00			
Prior Amendment(s) Amount(s): \$0				
Performance Indicators: Home visits will occur within twenty-four (24) hours for emergencies and three (3) working days for non-emergencies. Goals and objectives established in the case plan will be addressed during the service. Families will be linked with neighborhood (community-based) services or to mental health services. Families will remain intact for a year upon completion of the service. client.				
Actual performance versus performance inchas consistently met the timeframes established. The provider met expectations concerning link	d for emergency and non-emergency cases.			
Rating of Overall Performance of Contractor (Check One):				
 □ Superior □ Above Average √ Average □ Below Average □ Poor 				
Justification of Rating: Provider follows the s Reunification, Parent Teen Conflict and Reso responsive to the service needs of the agency	ource Family Support. Catholic Charities is			

CCDCFS User Department

referred to them.

October 13, 2012 Date

CONTRACT/AGREEMENT EVALUATION FORM (To be completed in its entirety by user department for all contract/agreement renewals or amendments.)

Contractor: Beech Brook – Child & Family-Focused Services	
Contract/Agreement No.: CE1000874-01 Time Period: 12/01/2010-11	/30/2012
Service Description: In-home therapeutic services with concrete interventions; we to eliminate the recidivism of abuse and neglect of children. Services are provided parents/caregivers, focusing on preserving the family unit through education, skill and linkage to community supports.	d to
Original Contract/Agreement Amount: \$400,000	
Prior Amendment(s) Amount(s): N/A	
Performance Indicators: Home visits will occur within twenty-four (24) hours for emergencies and three (3) working days for non-emergencies. Goals and objective established in the case plan will be addressed during the service. Families will be with neighborhood (community-based) services or to mental health services. Families are remain intact for a year upon completion of the service.	es : linked
Actual performance versus performance indicators (include statistics): Beech Brook has serviced one hundred twenty-eight (128) families during the corperiod. The provider has consistently met the timeframes established for emergen non-emergency cases. The provider met expectations concerning linkage to combased services.	cy and
Rating of Overall Performance of Contractor (Check One):	
 Superior Above Average √ Average Below Average Poor 	
Justification of Rating: Beech Brook has provided therapeutic services to on twenty-eight (128) families during the contract period. The service models are Intensive In-Home, Reunification, Parent Teen Conflict and Resource Family Support is responsive to the service needs of the agency and provides quality serfamilies referred to them.	port. Beech
C.C.D.C.F.S User Department Oct. 12, 2012 Date	1 ²⁷ %

s: evaluation

Page 202 of 403

(To be completed in its entirety by user department for all contract/agreement renewals or amendments.)

Contractor: Bellefaire Jewish Children's Bu	ureau - Child and Family-Focused Services
Contract/Agreement No.:CE1000873-01	Time Period: 12/01/2010 - 11/30/2012
Service Description: In-home therapeutic serve to eliminate the recidivism of abuse and neglect parents/caregivers, focusing on preserving the and linkage to community supports.	et of children. Services are provided to
Original Contract/Agreement Amount: \$240	0.000.00
Prior Amendment(s) Amount(s): \$0	
Performance Indicators: Home visits will occurrence and three (3) working days for no established in the case plan will be addressed with neighborhood (community-based) service remain intact for a year upon completion of the	on-emergencies. Goals and objectives during the service. Families will be linked as or to mental health services. Families will
Actual performance versus performance inchas consistently met the timeframes established. The provider met expectations concerning link	d for emergency and non-emergency cases.
Rating of Overall Performance of Contractor	(Check One):
☐ Superior☐ Above Average	
√ Average	
□ Below Average□ Poor	

Justification of Rating: Provider follows the service models as follows: Intensive In-Home, Reunification, Parent Teen Conflict and Resource Family Support. Provider is responsive to the service needs of the agency and provides quality service to the families referred to them.

CCDCFS
User Department

October 13, 2012 Date

<u>CONTRACT/AGREEMENT EVALUATION FORM</u> (To be completed in its entirety by user department for all contract/agreement renewals or amendments.)

Contractor: Catholic Charities Services Corp. Services	(Parmadale) – Child & Family-Focus ed
Contract/Agreement No.: CE1000875-02	Time Period: 12/01/2010-11/30/2012
Service Description: In-home therapeutic servi to eliminate the recidivism of abuse and neglect parents/caregivers, focusing on preserving the fa and linkage to community supports.	of children. Services are provided to
Original Contract/Agreement Amount: \$400,	.000
Prior Amendment(s) Amount(s): N/A	
Performance Indicators: Home visits will occur emergencies and three (3) working days for non established in the case plan will be addressed du with neighborhood (community-based) services remain intact for a year upon completion of the	-emergencies. Goals and objectives ring the service. Families will be linked or to mental health services. Families will
Actual performance versus performance indice Catholic Charities has serviced one hundred sixt period. The provider has consistently met the tin non-emergency cases. The provider met expect based services.	ty-three (163) families during the contract neframes established for emergency and
Rating of Overall Performance of Contractor (Check One):
 □ Superior □ Above Average √ Average □ Below Average □ Poor 	
Justification of Rating: Catholic Charities has pasixty-three (163) families during the contract partnersive In-Home, Reunification, Parent Teer Catholic Charities is responsive to the service service to the families referred to them.	period. The service models are as follows: 1 Conflict and Resource Family Support.
C.C.D.C.F.S User Department	Oct. 12, 2012 Date

CONTRACT/AGREEMENT EVALUATION FORM (To be completed in its entirety by user department for all contract/agreement renewals or amendments.)

	Contractor: Center For Families and Children (name was West Side Ecumenical Ministry) – Child & Family-Focused Services		
	Contract/Agreement No.: CE1000872-01 Time Period: 12/01/2010-11/30/2012		
	Service Description: In-home therapeutic services with concrete interventions; which help to eliminate the recidivism of abuse and neglect of children. Services are provided to parents/caregivers, focusing on preserving the family unit through education, skill building and linkage to community supports.		
Γ	Original Contract/Agreement Amount: \$240,000		
	Prior Amendment(s) Amount(s): Name Chane Only Amendment		
	Performance Indicators: Home visits will occur within twenty-four (24) hours for emergencies and three (3) working days for non-emergencies. Goals and objectives established in the case plan will be addressed during the service. Families will be linked with neighborhood (community-based) services or to mental health services. Families will remain intact for a year upon completion of the service.		
	Actual performance versus performance indicators (include statistics): West Side Ecumenical Ministry has serviced eighty-nine (89) families during the contract period. The provider has consistently met the timeframes established for emergency and non-emergency cases. The provider met expectations concerning linkage to community-based services.		
Ra	ating of Overall Performance of Contractor (Check One):		
	 Superior Above Average √ Average Below Average Poor 		
Justification of Rating: West Side Ecumenical Ministry has provided therapeutic services to eighty-nine (89) families during the contract period. The service models are as follows: Intensive In-Home, Reunification, Parent Teen Conflict and Resource Family Support. WSEM has been responsive to the service needs of the agency by continuously accepting cases and providing quality service to the families referred to them.			
	C.D.C.F.S Oct. 12, 2012 Partment Date		
Us	er Department Date		
s: e	valuation		

(To be completed in its entirety by user department for all contract/agreement renewals or amendments.)

Contractor: The Cleveland Christian Home, Services	, Incorporated – Child and Family-Focused
Contract/Agreement No.:CE1000876-01	Time Period: 12/01/2010 - 11/30/2012
Service Description: In-home therapeutic servito eliminate the recidivism of abuse and neglect parents/caregivers, focusing on preserving the fand linkage to community supports.	of children. Services are provided to
Original Contract/Agreement Amount: \$400	.000.00
Prior Amendment(s) Amount(s): \$0	
Performance Indicators: Home visits will occ emergencies and three (3) working days for non established in the case plan will be addressed do with neighborhood (community-based) services remain intact for a year upon completion of the	n-emergencies. Goals and objectives ouring the service. Families will be linked s or to mental health services. Families will
Actual performance versus performance indi- has consistently met the timeframes established. The provider met expectations concerning linka	for emergency and non-emergency cases.
ating of Overall Performance of Contractor (□ Superior □ Above Average √ Average □ Below Average □ Poor	Check One):

Justification of Rating: Provider follows the service models as follows: Intensive In-Home, Reunification, Parent Teen Conflict and Resource Family Support. Provider is responsive to the service needs of the agency and provides quality service to the families referred to them.

CCDCFS User Department

October 13, 2012 Date

(To be completed in its entirety by user department for all contract/agreement renewals or amendments.)

Contractor: Guidestone – Child and Family-	Focused Services
Contract/Agreement No.:CE1000870-01	Time Period: 12/01/2010 – 11/30/2012
Service Description: In-home therapeutic serv to eliminate the recidivism of abuse and neglec parents/caregivers, focusing on preserving the and linkage to community supports.	t of children. Services are provided to
Original Contract/Agreement Amount: \$240	0.000.00
Prior Amendment(s) Amount(s): Name Cha	nge Amendment
Performance Indicators: Home visits will occemergencies and three (3) working days for not established in the case plan will be addressed dwith neighborhood (community-based) services remain intact for a year upon completion of the	n-emergencies. Goals and objectives uring the service. Families will be linked s or to mental health services. Families will
Actual performance versus performance ind has consistently met the timeframes established. The provider met expectations concerning links	I for emergency and non-emergency cases.
Tating of Overall Performance of Contractor (Superior Above Average √ Average Below Average Poor	(Check One):

Justification of Rating: Provider follows the service models as follows: Intensive In-Home, Reunification, Parent Teen Conflict and Resource Family Support. Provider is responsive to the service needs of the agency and provides quality service to the families referred to them.

CCDCFS User Department

October 13, 2012 Date

(To be completed in its entirety by user department for all contract/agreement renewals or amendments.)

Contractor: Mental Health Services (MHS, Inc.) - Child and Family-Focused Services
Contract/Agreement No.: CE1000871-01 Time Period: 12/01/2010 - 11/30/2012
Service Description: In-home therapeutic services with concrete interventions; which help to eliminate the recidivism of abuse and neglect of children. Services are provided to parents/caregivers, focusing on preserving the family unit through education, skill building and linkage to community supports.
Original Contract/Agreement Amount: \$240.000.00
Prior Amendment(s) Amount(s): NA
Performance Indicators: Home visits will occur within twenty-four (24) hours for emergencies and three (3) working days for non-emergencies. Goals and objectives established in the case plan will be addressed during the service. Families will be linked with neighborhood (community-based) services or to mental health services. Families will remain intact for a year upon completion of the service. client.
Actual performance versus performance indicators (include statistics): The provider has consistently met the timeframes established for emergency and non-emergency cases. The provider met expectations concerning linkage to community-based services.
Rating of Overall Performance of Contractor (Check One): □ Superior □ Above Average √ Average □ Below Average □ Poor

Justification of Rating: Provider follows the service models as follows: Intensive In-Home, Reunification, Parent Teen Conflict and Resource Family Support. Provider is responsive to the service needs of the agency and provides quality service to the families referred to them.

CCDCFSUser Department

October 13, 2012 Date

Department Director Signature of Approval

Date



DEPARTMENT NAME: CI	CHILDREN & FAMILY SERVICES	ERVICES	PROP	OSAL I	PROPOSAL DUE DATE:	* October 22, 2012	
RFP TITLE: CI	CHILD & FAMILY FOCUSED SERVICES	SED SERVICES	RFP #	20	025318	SBE: 0%	
TO BE COMPLETED BY OPD	PD	TO BE COMPLETED BY SBE CONTRA	BY SBE C	ONTRA	CT COMPL	CT COMPLIANCE OFFICER	USER DEPT.
VENDOR NAME & ADDRESS	IG / ETHICS REGISTRATION COMPLETE	SBE SUBCONTRACTOR NAME AMOUNT & PERCENTAGE	SBE PRIME%	TOTAL	COMPLY	COMMENTS & INITIAL S	AWARD
Applewood Centers, Inc. 2525 E. 22 nd St.	VER 10-25-12	5,12					
Cleveland, OH 44115							
The Bair Foundation of Cleveland, Ohio 26401 Emery Road, Suite 107 Warrensville, OH 44128	DER OF			. 11			
	10000						
Beech Brook 3737 Lander Road Cleveland, OH 44124	Wiec or		w /		(S)		
	10-23-12			-7	7		
Bellefaire JCB	VER OK			10	/		
22001 Fairmount Boulevard Shaker Hts., OH 44118	8		1	5			
	10-23-12		U	//			
Catholic Charities Services	VER OC			•			
Corporation/Parmadale 6753 State Road Parma, OH 44134	(3)						
	10-23-12						

Department Director Signature of Approval

Date



DEPARTMENT NAME: CHILDRE	CHILDREN & FAMILY SERVICES	ERVICES	PRO	OSAL E	PROPOSAL DUE DATE:	- October 22, 2012	19
REP TITLE: CHILD &	CHILD & FAMILY FOCUSED SERVICES	SED SERVICES	RFP #	1	RQ25318	SBE: 0%	
TO BE COMPLETED BY OPD		TO BE COMPLETED BY SBE CONTRACT	BY SBE C	ONTRAC	CT COMPI	COMPLIANCE OFFICER	USER DEPT.
	IG/ETHICS REGISTRATION	SBE SUBCONTRACTOR NAME			COMPLY		award
VENDOR NAME & ADDRESS	COMPLETE	AMOUNT & PERCENTAGE	PRIME %	SBE %	N.N.	COMMENTS & INITIALS	Y/N
The Center for Families and Children 4500 Euclid Avenue Cleveland OH 44103	A TON ON						
	10-23-12						
Cleveland Christian Home 3146 Scranton Road Cleveland, OH 44109	NER 10-23-12						
Guidestone 202 East Bagley Road Berea, OH 44017	NEC 10-25-12	BO LOS	**	_			
MHS, Inc. 1744 Payne Avenue Cleveland, OH 44114	1/ER 6K			-			
Northern Ohio Recovery Association, Inc. 3746 Prospect Avenue Cleveland, OH 44115	16 535-12			/ ts			
	100010						

Department Director Signature of Approval

Date

TABULATION OF PROPOSALS RECEIVED

REP TITLE:

CHILD & FAMILY FOCUSED SERVICES

CHILDREN & FAMILY SERVICES

PROPOSAL DUE DATE:

October 22, 2012

RFP 排

RQ25318

SBE:

0%

DEPARTMENT NAME:

TO BE COMPLETED BY OPD		TO BE COMPLETED BY SBE CONTRACT	BY SBE	ONTRA		COMPLIANCE OFFICER	USER DEPT.
VENDOR NAME & ADDRESS	IG / ETHICS REGISTRATION COMPLETE	SBE SUBCONTRACTOR NAME AMOUNT & PERCENTAGE	SBE PRIME %	TOTAL	COMPLY	COMMENTS & INITTIALS	AWARD Y/N
Ohio Mentor Inc.,/dba Institute of Family Centered Services 9800 Rockside Road, Suite 800 Valley View, OH 44125	NER 18-23-12	78.				A	
Pressley Ridge 23701 Miles Road Cleveland, OH 44128	VER (10)		60		1	, , ,	140
SAFY 10100 Elida Road Delphos, OH 45833	JER OK	S.	Con Contraction of the Contracti	7	,		¥
		500			*		

County Council of Cuyahoga County, Ohio

Resolution No. R2012-0240

Sponsored by: County Executive	A Resolution authorizing the County	
FitzGerald/Department of	Executive and Department of Public Works to	
Public Works	release an amount not-to-exceed \$500,000.00	
	from the Medical Mart Contingency Fund to	
	the Cleveland Sports Commission when the	
	2013 Senior Games are held at the Convention	
	Center, and authorizing the County Executive	
	and Department of Public Works to negotiate	
	and execute any necessary contract or other	
	documents for same	

WHEREAS, under the Medical Mart Agreement Regarding Final Project Finance Approval dated December 1, 2010, a County Contingency Fund was established in the amount of \$3,485,435.00 ("Contingency Fund"); and,

WHEREAS, there is currently \$3,385,435.00 remaining in the Contingency Fund; and,

WHEREAS, the County Executive and Department of Public Works have requested the authority to release an amount not to exceed \$500,000.00 from the Contingency Fund to the Cleveland Sports Commission for the Senior Games to be held at the Cleveland Convention Center in Cleveland; and,

WHEREAS, the Contingency Fund is held at U.S. Bank, the Trustee for Medical Mart, and it is not a part of the County's General Fund; and,

WHEREAS, Section 3.10(a)(11) of the Cuyahoga County Contracting and Purchasing Procedures Ordinance provides that the Cuyahoga County Council may review and approve any "contract, purchase, sale, lease, grant made by the County, loan made by the County, or any exemption, procedure, or other action within the process of completing any such contract, purchase, sale, lease, grant, exemption, or loan that does not otherwise require Council approval, but the County Executive, Board of Control, or Contracts and Purchasing Board determines to submit to Council for its consideration, review, and approval; and,

WHEREAS, Section 3.18 of the Cuyahoga County Contracting and Purchasing Procedures Ordinance provides that the "County Executive may refer an item to the County Council for approval that otherwise would require approval from the Board of Control or the Contracts and Purchasing Board"; and,

WHEREAS, the County Executive has determined to submit this intended action to the County Council instead of the Cuyahoga County Board of Control for its review and approval pursuant to Section 3.10(a)(11) and Section 3.18 of the Cuyahoga County Purchasing and Procedures Ordinance; and,

WHEREAS, it is necessary that this Resolution become immediately effective in order that critical services provided by Cuyahoga County can continue and to provide for the usual, daily operation of a County entity.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. For the 2013 Senior Games to be held at the County's Convention Center in Cleveland, the Cuyahoga County Council hereby authorizes the Department of Public Works to release an amount not to exceed \$500,000.00 from the Medical Mart and Convention Center Contingency Fund to the Cleveland Sports Commission to aid with the Senior Games to be held at the Cleveland Convention Center.

SECTION 2. The County Executive is hereby authorized to negotiate and execute any necessary contract or other documents to effectuate the release of the funds under the terms specified in this Resolution.

SECTION 3. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by	, the forego	oing Resolution was
Yeas:			
Nays:			
	County Council	il President	Date
	 County Execut	ive	 Date

Clerk of Council	Date

First Reading/Referred to Committee: <u>November 13, 2012</u> Committee(s) Assigned: <u>Economic Development & Planning</u>

Journal CC009 January 22, 2013

County Council of Cuyahoga County, Ohio

Resolution No. R2013-0004

Sponsored by: County Executive	A Resolution declaring that public		
FitzGerald/Department of Public	convenience and welfare requires		
Works/Division of County Engineer	rehabilitation of East 49 th Street Bridge No.		
	00.54 over abandoned railroad tracks in the		
	City of Cleveland; total estimated		
	construction cost \$1,940,000.00; finding		
	that special assessments will neither be		
	levied nor collected to pay for any part of		
	the County's costs of said improvement;		
	and authorizing the County Executive to		
	enter into an agreement of cooperation with		
	said municipality in connection with said		
	project.		

WHEREAS, the County Executive FitzGerald/Department of Public Works/Division of County Engineer has declared that public convenience and welfare requires the rehabilitation of East 49th Street Bridge No. 00.54 over abandoned railroad tracks in the City of Cleveland; and

WHEREAS, the anticipated construction cost for the rehabilitation is \$1,940,000.00; and

WHEREAS, that special assessments are not to be levied nor collected to pay for any part of the County's costs of this improvement; and

WHEREAS, this projects will be funded 75% with the County's Road and Bridge Fund, and 25% from Issue I funds; and

WHEREAS, the primary goal of this project is to properly maintain the County's infrastructure; and

WHEREAS, the location of the project is East 49th Street Bridge No. 00.54 in the city of Cleveland, Council District 7; and

WHEREAS, the anticipated start date for construction of this project is 2014.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council declares that public convenience and welfare requires the rehabilitation of East 49th Street Bridge No. 00.54 over abandoned railroad tracks in the City of Cleveland.

SECTION 2. That special assessments are not to be levied nor collected to pay any part of the County's cost of this improvement.

SECTION 3. That the County Executive is hereby authorized to enter into and execute an agreement of cooperation with the City of Cleveland where the bridge is located.

SECTION 4. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.			the foregoing Resolution was	
Yeas:				
Nays:				
	County Cou	uncil President	Date	
	County Exe	cutive	Date	
	Clerk of Co	uncil	 Date	

First Reading/Referred to Committee: <u>January 8, 2013</u>

Committee(s) Assigned: Public Works, Procurement & Contracting

Journal CC009 January 22, 2013

Resolution No. R2013-0005

Sponsored by: County
Executive FitzGerald/
Department of Development
and Councilmembers Germana
and Miller

A Resolution approving the issuance by the Illinois Finance Authority of bonds in an aggregate principal amount not-to-exceed \$25,000,000.00, in one or more series, for the purpose of (1) refunding revenue bonds issued by Cuyahoga County, Ohio and (2) financing and refinancing facilities located in Cuyahoga County, Ohio; and declaring the necessity that this Resolution become immediately effective.

WHEREAS, the County of Cuyahoga, Ohio (the "County"), a county and political subdivision duly organized and validly existing under the laws of the State of Ohio and its Charter, is authorized and empowered, by virtue of the laws of the State of Ohio, including, without limitation, Chapter 140, Ohio Revised Code (the "Ohio Act"), to, among other things, (i) issue its revenue bonds for the purpose of paying the "costs of hospital facilities" (as such terms are defined in the Ohio Act), within the boundaries of the County, including refunding obligations previously issued for that purpose and (ii) enact this Resolution; and

WHEREAS, the County, pursuant to the foregoing authority and at the request of the Borrower, issued its \$10,205,000 Weekly Adjustable Rate Health Care Facilities Revenue Bonds, Series 2004E (Franciscan Communities, Inc. – Mount Alverna Project) (the "*Prior Bonds*") to provide Franciscan Communities, Inc., an Indiana nonprofit corporation authorized to do business in the States of Illinois and Ohio (the "*Borrower*"), with a portion of the funds necessary to (i) pay or reimburse the Borrower for the payment of certain costs of acquiring, constructing, renovating, remodeling and equipping the Borrower's senior care community known as Mount Alverna Village located in Parma, Ohio ("Mount Alverna Village"); (ii) pay a portion of the interest on the Prior Bonds; (iii) establish a debt service reserve fund and (iv) pay certain expenses incurred in connection with the issuance of the Prior Bonds, including certain fees of a credit facility provider; and

WHEREAS, the Borrower has requested approval from the Council of the County of Cuyahoga, Ohio (the "Council") for the issuance by the Illinois Finance Authority, a body politic and corporate of the State of Illinois (the "Illinois Authority") of bonds, in one or more series (the "Series 2013 Bonds") pursuant to the Illinois Finance Authority Act, 20 ILCS 3501/801-1, et seq., as amended (the "Illinois Act"), for the purpose of providing the Borrower with the

funds necessary to (i) refund the outstanding principal amount of the Prior Bonds; (ii) pay or reimburse the Borrower for, or refinance certain indebtedness the proceeds of which were used for, the payment of certain costs of acquiring, constructing, renovating, remodeling and equipping a "project" (as such term is defined in the Illinois Act) for Mount Alverna Village; (iii) pay a portion of the interest on the Series 2013 Bonds, if deemed necessary or desirable by the Borrower and the Illinois Authority; (iv) establish a debt service reserve fund with respect to the Series 2013 Bonds; (v) provide working capital, if deemed necessary or desirable by the Borrower and the Illinois Authority; and (vi) pay certain expenses incurred in connection with the issuance of the Series 2013 Bonds and the refunding of the Prior Bonds (collectively, the "Financing Purposes"); and

WHEREAS, a public hearing was held on December 20, 2012 pursuant to Section 147(f) of the Internal Revenue Code of 1986, as amended, for the purpose of informing interested parties of the proposed issuance of the Series 2013 Bonds by the Illinois Authority and to afford those parties an opportunity to express their views on the subject both orally and in writing; and

WHEREAS, this Council will approve the issuance of the Series 2013 Bonds by the Illinois Authority for the Financing Purposes which approval will better provide for the health and welfare of the people of the State of Ohio and the County by enhancing the availability, efficiency and economy of "Hospital Facilities" (as defined in the Ohio Act), and the services rendered thereby and provide efficient operation of Hospital Facilities by facilitating the financing of Hospital Facilities, to be available to or for the service of the general public without discrimination by reason of race, creed, religion, color, national origin or age; and

WHEREAS, it is necessary that this Resolution become immediately effective in order that the Financing Purposes may proceed in a timely manner.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. Approval of the Series 2013 Bond Issuance by the Illinois Authority. This Council hereby approves the issuance of the Series 2013 Bonds by the Illinois Authority in order to provide an amount not exceeding \$25,000,000 for the Financing Purposes located in Cuyahoga County, Ohio based on the proposal of the Borrower. This approval is intended to comply with the provisions of Section 147(f) of the Internal Revenue Code of 1986, as amended.

SECTION 2. Findings and Determinations. This Council finds and determines based upon representations of the Borrower, that: (a) there is a substantial need within the County to better provide for the health and welfare of the people of the State of Ohio and the County by enhancing the availability, efficiency and economy of Hospital Facilities and the services rendered thereby,

and to facilitate the financing and refinancing of those Hospital Facilities, to provide health care and other services to the residents of the County available for the service of the general public without discrimination by reason of race, creed, color or national origin, religion or age and at the lowest practicable cost; (b) the approval of the Illinois Authority to issue the Series 2013 Bonds for the Financing Purposes will promote the public purpose stated in Section 140.02 of the Ohio Revised Code and restated in the preceding clause (a), and the County will be duly benefited thereby and (c) it is necessary to, and the County shall, approve, as provided and authorized herein, the issuance of the Series 2013 Bonds by the Illinois Authority for the purpose of providing the Borrower with the funds necessary to (i) refund the outstanding principal amount of the Prior Bonds; (ii) pay or reimburse the Borrower for, or refinance certain indebtedness the proceeds of which were used for, the payment of certain costs of acquiring, constructing, renovating, remodeling and equipping a "project" (as such term is defined in the Illinois Act) for Mount Alverna Village; (iii) pay a portion of the interest on the Series 2013 Bonds, if deemed necessary or desirable by the Borrower and the Illinois Authority; (iv) establish a debt service reserve fund with respect to the Series 2013 Bonds; (v) provide working capital, if deemed necessary or desirable by the Borrower and the Illinois Authority; and (vi) pay certain expenses incurred in connection with the issuance of the Series 2013 Bonds and the refunding of the Prior Bonds.

SECTION 3. Notice and Public Hearing. A notice was published on December 5, 2012 for a public hearing to provide notice to the public of the issuance of the Series 2013 Bonds and the Financing Purposes which public hearing was held on December 20, 2012 at the Cuyahoga County Department of Development, 1701 East 12th Street, 1st Floor, Cleveland, Ohio 44114 for the purpose of hearing objections and/or public comment on the issuance of the Series 2013 Bonds as required by Section 147(f) of the Internal Revenue Code of 1986, as amended.

SECTION 4. Terms and Provisions of the Series 2013 Bonds. The terms and provisions of the Series 2013 Bonds including, but not limited to, the form, execution and interest rates, will be set forth under a bond resolution adopted by, and a bond trust indenture and loan agreement executed by, the Illinois Authority.

SECTION 5. Sale of the Series 2013 Bonds. The Series 2013 Bonds shall be sold by the Illinois Authority to BB&T Capital Markets and/or Huntington National Bank pursuant to a bond purchase agreement.

SECTION 6. Open Meeting. This Council hereby finds and determines that all formal actions of this Council and its committees concerning and relating to the adoption of this Resolution were adopted in an open meeting of this Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in

compliance with all legal requirements including Section 121.22, Ohio Revised Code.

SECTION 7. Effective Date. This Resolution shall be in full force and effect immediately upon the signature of the County Executive, provided this Resolution receives the affirmative vote of eight members of Council. It is hereby determined to be necessary that this Resolution become immediately effective in order that the Financing Purposes may proceed in a timely manner.

On a motion bywas duly adopted.	, seconded by, the f	oregoing Resolution
Yeas:		
Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date

First Reading/Referred to Committee: <u>January 8, 2013</u>

Committee(s) Assigned: Finance & Budgeting

Additional Sponsorship Requested: <u>January 14, 2013</u>

Journal CC009 January 22, 2013

Resolution No. R2013-0006

Sponsored by: County Executive FitzGerald/Department of Health and Human Services/Division of Community Initiatives/Family and Children First Council

A Resolution making an award on RQ24978 to Educational Service Center of Cuyahoga County in the amount of \$692,930.00 for fiscal agent services for the Families and Schools Together Program in connection with the Ohio Children's Trust Fund FY2013 grant program for the period 7/1/2012 - 9/30/2014; authorizing the County Executive to execute the agreement and all other documents consistent with said award and this Resolution.

WHEREAS, the County Executive/Department of Health and Human Services/Division of Community Initiatives/Family and Children First Council has recommended an award on RQ24978 to Educational Service Center of Cuyahoga County in the amount of \$692,930.00 for fiscal agent services for the Families and Schools Together Program in connection with the Ohio Children's Trust Fund FY2013 grant program for the period 7/1/2012-9/30/2014; and,

WHEREAS, the primary goals of the project are to a) build protective factors of families, b) help reduce the probability of child abuse and neglect in our community, and c) provide parent advocacy services to the county, schools and community based programs related to Family and Children First Council; and,

WHEREAS, the project is funded 85% by the State Fund (Ohio Children's Trust Fund), 10% by the Health and Human Services Fund, and 5% by the Temporary Assistance for Needy Families (TANF) fund.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council hereby makes an award on RQ24978 to Educational Service Center of Cuyahoga County in the amount of \$692,930.00 for fiscal agent services for the Families and Schools Together Program in connection with the Ohio Children's Trust Fund FY2013 grant program for the period 7/1/2012-9/30/2014.

SECTION 2. That the County Executive is authorized to execute an agreement in connection with said award and all documents consistent with this Resolution.

SECTION 3. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by, the f	Foregoing Resolution was
Yeas:		
Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	 Date

First Reading/Referred to Committee: <u>January 8, 2013</u> Committee(s) Assigned: <u>Health, Human Services & Aging</u>

Journal CC009 January 22, 2013

Resolution No. R2013-0007

Sponsored by: County Executive	A Resolution making awards on
FitzGerald on behalf of Juvenile	RQ21514 to various providers in the total
Court	amount not-to-exceed \$2,700,000.00 for
	residential treatment services for the
	Youth and Family Community
	Partnership Program for the period
	2/1/2013 - 1/31/2015, and authorizing the
	County Executive to execute a master
	contract and all other documents required
	in connection with said awards and
	consistent with this Resolution.

WHEREAS, the County Executive/Juvenile Court has recommended awards on RQ21514 to various providers in one master contract in the total amount not-to-exceed \$2,700,000.00 for residential treatment services for the Youth and Family Community Partnership Program for the period 2/1/2013 – 1/31/2015 with the following:

- A) Applewood Centers, Inc.
- B) Beech Brook
- C) Bellefaire Jewish Children's Bureau
- D) Carrington Youth Academy LLC
- E) Catholic Charities Services Corporation
- F) The Cleveland Christian Home Incorporated
- G) New Directions, Inc.
- H) The Village Network; and

WHEREAS, the primary goal of this project is to implement a comprehensive program to provide a wide range of services to the Juvenile Court's delinquent youth population in a series of residential programs. Youth referred to this program will have demonstrated unsuccessful participation and/or completion in other community-based programs; and

WHEREAS, a per unit rate for each service provided will be paid by the Juvenile Court to the vendors for each youth receiving program services; and,

WHEREAS, the total contract amount for each provider will not be predetermined, but will be decided based upon usage of services, so long as the total cost incurred for the usage of services provided under the master contract does not exceed the \$2,700,000.00; and,

WHEREAS, this project is funded 100% from the General Fund of the Health and Human Services Levy Funds.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. The Cuyahoga County Council hereby approves the County Executive's recommendation and awards on RQ21514 to various providers in one master contract in the total amount not-to-exceed \$2,700,000.00 for residential treatment services for the Youth and Family Community Partnership Program for the period 2/1/2013 - 1/31/2015 with the following:

- A) Applewood Centers, Inc.
- B) Beech Brook
- C) Bellefaire Jewish Children's Bureau
- D) Carrington Youth Academy LLC
- E) Catholic Charities Services Corporation
- F) The Cleveland Christian Home Incorporated
- G) New Directions, Inc.
- H) The Village Network.

SECTION 2. The County Executive is hereby authorized to execute the master contract and all other documents consistent with the award and this Resolution.

SECTION 3. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	_, seconded by	_, the foregoing Resolution was
Yeas:		

Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date

First Reading/Referred to Committee: <u>January 8, 2013</u>

Committee(s) Assigned: <u>Justice Affairs</u>

Journal CC009 January 22, 2013

Resolution No. R2013-0008

Sponsored by: County Executive	A Resolution making awards on
FitzGerald on behalf of Juvenile	RQ21516 to various providers in the total
Court	amount not-to-exceed \$2,250,000.00 for
	residential treatment services for the
	period 2/1/2013 - 1/31/2015, and
	authorizing the County Executive to
	execute a master contract and all other
	documents required in connection with
	said awards and consistent with this
	Resolution.

WHEREAS, the County Executive/Juvenile Court has recommended awards on RQ21516 to various providers in one master contract in the total amount not-to-exceed \$2,250,000.00 for residential treatment services for the period 2/1/2013-1/31/2015 with the following:

- A) Applewood Centers, Inc.
- B) Beech Brook
- C) Bellefaire Jewish Children's Bureau
- D) Carrington Youth Academy LLC
- E) Catholic Charities Services Corporation
- F) The Cleveland Christian Home Incorporated
- G) Cornell Abraxas Group, Inc.
- H) Guidestone, fka Berea Children's Home
- I) Keystone Richland Center, LLC dba Foundations for Living
- J) New Directions, Inc.
- K) Safe House Ministries, Inc.
- L) The Glen Mills Schools
- M) The House of Emmanuel, Inc.

N) The Village Network; and,

WHEREAS, the primary goal of this project is to implement a comprehensive program to provide a wide range of services to the Juvenile Court's delinquent youth population in a series of residential programs. Youth referred to this program will have demonstrated unsuccessful participation and/or completion in other community-based programs; and,

WHEREAS, a per unit rate for each service provided will be paid by the Juvenile Court to the vendors for each youth receiving program services; and,

WHEREAS, the total contract amount for each provider will not be predetermined, but will be decided based upon usage of services, so long as the total cost incurred for the usage of services provided under the master contract does not exceed the \$2,250,000.00; and,

WHEREAS, the project is funded 100% by general funds.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council hereby makes awards on RQ21516 to various providers in one master contract in the total amount not-to-exceed \$2,250,000.00 for residential treatment services for the period 2/1/2013-1/31/2015 with the following:

- A) Applewood Centers, Inc.
- B) Beech Brook
- C) Bellefaire Jewish Children's Bureau
- D) Carrington Youth Academy LLC
- E) Catholic Charities Services Corporation
- F) The Cleveland Christian Home Incorporated
- G) Cornell Abraxas Group, Inc.
- H) Guidestone, fka Berea Children's Home
- I) Keystone Richland Center, LLC dba Foundations for Living
- J) New Directions, Inc.
- K) Safe House Ministries, Inc.
- L) The Glen Mills Schools

- M) The House of Emmanuel, Inc.
- N) The Village Network; and,

SECTION 2. That the County Executive is authorized to execute a master contract in connection with said awards and all documents consistent with this Resolution.

SECTION 3. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by, th	e foregoing Resolution was
Yeas:		
Nays:		
	County Council Preside	ent Date
	County Executive	Date
	Clerk of Council	

First Reading/Referred to Committee: <u>January 8, 2013</u> Committee(s) Assigned: <u>Justice Affairs</u>

Committee(s) Assigned. <u>Sustice Artic</u>

Journal CC009 January 22, 2013

Resolution No. R2013-0009

Sponsored by: County Executive FitzGerald/Department of Health and Human Services/Division of Community Initiatives/Office of Early Childhood and Councilmember Conwell

A Resolution authorizing a contract with Starting Point in the amount not-to-exceed \$1,832,782.00 for administration of the Special Needs Child Care Program for the Invest in Children Program for the period 1/1/2013 - 12/31/2013, and authorizing the County Executive to execute the contract and all other documents required consistent with this Resolution.

WHEREAS, the County Executive/Department of Health and Human Services/ Division of Community Initiatives/Office of Early Childhood has submitted a contract for Council's approval with Starting Point in the amount not-to-exceed \$1,832,782.00 for administration of the Special Needs Child Care Program for the Invest in Children Program for the period 1/1/2013-12/31/2013; and,

WHEREAS, the primary goals of the project are the management and implementation of the Universal Pre-Kindergarten (UPK) supportive services components; and,

WHEREAS, the project is funded 100% by the Health and Human Services Levy.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council authorizes a contract with Starting Point in the amount not-to-exceed \$1,832,782.00 for administration of the Special Needs Child Care Program for the Invest in Children Program for the period 1/1/2013-12/31/2013.

SECTION 2. That the County Executive is authorized to execute a contract and all documents consistent with this Resolution.

SECTION 3. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion by duly adopted.	, seconded by	, the forego	oing Resolution wa
Yeas:			
Nays:			
	County Council	l President	Date
	County Executi	ive	Date
	Clerk of Counc	il	Date

First Reading/Referred to Committee: <u>January 8, 2013</u> Committee(s) Assigned: <u>Health, Human Services & Aging</u>

Additional Sponsorship Requested: <u>January 17, 2013</u>

Journal CC009 January 22, 2013

Resolution No. R2012-0260

Sponsored by:	County Executive
FitzGerald/De	partment of Public
Works	

A Resolution making an award on RQ23209 to R.W. Armstrong & Associates, Inc. in the amount of \$4,200,000.00 for general engineering services for various projects in connection with the Master Plan Update and Airport Layout Plan for the Cuyahoga County Airport, and authorizing the County Executive to execute the contract and all other documents required in connection with said award and consistent with this Resolution.

WHEREAS, the County Executive/Department of Public Works has recommended an award on RQ23209 to R.W. Armstrong & Associates, Inc. in the amount of \$4,200,000.00 for general engineering services for various projects in connection with the Master Plan Update and Airport Layout Plan for the Cuyahoga County Airport; and,

WHEREAS, County Council has determined that awarding RQ23209 to R.W. Armstrong & Associates, Inc is in the best interest of the County; and,

WHEREAS, this project is funded 90% by the FAA and 10% by the County.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council hereby makes an award on RQ23209 to R.W. Armstrong & Associates, Inc. in the amount of \$4,200,000.00 for general engineering services for various projects in connection with the Master Plan Update and Airport Layout Plan for the Cuyahoga County Airport.

SECTION 2. That the County Executive is authorized to execute a contract in connection with said award and all documents consistent with this Resolution.

SECTION 3. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly adopted.	, seconded by, the foreg	going Resolution was
Yeas:		
Nays:		
	County Council President	- Date
	County Executive	Date
	Clerk of Council	Date
_	to Committee: <u>December 11, 2012</u> I: Economic Development & Planning	

Committee Report/Second Reading: <u>January 8, 2013</u>

Journal CC009 January 22, 2013

2

Resolution No. R2012-0264

Sponsored by: County Executive
FitzGerald/Department of Public
Works and Councilmember
Germana

A Resolution making awards on RQ24534 to Geis Property Management, LLC for sale of the Ameritrust Complex and entry of a lease for a consolidated County headquarters at the site of the P & H Buildings; authorizing the County Executive to execute all documents in connection with said awards and consistent with this Resolution; authorizing the Director of Public Works to administer said project; and declaring the necessity that this Resolution become immediately effective.

WHEREAS, Cuyahoga County engaged the services of Allegro Realty Advisors ("Allegro") to conduct an audit of the County's real-estate needs and come forth with a set of recommendations for the County; and,

WHEREAS, Allegro's recommendations included a recommendation for the County to consolidate multiple functions of County government into a consolidated headquarters; and,

WHEREAS, Allegro recommended that the County place 13 properties currently owned by the County for sale, including the Ameritrust Complex, which consists of seven parcels commonly known as the Rotunda, the Swetland, the Tower, the P Building, the H Building, the Ameritrust Garage, and the Oppman Garage; and,

WHEREAS, the County engaged the services of CBRE, Inc. ("CBRE"), a commercial real estate services firm, to serve as the real estate portfolio program manager, provide brokerage services, and represent the County in connection with the real estate portfolio program; and,

WHEREAS, the County required CBRE, and CBRE expressly agreed in its Contract with the County, to "[p]erform all services hereunder in a fiduciary, professional and efficient manner in accordance with the highest industry standards and practices applicable to the performance of such services"; and,

WHEREAS, the County also retained the services of URS Corporation to provide professional architectural and engineering services in connection with the real property consolidation project; and,

WHEREAS, CBRE worked with the County to prepare and issue a request for proposals, RFP24534, (the "RFP") for the real property consolidation project in accordance with the County's Contracting and Purchasing Procedures Ordinance; and,

WHEREAS, the RFP contains three initiatives: Initiative I sought proposals for the County Headquarters; Initiative II sought proposals for a storage facility; and Initiative III sought proposals for the disposition and sale of the County properties that Allegro recommended be placed for sale; and,

WHEREAS, Geis Companies LLC proposed to purchase, through its affiliate Geis Property Management, LLC ("Geis") either directly as the party or as a guarantor of a special purpose entity, the Ameritrust Complex for \$27,000,000.00, to lease an administrative headquarters to the County in a new 8-story building to be constructed on the site of the P&H buildings and having a minimum useable square footage of 220,342, and to provide the County with the option to purchase the administrative headquarters in 2040 at the end of the 26 year lease term at the cost of \$1 dollar; and,

WHEREAS, Geis plans to utilize financing through an Ohio port authority to finance the county administration headquarters, which may result in an assignment of the County's lease with Geis to a public entity; and,

WHEREAS, Geis proposed rent in an annual rent payment of \$5,774,220.00 for the first year, which increases annually by two percent, with the first year rent abated by \$4,007,100.00, meaning that the County would only pay \$1,767,120.00 for the first year, provided that it takes occupancy of the Building on July 15, 2014; and,

WHEREAS, because of the port financing and the issuance of bonds, Geis proposed that the annual rent of \$5,774,220.00 be divided into two components, each increasing 2 percent annually: an annual "Base Rent" of \$4,007,100.00 for debt service against which the County does not have the right of setoff and a "Service Rent" of \$1,767,120.00, against which the County has the right of setoff; and,

WHEREAS, the County will also be obligated to pay 50% of the annual ground rent payable under an underlying ground lease as ascertained and fixed pursuant to and in accordance with the arbitrators' decision in that certain lawsuit styled Timothy F. Hagan, et al. v. Cleveland Times Square Holdings at Six Points LLC, et al, Case No. CV-09-681954 filed in the Cuyahoga County Court of Common Pleas, less \$11,000.00 per year, and otherwise on the terms and conditions set forth in the County's lease. The County will not have any right of setoff against such amount; and,

WHEREAS, the County determined that it is more economical for it to directly assume the costs of utilities, tenant improvements above the allowance provided by the landlord, and insurance instead of having the landlord finance these costs and add them to the annual rent obligations;

WHEREAS, Geis committed to deliver the building for the County's immediate use on July 15, 2014, and to have the rent commencement date for the Base Rent to commence on July 15, 2015; and,

WHEREAS, Geis proposed to provide the County with certain rights to expand into the third floor of the Ameritrust Tower, the Rotunda, and three floors of the Swetland Building should the need for such expansion arise in the future; and,

WHEREAS, CBRE and URS analyzed the proposals received in response to the RFP as part of the County's evaluation team, which recommended that the County award Initiative I and the sale of the Ameritrust Complex under Initiative III to Geis; and,

WHEREAS, the proposals submitted by Geis for Initiatives I and III received the highest evaluations by the County's evaluation team; and,

WHEREAS, it is necessary that this Resolution become immediately effective in order that critical services provided by Cuyahoga County can continue and to provide for the usual, daily operation of a County entity.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. That the Cuyahoga County Council hereby awards the sale of the Ameritrust Complex, consisting of the seven parcels commonly known as the Rotunda, the Swetland, the Tower, the P Building, the H Building, the Ameritrust Garage, and the Oppman Garage, pursuant to Initiative III of the RFP, to Geis for a purchase price of \$27,000,000.00.

SECTION 2. That the Cuyahoga County Council awards Initiative I of the RFP to Geis for the County to lease a new county headquarters to be constructed on the site of the P & H buildings in the Ameritrust Complex with the option to purchase at the end of the lease term in the year 2040 for \$1 dollar. The Base Rent commencing on July 15, 2015, shall be \$4,007,100.00, with annual increases of 2%; and the "Service Rent", which shall begin upon the County's taking occupancy of the building, shall be \$1,767,120.00, with annual 2% increases. The County will also be obligated to pay 50% of the annual ground rent payable under an underlying ground lease as ascertained and fixed pursuant to and in accordance with the arbitrators' decision in that certain lawsuit styled *Timothy F. Hagan, et al. v. Cleveland Times Square Holdings at Six Points LLC*, et al, Case No. CV-09-681954 filed in the Cuyahoga County Court of Common Pleas, less \$11,000.00 per year, and otherwise on the terms and conditions set forth in the County's Lease.

SECTION 3. All of the County's obligations under the Lease shall be subject to the County's annual appropriations.

SECTION 4. That the County Executive or his authorized designee is authorized to (a) take all actions, and to execute, acknowledge, deliver and/or file for record (as and where appropriate) (i) all documents and instruments necessary or desirable to facilitate and/or consummate the transactions contemplated hereby, convey and transfer acceptable title to the Ameritrust Complex, and facilitate the issuance of title insurance with respect thereto, including, but not limited to, a purchase and sale agreement and all documents to be executed by the County thereunder, a lease agreement and all documents to be executed by the County thereunder, and all financing-related documents (including but not limited to subordination, non-disturbance and attornment agreements, pledges, and security agreements), (ii) all other and further documents, instruments, certificates, agreements, amendments, assignments, subleases, consents, affidavits, certifications, disbursement authorizations, settlement statements, closing statements, proration statements, escrow agreements, escrow instructions, and notices, and (iii) amendments, modifications and supplements to any of the foregoing, that the County Executive may deem necessary or advisable in connection with the consummation of the transactions contemplated hereby, in all cases containing such terms and conditions as may be approved by the County's Director of Law, (b) agree to such payments, prorations, credits, deposits, holdbacks, escrows and other arrangements from (or relative to) sales proceeds as may be necessary or advisable in connection therewith to facilitate and/or consummate such transactions, and (c) prosecute and/or defend any actions or proceedings that may be necessary or advisable relative to any of the foregoing matters.

SECTION 5. That the Director of Public Works is authorized to administer the project through its milestones and be responsible for any ongoing approval rights under the Lease with Geis.

SECTION 6. That all documents to be executed in connection with this transaction be subject to the Law Director's approval as to legal form and correctness.

SECTION 7. It is necessary that this Resolution become immediately effective in order that critical services provided by Cuyahoga County can continue, and to continue the usual and daily operation of a County entity. Provided that this Resolution receives the affirmative vote of eight members of Council, this Resolution shall become immediately effective upon the signature of the County Executive.

SECTION 8. It is found and determined that all formal actions of this Council relating to the adoption of this Resolution were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byadopted.	, seconded by	, the foregoing Resolution was duly
Yeas:		

Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date
Legislation was accepted on the Flo December 11, 2012	or for its anticipated placement on ago	enda:
First Reading/Referred to Committee Committee(s) Assigned: Committee		
Committee Report/Second Reading/ Committee(s) Assigned: Committee	Referred to Committee: January 8, 2 e of the Whole	013
Additional Sponsorship Requested	on the Floor: January 8, 2013	
Journal CC009 January 22, 2013		

Ordinance No. O2013-0003

Sponsored by: County Executive	An Ordinance providing for modifications to	
FitzGerald/Department of	and adoption of the Cuyahoga County Human	
Human Resources	Resources Personnel Policies and Procedures	
	Manual to be applicable to all County	
	employees, and declaring the necessity that	
	this Ordinance become immediately effective.	

WHEREAS, the County Executive/Department of Human Resources has recommended to amend the Human Resources Personnel Policies and Procedures Manual; and,

WHEREAS, Cuyahoga County Charter Section 9.01 provides that the County's Human Resources Policies and Systems for County employees shall be established by ordinance and shall be administered in such a manner as will eliminate unnecessary expense and duplication of effort, while ensuring that persons will be employed in the public service without discrimination on the basis of race, color, religion, sex, national origin, sexual orientation, disability, age or ancestry; and,

WHEREAS, on April 5, 2011, County Council adopted the Human Resources Personnel Policies and Procedures Manual ("the Policy Manual") for a period not-to-exceed June 30, 2011 (Ordinance No. O2011-0015); and,

WHEREAS, on June 28, 2011, County Council extended the effective period for the Policy Manual for a period not-to-exceed December 31, 2011 (Ordinance No. O2011-0028); and,

WHEREAS, on January 24, 2012, County Council adopted the amended version of the County's Human Resources Personnel Policies and Procedures Manual (Ordinance No. O2011-0061); and,

WHEREAS, the Cuyahoga County Human Resource Commission met on December 19, 2012, and has recommended approval of the proposed revisions to the County's Human Resources Personnel Policies and Procedures Manual; and,

WHEREAS, it is necessary that this Ordinance become immediately effective in order that critical services provided by Cuyahoga County can continue and to provide for the usual, daily operation of a County entity.

NOW, THEREFORE, BE IT ENACTED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

- **SECTION 1.** Adoption of Revised Human Resources Personnel Policies and Procedures Manual: Council hereby adopts the amended version of the County's Human Resources Personnel Policies and Procedures Manual that is attached hereto as Exhibit "A" as effective for all County employees (added language is <u>underlined</u>; deleted language in <u>strikethrough</u>), and shall remain in force and effect and shall be followed by County employees under the authority of the County Council and the County Executive.
- **SECTION 2.** Administrative Rules: The Cuyahoga County Administrative Rules, adopted by the Cuyahoga County Board of Commissioners and last revised 7/23/2009, are hereby rescinded, as superseded by the Personnel Policies & Procedures Manual of the Cuyahoga County Human Resources.
- **SECTION 3.** Administrative Rules: Section 2.02, the following language shall be redacted: "the Cuyahoga County Administrative Rules," and replaced with "the Human Resource Commission and Cuyahoga County Council."
- **SECTION 4.** Ethics Policy: Section 3.01 shall be modified to incorporate bulleted points to highlight keywords for easier searching of the electronic versions of the Policy Manual. Keywords included in the bulleted points are: Secondary Employment Information and Reporting Requirements, Candidacy for Public Office, Employment of Relatives, Nepotism, and Sexual Harassment.
- **SECTION 5.** <u>Background Check:</u> Section 5.05, the word "criminal" shall be redacted from the section. The language relating to the disclosure of convicted crimes shall be omitted and in its place, a list of possible employee background checks the County may perform has been included. The following language shall be redacted: "Conviction of a crime or a history of criminal convictions may prevent an applicant from being offered employment," and shall be replaced with: "A misdemeanor or felony may preclude an individual from employment."
- **SECTION 6.** <u>Performance Evaluation:</u> Section 5.08, the following language shall be redacted: "Each evaluation shall measure the employee's performance for the year immediately preceding the evaluation date," and shall be replaced with "Employee evaluations may occur more than once per year."
- **SECTION 7.** <u>Non-Bargaining Position Audits:</u> Section 5.09 shall be modified to included language that if a position audit results in reclassification to a higher pay range/no change in a classification that the employee will be adjusted "in accordance with the County's current Pay Equity Ordinance."
- **SECTION 8.** <u>Bi-Lingual Pay Supplement:</u> In accordance to Section 5.21, the following language shall be included within this new Section: "The County will pay a supplemental 5% pay increase to an employee's base pay for specifically designated job classifications. Employees must perform the mandatory duties in written or verbal fashion at least 20% of the time. Also, use of Braille or hand sign language would apply."

- **SECTION 9.** <u>Direct Deposit Program:</u> Section 6.12 shall include language from Ordinance No.O2012-0025.
- **SECTION 10.** Smoke & Tobacco Free Workplace: Section 7.02 shall include language that allows for inclusion of all tobacco products, in which this Section applies to "all employees, clients, and visitors or persons otherwise required to be in a County facility, vehicle or those driving their personal vehicles when transporting persons on company authorized business." Section 7.02 shall also include language about the incentive program and assistance to help tobacco users quit.
- **SECTION 11.** Fitness for Duty Drug and Alcohol Testing: Section 7.04 shall be modified to illustrate that the County offers no guaranteed second chance or last chance. Section 7.04 shall also include language in which employees who test positive will be offered enrollment in the Fitness for Duty program on a case-by-case basis.
 - **SECTION 12.** Sustainable Cuyahoga County: Section 7.10 shall be removed.
- **SECTION 13.** Lactation Policy: In compliance with the amended FLSA by the Patient Protection and Affordable Care Act, the new Section 7.10, *Lactation Policy*, shall be added to the County's Human Resources Personnel Policies and Procedures Manual. The language in this section illustrates the objective of the County to provide comprehensive service including lactation time, a private separate space, and workplace support for any employee who is breastfeeding her child up to one year after the child's birth.
- **SECTION 14.** FMLA Coordination with Other Leaves of Absence: Section 8.05 shall include the language: "when the County offers or mandates furlough time, FMLA will only commence after all furlough time has been exhausted."
- **SECTION 15.** <u>Sick Leave:</u> Section 9.03 shall be modified to replace the reference to "Section 2.04" to read "Section 2.03."
- **SECTION 16.** Maternity/Paternity Leave Policy: In compliance with TITLE VII as amended by the Pregnancy Discrimination Act, Section 9.10, *Maternity/Paternity Leave Policy*, shall be added to the County's Human Resources Personnel Policies and Procedures Manual. The language in this section illustrates that employees who do not meet the minimum eligibility requirements for FMLA consideration are eligible for this maternity/paternity leave. According to this Section, employees may receive a maximum of 12 weeks unpaid leave (with ability to use any accrued sick time concurrent with the 12 weeks). Also, included is the language in which employees are eligible to reserve 40 hours of sick time to be utilized upon their return from approved FMLA or maternity/paternity leave.
- **SECTION 17.** Workers' Compensation: Section 12, shall include modified language to reflect the County's new Managed Care Organization as University Hospitals CompCare and to include their contact information. Also, the "Worker's

Compensation Injury Reporting Packet" in Section 12.03, shall be renamed as a "Cuyahoga County Injury Care Packet."

- **SECTION 18.** Criminal Conviction: Section 13.07, the following language shall be redacted: "pursuant to Section 5.05 of this Manual, all applicants for appointment to positions under the County are required to declare whether they have been previously convicted of certain crimes. Any employee that fails to disclose a previous criminal conviction or current arrest for one of the listed crimes on their application is subject to disciplinary action, up to and including removal for falsification of their application." Additional language in Section 13.07 shall be included that if the employees are "arrested and/or charged" they need to report said arrest or change to HR.
- **SECTION 19.** <u>Inappropriate Conduct/Grounds for Discipline:</u> Section 13.08 shall incorporate additional examples of inappropriate conduct and the grounds for discipline.
- **SECTION 20.** <u>Progressive Discipline Program:</u> Section 13.09 shall incorporate new definitions of verbal reprimands and working suspensions.
- **SECTION 21.** <u>Appeals:</u> Sections 13.12 and 14.04 shall incorporate new appeals language on filing at the Cuyahoga County Human Resource Commission, how the appeal shall be filed, claims information and the time frame in which the appeal shall be filed.
- **SECTION 22.** Attendance Control Plan: Section 14.03 shall include "working suspensions" for Stage 2 AWOL infractions, as well as language reflecting AWOL stages which shall start at 10 hours, 20 hours, and 36 hours.
- **SECTION 23.** <u>Driving on County Business/Use of County Fleet Vehicles:</u> Section 16.03 shall include the following language: "Smoking is strictly prohibited within an employee's personal vehicle when transporting other employees and/or the public throughout the course of County business."
- **SECTION 24.** It is necessary that this Ordinance become immediately effective in order that critical services provided by Cuyahoga County can continue, and to continue the usual and daily operation of a County entity. Provided that this Ordinance receives the affirmative vote of eight members of Council, this Ordinance shall become immediately effective upon the signature of the County Executive.
- **SECTION 25.** It is found and determined that all formal actions of this Council relating to the adoption of this Ordinance were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byenacted.	, seconded by, the	e foregoing Ordinance was duly
Yeas:		
Nays:		
	County Council President	Date
	County Executive	Date
	Clerk of Council	Date
First Reading/Referred Committee(s) Assigne		
Journal		

Personnel Policies and Procedures Manual



Cuyahoga County Department of Human Resources

Initial Effective Date: 4/5/11 (Ordinance No. O2011-0015)

1st Revision: 4/5/11(Ordinance No. O2011-0028)

2nd Revision: 6/28/11 (Ordinance No. O2011-0043)

3rd Revision: 1/24/12 (Ordinance No. O2011-0061)

4th Revision: 2/14/12 (Ordinance No.

5th Revision: 2/1-

Cuyahoga County

Department of Human Resources



Policies and Procedures Manual

Last Revised: 1/24/12

Comment [jjs1]: Updated with new sections

Table of Contents

INTRODUCTION	SECTION 1: INTRODUCTION / GENERAL INFORMATION	
DISCLAIMER / RESERVATION OF RIGHTS 1.03	Introduction	1.01
RELATIONSHIP TO COLLECTIVE BARGAINING AGREEMENTS. 1.04 RELATIONSHIP TO DEPARTMENTAL POLICY AND PROCEDURE MANUALS. 1.05 BUILDING CLOSURE POLICY 1.06 EMPLOYEE COMMUNICATIONS WITH THE MEDIA 1.07 ACKNOWLEDGEMENT OF RECEIPT 1.08 SECTION 2: CIVIL SERVICE. 2.01 CLASSIFICATION 2.02 ADMINISTRATIVE RULES 2.02 PRIOR SERVICE 2.03 SECTION 3: CODE OF ETHICS. 3.01 ETHICS POLICY 3.01 SECTION 4: EQUAL EMPLOYMENT OPPORTUNITY 4.01 DIVERSITY STATEMENT 4.01 AFFIRMATIVE ACTION POLICY 4.02 EQUAL EMPLOYMENT OPPORTUNITY POLICY 4.03 WORKPLACE HARASSMENT POLICY 4.04 AUGUAL EMPLOYMENT OPPORTUNITY COMPLAINT PROCEDURE 4.05 ANTI-RETALLATION POLICY 4.06 THE AMERICANS WITH DISABILITIES ACT ("ADA") 4.07 RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY APPLICATION 5.02 EMPLOYMENT ELIGIBLITY VERIFICATION 5.02 APPLICATION		
RELATIONSHIP TO DEPARTMENTAL POLICY AND PROCEDURE MANUALS 1.05 BUILDING CLOSURE POLICY 1.06 EMPLOYEE COMMINICATIONS WITH THE MEDIA 1.07 ACKNOWLEDGEMENT OF RECEIPT 1.08 SECTION 2: CIVIL SERVICE 1.08 CLASSIFICATION 2.01 ADMINISTRATIVE RULES 2.02 PRIOR SERVICE 2.03 SECTION 3: CODE OF ETHICS ETHICS POLICY ETHICS POLICY 3.01 SECTION 4: EQUAL EMPLOYMENT OPPORTUNITY 4.02 DIVERSITY STATEMENT 4.01 AFFIRMATIVE ACTION POLICY 4.03 WORKPIACE HARASSMENT POLICY 4.03 WORKPIACE HARASSMENT POLICY 4.04 EQUAL EMPLOYMENT OPPORTUNITY COMPLAINT PROCEDURE 4.05 ANTI-RETALIATION POLICY 4.06 FOR ANTI-RETALIATION POLICY 4.06 RELIGIOUS ACCOMMODATION 4.07 RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY 5.01 EXAMINATIONS 5.02 EVALUATION 5.03 APPLICATION (ID BADGES 5.04	DISCLAIMER / RESERVATION OF RIGHTS	<u>1.03</u>
BUILDING CLOSURE POLICY	RELATIONSHIP TO COLLECTIVE BARGAINING AGREEMENTS	<u>1.04</u>
EMPLOYEE COMMUNICATIONS WITH THE MEDIA ACKNOWLEDGEMENT OF RECEIPT 1.08	RELATIONSHIP TO DEPARTMENTAL POLICY AND PROCEDURE MANUALS	<u>1.05</u>
ACKNOWLEDGEMENT OF RECEIPT 108 SECTION 2: CIVIL SERVICE 2.01 CLASSIFICATION 2.02 ADMINISTRATIVE RULES 2.03 SECTION 3: CODE OF ETHICS 2.03 ETHICS POLICY 3.01 SECTION 4: EQUAL EMPLOYMENT OPPORTUNITY 4.01 DIVERSITY STATEMENT 4.02 AFFIRMATIVE ACTION POLICY 4.02 EQUAL EMPLOYMENT OPPORTUNITY POLICY 4.03 WORKPLACE HARASSMENT POLICY 4.04 EQUAL EMPLOYMENT OPPORTUNITY COMPLAINT PROCEDURE 4.05 ANTI-RETALIATION POLICY 4.06 THE AMERICANS WITH DISABILITIES ACT ("ADA") 4.07 RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY 5.01 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICATION 5.03 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICATION 5.03 EXAMINATIONS 5.02 NEW HIRE ORIENTATION / ID BADGES 5.05 NEW HIR	BUILDING CLOSURE POLICY	<u>1.06</u>
SECTION 2: CIVIL SERVICE. 2.01 CLASSIFICATION 2.02 ADMINISTRATIVE RULES 2.02 PRIOR SERVICE 2.03 SECTION 3: CODE OF ETHICS		
CLASSIFICATION 2.01 ADMINISTRATIVE RULES 2.02 PRIOR SERVICE 2.03 SECTION 3: CODE OF ETHICS		
ADMINISTRATIVE RULES PRIOR SERVICE 2.03 SECTION 3: CODE OF ETHICS		
PRIOR SERVICE 2.03 SECTION 3: CODE OF ETHICS	CLASSIFICATION	<u>2.01</u>
SECTION 3: CODE OF ETHICS 3.01	ADMINISTRATIVE RULES	2.02
ETHICS POLICY 3.01 SECTION 4: EQUAL EMPLOYMENT OPPORTUNITY	PRIOR SERVICE	<u>2.03</u>
SECTION 4: EQUAL EMPLOYMENT OPPORTUNITY	SECTION 3: CODE OF ETHICS	
DIVERSITY STATEMENT 4.01 AFFIRMATIVE ACTION POLICY 4.02 EQUAL EMPLOYMENT OPPORTUNITY POLICY 4.03 WORKPLACE HARASSMENT POLICY 4.04 EQUAL EMPLOYMENT OPPORTUNITY COMPLAINT PROCEDURE 4.05 ANTI-RETALIATION POLICY 4.06 THE AMERICANS WITH DISABILITIES ACT ("ADA") 4.07 RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY 5.01 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15	ETHICS POLICY	3.01
AFFIRMATIVE ACTION POLICY 4.02 EQUAL EMPLOYMENT OPPORTUNITY POLICY 4.03 WORKPLACE HARASSMENT POLICY 4.04 EQUAL EMPLOYMENT OPPORTUNITY COMPLAINT PROCEDURE 4.05 ANTI-RETALIATION POLICY 4.06 THE AMERICANS WITH DISABILITIES ACT ("ADA") 4.07 RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY 5.01 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15	SECTION 4: EQUAL EMPLOYMENT OPPORTUNITY	
AFFIRMATIVE ACTION POLICY 4.02 EQUAL EMPLOYMENT OPPORTUNITY POLICY 4.03 WORKPLACE HARASSMENT POLICY 4.04 EQUAL EMPLOYMENT OPPORTUNITY COMPLAINT PROCEDURE 4.05 ANTI-RETALIATION POLICY 4.06 THE AMERICANS WITH DISABILITIES ACT ("ADA") 4.07 RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY 5.01 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15	DIVERSITY STATEMENT	4.01
EQUAL EMPLOYMENT OPPORTUNITY POLICY 4.03 WORKPLACE HARASSMENT POLICY 4.04 EQUAL EMPLOYMENT OPPORTUNITY COMPLAINT PROCEDURE 4.05 ANTI-RETALIATION POLICY 4.06 THE AMERICANS WITH DISABILITIES ACT ("ADA") 4.07 RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY 5.01 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
WORKPLACE HARASSMENT POLICY 4.04 EQUAL EMPLOYMENT OPPORTUNITY COMPLAINT PROCEDURE 4.05 ANTI-RETALIATION POLICY 4.06 THE AMERICANS WITH DISABILITIES ACT ("ADA") 4.07 RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY ** APPLICATION 5.01 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTIONAL PROBATIONARY PERIOD 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15	EQUAL EMPLOYMENT OPPORTUNITY POLICY	4.03
EQUAL EMPLOYMENT OPPORTUNITY COMPLAINT PROCEDURE 4.05 ANTI-RETALIATION POLICY 4.06 THE AMERICANS WITH DISABILITIES ACT ("ADA") 4.07 RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY SECTION 5: EMPLOYMENT WITH THE COUNTY APPLICATION 5.01 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
THE AMERICANS WITH DISABILITIES ACT ("ADA") 4.07 RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY	EQUAL EMPLOYMENT OPPORTUNITY COMPLAINT PROCEDURE	
RELIGIOUS ACCOMMODATION 4.08 SECTION 5: EMPLOYMENT WITH THE COUNTY 5.01 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15	ANTI-RETALIATION POLICY	
SECTION 5: EMPLOYMENT WITH THE COUNTY 5.01 APPLICATION 5.02 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15	THE AMERICANS WITH DISABILITIES ACT ("ADA")	<u>4.07</u>
APPLICATION 5.01 EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15	RELIGIOUS ACCOMMODATION	<u>4.08</u>
EXAMINATIONS 5.02 EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15	SECTION 5: EMPLOYMENT WITH THE COUNTY	
EMPLOYMENT ELIGIBILITY VERIFICATION 5.03 APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15	APPLICATION	<u>5.01</u>
APPLICANT DRUG TESTING 5.04 CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
CRIMINAL BACKGROUND CHECK 5.05 NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
NEW HIRE ORIENTATION / ID BADGES 5.06 NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
NEW-HIRE PROBATIONARY PERIOD 5.07 PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
PERFORMANCE EVALUATION 5.08 NON-BARGAINING POSITION AUDITS 5.09 PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
Non-Bargaining Position Audits 5.09 Promotion 5.10 Promotional Probationary Period 5.11 Temporary Working Level (TWL) 5.12 Lateral Class Change 5.13 Demotion 5.14 Reduction 5.15	NEW-HIRE PROBATIONARY PERIOD	<u>5.07</u>
PROMOTION 5.10 PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
PROMOTIONAL PROBATIONARY PERIOD 5.11 TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
TEMPORARY WORKING LEVEL (TWL) 5.12 LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
LATERAL CLASS CHANGE 5.13 DEMOTION 5.14 REDUCTION 5.15		
DEMOTION		
REDUCTION 5.15		

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



REMOVAL - INVOLUNTARY TERMINATION OF EMPLOYMENT	
LAYOFFS	
DISABILITY SEPARATION	
DISABILITY RETIREMENT	
BI-LINGUAL PAY SUPPLEMENT	5.21
SECTION 6: WORKWEEK AND HOURS / PAYROLL	
COMPLIANCE WITH THE FAIR LABOR STANDARDS ACT (FLSA)	6.01
ATTENDANCE RECORDS	
TIMESHEET RECORDKEEPING	
AUTOMATED TIME RECORDKEEPING	
STANDARD WORKWEEK AND HOURS	<u>6.05</u>
FLEXTIME / REDUCED WORK HOURS	6.06
LUNCH / BREAKS	<u>6.07</u>
OVERTIME, COMPENSATORY TIME AND EXCHANGE TIME (ACCRUAL)	<u>6.08</u>
HOLIDAYS	
PAYROLL SYSTEM	
PAYROLL DEDUCTIONS	
DIRECT DEPOSIT PROGRAM	
SAVINGS PROGRAMS	
PAYMENT UPON SEPARATION FROM EMPLOYMENT WITH COUNTY	<u>6.14</u>
SECTION 7: WORKPLACE WELLNESS	
Drug-Free Workplace	
SMOKE AND TOBACCO-FREE WORKPLACE	
EMPLOYEE ASSISTANCE PROGRAM (EAP)	
FITNESS FOR DUTY - DRUG AND ALCOHOL TESTING	
FITNESS FOR DUTY - PHYSICAL AND MENTAL HEALTH	<u>7.05</u>
SAFETY POLICY	
WORKPLACE VIOLENCE	
MEDIATION PROGRAM	
WELLNESS PROGRAMS	
SUSTAINABLE CUYAHOGA COUNTY	
LACTATION POLICY	
SECTION 8: FAMILY MEDICAL LEAVE ACT (FMLA)	
DEFINITIONS	<u>8.01</u>
ELIGIBILITY REQUIREMENTS	
QUALIFYING CONDITIONS	<u>8.03</u>
AMOUNT OF LEAVE	
COORDINATION WITH OTHER LEAVES OF ABSENCE	<u>8.05</u>
FMLA APPLICATION PROCEDURE	<u>8.06</u>
CONTINUATION OF BENEFITS	
INTERMITTENT LEAVE AND REDUCED LEAVE SCHEDULES	<u>8.08</u>
RETURN TO WORK	
SECTION 9: EMPLOYEE LEAVE (GENERAL)	
GENERAL	
VACATION LEAVE	<u>9.02</u>
SICK LEAVE	
COMPENSATORY TIME	
EXCHANGE TIME	
FUNERAL/BEREAVEMENT LEAVE	
COURT LEAVE/HRC HEARINGS/JURY DUTY	<u>9.07</u>
Cuyahoga County	
Department of Human Resources	
Policies and Procedures Manual	Last Revised: 4/5/11

LEAVE DONATION PROGRAM UNPAID LEAVES OF ABSENCE	
MATERNITY/PATERNITY LEAVE.	
SECTION 10: MILITARY LEAVE	
MILITARY LEAVE WITH PAY	<u>10.01</u>
UNPAID MILITARY LEAVE	<u>10.02</u>
SECTION 11: BENEFITS	
HEALTH CARE BENEFITS.	
OPEN ENROLLMENT	
BENEFITS DURING UNPAID LEAVES OF ABSENCE	
CHANGES IN MEDICAL BENEFITS COVERAGE	
CONTINUATION OF COVERAGERETIREMENT – OHIO PUBLIC EMPLOYEE RETIREMENT SYSTEM (OPERS)	<u>11.05</u>
DEATH BENEFIT	<u>11.06</u>
OPTIONAL ADDITIONAL EMPLOYEE BENEFITS	
FLEXIBLE SPENDING ACCOUNTS (FSAS)	
SECTION 12: WORKERS' COMPENSATION	
Workers' Compensation Policy	
EMPLOYEE PROCEDURE	
SUPERVISOR PROCEDURE	
RELATIONSHIP TO LEAVE TIME	
ALTERNATIVE WORK PROGRAM	
WAGE CONTINUATION PROGRAM	<u>12.06</u>
SECTION 13: EMPLOYEE CONDUCT POLICY	
GUIDELINES FOR APPROPRIATE CONDUCT	
EMPLOYEE DRESS/APPEARANCE	<u>13.02</u>
EMPLOYEE RESPONSIBILITY FOR COUNTY PROPERTY	
RECORDING OF CONVERSATIONS	
WORKPLACE SEARCH/PRIVACY POLICY	
PROFESSIONAL LICENSES	
CRIMINAL CONVICTION	
INAPPROPRIATE CONDUCT / GROUNDS FOR DISCIPLINE	
CONDUCT/PERFORMANCE IMPROVEMENT MEASURES	
PRE-DISCIPLINARY CONFERENCE (PDC)	
APPEALS	
SECTION 14: ATTENDANCE CONTROL PLAN	
Purpose	14.01
AWOL (ABSENT WITHOUT OFFICIAL LEAVE)	<u>14.01</u>
ATTENDANCE CONTROL PLAN	<u>14.02</u> 14.03
APPEALS	
SECTION 15: ELECTRONIC EQUIPMENT AND COMMUNICATIONS	
ELECTRONIC EQUIPMENT AND COMMUNICATIONS POLICY	
PRIVACY EXPECTATIONS	
SECURITY AND PROPRIETARY INFORMATION	
PROHIBITED USES OF ELECTRONIC EQUIPMENT AND COMMUNICATIONS	15 04

Cuyahoga County Department of Human Resources



Policies and Procedures Manual

SECTION 16: TRAVEL / DRIVING ON COUNTY BUSINESS	
TRAVELING ON COUNTY BUSINESS	16.0
SAFETY BELT USAGE	
DRIVING ON COUNTY BUSINESS / USE OF COUNTY FLEET VEHICLES	
VEHICLE ACCIDENT REPORTING PROCEDURE	
SECTION 17: EMPLOYEE COMPLAINT PROCEDURES	·
EQUAL EMPLOYMENT OPPORTUNITY COMPLAINTS	
FAIR LABOR STANDARDS ACT (FLSA) COMPLAINTS	<u>17.02</u>
GENERAL COMPLAINTS	
SECTION 18: RECORDS MANAGEMENT	
CONFIDENTIAL INFORMATION	18.0
COMPLIANCE WITH RECORDS RETENTION SCHEDULE	18.02
PERSONNEL RECORDS	
ELECTRONIC MAIL RECORDS	18.04
COMBINATION WITH DEGLIEGE TO INCRECT AND CORV. COLINEY DECORDS	

Cuyahoga County Department of Human Resources



Policies and Procedures Manual

Page 247 of 403

Section 1:

INTRODUCTION / GENERAL INFORMATION

1.01 Introduction

Cuyahoga County ("County") is responsible for providing many vital services to the citizens of Cuyahoga County through its employees. The County expects employees to fulfill their role in providing these services and to perform their job in a professional, fair, honest, and thorough manner. The County expects that employees will strive for excellence in advancing the County's mission, demonstrating integrity and professional standards.

1.02 Administration

All matters relating to the administration of the policies and procedures in this Manual will be under the general supervision of the Director of the Department of Human Resources ("Human Resources").

Questions regarding interpretation and application of this Manual should be directed to <u>Human</u> Resources.

1.03 <u>Disclaimer / Reservation of Rights</u>

The policies and procedures set forth in this Manual supersede all previous written and unwritten County personnel policies. This Manual does not constitute a contract of employment, expressed or implied, and should not be considered as such. Furthermore, this Manual is not a limitation on the County's right to direct its workforce. Unless restricted by law, the County reserves all rights to manage its workforce. The policies and procedures contained in this Manual do not reflect or represent every conceivable factual situation, but those most often encountered. Situations may differ and will be handled as warranted by the circumstances and at the discretion of the County. The policies and procedures in this Manual are intended to be guidelines to employees and managers. The County retains the right to amend, add to or change the policies and procedures in this Manual at any time.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Sections 1.01 & 1.03

1.04 Relationship to Collective Bargaining Agreements

For employees covered by a <u>collective bargaining agreement</u>, the terms and conditions of the <u>collective bargaining agreement</u> supersede this Manual on any subject covered by the <u>collective bargaining agreement</u>. The subjects covered in this Manual do not diminish the County's management rights and should not in any way be considered a waiver of these rights. Employees should contact their union representative or Human Resources should they have any questions regarding conflicts between your <u>collective bargaining agreement</u> and this Manual.

1.05 Relationship to Departmental Policy and Procedure Manuals

The terms and conditions of this Manual supersede any contradictory terms or conditions set forth in any Departmental Policy and Procedure Manual.

Sections 1.04 & 1.05

Cuyahoga County Department of Human Resources



Policies and Procedures Manual

1.06 Building Closure Policy

Timesheet Code: AWD

Closing for the Full Day

In the event that it appears necessary to close County Buildings for a complete day, either the evening before or prior to 6:30 a.m. on the day of the proposed closing, the County Executive or designee will authorize the placing of a recording on the County's main number (216) 443-7000 regarding the closing of County Buildings. The recording will also identify the County's 24-hour security number (216) 443-2141. This number will provide person-to-person solutions in unusual cases of emergency.

In addition, arrangements for dissemination of County building status information have been made with local television and radio stations. These announcements can be heard/found on the following radio and television stations/websites:

mowing radio and television st	ations/websites:	
Television Stations	Radio Stations	Websites
WKYC – TV, Channel 3	WTAM, 1100 AM	www.cuyahogacounty.us
WOIO-TV, Channel 19	WMVX, 106.5 FM	www.wkyc.com
WUAB-TV, Channel 43	WMJI, 105.7 FM	www.wtam.com
WVIZ/PBS, ideastream	WGAR, 99.5 FM	www.wmvx.com
	WAKS, 96.5 FM	www.wmji.com
	WMMS, 100.7 FM	www.wgar.com
	WHLO, 640 AM	www.kisscleveland.com
	WKDD, 98.1 FM	www.wmms.com
	WCRF, 103.3 FM	www.640whlo.com
	WCPN, 90.3 FM	www.wkdd.com
	WCLV, 104.9 FM	www.1350radiofreeohio.com
	Radio Free Ohio, 1350 AM	

The announcement will state "CUYAHOGA COUNTY GOVERNMENT BUILDINGS CLOSED." In the absence of an announcement, employees should assume the County's buildings are open for operation.

Early Closings / Abbreviated Work Day

Employees will be notified by their respective Department's management should it be necessary to close buildings early because of weather or other emergency. Employees will be instructed by management as to the actual closing time of the building and the proper procedure for completing their timesheet for that day.

"Essential Employees"

Any notice of closing will not apply to "essential employees" (i.e., employees necessary to protect the immediate safety and/or security of person or property for which the County has direct responsibility, or the provision of emergency services which cannot be interrupted.) All elected officials, agency heads, and Department Directors will identify their essential employees and any buildings that must remain open and operating during an emergency building closing.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual

Section 1.06

Last Revised: 4/5/11

1.07 Employee Communications with the Media

Any news media request for comment (e.g., on-camera interview, interview by a newspaper) shall be referred immediately to the Director of Communications or applicable Department Director and/or designee for official response.

1.08 Acknowledgement of Receipt

Employees are required to acknowledge receipt of this Manual (either electronically or in writing) and are required to make themselves familiar with its contents. A copy of each employee's acknowledgement will be retained by Human Resources.

Cuyahoga County Department of Human Resources

(4)

Sections 1.07 & 1.08

Last Revised: 4/5/11

Policies and Procedures Manual

Section 2: CIVIL SERVICE

2.01 Classification

Pursuant to the <u>Ohio Revised Code</u> ("Revised Code"), County employees are divided into either unclassified service or classified service.

Unclassified Service

The unclassified service is comprised of the specific positions listed in the Revised Code, which includes, but is not limited to, Department Directors, the clerical and administrative staff of the County Executive, student interns, and temporary employees. All offices and positions in the unclassified service are exempt from civil service examination and have no tenure under the law. Unclassified employees serve at the pleasure of the County.

Classified Service

All other employees are deemed to be classified employees and their employment is subject to the provisions of the Ohio Revised Code, the Ohio Administrative Code, the Cuyahoga County Administrative Rules and this Manual.

2.02 Administrative Rules

The County has adopted the rules applicable to county government contained in the <u>Ohio Administrative Code Chapter 123</u> and <u>O.R.C 124</u> except as they are modified by the <u>Human Resource Commission and Cuyahoga County Administrative RulesCouncil</u>.

2.03 Prior Service

Employees who have prior service with the County or another political subdivision of the State of Ohio may be eligible for a higher vacation accrual rate or credit for unused accumulated sick leave.

Employees must provide Human Resources with a letter from their former employer(s), on their letterhead, with qualifying start and end dates of employment and with verification of any unused accumulated sick leave. Vacation accrual rates and available sick leave balances shall be adjusted and be effective upon receipt of documentation deemed acceptable by the County.

NOTE: See Section 9.02 for more information on the impact of prior service on vacation accrual rates and Section 9.03 for the impact on sick leave balances.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 2.01 & 2.03

Section 3: CODE OF ETHICS

3.01 Ethics Policy

It is the policy of the County to carry out its mission in accordance with the strictest ethical guidelines and to ensure that County members and employees conduct themselves in a manner that fosters public confidence in the integrity of the County, its processes, and its accomplishments. Failure to adhere to the standards of ethical conduct may subject an employee to discipline, up to and including removal, pursuant to Section 13 of this Manual as well as criminal prosecution in certain cases.

<u>Please consult the Ethics Policy found See complete policy</u> at hr.cuyahogacounty.us. Hard copies are available upon request to Human Resources.

<u>Information regarding</u>, but not limited to the following can be found in the Ethics Policy:

- Secondary Employment Information and Reporting Requirements
- Candidacy for Public Office
- Employment of Relatives
- Nepotism
- Sexual Harassment

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 3.01

Section 4: EQUAL EMPLOYMENT OPPORTUNITY

<u>4.01</u> <u>Diversity Statement</u>

The County recognizes the value of diversity and all the benefits of fostering an inclusive work environment. All County employees should strive to create and support a work environment representative of the citizens we serve and reflective of the demographics of Cuyahoga County.

Diversity represents those human qualities that are different from our own and outside the groups to which we personally belong. Diversity incorporates the primary characteristics of age, ethnicity, gender, physical abilities and challenges, race and sexual orientation. Diversity dimensions also include educational background, geographic location, parental status, military experience, religious beliefs, and social, economic and political affiliation.

<u>4.02</u> <u>Affirmative Action Policy</u>

Through affirmative action, the County seeks to enhance its equal employment opportunity goal and achieve equitable and sufficient representation of protected class members who have traditionally been underrepresented at all levels of employment and specifically where under-utilization exists.

The County shall make good faith efforts to recruit, train, hire, and promote members of these groups at sufficient levels. This includes preventing discrimination in hiring and promoting, providing access to varieties of jobs at all levels of pay and enhancing opportunities for these groups.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 4.01 & 4.02

4.03 Equal Employment Opportunity Policy

The County is committed to maintaining a professional work environment in which all individuals are treated with respect and dignity. Each employee has the right to work in an atmosphere which promotes equal opportunities and prohibits discriminatory practices, including harassment. It is the policy of the County to assure equal employment opportunities to all persons. The County strictly prohibits discrimination against any person in the recruitment, training, examination, appointment, retention, discipline or any other aspect of personnel administration based on any of the following characteristics ("protected characteristics"):

- Age
- Ancestry
- Disability
- Genetic Information
- Military Status
- National Origin

- Race
- Religion
- Sex / Gender
- Sexual Orientation
- Veteran Status

The County will not condone nor tolerate acts of discrimination and/or harassment, including sexual harassment, by any supervisor or employee under any circumstances. Appropriate corrective action will be taken if any employee is in violation of this policy.

Employees who feel that they have been discriminated against in violation of this Policy should immediately report their complaint in accordance with the County Equal Employment Opportunity Complaint Procedure (Section 4.05).

Cuyahoga County Department of Human Resources



Section 4.03

4.04 Workplace Harassment Policy

In furtherance of its Equal Employment Opportunity Policy, the County strictly prohibits any form of "workplace harassment". "Workplace harassment" is defined as:

- Conduct, whether verbal, non-verbal or physical;
- That is based on a person's protected characteristic (see above); and
- Creates an intimidating, hostile or offensive work environment that unreasonably interferes with work or negatively affects an individual's employment opportunities.

Workplace harassment can occur between co-workers, an employee and a supervisor/manager, or an employee and a non-employee who conducts business with the County.

Sexual Harassment

The County's prohibition against "workplace harassment" includes a zero tolerance policy prohibiting sexual harassment. Sexual harassment is defined as unwelcome sexual advances, requests for sexual favors, and other verbal, non-verbal, or physical conduct of a sexual nature that takes place under either of the following conditions:

- Submission to or rejection of this conduct by an individual is used explicitly or implicitly
 as a factor in decisions affecting hiring, evaluation, promotion, pay increases, termination
 or other aspects of employment; or
- This conduct substantially interferes with an individual's employment or creates an
 intimidating, hostile, or offensive work environment.

Examples of sexual harassment include, but are not limited to:

- Unwanted sexual advances:
- Demands for sexual favors in exchange for favorable treatment or continued employment;
- Repeated sexual jokes, flirtations, advances or propositions;
- Verbal abuse of a sexual nature graphic, verbal commentary about an individuals' body, sexual prowess or sexual nature;
- Whistling or leering;
- Touching, pinching or assault;
- Coerced sexual acts;
- Suggestive insulting, obscene comments or gestures; and
- Display in the workplace of sexually suggestive objects, pictures or written material.

All forms of workplace harassment are unacceptable in the workplace itself and by any employee, in any setting involving business and outside the workplace, including but not limited to, other work-related settings such as business trips, holiday parties, office picnics and County approved events which employees attend.

Employees who feel that they have been subjected to harassment in violation of this Policy should immediately report their complaint in accordance with the County Equal Employment Opportunity Complaint Procedure (Section 4.05).

Cuyahoga County
Department of Human Resources



Section 4.04

Policies and Procedures Manual

4.05 Equal Employment Opportunity Complaint Procedure

Employees who feel that they have been subject to discrimination or harassment in violation of the policies set forth in this Manual should immediately report their complaint in accordance with the procedure in this section. An employee or applicant wishing to file a discrimination, harassment, and/or retaliation complaint should contact Human Resources at (216) 443-7190 or TTY (216) 443-7002.

Employees may also contact their immediate supervisor or their Department Director if they so choose. The supervisor or Director will then refer the matter to Human Resources.

If, after initial review of the complaint, it is determined that the complaint involves equal employment opportunity issues, Human Resources will initiate a thorough investigation into the complaint to determine whether there has been a violation of the Equal Employment Opportunity Policy, Workplace Harassment Policy, and/or Anti-Retaliation Policy. The investigation by Human Resources may include:

- Interviews of the complainant, the charged party and any other relevant witnesses;
- Requests for written statements by the complainant, the charged party and any other relevant witnesses; and
- Review of relevant documentation and personnel files.

Efforts will be made to complete the investigation in a prompt manner. The length of the investigation will vary based on the circumstances surrounding the investigation.

Information obtained in the course of investigations will remain confidential to the extent required by law. Information will be kept confidential unless disclosure is required to further the investigation or unless the County is required to release such information as a result of judicial, administrative or grievance proceedings or under the Ohio Public Records Law.

After obtaining and reviewing all necessary information, Human Resources will issue a determination as to whether there is probable cause to believe that the charged party has violated any of the County's equal employment opportunity policies. The complainant and the charged party will be informed in writing of the determination.

If Human Resources finds that there is probable cause, Human Resources, in consultation with Department management, will determine the appropriate corrective action to remedy the violation. Corrective action for violations of County Equal Employment Opportunity, Workplace Harassment, and Retaliation Policies, includes, but is not limited to:

- Discipline, up to and including removal, of the charged party pursuant to the County Discipline Policy (Section 13);
- Mediation referral (Section 7.08);
- Training (Section 13.10); and/or
- Administrative transfers of employees to separate certain individuals.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 4.05

4.06 Anti-Retaliation Policy

The County strictly prohibits retaliatory action against an individual who opposes discrimination and/or harassment.

"Opposing discrimination and/or harassment" may include, but is not necessarily limited to:

- Filing a charge of discrimination or harassment;
- Cooperating with an internal or external investigation of alleged discrimination or harassment;
- Complaining to anyone about alleged discrimination against oneself or others;
- Threatening to file a charge of discrimination or harassment;
- Refusing to obey an order reasonably believed to be discriminatory;
- Picketing in opposition to discrimination; or
- Requesting a reasonable accommodation based on religion or disability.

"Retaliatory action" may include, but is not necessarily limited to:

- Employment actions, such as termination, refusal to hire and denial of promotions;
- Other actions affecting employment such as threats, unjustified negative evaluations, unjustified negative references or increased surveillance; or
- Any other action such as an assault or unfounded civil or criminal charges that is likely to deter reasonable people from pursuing their rights.

Any action that is perceived to be retaliatory against an individual should be immediately reported in accordance with the County Equal Employment Opportunity Complaint Procedure (Section 4.05).

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 4.06

4.07 The Americans With Disabilities Act ("ADA")

The County, in accordance with the Rehabilitation Act of 1973, Section 504, and the Americans with Disabilities Act of 1990 ("ADA"), seeks to assure its employees that no individual will be discriminated against in a county workplace due to a disability. Under the ADA, the term disability means: "(a) A physical or mental impairment that substantially limits one or more major life activities of an individual; (b) a record of such an impairment; or (c) being regarded as having such an impairment." Simply having a disability does not ensure ADA coverage.

An "ADA Reasonable Accommodation" may be provided to qualified employees. What constitutes a reasonable accommodation will vary from case to case depending on the needs of the position and of the qualified employee. There is no guarantee that a reasonable accommodation will exist for every situation.

The County seeks to ensure that individuals with a disability are hired and retained, and have reasonable access to County owned and leased facilities, as well as facilities occupied by providers under contract with the County to provide services.

ADA Reasonable Accommodation Request Procedure

The County's ADA Reasonable Accommodation Request Procedure serves to verify the existence of a disability and the need for an accommodation in testing, program accessibility, or employment.

To request an ADA accommodation, the employee is required to contact Human Resources (216-443-7190) to request a copy of the required forms. The forms need to be returned to Human Resources, who will then contact the medical provider for verification of disability information provided by the employee.

Upon receipt of the required medical information, Human Resources will engage the employee in an interactive process to:

- Ascertain the individual's precise job-related limitations and how they can be overcome with reasonable accommodation; and
- To identify potential accommodations and assess the effectiveness of each. Human Resources, with input from the employee, department management and other necessary professionals, will determine whether or not a reasonable accommodation may be fashioned, and whenever possible, will recommend an appropriate accommodation. The appropriate accommodation need not be the one desired by the employee.

Human Resources will forward a copy of their recommendation(s) to the Department Director. Once the Director approves the recommendation, Human Resources will then inform the employee. The Employment Relations Division will monitor the situation to ensure that the agreed upon accommodation is achieved and maintained as long as required.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 4.07

4.08 Religious Accommodation

In accordance with federal and state laws and County policy, the County prohibits discrimination on the basis of religion. The County provides reasonable accommodations for sincerely held religious beliefs and/or practices unless doing so would impose an undue hardship on the County. A reasonable religious accommodation is an adjustment to the work environment that will allow an employee to comply with his or her religious beliefs. An employee seeking a reasonable religious accommodation should initially approach their immediate supervisor to discuss their request. If the employee is not satisfied with their supervisor's response, the employee can forward their request to Human Resources for review. Human Resources will respond to the employee's request within a reasonable time.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 4.08

Section 5: EMPLOYMENT WITH THE COUNTY

5.01 Application

All persons applying for original appointment to a position under the County shall complete and file with Human Resources an "Employment Application Form".

No applicant shall be required to disclose their religious or political affiliation or racial or ethnic origin, except as necessary to gather equal employment opportunity or other statistics that, when collected, will not identify any specific individual.

5.02 Examinations

Appointments and promotions for positions in the classified service are made according to merit and fitness, which are evaluated through competitive examinations and/or other appropriate mechanisms as determined by the Director of Human Resources. An examination may include an evaluation of such factors as education, training, capacity, knowledge, manual dexterity, and physical or psychological fitness. An examination may involve structured interviews, assessment centers, work simulations, examination of knowledge, skills and abilities, and any other acceptable testing methods.

5.03 Employment Eligibility Verification

In accordance with the Immigration Reform and Control Act of 1986, the County is committed to employ only United States citizens and aliens lawfully authorized to work in the United States.

Employees hired after June 1, 1987, are required to complete Section 1 of Form I-9 Employment Eligibility Verification and present documentation within three working days to Human Resources to verify identity and employment eligibility.

Human Resources shall examine the documentation and, if satisfied, shall complete Section 2 of Form I-9 to record information. The completed Form I-9 shall be retained for three years or for one year past the employment of the individual, whichever is longer.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Sections 5.01 & 5.03

5.04 Applicant Drug Testing

All final applicants tentatively offered a position under the County will be required to submit to urinalysis to test for illegal drug use prior to final appointment. An applicant with a positive test may not be offered employment for a period of one (1) year from the date of the positive test. Applicants shall have an opportunity to submit medical documentation of legally prescribed medications which may explain a positive test result.

NOTE: See Section 7.04 for more information on County drug/alcohol testing policies and procedures.

5.05 Criminal Background Check

All outside applicants for employment may be required to submit to fingerprinting for a eriminal background check. A eriminal background check may also be required for current employees who are applying for certain types of positions under the County (e.g., positions working with children or seniors). The County in its discretion may also periodically conduct eriminal background checks of current employees.

In addition to the performance of a criminal background check, all applicants for employment are required to disclose whether they have been convicted of any of the following crimes: Felonies (or under arrest for crime punishable as a felony);

Crimes involving dishonesty, fraud, or theft (e.g., forgery, burglary, robbery, eredit card fraud, perjury, bribery, tax evasion);

Crimes of moral turpitude (e.g., sex offenses, pandering, prostitution, importuning, public indecency)

The following types of employee background checks will be performed:

- Prior employment verification
- Personal and professional references
- Educational verification
- Criminal history
- Motor vehicle (if applicable)

NOTE: Current employees are required to disclose if they are arrested <u>and/-or charged</u> for certain crimes. *See* <u>Section 13.07</u>.

A misdemeanor or felony may preclude an individual from employment. Conviction of a crime or a history of criminal convictions may prevent an applicant from being offered employment. Determination of such action will be based on an analysis of the responsibilities of the position and the nature and time frame of the conviction. Current employees may also be subject to

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Sections 5.04 & 5.05

discipline, up to and including removal, pursuant to <u>Section 13</u> of this Manual, if a background check reveals a criminal conviction.

5.06 New Hire Orientation / ID Badges

All newly appointed employees are required to attend a general County employee orientation. Newly appointed employees may also be required to attend additional employee orientations as required by their respective department.

New employees will receive an identification badge from the County at no cost. Employees are required to wear their identification badges at all times while on County property or County business. It is the responsibility of the employee to contact their supervisor immediately if an identification badge is lost, stolen or damaged. Employees are required to request a new badge within five (5) working days. Employees will be charged a replacement fee to obtain a new badge. Non-compliance with this Section could subject the employee to disciplinary action under Section 13 of this Manual.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 5.06

5.07 New-Hire Probationary Period

All newly appointed employees to a non-bargaining, full or part-time classified position shall be required to successfully complete a probationary period of no less than one hundred eighty (180) calendar days, with its effective date beginning the date of appointment. No appointment is final until the employee has satisfactorily served the probationary period. A probationary employee may be removed or reduced at any time during the probationary period when, in the sole judgment of the County, the employee's fitness and/or quality of work are not such as to merit continuation in the position. A probationary employee has no right to appeal the removal or reduction under the Revised Code.

NOTE: The length of a probationary period for a part-time employee shall be determined in accordance with <u>Ohio Administrative Code Section 123:1-19-04</u>.

A longer probationary period, not in excess of one (1) year, may be established for specific job classifications. The Director of Human Resources may extend an employee's probationary period to allow additional time to review the employee's performance. No extension may be granted, however, if the extension would cause the total probationary period to exceed one (1) year.

Probationary employees will be evaluated with respect to performance efficiency twice during their probationary period. The first performance evaluation will be completed within thirty (30) days of the conclusion of the first half of the probationary period. The second evaluation will be completed within thirty days of completion of the probationary period, unless the employee is given a probationary removal or reduction, in which case the final evaluation will be made at the time of the removal or reduction. The final probationary evaluation shall state whether the employee is to be retained, removed or reduced.

The following time shall <u>not</u> be counted as part of the probationary period:

- Days spent on any unpaid leave of absence; or
- Days spent on a paid leave of absence for more than five (5) consecutive working days.

The employee's probationary period shall be extended by a number of days equal to the number of days that are not counted as part of the probationary period. For example, an employee who misses 10 consecutive working days on paid sick leave will have their probationary period automatically extended for 10 working days.

NOTE: Probationary periods and probationary evaluation requirements for bargaining unit employees are set forth in each applicable <u>collective bargaining agreement</u>.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 5.07

5.08 Performance Evaluation

Performance evaluations are an opportunity for management to assess work performance of employees and to provide them with instructive feedback. Classified employees will be evaluated with respect to performance efficiency on a schedule determined by the Director of Human resources. Each evaluation shall measure the employee's performance for the year immediately preceding the evaluation date. Employee evaluations may occur more than once per year.

Each employee will be evaluated by their immediate supervisor. Upon completion of their evaluation, the evaluator shall prepare a <u>Performance Evaluation Form</u> and review it with the employee. The employee shall sign the evaluation to indicate that he or she has received a copy of the completed form. The employee's signature on the form does not indicate agreement with its contents; it merely acknowledges receipt of the form. Refusal of the employee to sign the form shall constitute waiver of the employee's appeal rights outlined below.

Employees who disagree with their performance evaluation are entitled to prepare a rebuttal statement that will be attached to their evaluation prior to placement in the employee's personnel file.

NOTE: Bargaining unit employees should refer to their <u>collective bargaining agreement</u> for information on performance evaluations.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 5.08

Non-Bargaining Position Audits

Non-probationary, non-bargaining, classified employees who believe that their duties have changed significantly and feel that their classification is no longer appropriate may request a position audit.

An employee wishing to initiate a position audit shall submit their request in writing to the Director of Human Resources. Upon receipt of the employee's request, Human Resources shall forward a Comprehensive Position Questionnaire (CPQ) to the employee for completion. The CPQ will be accompanied by a cover letter and instructions for completing the CPQ. The cover letter will also be sent to the employee's Department Director.

The employee shall return the completed CPQ forms to Human Resources within thirty (30) days of receipt. The employee may request in writing to the Director of Human Resources a **onetime** extension of time in which to complete the CPQ. The deadline, however, may not be extended beyond thirty (30) days after the original due date.

Human Resources will review all submitted information and will make a determination as to the appropriate classification. To meet the classification criteria, an employee must perform the mandatory duties stated in the classification function at least 20% of the time. Prior to rendering a decision on the position audit request, Human Resources reserves the right to conduct an onsite audit. Upon completion of their review, Human Resources will send results of the position audit to the employee by certified mail to the address listed on the employee's Request for Position Audit Form with a copy to the Department Director. The employee has the right to appeal the decision rendered by Human Resources. This must be done in writing to the Cuyahoga County Human Resources Commission (HRC) within thirty (30) days of receipt of the letter of notification from Human Resources.

NOTE: Employees may lose their appeal rights if their own negligence caused them to fail to receive the position audit decision from Human Resources (i.e., failure to list correct mailing address on Request for Position Audit Form or failure to sign for the certified letter).

If the position audit results in the employee being reassigned to a classification in a higher pay range, the employee's rate of pay shall be adjusted in accordance with the County's current Pay Equity Ordinance. to either the minimum of the new pay range or to the lowest step in the range which will reflect an increase of at least 5%, whichever is the greater increase.

If the position audit results in the employee being reassigned to a classification in a lower pay range, the employee's rate of pay may be reduced.

If the position audit results in no change in the employee's classification, but a change in the pay range assignment of the classification, the employee will be adjusted in accordance with the County's current Pay Equity Ordinance. to the same step in the new pay range.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 5.09 (1)

If the position audit results in the employee being reassigned to a classification in a higher pay range, any salary adjustment will be computed retroactive to the beginning of the first pay period following the date that the written request for the position audit was received by Human Resources. If the position audit results in the creation of a new classification and/or pay range, any salary adjustment will be effective the first pay period following approval of the new classification and/or pay range by the County (i.e., not retroactive).

If, after conducting a position audit on an employee, Human Resources becomes aware of similarly situated employees, it will reassign the other employees, effective the beginning of the first pay period following the date that the decision was rendered on behalf of the employee that had requested the audit. Similarly situated employees will not be entitled to retroactive pay increases.

An employee may withdraw the request for a position audit at any time prior to a decision being rendered. An employee may not request a position audit more than once in a twelve-month (12) rolling period unless documentation acceptable to the Director of Human Resources is provided at the time of the second request that the job has substantially changed since the date of the completion of the previous audit.

An employee who has received a classification change pursuant to this Section is not required to serve a new probationary period.

NOTE: Bargaining unit employees should refer to their <u>collective bargaining agreement</u> for information on position audits.

Cuyahoga County Department of Human Resources



Section 5.09 (2)

Policies and Procedures Manual

5.10 Promotion

Promotions for positions in the classified service shall be made according to merit and fitness. Competitive examinations or other appropriate mechanisms to determine merit and fitness shall be utilized unless deemed impracticable by the Director of Human Resources.

Whenever a current employee is selected to fill a position which is in a higher pay range, the employee will be adjusted to the minimum of the new pay range, or be placed on a step in the higher pay range that reflects at least a five percent (5%) increase in his or her salary, whichever is greater. At no time will an employee be eligible for promotion while serving a probationary period.

Comment [E2]: This language needs to be reviewed

5.11 Promotional Probationary Period

All newly promoted classified employees shall be required to successfully complete a probationary period of one hundred eighty (180) days. The promotional probationary period follows the same procedures as the New Employee Probationary Period (Section 5.07) with regard to extension of the period, applicability of time spent on unpaid leave of absence and evaluation. No promotion is final until the employee has served the probationary period.

If an employee accepts a promotion and is found to be unsatisfactory in the advanced position, the employee shall be returned to the position from which the employee was promoted or to a similar position. Upon such return, the employee's salary shall be the same that the employee was receiving prior to promotion, except for changes in pay range that may have occurred or any step increase to which the employee would have been entitled in the lower classification.

Any employee failing a promotional probationary period shall have the right to appeal through the Employee Complaint Procedure (Section 17.03) within five days following issuance of notice of failure of probationary period. A probationary employee, however, has no right to appeal the removal or reduction under the Revised Code.

5.12 <u>Temporary Working Level (TWL)</u>

A TWL pay adjustment may be granted when an employee is temporarily assigned a significant amount of the duties of a position with a higher pay range for a minimum of a two (2) week period, but not to exceed one (1) year.

The employee must meet the minimum requirements for the position in the higher pay range to be granted a TWL pay adjustment. An employee that is granted a TWL pay adjustment will be placed on a step that reflects at least a 5% increase in his or her salary, or the lowest step in the higher pay range, whichever is greater.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 5.10 & 5.12

5.13 Lateral Class Change

A lateral classification change involves the movement of an employee, with the employee's consent, from one classification to another classification that is assigned to the same pay range or to a pay range in which the step one rate is the same as the step one rate as the classification from which the employee moves. An employee who accepts a lateral classification change will not have their pay adjusted and will not be subject to a probationary period.

NOTE: The filing of a Request for Position Audit (<u>Section 5.10</u>) constitutes the employee's consent to a lateral classification change should that be the ultimate decision at the end of the position audit process.

5.14 Demotion

A demotion is a reduction in rank or position to a classification which carries a lower salary range than that previously held. A demotion shall only be made for one or more of the offenses set forth in <u>Section 13.08</u> of this Manual, except when the employee voluntarily agrees to a demotion in writing.

An employee who is demoted will be placed on a step in the lower pay range which reflects at least a 5% decrease in salary or be adjusted to the maximum of the pay range of the new position, whichever is the greater reduction in salary.

Comment [E3]: This language needs to be reviewed

5.15 Reduction

"Reduction" means a change of the classification held by an employee to one having a lower base pay range, a change to lower step within a salary range with an accompanying diminution in compensation, or any decrease in compensation for an employee. A reduction shall only be made for one or more of the offenses set forth in the <u>Section 13.08</u> of this Manual, except when the employee voluntarily agrees to a reduction in writing.

Cuyahoga County Department of Human Resources



Sections 5.13 & 5.15

Last Revised: 4/5/11

Policies and Procedures Manual

5.16 Resignation - Voluntary Termination of Employment

An employee may resign his or her employment with the County by submitting sufficient notice to the Department or Human Resources. Upon receipt of an employee's notice of Resignation, the Director of Human Resources will confirm acceptance of the employee's resignation to the employee in writing. A resignation notice may not be rescinded by an employee after acceptance by the Director, unless, in their sole discretion, the County accepts the employee's request to rescind.

Employees are requested, where possible, to provide fourteen (14) days advance written notice of their intended resignation.

5.17 Removal – Involuntary Termination of Employment

Removal is an involuntary termination of County employment. A removal shall only be made for one or more of the offenses set forth in <u>Section 13.08</u> of this Manual. Employees who are removed may appeal the removal to the HRC or in accordance with the terms of their <u>collective bargaining agreement</u> (*See* <u>Section 13.12</u> of this Manual for more information regarding an employee's right to appeal a removal action.)

5.18 Layoffs

County employees in the classified service may be laid off whenever a reduction in force is necessary due one or more of the following reasons:

- · Lack of funds;
- Lack of work; or
- The abolishment of positions as a result of reorganization for the efficient operation of the County, for reasons of economy, or for lack of work.

Whenever it becomes necessary for the County to reduce its workforce, the County shall lay off employees or abolish their positions in accordance with the Revised Code and the administrative rules promulgated thereunder. Employees should refer to the applicable Ohio Revised Code and Ohio Administrative Code sections for detailed information regarding retention points, order of layoff, displacement rights and reinstatement rights.

NOTE: Bargaining employees should refer to their respective <u>collective bargaining agreement</u> for information regarding to layoffs.

Cuyahoga County
Department of Human Resources



Sections 5.16 & 5.18

Policies and Procedures Manual

Last Revised: 4/5/11

Page 270 of 403

Comment [E4]: Need to review this section

5.19 Disability Separation

An employee who is unable to perform the essential job duties of his or her position due to a disabling illness, injury or condition may be disability separated. A disability separation may be voluntary or involuntary.

Eligibility

An employee is eligible for disability separation if both of the following conditions are met:

- 1. The employee is unable to perform the essential job duties of his or her position due to a disabling illness injury or condition; and
- 2. The employee has exhausted all of his or her paid sick leave and applicable unpaid leave.

When the employee does not dispute his or her inability to perform the essential job duties of his or her position due to a disabling illness, injury or condition, the disability separation process is considered "**voluntary**" and the following conditions apply:

- The County may grant the employee's request for voluntary disability separation or may require the employee to submit to a medical or psychological fitness for duty examination with a physician chosen by the County. If the examination supports the employee's request, the County shall grant the employee's request. If the medical examination does not support the employee's request, the County will not approve the request.
- An employee who is granted a voluntary disability separation waives his or her right to a pre-separation hearing and to any appeal of the decision to approve his or her request.
- An employee who is granted a voluntary disability separation shall retain the right to be reinstated to his or her position for two (2) years from the date that the employee is no longer in active work status due to a disabling illness, injury or condition. An employee may submit a written request for reinstatement from a voluntary disability in accordance with the procedure established below.

When there is a dispute between the County and the employee as to the employee's ability to perform the essential job duties of his or her position, the disability separation process is considered "**involuntary**" and the following conditions apply:

- The County must have substantial credible medical evidence of an employee's disabling illness, injury or condition (e.g., a fitness for duty examination finding the employee unfit for duty).
- The County must determine that the employee is unable to perform any of his or her essential job duties.
- The County will institute a hearing prior to involuntarily disability separating an employee. The employee shall be provided written notice at least seventy two (72) hours in advance of the hearing. If the employee does not waive his or her right to the hearing in writing, the hearing shall go forward and the employee has the right to examine the County's evidence of disability, to rebut that evidence, and to present testimony and evidence on his or her own behalf.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 5.19 (1)

- If the County determines, after weighing the testimony presented and evidence admitted at the pre-separation hearing, that the employee is unable to perform his or her essential job duties, then the County shall issue an involuntary disability separation order.
- An involuntarily disability separated employee shall have the right to appeal in writing to HRC within ten (10) days following the date the order is served.
- An employee shall have the right to request reinstatement to his or her position for two (2) years from the date of separation.

Reinstatement

An employee on disability separation may make a written request to the County for reinstatement from the separation. An employee may not make a first request for reinstatement until three (3) months from the date the employee was no longer in active work status. The County shall notify the employee of its decision to approve or deny the reinstatement request no later than sixty (60) days after it receives the employee's written request. The employee shall not make subsequent requests for reinstatement more than once every three (3) months from the date the employee is notified of a reinstatement denial.

An employee is not eligible for reinstatement if the request occurs later than two (2) years from the date that the employee was no longer in active work status due to the disabling illness, injury or condition.

NOTE: For purposes of reinstatement under this Section, the date of separation is the date in which the employee was no longer performing active work status due to the disabling illness, injury or condition. If an employee attempts to return to work but fails to perform the essential job duties for six (6) consecutive months, the employee's effective date of separation does not change.

An employee's request for reinstatement shall be accompanied by substantial, credible medical evidence that the employee is once again capable of performing the employee's essential job duties. Upon receipt of this evidence, the County shall either reinstate the employee or require the employee to submit to a medical or psychological fitness for duty examination. If the County determines that the employee is unable to perform his or her essential job duties, the County will institute a pre-reinstatement hearing.

An employee shall be provided written notice at least seventy two (72) hours in advance of the pre-reinstatement hearing. If the employee does not waive the right to the hearing, then at the hearing the employee has a right to examine the County's evidence of continuing disability, to rebut that evidence, and to present testimony and evidence on his or her own behalf.

If the County finds the employee incapable of performing essential duties, the County will not reinstate the employee. The employee will be notified of this decision in writing and shall have the right to appeal in writing to HRC within thirty (30) days of receiving the decision.

Cuyahoga County Department of Human Resources



Section 5.19 (2)

Last Revised: 4/5/11

Policies and Procedures Manual

If the County determines that the employee is to be reinstated, the employee has a right to be assigned to a position in the classification the employee held at the time of disability separation. If the classification the employee held at the time of disability separation no longer exists or is no longer utilized by the County, the employee shall be placed in a similar classification. If no similar classification exists, or if the employee no longer meets the minimum qualifications, the employee may laid off in accordance with the Ohio Revised Code or in accordance with an applicable collective bargaining agreement.

County employees who are unable to perform the essential job duties of his or her position on a permanent basis due to a disabling illness, injury or medical condition, may be eligible for disability retirement through <u>OPERS</u>. Employees must contact <u>OPERS</u> to initiate the disability retirement process. If the employee has been granted disability benefits by <u>OPERS</u>, the reinstatement requirements of this Section may apply for up to five years.

NOTE: If the employee has been granted disability benefits by <u>OPERS</u>, the reinstatement requirements of this Section may apply for up to five years. *See* <u>Section 5.20</u> of this Manual for more details.

Cuyahoga County
Department of Human Resources



Section 5.19 (3)

Policies and Procedures Manual

5.20 Disability Retirement

County employees who are unable to perform the essential job duties of his or her position on a permanent basis due to a disabling illness, injury or medical condition, may be eligible for disability retirement through <u>OPERS</u>. Employees must contact <u>OPERS</u> to initiate the disability retirement process. Contact information for <u>OPERS</u> can be found in <u>Section 11.06</u> of this Manual.

5.21 Bi-Lingual Pay Supplement

Comment [jjs5]: New Section

The County will pay a supplemental 5% pay increase to an employee's base pay for specifically designated job classifications. Employees must perform the mandatory duties in written or verbal fashion at least 20% of the time. Also, use of Braille or hand sign language would apply.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual

(\$

Section 5.20 <u>& 5.21</u>

Section 6:

WORKWEEK AND HOURS / PAYROLL

6.01 Compliance with the Fair Labor Standards Act (FLSA)

The FLSA is a federal law which requires that certain employees be paid at least the federal minimum wage for all hours worked and overtime for all hours worked over forty (40) hours in a workweek. It is County policy to comply with all requirements of the FLSA. In accordance with this policy, all positions with the County are categorized as either "non-exempt" or "exempt" based on the nature of the positions:

- Non-Exempt Employees Non-exempt employees receive compensatory time or overtime for time worked beyond forty (40) hours in a workweek.
- Exempt Employees Exempt employees do not receive compensatory time or overtime for time worked beyond forty (40) hours in a workweek. Subject to the provisions of 6.08 of this Manual, however, exempt employees may be eligible to receive exchange time for time worked beyond (40) hours.

An employee who believes that he or she has been improperly categorized as exempt or his or her paycheck has been improperly docked may file a complaint in accordance with the County FLSA Complaint Process (Section 17.02). If the County determines that the employee has been improperly categorized or docked, the employee will be re-categorized and/or reimbursed.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 6.01

6.02 Attendance Records

Human Resources utilizes two methods for capturing time worked by employees: (1) timesheets – including paper and electronic forms; and (2) automated time recording system. The following attendance records policies and procedures are established pursuant to principles of public accountability. Attendance records are public records.

Each employee is generally required to record or clock his or her time at the moment of arriving to commence work and departing for the workday. Supervisors will arrange for a sign in/out procedure to be in place for all employees who are required to report directly to a worksite away from the location where the normal sign in/out procedure takes place. The employee, however, is solely responsible for accurately completing and signing his or her own timesheet.

Falsification of a timesheet is grounds for discipline, up to and including removal.

Employees who do not report to work on time are in violation of the Attendance Control Policy (Section 14.03). When an employee is tardy without approved leave (i.e., sick/vacation/FMLA), the time taken will be without pay and will be cumulative toward the employee AWOL hours (*see* Section 14.02).

Cuyahoga County
Department of Human Resources



Section 6.02

Policies and Procedures Manual Last Revised: 4/5/11

6.03 Timesheet Recordkeeping

Timesheets (paper or electronic) are to be used to record reasons for absences and additional hours worked which qualify for overtime, compensatory time and/or exchange time. Timesheets will include the following codes for all the various possibilities for absences:

Reason for Absence	Timesheet	Policy Manual
	Code	Section
Abbreviated Work Day	AWD	1.06
Absent w/out Official Leave	AWOL	14.02
Compensatory Time Leave	CT	<u>9.04</u>
Court/Jury Duty/HRC Leave	CL	<u>9.07</u>
Exchange Time Leave	ET	<u>9.05</u>
Extended Unpaid Sick/Medical Leave	LOA	<u>9.09</u>
FMLA Leave	FMLA	<u>8</u>
Holiday	Н	6.09
Holiday w/out Pay	HX	N/A
Leave Donation	SLD	9.08
New-Hire Administrative Leave	ALX	<u>9.09</u>
Paid Administrative Leave	AL	<u>13.09</u>
Paid Military Leave	ML	<u>10.01</u>
Personal Day	PD	N/A
Personal Leave of Absence	LOA	9.09
Sick Leave	S	9.03/9.06
Suspension w/out Pay	SUSX	13.09
Suspension w/Pay	SUS	13.09
Unpaid Military Leave	MLX	10.02
Vacation Leave	V	9.02
Voluntary Work Reduction	VWR	6.06
Worker's Compensation Leave w/out Pay	WCX	<u>12.04</u>

Each employee is responsible for ensuring that the employee has adequate available leave time when requesting/claiming leave time on his or her timesheet. Employees who claim leave time that they do not have will be considered AWOL for the claimed period of time (See Section 14.02).

Employees shall adhere to the following procedure for accurately completing their timesheet:

- Accurately record the time they began work and the time that they ended working for each regular working day;
- Determine the number of hours and minutes worked each day (only if utilizing paper timesheet);
- Convert the time worked or time off to a decimal (only if utilizing paper timesheet);

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual

Section 6.03 (1)

Last Revised: 4/5/11

- If time worked is less than the scheduled workday, note the hours off and indicate the reason for the absence (only if utilizing paper timesheet);
- Add the weekly regular, overtime, compensatory and exchange hours for each week to reflect in the breakdown of hours worked on the timesheet (only if utilizing paper timesheet); and
- Sign or electronically submit the timesheet to indicate their confirmation of the times noted as the arrival and departure for each day.

The sum of work time, authorized leave and time without pay shall not exceed the amount of time an employee would have been regularly scheduled to work in any day, unless overtime/compensatory time/exchange time was required and authorized by the supervisor/department head.

Upon the validation of the timesheet by the employee, the supervisor shall review the timesheet to certify that the employee rendered the time indicated on the timesheet. Supervisors are accountable for their employees' whereabouts at all times. If the supervisor is satisfied that the employee's timesheet accurately reflects the employee's attendance for the week, the supervisor shall sign or electronically submit the timesheet to reflect the authorization of hours to be paid.

Upon the completion and validation of timesheets, the supervisor (or designee) must ensure that signatures by each employee and the supervisor are included. All completed timesheets must be forwarded on a weekly basis to the appropriate payroll administrator for record keeping and processing.

Falsification of a timesheet by an employee or a supervisor is a violation of the County's Discipline Policy (Section 13), which may result in disciplinary action, up to and including removal from County employment.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 6.03 (2)

6.04 Automated Time Recordkeeping

The second method for time recording utilized by the County is by automated time recording mechanism (e.g., Timelink). The automated payroll system is a precise measure of time and attendance which is based upon keypunches and/or card swipes. No grace period exists for employees who utilize the automated time recording mechanism. Therefore, should an employee key or swipe in after their scheduled time to work or before their scheduled shift ends, the system will note the late arrival or early departure and accordingly dock the earnings of the employee.

Employees who utilize the automated time recording mechanism must do so at their assigned worksite. It is the responsibility of each employee to punch in at the start of the shift and out at the conclusion of the shift. Early arrivals and late departures will not be counted as overtime unless previously approved by a supervisor.

Under no circumstances may an employee swipe/key another employee in at the beginning of a shift or out for the end of a shift. Violation of this provision may result in disciplinary action, up to and including removal.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 6.04

<u>6.05</u> <u>Standard Workweek and Hours</u>

The normal workweek for full-time County employees is five (5) days per week, usually Monday through Friday. The normal workday is from 8:30 a.m. to 4:30 p.m., with a paid lunch period in accordance with Section 6.07 of this Manual. The normal workweek and standard hours of work may vary based on operation needs, and/or in cases of 7-day/24-hour-a-day operations or in situations covered by a collective bargaining agreement.

NOTE: A non-exempt employee may not work beyond the standard hours of work so as to exceed the normal forty-hour workweek without prior authorization by the employee's supervisor.

In the case of unexcused absences or absence without approved leave, the time missed will be without pay and will be recorded as AWOL pursuant to <u>Section 14.02</u>.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 6.05

Last Revised: 1/24/12

6.06 Flextime / Reduced Work Hours

Department Directors, in their sole discretion, may authorize employees to participate in flextime. Those authorized to participate in flextime are required to comply with all other policies and procedures, including, but not limited to, the requirements of this Section and the Attendance Control Policy (Section 14). Participation in flextime does not eliminate any of the work rule requirements mandated in this Manual. Department Directors, in their sole discretion, may revoke the offer of flextime at any time.

NOTE: No flextime options permit employees to work before 7:00 a.m. or beyond 6:00 p.m. unless absolutely necessary for County operations.

Flextime Options

OPTION A - Core Hours

- Permits employees to choose, on a day to day basis, a work schedule beginning as early as 7:00 a.m. and as late as 10:00 a.m.
- An employee's workday will end eight (8) hours from the time they arrive.
- All employees are required to be present during the "core hours" of 10:00 a.m. to 3:00 p.m.

OPTION B - Four (4) Day Work Week

- Permits employees to work four (4) ten (10) hour days.
- Directors and Division Managers are excluded from participation in this Option.
- Department heads may select from three different plans for reallocating the number of workdays from five (5) to four (4) days:
 - o Plan 1 Department as a whole operates four days. All employees are scheduled to work the same days (i.e., Monday through Thursday; Tuesday through Friday or Thursday through Sunday).
 - Plan 2 Department operates five (5) days a week with each employee working only four days – days off are staggered among teams of employees.
 - Plan 3 Department operates seven days a week, each employee working four days – schedule alternates as necessary.
- Holiday Scheduling Employees on the four (4) day work week will revert to a five (5) day, eight (8) hour schedule during each week in which a holiday occurs

OPTION C - Selected Daily Work Hours

- Employee may select a different number of work hours each day.
- Total selected hours must equal forty (40) hours each workweek.
- No less than five (5) hours or more than ten (10) hours may be worked in one day.
- This plan may include Saturday as a workday.
- Holiday Scheduling Employees who are on selected daily work hours will revert to a five (5) day, eight (8) hour schedule during each week in which a holiday occurs.

Cuyahoga County
Department of Human Resources



Section 6.06 (1)

Policies and Procedures Manual

Voluntary Work Reduction

Timesheet Code: VWR

Department Directors, in their sole discretion, may authorize employees to participate in a voluntary work reduction program. A voluntary work reduction involves a decrease in an employee's amount of hours worked. Those authorized to participate in a voluntary work reduction program are required to comply with all other policies and procedures, including, but not limited to, the requirements of this Section and the Attendance Control Policy (Section 14). Participation in a voluntary work reduction program does not eliminate any of the work rule requirements mandated in this Manual. The Department Director, in their sole discretion, may revoke the offer of a voluntary work reduction program at any time.

NOTE: Exempt employees who participate in a voluntary work reduction program shall not be eligible to accumulate exchange time.

Cost Savings Days

In their sole discretion, the County may establish mandatory or voluntary cost savings programs in accordance with the authority granted by the Revised Code. A cost savings program involves the taking of unpaid cost savings days ("furlough days") by County employees.

Cuyahoga County Department of Human Resources



Section 6.06 (2)

Policies and Procedures Manual

6.07 Lunch / Breaks

Unless otherwise specified in a <u>collective bargaining agreement</u>, County employees will receive a one-hour paid lunch period. In addition, County employees may receive two paid rest breaks of fifteen (15) minutes in duration. All rest breaks and lunch periods are to be scheduled by the employee's immediate supervisor based on the operational needs of their unit and in accordance with the following provisions:

- One rest break shall be taken in the first half of the work day and one shall be taken in the second half of the work day;
- Rest breaks shall not abut the end or beginning of the lunch period;
- Rest breaks and lunch periods cannot be used to make-up tardiness or quitting early. For example, an employee who is scheduled to end his or her day at 4:30 may not leave for the day at 3:30 p.m. and take his or her lunch from 3:30 to 4:30 p.m.; and
- An employee must return to work after a lunch period for that period to be considered a lunch period. For example, an employee may not take his or her lunch period from 12 p.m. to 1 p.m. and then take sick leave from 1 p.m. until the end of the day. The employee will be required to use his or her own leave time to cover the period from 12 p.m. to 1 p.m. If, however, the employee only used sick leave from 1 p.m. until 2 p.m. and returned to work for the remainder of the day, the 12 p.m. to 1 p.m. period would be considered a proper lunch period.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 6.07

6.08 Overtime, Compensatory Time and Exchange Time (Accrual)

The accrual and use (See Sections 9.04 and 9.05) of all overtime/compensatory time/exchange time must be prior approved by an employee's supervisor.

Employees are not permitted to accrue or use compensatory time or exchange time without prior approval by the employee's supervisor

Compensatory Time / Overtime

In accordance with <u>Section 6.01</u> of this Manual, employees in non-exempt positions who are required to or are granted prior authorization to work beyond forty (40) hours in a work week shall accrue compensatory time at a rate of one and one-half (1½) hours for every hour worked beyond forty (40) hours. It is incumbent upon management (including front-line supervisors) to direct their unit operations and/or employee workload so that non-exempt employees do not perform unscheduled work that results in overtime or compensatory time liability. It is the duty of unit management to exercise control so that work is not performed by non-exempt employees outside of their normal scheduled work hours unless the additional work time has been authorized.

To avoid overtime or compensatory time liability, a supervisor may direct a "temporary schedule change" within the same workweek so that a non-exempt employee's hours in a paid status within the workweek do not exceed forty hours. For example: Employee (a non-exempt employee) has a typical M-F, 8:30 - 4:30 schedule. The supervisor asks Employee to stay 2 hours late on Wednesday to finish a critical project. If Employee works the remainder of the standard work schedule through the end of the week, Employee will have worked 42 hours, and shall be entitled to overtime or compensatory time for the additional 2 hours. To avoid the overtime or compensatory time liability, the supervisor may direct Employee to work two fewer hours on Thursday or Friday of the same week to maintain the employee's work week at the standard forty hours. The extra time worked and the reduced work hours must be within the same week (Sunday to Saturday) in order for a temporary schedule change to be effective to avoid unnecessary overtime or compensatory time liability.

NOTE: A "temporary schedule change" is a separate and distinct concept from a flexible work schedule (Section 6.06). A temporary schedule change must be pre-approved by an employee's supervisor and does not constitute a permanent change to an employee's regular work schedule. Non-exempt employee's who work beyond their regular work schedule without prior authorization may be subject to disciplinary action, up to and including termination of employment. A supervisor shall only approve a "temporary schedule change" as a means to avoid overtime or compensatory time liability in extraordinary situations in which a non-exempt employee is required to work additional hours on a particular day during a workweek.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 6.08 (1)

Supervisors must ensure that the employees for whom they sign timesheets understand and comply with the policy on non-exempt overtime/compensatory time. Employees who do not comply with the policy and proceed to work unauthorized overtime should be counseled, the discussion should be documented, and progressive discipline should be initiated with Human Resources for further violations. Supervisors who do not regulate their employees' work hours may also be subject to disciplinary action.

The maximum amount of compensatory time a non-exempt employee may accrue is two hundred forty (240) hours, unless the employee works in a public safety, emergency response position, in which case the maximum is four hundred eighty (480) hours. Employees who have accrued unused compensatory time up to these limits will be paid overtime on a time and one-half basis for all additional overtime worked.

Accrued compensatory time may be used in accordance with <u>Section 9.04</u> of this Manual. At the sole discretion of the County, non-exempt employees may be paid overtime on a time and one-half basis in lieu of accrual of compensatory time.

Exchange Time

Employees in overtime-exempt positions may receive exchange time on an hour-for-hour basis for time worked beyond forty (40) hours in a work week, provided the following:

- There is a compelling, extraordinary reason for the exempt employee to be required to stay beyond normal working hours staying to finish up normal assignments does not qualify;
- There is a "meeting of the minds" between the exempt employee and the supervisor the employee must either be required to work the overtime or be granted **prior** authorization by their supervisor to work the additional hours; and
- The required or authorized increment of additional time is greater than one-half hour per day of required additional time.

After the initial half-hour period is accrued, exchange time is accrued in increments of oneminute. The maximum balance of exchange time an exempt employee may maintain is forty (40) hours.

Accrued exchange time may be used in accordance with <u>Section 9.05</u> of this Manual.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 6.08 (2)

6.09 Holidays

Timesheet Code: H

All County employees are entitled to eight (8) hours of holiday pay for:

New Year's Day
Martin Luther King Day
President's Day
Memorial Day
Independence Day
Labor Day
Columbus Day
Veterans Day
Thanksgiving Day
Day after Thanksgiving
Christmas Day

In the event that any of the aforementioned holidays falls on Saturday, the Friday immediately preceding shall be observed as the holiday. In the event that any of the aforementioned holidays fall on Sunday, the Monday immediately following shall be observed as the holiday.

NOTE: Bargaining unit employees should refer to their respective <u>collective bargaining</u> <u>agreement</u> for applicable provisions on holidays and holiday pay.

If the employee's work schedule is other than Monday through Friday, he or she is entitled to holiday pay for holidays observed on the employee's day off regardless of the day of the week on which they are observed.

An employee in a non-exempt position who is required to work on a day designated as holiday by this Section shall be entitled to pay for such time worked at overtime rates in addition to his or her regular pay. An employee in an exempt position who is required to work on a day designated as a holiday shall be entitled to exchange time on an hour-for-hour basis for all hours worked in addition to their regular pay.

Temporary employees are not entitled to holiday pay. Part-time/permanent employees are entitled to holiday pay for a portion of the holiday based on the number of hours that they are regularly scheduled to work on the day that the holiday falls.

Employees who are in an unpaid status on the regular work day immediately preceding or following the holiday are not eligible for holiday pay.

Cuyahoga County Department of Human Resources



Section 6.09

Policies and Procedures Manual Last Revised: 4/5/11

6.10 Payroll System

The County's compensation schedule is based on a two-week or eighty-hour (80) pay period, which begins on Sunday morning and ends on Saturday night. Paychecks are distributed on Friday's of pay weeks.

6.11 Payroll Deductions

There are two types of payroll deductions that the County shall be authorized to make from employee wages:

Mandatory

- Federal, state and local withholding taxes
- Ohio Public Employees Retirement System (OPERS)
- Garnishments
- Court ordered child or spousal support
- Medicare portion of Social Security for employees hired after April 1, 1986
- Union dues / Fair Share Fees

Voluntary: Employees may authorize deductions from wages for (these options are subject to change periodically at management's discretion):

- Benefit selections
- Charity Choice contributions
- RTA bus passes and vehicle parking, subsidized monthly
- Purchase of prior Ohio service time (contact OPERS @ 614-466-2085)
- Deferred compensation contributions
 - <u>County Commissioners Association of Ohio Deferred Compensation</u> <u>Program</u> (800-423-3699)
 - o Ohio Public Employees Deferred Compensation Program (877-644-6457)
- Credit union deductions
 - o <u>Cleveland Postal Employees Credit Union</u> (located in basement of County Administration Building 443-7290)
 - o United Telephone Credit Union (440-333-6366)

Employees interested in taking advantage of any of these voluntary deductions should contact the Payroll Division at (216) 443-7380 for more information.

Cuyahoga County Department of Human Resources



Sections 6.10 & 6.11

6.12 <u>Direct Deposit Program</u>

In lieu of paper paychecks, the County can electronically transfer net pay into an employee's checking or savings account at the financial institution of the employee's choice._Enrollment in the Direct Deposit Program will be mandatory. Employees not eligible to secure direct deposit via a bank account will be offered alternative means. For further information, please contact vourthe Time & Attendance Department at (216) 443-7380. —Employees interested in participating in the Direct Deposit Program should contact the Payroll Division at (216) 443-7380 for more information or complete a Direct Deposit Authorization Form on the MyHR website and submit it to the Payroll Division.

<u>6.13</u> Savings Programs

Employees of the County may be eligible to participate in various savings programs through payroll deduction, including:

- Deferred compensation investment; and
- Credit union savings accounts.

Employees interested in any of these programs should contact the Payroll Division at (216) 443-7380 or refer to the County intranet for more information.

6.14 Payment Upon Separation from Employment with County

Compensation for accrued vacation leave and compensatory time will be included in the last paycheck.

No employee shall be paid for any unused sick leave upon termination of employment, except that, upon retirement, an employee with ten (10) or more years of service may receive cash payment for one-fourth (1/4) the value of the accrued, but unused sick leave credit, not to exceed 240 hours (30 days).

Cuyahoga County Department of Human Resources

(4)

Policies and Procedures Manual

Section 7: WORKPLACE WELLNESS

7.01 Drug-Free Workplace

The County is committed to maintaining a drug-free workplace. All County employees are required to abide by the County Drug Free Workplace Policy. Employees who abuse alcohol and other controlled substances or who use illegal drugs violate this policy and undermine the County's goal of providing responsible and professional service to the public. Employees who violate the County's Drug Free Workplace Policy are subject to discipline up to and including removal.

All individuals who have received a job offer with the County are required to submit to a test for alcohol and/or illegal drugs pursuant to <u>Section 5.04</u> of this Manual. Current employees may be tested pursuant to the County Fitness for Duty Program (<u>Section 7.04</u>) when there is a reasonable suspicion of impairment or post-accident.

Employees whose jobs require them to possess a commercial driver's license (CDL) and who drive vehicles for which testing is required under Federal regulations are also subject to alcohol and drug testing pursuant to Federal law.

In addition, since early identification and treatment are safeguards to job investment for both the County and its employees, employees who have a substance abuse problem are encouraged to voluntarily seek treatment.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 7.01

7.02 Smoke & Tobacco Free Workplace

POLICY

In an effort to promote and contribute to the health and well being of all County employees and members of the public, the County passed a resolution approving a smoke-free workplace policy effective August 1, 1994. It provides (in relevant part) that:

"...Be it further resolved that smoking will be strictly prohibited within all areas of County owned or leased buildings including: offices, hallways, entrance ways, lobbies and waiting areas, reception areas, rest rooms, cafeterias, snack bars, elevators, stairwells, meeting and conference rooms, kitchen areas, parking garages and areas immediately adjacent to any entrances..."

<u>Furthermore</u>, the Revised Code prohibits smoking in any "public place" or "place of employment", including any place adjacent to locations of ingress or egress to the public place or place of employment.

The county has included the Smoke Free Workplace to include "Tobacco Free" which is not limited to, smoking (e.g. cigarettes, pipes, cigars, etc.) and/or use of smokeless tobacco (e.g. snuff, chew, etc.) This directive shall apply to all employees, clients, and visitors or persons otherwise required to be in a County facility, vehicle or those driving their personal vehicles when transporting persons on company authorized business.

INCENTIVE PROGRAM

Employees who are "tobacco free" and receive health care benefits through the County may qualify for a health care benefit incentive. For more information on this program, contact the Department of Human Resources. Benefits Department

ASSISTANCE TO SMOKERS

The County has a variety of smoking cessation resources available to employees upon request. Employees who wish to "kick the habit" should contact their medical provider for information on smoking cessation resources. The County also offers an Employee Assistance Program or you can contact the Ohio Tobacco Quit Line at 1-800-QUIT NOW.

NOTE: Smoking is also prohibited in all County vehicles (See Section 16.03).

ENFORCEMENT OF POLICY

An employee wishing to report a violation of this Section should:

• If the violation is occurring on County-owned property or is being committed by a County Employee – contact Protective Services at (216) 443-2141 or 1-888-81-HELPU (43578).

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 7.02

If the violation is occurring on non-County-owned property by a non-County
employee (i.e., County-leased property) – contact the Ohio Department of Health
Smoking Ban Hotline at
1-866-559-OHIO (6446).

NON-COMPLIANCE

In addition to the penalties proscribed by the Revised Code, employees of the County who violate this section may be subjected to discipline, up to and including removal.

7.02 Smoke-free Workplace

The County passed a resolution approving a smoke free workplace policy effective August 1, 1994. It provides (in relevant part) that:

"...Be it further resolved that smoking will be strictly prohibited within all areas of County owned or leased buildings including: offices, hallways, entrance ways, lobbies and waiting areas, reception areas, rest rooms, cafeterias, snack bars, elevators, stairwells, meeting and conference rooms, kitchen areas, parking garages and areas immediately adjacent to any entrances..."

Furthermore, the Revised Code prohibits smoking in any "public place" or "place of employment", including any place adjacent to locations of ingress or egress to the public place or place of employment.

NOTE: Smoking is also prohibited in all County vehicles (See Section 16.03).

An employee wishing to report a violation of this Section should:

- If the violation is occurring on County owned property or is being committed by a
 County Employee contact Protective Services at (216) 443-2141 or 1-888-81
 HELPU (43578).
- If the violation is occurring on non County owned property by a non County employee (i.e., County leased property) contact the Ohio Department of Health Smoking Ban Hotline at 1-866-559 OHIO (6446).

In addition to the penalties proscribed by the Revised Code, employees of the County who violate this section may be subjected to discipline, up to and including removal.

Cuyahoga County Department of Human Resources



Policies and Procedures Manual

7.03 Employee Assistance Program (EAP)

Employee Assistance Program (EAP) services assist County employees with resolving personal and family related problems which adversely affect work, relationships, and quality of life. EAP services also help identify and alleviate obstacles to maintaining good mental and personal health.

Participation in the program may be mandated by Management with approval from Human Resources or voluntary and is available to all employees, their family members and County retirees. The County pays all costs related to EAP enrollment. EAP services include:

- Assessment of problem areas;
- Counseling, if necessary; and
- Referral to additional support services for more comprehensive treatment.

If extended treatment or counseling is required, an employee's health plan may cover most costs.

All communications with the EAP are confidential and employees who participate are assured that their job, future, and reputation will neither be jeopardized nor benefited by utilizing the EAP program. It is the responsibility of the employee to follow through with referrals for assessment of a problem and to follow recommendations of the diagnostician or counseling agent.

Employees interested in using the EAP services or who have questions regarding the program should contact Human Resources or the County's EAP provider:

Ease@Work
(216) 241-EASE (3273) or (800) 521-3273
www.easeatwork.com/EASEy
Username: COUNTY
Password: EASE

Cuyahoga County Department of Human Resources

 \Diamond

Section 7.03

Policies and Procedures Manual

7.04 <u>Fitness for Duty – Drug and Alcohol Testing</u>

In accordance with its Drug Free Workplace Policy, the County has established the following Fitness for Duty/Drug and Alcohol Testing Program.

NOTE: Bargaining unit employees should refer to their <u>collective bargaining agreement</u> for information regarding drug and alcohol testing.

Reasonable Suspicion

If a supervisor or Department manager/Director has a reasonable suspicion that an employee may be under the influence of drugs and/or alcohol, the supervisor shall immediately:

- If during regular business hours (8:30 a.m. 4:30 p.m.), contact the Fitness for Duty Coordinator at 216-443-7630.
- If before or after regular business hours, contact the Protective Services Division at (216) 443-2141

The Fitness for Duty Coordinator will come to the worksite and perform an evaluation of the employee. If the Fitness for Duty Coordinator, in consultation with Department management, determines that a drug and/or alcohol test is appropriate, the employee's immediate supervisor and another member of Department management will be required to sign a <u>Reasonable Suspicion Form</u>.

The employee will then be asked whether he or she is willing to submit to a drug and/or alcohol test. If the employee refuses, the employee will be treated as having tested positive and will be referred to and disciplined in accordance with the Fitness for Duty Program (*see* <u>below</u>).

If the employee agrees to the drug and/or alcohol test, he or she will be escorted by either the Fitness for Duty Coordinator or a Protective Services representative to an approved testing site. If the employee leaves the testing site at any point prior to being tested, the employee will be treated as having tested positive and will be referred to and disciplined in accordance with the Fitness for Duty Program (*see* <u>below</u>). Upon conclusion of the test, the Fitness for Duty Coordinator will determine whether the employee will be taken back to their worksite or to their residence. The leave status of the employee during this period of time will be determined by the County in light of the particular circumstances of each case.

If the employee's test results are positive, the employee will be referred to and disciplined in accordance with the Fitness for Duty Program (*see* <u>below</u>). If the employee's test results are negative, the investigation will be closed by the Fitness for Duty Coordinator.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 7.04 (1)

Post-Accident

Employees involved in a vehicle accident as defined under <u>Section 16.04</u> of this Manual, must follow the reporting procedures outlined in that Section. Upon being notified of a vehicle accident by the employer's supervisor and/or Protective Services, the Fitness for Duty Coordinator will consult with the appropriate parties and determine whether a drug and/or alcohol test is appropriate.

If the Fitness for Duty Coordinator determines that a drug and/or alcohol test is appropriate, the Fitness for Duty Coordinator or a representative from Protective Services will meet the employee at the accident site and take him or her directly to an approved testing facility. If the employee needs to seek immediate medical attention, the Fitness for Duty Coordinator will attempt to contact the medical facility where the employee seeks treatment in order to arrange for a drug and/or alcohol test of the employee.

If the employee refuses to submit to the test or leaves the testing site at any point prior to being tested, the employee will be treated as having tested positive and will be referred to and disciplined in accordance with the Fitness for Duty Program (*see* <u>below</u>). Upon conclusion of the test, the Fitness for Duty Coordinator will determine whether the employee will be taken back to his or her worksite or residence. The leave status of the employee during this period of time will be determined by the County in light of the particular circumstances of each case.

If the employee's test results are positive, the employee will be referred to and disciplined in accordance with the Fitness for Duty Program (*see* <u>below</u>). If the employee's test results are negative, the investigation will be closed by the Fitness for Duty Coordinator.

Employees involved in a non-vehicle related accident on work time may be required to submit to a drug and/or alcohol test pursuant to the County's Workers' Compensation Procedures (Section 12.02). The results of the drug and/or alcohol test, or the employee's refusal to submit to the test may affect the employee's eligibility for compensation and benefits pursuant to Revised Code Section 4123.54 and Chapter 4121 of the Revised Code and will subject the employee to the provisions of the County's Fitness for Duty Program (see below).

Commercial Driver License

In accordance with rules issued by the U.S. Department of Transportation (<u>49 C.F.R. 382</u>), the Ohio Department of Transportation (<u>Section 4506 of the Ohio Revised Code</u>) and the County's Drug-Free Workplace Program, employees who are required by the County to have a Commercial Driver License (CDL) are subject to drug and alcohol testing.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 7.04 (2)

Fitness for Duty Program

New-hire probationary employees are not eligible to participate in the Fitness for Duty Program. If a new-hire probationary employee tests positive for drugs or alcohol under any of the above-cited testing procedures, he or she will be immediately removed from employment with the County.

All non-new-hire probationary employees who test positive for drugs and/or alcohol pursuant to a "reasonable suspicion" or "post-accident" test will be <u>offered a chance to enrollenrolled</u> in the Fitness for Duty Program <u>on a case-by-case basis</u>. If the employee declines <u>or fails to successfully complete to enroll in the Program</u>, he or she will be removed from their employment with the County based on the positive test. <u>If the employee enrolls in the program</u>, they will be required to sign a "last chance agreement".

Employees enrolled in the Fitness for Duty Program will be required to submit to intensive outpatient treatment through the County's EAP. Employees who refuse or neglect to seek treatment will be in violation of the terms of their Fitness for Duty Agreement and will be removed from employment with the County. The employee will also be subject to random drug/alcohol tests for the duration of the two-year (2) program. If the employee tests positive for drugs and/or alcohol at any time during that two-year (2) period or refuses to submit to a random test, the employee will be considered in violation of his or her last chance agreement of the Fitness for Duty policy and will be removed from employment with the County. If the employee complies with all requirements of the program and does not test positive during the two-year (2) period, he or she will return to regular status.

An employee who is concerned about a substance abuse problem may voluntarily enroll in the Fitness for Duty Program at any time. Employees who voluntarily enroll, will not be subjected to the above reference policy however, will not be required to sign a "last chance agreement" unless they test positive for drugs or alcohol while in the program.

An employee whose performance is below standard or who has violated a work rule cannot avoid discipline by enrolling in a treatment program or by enrolling in the Fitness for Duty Program.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual

(4)

Section 7.04 (3)

7.05 Fitness for Duty – Physical and Mental Health

The County may require that an employee submit to medical or psychological examinations for purposes of determination of the employee's fitness for duty. If a Department Director, in consultation with Human Resources, determines that an employee's behavior and/or condition warrants a Fitness for Duty Examination, the Department Director shall complete a Medical Examination Request Form and submit it to Human Resources. Upon receipt of the Examination Request Form, the employee will be scheduled for an examination with an appropriate licensed practitioner. The employee will be notified of the date and time of their examination. At the sole discretion of the Director of Human Resources, the employee may be placed on paid administrative leave pending the results of the examination. The employee will be required to adhere to all requirements of employees on paid-administrative leave (see Note Box on p. 101).

An employee's refusal to submit to an examination or the unexcused failure to appear for an examination amounts to insubordination, which may result in discipline, up to and including removal. An employee will be responsible for the costs associated with an unexcused failure to appear at a scheduled examination.

Upon completion of the examination, the County will receive a report outlining the physician's opinion regarding the employee's fitness for duty. The employee will also receive a copy of the physician's report unless the physician determines that the disclosure of the information is likely to have an adverse effect on the employee, in which case the information shall be released to a physician, psychiatrist, or psychologist who is designated by the person or by the person's legal guardian. If the physician determines that the employee is fit for duty, the employee shall return to work on the day immediately following the day they are notified of the physician's findings or as soon as practicable. Failure to return to work will subject the employee to discipline pursuant to the Attendance Control Policy (see Section 14.03).

If the physician determines that the employee is not fit for duty, the employee will not be returned to work and will be taken off paid administrative leave and will be required to use their accrued paid leave (sick, vacation, exchange time, and/or compensatory time) or go on an unpaid leave in accordance with Section 9.09 of this Manual. The employee will remain off work while the medical conditions identified by the physician remain unresolved. The employee should submit medical documentation from their treating physician to the Fitness for Duty physician. The Fitness for Duty physician will determine whether the treatment provided renders the employee fit for duty and able to return to work in their job classification.

Once the employee exhausts all of their paid and unpaid leave, the County will move forward with disability separation proceedings pursuant to <u>Section 5.19</u> of this Manual.

NOTE: An employee that is determined to be not fit for duty may be eligible for disability retirement through <u>OPERS</u>. Please refer to <u>Section 5.20</u>. for more information on disability retirement through <u>OPERS</u>.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 7.05

7.06 Safety Policy

All County employees are charged with the responsibility of promoting the safety and security of fellow employees and the general public who come in contact with the County services and facilities.

All employees and contractors of the County are expected to conduct their duties in a safe manner, aimed at preventing or minimizing injuries and property damage throughout all County operations. Each employee must operate safely, use equipment, tools and materials properly and be totally familiar with work rules and procedures for his or her areas of responsibility. Each employee shall take an active part in the identification and reporting of hazards. Supervisors shall actively participate in the assessment and resolution of hazards and shall fully cooperate with Central Services and Protective Services to eliminate or control hazards in all areas of County facilities.

7.07 Workplace Violence

Threats, threatening behavior, or acts of violence against employees, clients, or guests by anyone on County property will not be tolerated. In addition to applicable criminal penalties, employees who violate this section may be subject to discipline, up to and including removal. The County reserves the right to remove anyone who engages in violent or threatening behavior from the premises and may require that they remain off the premises pending the outcome of an investigation and/or as long as they are still considered a threat.

Employees should immediately contact Protective Services at (216) 443-2141 or 1-888-81-HELPU (43578), and/or the appropriate law enforcement authority (911) if an employee (or guest/client) is creating or communicating a threat of violence. Employees that fail to report such threats will be subject to disciplinary action, up to and including termination of employment.

Employees with a criminal or civil protective or restraining order which lists company locations as being protected areas are responsible for providing a copy of such documents to Protective Services and Human Resources. Human Resources may meet with the employee and offer EAP or other services that might assist the employee.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 7.06 & 7.07

7.08 Mediation Program

Mediation is a voluntary process in which a neutral third person ("the mediator") assists individuals in resolving their conflicts. The mediator assists the parties in clarifying, discussing, and resolving the issues that created the conflict. The mediator, however, will not judge the parties or make decisions for them. The County has established a mediation program to prevent the progression and escalation of inter-personal conflicts and to help employees respond to conflict in an effective and positive manner. It is the desire of the County that employees make use of the mediation service before conflict affects the quality of work relationships or results in workplace violence.

The County mediation program is available to all County employees at all levels and can be requested by either party at any time. County management may also refer employees to the mediation program when appropriate. Mediation communication is confidential to the extent allowed by law. Communication that is never confidential, however, includes signed agreements between the parties, threats of bodily injury, plans to commit a crime and complaints of mediator misconduct. Participation in the County Mediation Program does not exempt employees from following County policies and procedures and will not delay any pending discipline. The County reserves the right to take disciplinary measures where standard rules of conduct are violated or where behavior deteriorates to unacceptable levels.

Employees interested in participating in or learning more about the County Mediation Program should contact Human Resources, Employment Relations Division, at (216) 443-7190 or TTY (216) 443-7002.

7.09 Wellness Programs

The County is committed to their employees maintaining good health. To that end, at the sole discretion of the County, wellness opportunities may be periodically offered to employees, including:

- Lunch and learn programs;
- Fitness classes (e.g., aerobics or body sculpting);
- Flu shots at discounted prices;
- Discounted fitness club memberships;
- Health screenings (e.g., mammograms, cholesterol, hypertension);
- Annual wellness fair;
- Health risk assessment;
- Certain preventative care covered at 100% through benefits; and/or
- Discounts on medications for asthma, cholesterol, diabetes and hypertension;

Employees will be notified of these offers as they become available.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 7.08 & 7.09

7.10 Sustainable Cuyahoga County

Comment [c6]: To Be Redacted

The County is committed to sustainability and sustainable development as defined in the Brundtland Commission Report, *Our Common Future*, 1987:

"Sustainable development is development that meets the needs of the present without compromising the ability of future generations to meet their own needs."

Critical to this is the financial balance of the "triple bottom line" measuring success through economic, environmental, and social equity.

Office Practice

The County encourages all employees to seek opportunities to incorporate sustainable principles in work and home life. Some examples include:

- Conserve energy by turning off lights, adjusting thermostats, and turning off equipment when not in use;
- Reduce, Reuse and Recycle products to reduce waste and conserve natural resources;
- Buy local;
- Use public transportation, reduce vehicle idling, and use fuel efficient vehicles; and
- Increase personal wellness.

Information regarding county sustainability programs such as recycling may be found on the County intranet, Sustainability Office website links, as well as by direct notifications.

Climate Change and the "Carbon Footprint"

On November 15, 2007, the County adopted a resolution making a Climate Stabilization Declaration to:

"Work closely with local, state, and federal governments and other leaders to reduce county geographical greenhouse gas (GHG) emissions to 80 percent below current levels by 2050..."

As a result of this declaration, a mid term goal was set to achieve 20 percent reduction in emissions by June 22, 2019. In furtherance of this declaration, the County is reducing its annual "carbon footprint" or ecological footprint through reduction in use of fossil fuels such as coal, oil & gas that produce greenhouse gas emissions. The sum total direct use of fuels and indirect use through procurement of products and services equal ones "carbon footprint".

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 7.10

Last Revised: 1/24/12

7.101 Lactation Policy

Comment [jjs7]: New Section

In compliance with the Fair Labor Standards Act amended by the Patient Protection and Affordable Care Act (Effective March 23, 2010), the County will provide comprehensive service including unpaid lactation time, a private space and some workplace support for any employee who is breastfeeding her child for up to one year after the child's birth.

Time to Express Milk:

An employee typically will be given two or three pumping sessions of about 20 minutes each during an eight hour shift, excluding any time needed to go to and from the lactation room. Employees can decide their own schedule. Generally, the lunch break and two paid rest breaks (Section 6.07) are considered lactation sessions. Any additional session which the employee thinks is necessary with the physician's certification but will not affect her work efficiency and quality during the work hours will be reviewed by the supervisor; upon approval will be unpaid lactation break. The supervisor shall help do the time registration of the employee and then report it with the employees' signature to the Human Resource Department.

Note: Since expressing milk is not associated with a serious health condition or bonding with caring for a newborn child under FMLA (Section 8.0), the breaks taken to express milk should not be considered FMLA leave or counted against an employee's FMLA leave entitlement although the FMLA entitles employees to take unpaid, job-protected leave in order to care for a newborn child.

Room Requirements:

Since Cuyahoga County has different working sites, the County will allocate* one room for lactation use at each building in order to provide convenience for the employees. The room can be put into other use during non-lactation periods. The room should be reserved in advance to make sure there will not be any schedule conflicts.

For those employees working at offsite locations they are to work with their managers to identify an appropriate space for expressing milk. The space can be an empty office or other available room provided it is not accessible to the public or other employees while in use. The room or space cannot be a restroom or toilet stall as well.

Room Scheduling:

For the female employee who has breastfeeding needs when she returns to work, she should inform the Human Resource Department first. The Human Resource Department will inform the employee's Agency Director and the Agency Director, or appointed designee, will ensure that the employee registers for the room based on her schedule (and make some adjustments if there are multiple users), notify her about the reasonable break time rule, room location and provide training to all supervisory-level employees to ensure that they understand the County's compliance obligations. At the time of registration, the employee will be required to complete the required form documenting the employee's understanding and compliance with the Lacatation Policy. Contact the Human Resource Department for more information.

Cuyahoga County Department of Human Resources



Section 7.11

Policies and Procedures Manual Last Revised: 4/5/11

If large numbers of breastfeeding employees will use the room, Cuyahoga County will schedule room usage based on each individual need. In order to maintain privacy when scheduling, each user will be given a personal identification number.

Note: The lactation room should be kept locked between uses to safeguard equipment, supplies and milk that may be stored in a refrigerator. And an "Occupied" sign will be hung outside the room when the room is in use.

For the departments with large numbers of users but limited lactation rooms, the County will create lactation "suites" by installing partitions or curtains in the lactation room to accommodate multiple users at one time.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual

Section 7.11 (2)

Section 8:

FAMILY MEDICAL LEAVE ACT (FMLA)

Timesheet Code: FMLA

8.01 Definitions

Active Duty Duty under a call or order to active duty under a provision of law

referred to in section 101(a)(13)(B) of title 10, United States Code.

Adoption Legally and permanently assuming the responsibility of raising a

child as one's own. The source of an adopted child is not a factor in

determining eligibility for FMLA leave.

Chronic Serious Health Condition A health condition that meets the following criteria:

- Requires periodic visits (at least twice a year) for treatment by a health care provider, or by a nurse under the direct supervision of a health care provider;
- Continues over an extended period of time (including recurring episodes of a single underlying condition); and
- May cause episodic rather than a continuing period of incapacity (e.g., asthma, diabetes, epilepsy, etc.).

Conditions Requiring Multiple Treatments

Any period of absence to receive multiple *treatments* (including any period of recovery therefrom) by a *health care provider* or by a provider of health care services under orders of, or on referral by, a *health care provider*, for either:

- Restorative surgery after an accident or other injury; or
- A condition that would likely result in a period of *incapacity* of more than 3 consecutive days in the absence of medical intervention or *treatment*, such as cancer (chemotherapy, radiation, etc.), severe arthritis (physical therapy), or kidney disease (dialysis).

Contingency Operation

Same meaning as definition provided in section 101(a)(13) of title 10, United States Code.

Cuyahoga County Department of Human Resources



Section 8.01 (1)

Policies and Procedures Manual Last Revised: 4/5/11

Covered Servicemember

Current member of the Armed Forces, including a member of the National Guard or Reserves, who is undergoing medical *treatment*, recuperation, or therapy, is otherwise in *outpatient status* or is otherwise on the temporary disability retired list, for a *serious injury or illness* incurred in the line of duty on active duty.

Daughter

For purposes of FMLA leave taken for birth or *adoption*, or to care for a family member with a *serious health condition*, daughter means a biological, *adopted*, or *foster child*, a stepchild, a legal ward, or a child of a person standing *in loco parentis*, who is either <u>under age 18</u>, or age 18 or older and "*incapable of self-care* because of a *mental or physical disability*" <u>at the time that FMLA leave is to commence</u>.

Forward Rolling Twelve (12) Month Period

The twelve month period measured forward from the date an employee's first FMLA leave begins.

Foster Care

24-hour care for children in substitution for, and away from, their parents or guardian. Such placement is made by or with the agreement of the State as a result of a voluntary agreement between the parent or guardian that the child be removed from the home, or pursuant to a judicial determination of the necessity for foster care, and involves agreement between the State and foster family that the foster family will take care of the child.

Health Care Provider

A health care provider for purposes of the FMLA includes:

- A doctor of medicine or osteopathy who is authorized to practice medicine or surgery by the State in which the doctor practices.
- Podiatrists, dentists, clinical psychologists, chiropractors (limited to treatment consisting of manual manipulation of the spine to correct a subluxation as demonstrated by X-ray to exist), and optometrists.
- Nurse practitioners, nurse-midwives, clinical social workers, and physician assistants.
- Christian Science Practitioners listed with the First Church of Christ, Scientist in Boston, Massachusetts.
- A health care provider listed above who practices in a country
 other than the United States, who is authorized to practice in
 accordance with the law of that country, and who is performing
 within the scope of his or her practice as defined under such law.

Inpatient Care

An overnight stay in a hospital, hospice, or residential medical care facility, including any period of *incapacity*.

Cuyahoga County
Department of Human Resources



Section 8.01 (2)

Policies and Procedures Manual

Incapable of Self-Care

Where the individual requires active assistance or supervision to provide daily self-care in three or more of the "activities of daily living" (ADL's) or "instrumental activities of daily living" (IADL's). ADL's include adaptive activities such a scaring appropriately for one's grooming and hygiene, bathing, dressing and eating. IADL's include cooking, cleaning, shopping, taking public transportation, paying bills, maintaining a residence, using telephones and directories, using a post office, etc.

Incapacity and Treatment

A period of *incapacity* of more than 3 consecutive, full calendar days, and any subsequent *treatment* or period of *incapacity* relating to the same condition, that also involves one of the following:

- Treatment 2 or more times, within 30 days of the first day of incapacity, by a health care provider.
 - Both *treatments* must involve an in-person visit to a *health care provider*.
 - The first in-person visit must take place within 7 days of the first day of *incapacity*.
 - Whether the second *treatment* visit is necessary must be determined by the *health care provider* (i.e., the doctor must request that the employee follow-up).
- Treatment by a health care provider on at least one occasion, which results in a regimen of continuing treatment under the supervision of the health care provider.
 - The initial treatment requires an in-person visit to a health care provider.
 - The initial in-person visit must take place within 7 days of the first day of *incapacity*.
 - Whether the *regimen of continuing treatment* is necessary must be determined by the *health care provider* (i.e., doctor must prescribe the regimen).

Incapacity

Inability to work, attend school or perform other regular daily activities due to the *serious health condition*, *treatment* therefore, or recovery therefrom.

In Loco Parentis

Those with day-to-day responsibilities to care for and financially support a child, or, in the case of an employee, who had such responsibilities for the employee when the employee was a child. A biological or legal relationship is not necessary.

Intermittent Leave

Leave taken in separate blocks of time due to a single illness or injury rather than for one continuous period of time.

Cuyahoga County Department of Human Resources



Section 8.01 (3)

Policies and Procedures Manual

Next of Kin of a Covered Servicemember The nearest blood relative other than the *covered servicemember's* spouse, parent, son, or daughter.

Outpatient Status

With respect to a *covered servicemember*, the status of a member of the Armed Forces assigned to either a military medical treatment facility as an outpatient; or a unit established for the purpose of providing command and control of members of the Armed Forces receiving medical care as outpatients.

Parent

A biological, *adoptive*, step or *foster father or mother*, or any other individual who stood *in loco parentis* to the employee when the employee was a *son* or *daughter* as defined in this section. In-law family members are not covered.

Parent of a Covered Servicemember A *covered servicemember's* biological, *adoptive*, step or *foster father or mother*, or any other individual who stood *in loco parentis* to the covered servicemember. This term does not include parents "in law".

Permanent or Long-Term Conditions A period of *incapacity* which is permanent or long-term due to a condition for which *treatment* may not be effective (e.g., Alzheimer's, a severe stroke, or the terminal stages of a disease). The individual must be under the continuing supervision of, but need not be receiving active *treatment* by a *health care provider*.

Physical or Mental Disability

A physical or mental impairment that substantially limits one or more of the major life activities of an individual.

Qualifying Exigency

The following types of military-related events may qualify as a "qualifying exigency" for purposes of eligibility for purposes of military-related FMLA leave under this Section:

- Short-Notice Deployment
- Military Events and Related Activities
- Childcare and School Activities
- Financial and Legal Arrangements
- Counseling
- Rest and Recuperation
- Post-Deployment Activities
- Additional Activities

Reduced Leave Schedule Leave schedule that reduces an employee's usual number of working hours per week or hours per day for a period of time.

Cuyahoga County
Department of Human Resources



Section 8.01 (4)

Policies and Procedures Manual

Regimen of Continuing Treatment

Includes a course of prescription medication (e.g., antibiotic) or therapy requiring special equipment to resolve or alleviate the health condition (e.g., oxygen). A regimen of continuing treatment that includes the taking of over-the-counter medications such as aspirin, antihistamines, or salves, or bed-rest, drinking fluids, exercise, and other similar activities that can be initiated without a visit to a health care provider, is not, by itself, sufficient to constitute a regimen of continuing treatment for FMLA purposes.

Serious Health Condition

An illness, injury, impairment or physical or mental condition that involves at least one of the following:

- Inpatient Care;
- "Incapacity and Treatment";
- Pregnancy or Prenatal Care;
- Chronic Serious Health Condition;
- Permanent or Long-Term Conditions; or
- Conditions Requiring Multiple Treatments.

Serious Illness or Injury An injury or illness incurred by a *covered servicemember* in the line of duty on active duty that may render the servicemember medically unfit to perform the duties of his or her office, grade, rank or rating.

Son

For purposes of FMLA leave taken for birth or *adoption*, or to care for a family member with a *serious health condition*, son means a biological, *adopted*, or *foster child*, a stepchild, a legal ward, or a child of a person standing *in loco parentis*, who is either <u>under age 18</u>, or age 18 or older and "*incapable of self-care* because of a *mental or physical disability*" at the time that leave is to commence.

Son or Daughter of a Covered Servicemember The *covered servicemember's* biological, *adopted*, or *foster child*, stepchild, legal ward, or a child for whom the servicemember stood *in loco parentis*, and who is of any age.

Son or Daughter on Active Duty or Call to Active Duty The employee's biological, *adopted*, or *foster child*, stepchild, legal ward, or a child for whom the employee stood in *loco parentis*, who is on active duty or a call to active duty status, and who is of any age.

Spouse

A husband or wife as defined or recognized under Ohio law for purposes of marriage in Ohio.

Treatment

Includes, but is not limited to, examinations to determine if a *serious health condition* exists and evaluations of the condition. Treatment <u>does not</u> include routine physical examinations, eye examinations, or dental examinations.

Cuyahoga County
Department of Human Resources



Section 8.01 (5)

Policies and Procedures Manual

8.02 Eligibility Requirements

Prior to making an application for FMLA under this Section, an employee must meet both of the following conditions to be considered eligible:

- The employee must have completed twelve (12) months of employment with the County. These twelve (12) months do not need to be consecutive; and
- The employee must have worked a minimum of one-thousand two-hundred-fifty (1,250) hours over the twelve (12) month period immediately preceding the commencement of the leave.

An employee returning from National Guard or Reserve military service will be credited with the hours of service that would have been worked, but for the period of military service, towards his or her one-thousand two-hundred-fifty (1,250) hour requirement.

8.03 Qualifying Conditions

Eligible employees shall be entitled to a certain amount of FMLA leave (*see* Section 8.04) under any of the following six circumstances:

- Birth of a *son* or *daughter*, and to care for the newborn child.
- Placement with the employee of a **son** or **daughter** for **adoption** or **foster care**.

NOTE: Eligibility for leave for birth or placement for *adoption* or *foster care* expires at the end of the twelve (12) month period beginning on the date of birth or placement.

- Because of a *serious health condition* that makes the employee unable to perform the functions of the employee's job.
- To care fore the employee's spouse, son, daughter, or parent with a serious health condition.
- Because of any qualifying exigency arising out of the fact that the employee's spouse, son, daughter, or parent is on active duty (or has been notified of an impending call or order to active duty) in the Armed Forces in support of a contingency operation.
- To care for a *covered servicemember* with a *serious injury or illness* if the employee is the *spouse*, *son*, *daughter*, *parent* or *next of kin* of the *covered servicemember*. The leave described in this paragraph shall only be available during a single twelve (12) month period.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Sections 8.02 & 8.03

8.04 Amount of Leave

Except in the case of leave to care for a *covered servicemember* with a *serious injury or illness*, an eligible employee is limited to a total of twelve (12) workweeks of FMLA leave during any *forward rolling twelve* (12) *month period*.

An eligible employee who is the *spouse, son, daughter, parent,* or *next of kin* of a *covered servicemember* shall be entitled to a total of twenty six (26) workweeks of leave during a single twelve (12) month period to care for the *covered servicemember*. If an employee uses *covered servicemember* family leave in combination with another form of FMLA leave (e.g., pregnancy leave), the maximum amount of combined FMLA leave that an employee can take in a single twelve (12) month period is twenty six (26) workweeks. In this scenario, though, the other form of FMLA leave can only account for a maximum of twelve (12) workweeks of the combined twenty six (26) workweeks of leave.

NOTE: When both spouses are employed by the County, FMLA leave is limited to a combined twelve (12) work weeks during any rolling twelve (12) month period if leave is taken for birth, *adoption*, or placement in *foster care*. This limitation does not apply to leave taken by either spouse to care for the other or for a child if the spouse or child has a *serious health condition*.

In addition, the aggregate number of leave to which both spouses are entitled to is limited to twenty six (26) workweeks during a single twelve (12) month period for *covered servicemember* leave, if the leave is leave for the *covered servicemember* or a combination of leave for the *covered servicemember* and the other forms of FMLA leave.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 8.04

8.05 Coordination With Other Leaves of Absence

Employers are mandated by law to grant up to twelve (12) weeks of unpaid leave to eligible employees under the FMLA. If an employee qualifies for FMLA leave (continuous or intermittent), any paid leave accrued by the employee shall be applied concurrently to the FMLA leave in the following order:

TYPE OF PAID LEAVE	ASSOCIATED POLICY	
	MANUAL SECTION	
Sick	9.03	
Vacation	<u>9.02</u>	
Compensatory Time	<u>9.04</u>	
Exchange Time	<u>9.05</u>	

If the medical condition does not qualify under the FMLA, then sick time shall apply in accordance with Section 9.03 of this Manual.

In addition, FMLA will automatically be applied to all employees who meet the FMLA qualifying criteria and utilize the following types of leave:

- Unpaid medical leaves of absence provided to employees by County policy or by any of the <u>collective bargaining agreements</u>;
- Leaves of absence using sick leave that is provided through the Leave Donation Program (Section 9.08); or
- Workers' Compensation leave of absence.

NOTE: As is the case with all unpaid leaves, an employee is not entitled to accrue sick or vacation benefits during any unpaid portion of FMLA leave. When the County offers or mandates furlough time, FMLA shall commence after all furlough time has been exhausted

Cuyahoga County Department of Human Resources

Policies and Procedures Manual

Section 8.05Last Revised: 1/24/12

8.06 FMLA Application Procedure

An employee that would like to request the use of FMLA leave should contact Human Resources. At least thirty (30) days advance notice must be provided before FMLA leave is to begin if the need for leave is foreseeable. In case of emergency, notice must be given as soon as practicable. If an employee fails to give thirty (30) days notice for foreseeable leave with no reasonable excuse for the delay, the County may delay the taking of FMLA leave until thirty (30) days after the date that the employee actually provided notice of the need for FMLA leave.

Completion of an FMLA Medical Certification Form is required whenever an employee requests FMLA leave due to the *serious health condition* of the employee or their *spouse, son, daughter,* or *parent*. Employees must return the FMLA Medical Certification Form to Human Resources within fifteen (15) calendar days of the employee's receipt of the form. If the certification is incomplete or insufficient, the employee may be required to provide additional information with in seven (7) calendar days. Failure to cure the deficiencies identified by the County may result in a denial of FMLA leave. After the FMLA Medical Certification Form has been returned to Human Resources, employees will be notified in writing whether their FMLA request is approved or denied.

For *adoption* or *foster care* placement, legal documentation should be submitted from a court or social service agency.

In any case in which the necessity for leave under the *qualifying exigency* condition is foreseeable, whether because the *spouse*, *or a son*, *daughter*, or *parent* of the employee is on *active duty*, or because of notification of an impending call or order to *active duty* in support of a *contingency operation*, the employee shall provide such notice to the employer as is reasonable and practicable. In addition the request for leave under the *qualifying exigency* condition must be supported by certification deemed sufficient by the Department of Labor.

The County reserves the right to require employees to recertify their conditions periodically upon request. Failure to submit a certification of health care provider form for the re-certification within fifteen (15) calendar days will result in delay or discontinuation of approved FMLA leave until the completed form is submitted.

If an employee needs an extension of FMLA leave, the employee must provide Human Resources notice as soon as practicable of the changed circumstances. Employees shall not be granted an extension beyond the applicable maximum amount of FMLA entitlement.

NOTE: Only persons who are authorized by the County may contact the health care provider of an employee to clarify the information in a FMLA Medical Certification Form or to verify its authenticity. **Supervisors and other management representatives may not contact an employee's health care provider.**

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 8.06

8.07 Continuation of Benefits

The County will continue to pay its portion of medical and supplemental benefits (vision & dental) when an employee is on FMLA leave. Employees on FMLA leave are still responsible to pay the employee's contribution for these benefits. Employees who take paid leave will continue to have their usual payroll deduction for health insurance benefits. Employees on unpaid FMLA leave will be billed for their usual contribution for medical and supplemental benefits. The amount billed will be the monthly amount that the employee would normally have paid for health insurance benefits through payroll deduction. Any new or additional coverage or changes in health benefits will be made available to employees on FMLA leave.

Employees on unpaid FMLA leave will also be billed for their life insurance deductions. If a death occurs and life insurance deductions have not been paid while on unpaid leave, death benefits will not be payable to the beneficiary.

8.08 Intermittent Leave and Reduced Leave Schedules

Intermittent leave or a reduced leave schedule may only be granted if:

- There is a medical necessity that can be accommodated by such leave; or
- Because of any *qualifying exigency* arising out of the fact that the *spouse*, *son*, *daughter*, or *parent* of the employee is on *active duty* (or has been notified of an impending call or order to *active duty*) in the Armed Forces in support of a *contingency operation*.

Eligible employees are required to make every reasonable effort to schedule leave so as not to disrupt the operations of their Department. Accordingly, the County may temporarily assign an employee to an alternate position with equivalent pay and benefits if the employee is qualified and the County determines that the alternative position will better accommodate the employee's intermittent leave or reduced leave schedule.

The actual amount of leave taken under an *intermittent* or *reduced leave schedule* shall be counted towards the twelve (12) weeks of FMLA entitlement. For example, a full-time employee who reduces from an eight (8) hour work day to a four (4) hour work day will use (20) hours of FMLA leave each week.

Employees on *intermittent* FMLA leave may be required to periodically provide an updated medical certification form.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Sections 8.07 & 8.08

8.09 Return to Work

Upon return from FMLA leave, an employee is entitled to be returned to the same position held when leave commenced, or to an equivalent position with equivalent pay, benefits and other terms and conditions of employment. Failure to return to work upon the expiration of the approved leave, without reasonable explanation, may subject the employee to discipline, up to and including removal, under Sections 13 and/or 14 of this Manual.

An employee who has been on FMLA leave because of his or her own *serious health condition* may be required to present medical documentation of his or her ability to return to work. The County may require that an employee obtain a second medical opinion from a physician selected by the County and at the County's expense. If the second medical opinion disagrees with the employee's health care provider, the County's health care provider and the employee's health care provider will select a third health care provider whose opinion shall be final. The County shall pay the cost of the third opinion.

Benefit deductions through payroll resume the first pay period of the month after an employee returns from leave. Benefits requires documentation of the return through use of either a Personnel Action Form (PAN) or Benefits Office Change Form (BOC).

Cuyahoga County Department of Human Resources



Section 8.09

Policies and Procedures Manual

Last Revised: 4/5/11

Section 9: EMPLOYEE LEAVE (General)

9.01 General

All leaves described in this section are subject to prior approval, unless otherwise noted. Depending on the practice of an employee's respective Department, the employee may be required to complete an Employee Request for Leave Form.

Every request for leave will be given fair consideration in accordance with the following policies and procedures and the staffing needs of the departmental unit. Any leave approved upon a false statement is invalid and any approved leave shall terminate if the reason for granting the leave is no longer applicable. Moreover, employees providing false statements or documentation are subject to discipline, up to and including removal, under Section 13 of this Manual.

Employees are solely responsible to ensure that they have adequate vacation leave, sick leave and/or compensatory/exchange time when taking paid leave.

NOTE: Bargaining unit employees should refer to their <u>collective bargaining agreement</u> for information on paid leave.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 9.01

9.02 Vacation Leave

Timesheet Code: V

Each full-time permanent employee, after service of one (1) year with the county or any political subdivision of the State of Ohio, is eligible for vacation leave. Employees accrue vacation leave based on years of service. Vacation accrual for eligible employees is pro-rated based on the number of hours paid in a pay period. One year of service shall be computed on the basis of twenty six (26) bi-weekly pay periods. Overtime hours are not included in the accrual of vacation leave.

Vacation leave is earned during the time the employee is in active pay status. Time spent on unpaid leaves of absence count toward service credit, but employees in unpaid status do not accrue vacation leave.

Employees may carry their vacation leave from year to year, up to a maximum of three years. Once an employee accumulates the maximum allowable vacation balance for the employee's current accrual rate, the employee has a period of one (1) year from the date on which the maximum balance was attained to use the accrued time in excess of the maximum allowable balance. Upon the end of the year period, any time over the maximum amount will be forfeited.

Vacation accrual is computed as follows:

Length of Ohio Public Service Completed	Accrual Rate (hours earned per 80 hours in active status)	Annual Amount (hours earned per 2080 hours in active pay status)	Maximum Accrual Balance (total hours)
Less than 1 year	3.1	80 hours (not awarded until completion of one year of Ohio public service)	N/A
1 year – less than 5 years	3.1	80	240
5 years – less than 15 years	4.6	120	360
15 years – less than 25 years	6.2	160	480
25 years or more	7.7	200	600

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 9.02 (1)

NOTE: An individual who is appointed by the County as a non-bargaining employee and provides proper documentation of prior service (*See* Section 2.04) is entitled, immediately upon submission of the appropriate documentation, to begin accruing vacation benefits at the rate set forth in this Manual that reflects their prior service level. The employee is also entitled to use such vacation leave at any time during the year in which it is accrued, upon approval from their supervisor.

However, an employee who has retired in accordance with the provisions of any retirement plan offered by the State of Ohio and who is employed by the County on or after June 24, 1987, shall not have prior service for the purpose of computing vacation leave. All employees currently receiving credit for service time prior to June 24, 1987, though, shall continue to receive service credit for such time.

Upon separation from County service, an employee is entitled to compensation for any earned and unused vacation leave to his or her credit at the time of separation.

Cuyahoga County Department of Human Resources

(4)

Section 9.02 (2)

Policies and Procedures Manual

9.03 Sick Leave

Timesheet Code: S

Sick leave may only be used for the purposes set forth in this Section. The use of sick leave is not an entitlement. Sick leave is not another form of vacation or personal leave. Supervisors and management should ensure that sick leave is properly used.

All County full-time, part-time, permanent and temporary employees accrue paid sick leave at the rate of .0575 hours for each hour of service in active pay status. For purposes of this Section, "active pay status" includes vacation, sick, overtime, compensatory time, and holidays. Employees in an unpaid leave status do not accrue sick leave.

NOTE: An individual who is appointed by the County as a non-bargaining employee who has a balance of sick leave earned while employed by a political subdivision of the State of Ohio within the past ten years may have the unused sick leave balance credited to their current sick leave balance. *See* Section 2.04 Section 2.03 for more information on the proper procedure for submitting documentation of prior service.

Sick leave is charged in minimum units of one minute. Sick leave payment shall not exceed the normal, scheduled workday earnings, or a maximum of forty (40) hours per week. Employees may not use sick leave that was accrued in the same pay period. In addition, sick leave may not be used to cover absences that occurred prior to the accrual of the leave.

Permissible Uses

Sick leave shall be granted for absences due to the following reasons and must be approved by management:

- Illness, injury, or pregnancy-related condition of the employee;
- Exposure of an employee to a contagious disease which could be communicated to and jeopardize the health of other employees;
- Death of a member of the employee's immediate family* (see Section 9.06);
- Examination of the employee, including medical, psychological, dental or optical examination by an appropriate practitioner;
- Illness, injury or pregnancy-related condition of a member of the employee's immediate family*; or
- Examination including medical, psychological, dental, or optical examination for a
 member of the employee's immediate family* by an appropriate practitioner where the
 employee's presence is reasonably necessary.

[* For purposes of this Section, Section 3.01 and Section 9.06, immediate family includes spouse, domestic partner, mother, father, sister, brother, son, daughter, grandmother, grandfather, grandson, granddaughter, mother-in-law, father-in-law, brother-in-law, sister-in-law, a legal guardian or other person who has an "in loco parentis" relationship with the employee, or any other relative residing with the employee.]

Cuyahoga County Department of Human Resources



Section 9.03 (1)

Policies and Procedures Manual

Procedure

An employee who is ill and unable to report to work shall notify their supervisor or designee on the first day of absence and each day thereafter, unless emergency conditions prevent such notification or unless the employee submits medical evidence of a planned extended medical-related absence. An employee working in a seven day continuous operation must report their absence two hours prior to their scheduled starting time except when emergency conditions prevent such notification. Other employees must give notice within one half-hour **before** the time the employee is scheduled to report for work unless emergency conditions prevent such notification.

NOTE: Upon being put on notice that an employee has been out on sick leave for three or more consecutive days, the employee's supervisor should immediately contact Human Resources so that they may send out FMLA information to the employee's home address. Employees should refer to Section 8 for more information on FMLA leave.

Upon return to work, the employee shall follow their Department's guidelines with regard to completion of the Employee Request for Leave Form. If the need for leave is foreseeable, sick leave should be requested in writing prior to the date or dates that an employee will be scheduled off from work. The County reserves the right to require documentation to support the employee's sick leave request (i.e., a doctor's statement) for all absences of three (3) or more days.

An employee's sick leave balance is provided bi-weekly. The employee is solely responsible for knowing his or her balance. Taking time off without sufficient leave to cover the absence may lead to the accrual of AWOL time (*See* Attendance Control Policy, Section 14).

Prohibited Use

The use of sick leave for any purpose other than its intended purposes is considered abuse of sick leave. Examples of abuse of sick leave include, but are not limited to the following:

- Calling in sick when one of the above cited reasons for using sick leave does not exist;
- Reporting illness in the immediate family when such illness does not exist;
- Reporting off sick to participate in some other activity or to take care of personal business;
- Establishing a pattern of reporting off sick, reporting tardy, or leaving early on certain days of the week, on days following regular days off (e.g., weekends), or on the day before or after a holiday or previously scheduled vacation; or
- Failure to follow the rules and regulations regarding the use of sick leave and reporting procedures.

Cuyahoga County Department of Human Resources



Section 9.03 (2)

Policies and Procedures Manual

If a supervisor or management official has a reasonable suspicion that sick time is being abused, either on an individual case or by virtue of a demonstrated pattern, he or she should immediately consult with Human Resources. Human Resources may then require an employee to provide a physician's statement or other appropriate documentation to support the use of sick leave. If it is determined that the employee abused his or her sick leave, Human Resources may charge the employee with AWOL and may pursue disciplinary action against the employee, up to and including removal.

Unused Sick Leave

Employees will not be paid for unused sick leave upon termination of employment with the exception that upon retirement, an employee with ten (10) or more years of service will receive cash payment for one-fourth (1/4) of the accrued, unused sick leave balance up to a maximum of two-hundred forty (240) hours.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 9.03 (3)

9.04 Compensatory Time

Timesheet Code: CT

NOTE: See Section 6.08 for information regarding accrual of compensatory time.

As stated in <u>Section 6.08</u>, compensatory time is granted to non-exempt (i.e., overtime eligible) employees. An employee who has requested the use of compensatory time from their direct supervisor shall be permitted to use it within a reasonable time after the request, so long as it does not unduly disrupt the operations of the agency.

Compensatory time not used within one hundred eighty (180) days after it is earned shall be paid to the employee. Upon termination all unused compensatory time is paid out to the employee.

Employees moving from a non-exempt position to an exempt position will receive payment for their accrued compensatory time at the time of the move.

9.05 Exchange Time

Timesheet Code: ET

NOTE: See Section 6.08 for information regarding accrual of exchange time.

As stated in <u>Section 6.08</u>, exchange time is granted to exempt (i.e., not overtime eligible) employees. Use of accrued exchange time may be granted to the employee by the employee's immediate supervisor at a time mutually convenient to the operation of the County and the employee. Exchange time may not be used in the same pay period in which it was earned.

Exchange time not used within six (6) months of accrual shall expire. At no time shall exchange time be paid out to the employee.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 9.04 & 9.05

9.06 Funeral/Bereavement Leave

Timesheet Codes: S/V/CT/ET/ALX

Sick leave shall be granted for absences due to the death of a member of an employee's immediate family*. In the event of the death of a member of his or her immediate family*, an employee shall be granted time off with pay, to be charged against his or her accumulated and unused sick leave, not to exceed five (5) days.

NOTE: In the event the employee has no sick time available, the employee may use vacation, compensatory or exchange time. If the employee has no vacation, compensatory or exchange time, the employee may request an unpaid Personal Leave of Absence.

In the event of the death of a relative other than a member of their immediate family*, an employee shall be granted a leave of absence with pay, to be charged against their accumulated and unused sick leave, for one (1) day to attend the funeral if within the State of Ohio or two (2) days when the funeral is outside the State of Ohio. Any additional time off requested by the employee is subject to the approval of his or her supervisor, manager and/or Department Director.

To be eligible for the above described funeral leave, an employee must attend the funeral, or other obligations related to the death and/or estate, etc. The failure to do so, or a misrepresentation of facts related to funeral leave shall be proper cause for disciplinary action, up to and including removal, as well as forfeiture of pay for the time away from work. The County reserves the right to request documentation of the family member's death and/or documentation establishing the familial relationship between the employee and the deceased family member.

[* See Section 9.03 for definition of "immediate family".]

Cuyahoga County Department of Human Resources



Section 9.06

Policies and Procedures Manual

9.07 Court Leave/HRC Hearings/Jury Duty

Timesheet Code: CL

General Provisions

- Any compensation or reimbursement for jury duty or for court attendance compelled by subpoena, when such duty is performed during an employee's normal working hours, shall be remitted by the employee to their Department Director or supervisor for transmittal to the County Treasurer.
- In order to be approved for court leave under this Section, the employee shall furnish a copy of the court/HRC notification or jury summons to their supervisor prior to their appearance date(s).
- Unless there will be less than one (1) hour left in the employee's regularly scheduled work day upon reporting to work, an employee out on approved court leave is required to immediately report to work after the court appearance is completed. (e.g., an employee on an 8:30 4:30 schedule who is released from jury duty at 1 p.m. is required to immediately report to work.)

Court Leave

An employee may be granted court leave with full pay if all of the following conditions are met:

- The legal matter is work related;
- The employee is not a party to the legal matter; and
- The employee is requested by the County to appear or is subpoenaed to appear before any court, commission, board or other legally constituted body authorized by law to compel the attendance of witnesses.

If any of these conditions are not met, the employee is not entitled to court leave. An employee not entitled to court leave, however, may use vacation leave, compensatory/exchange leave for purposes of appearing before a court or other legally constituted body in a matter in which he or she is a party or witness. Such instances would include, but not be limited to, criminal or civil cases, traffic court, divorce proceedings, custody, or appearing as directed as parent or guardian of juveniles.

HRC Hearings

An employee who is the appellant or is requested to appear in any action before the HRC and is in active pay status at the time of the scheduled hearing before the Commission, shall be granted court leave with full pay for purposes of attending the hearing. The employee, however, shall not be granted any mileage, travel or other related expenses for the hearing.

Jury Duty

Court leave with full pay shall be granted to any employee who is summoned for jury duty by a court of competent jurisdiction.

NOTE: For information regarding appearance at Industrial Commission (i.e., workers' compensation) hearings, please see Section 12.02.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 9.07

9.08 Leave Donation Program

Timesheet Code: SLD

County employees may donate accrued sick or vacation leave to a fellow County employee who has a serious health condition as defined under the FMLA and are in critical need of time off due to the condition. FMLA leave will run concurrently with the donated leave time.

An employee may receive donated leave provided that:

- The employee is first approved for FMLA leave that removes them from the workplace for fifteen (15) consecutive work days for their *own* serious health condition. Employees on intermittent FMLA leave do not qualify to participate in the program;
- Has exhausted all sick leave, vacation leave and compensatory/exchange time; and
- Has not had any discipline for attendance within the previous year.

An employee may donate accrued sick or vacation leave provided the donating employee:

- Voluntarily elects to donate leave to a designated recipient, who has qualified for the
 donation program, and does so with the understanding that donated leave which is used
 by the designated recipient will not be returned;
- Retains a sick balance of at least one hundred twenty (120) hours <u>after their donation</u>;
 and
- Is in active pay status at the time their sick time is to be used;

Employees interested in receiving donated time should contact Human Resources. Employees who qualify may receive leave through the Leave Donation Program for up to twelve (12) weeks. An employee's supervisor or representative may, with the written permission of the employee who is in need of leave or a member of the employee's immediate family, inform an employee's co-workers of such need for leave. When an employee has exhausted all donated time they will be placed on an unpaid extended sick/medical leave or may be separated in accordance with this Manual or their collective bargaining agreement.

Donation of leave time is strictly voluntary. Any employee who feels that they are being pressured to donate leave time should file a General Employee Complaint (*See* Section 17.03).

NOTE: Employees that are denied participation in the Leave Donation Program may file an appeal in writing to the Director of Human Resources.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 9.08

9.09 Unpaid Leaves of Absence

The County offers employees three types of unpaid leave: (1) New-Hire Administrative Leave of Absence; (2) Personal Leave of Absence; and (3) Extended Unpaid Sick/Medical Leave. Depending on the practice of an employee's respective Department, the employee may be required to complete an Employee Request for Leave Form to request an unpaid leave of absence.

Any leave approved upon a false statement is invalid and any approved leave shall terminate if the reason for granting the leave is no longer applicable. Moreover, employees providing false statements or documentation are subject to discipline, up to and including removal, under Section 13 of this Manual.

An employee on an unpaid leave of absence does not accrue sick or vacation leave. In addition, time spent on an unpaid leave of absence does not count towards an employee's new hire or promotional probationary period. However, the time spent on an unpaid leave of absence is counted in determining length of service for purposes of vacation eligibility, annual step increases and layoff purposes, provided the employee is properly returned to service and is not serving a probationary period.

NOTE: With the exception of Extended Unpaid Sick/Medical Leave taken concurrently with FMLA leave, an employee in an unpaid status on the first day of a month will not receive health care coverage and related benefits unless he or she elect to pay the entire expense. The employee may be required to pay up to one-hundred and two percent (102%) of the entire health insurance and related benefits premium costs to retain his or her benefits status.

The total amount of time for all leaves under this section shall not exceed six (6) months in any forward rolling twelve month period. Employees are responsible to ensure that they have received proper approval when taking unpaid leave.

Failure to return to work upon the expiration of the approved leave, without reasonable explanation, may subject the employee to discipline, up to and including removal.

NOTE: An employee who fails to return to service from an unpaid leave of absence and who is subsequently removed or voluntarily resigns from service shall not receive service credit for the time spent on such leave and is deemed to have a termination date corresponding to the starting date of the unpaid leave of absence.

An employee may be permitted to return to work prior to the originally scheduled expiration of the leave if such earlier return is agreed to by both the employee and Human Resources. Upon completion of an unpaid leave of absence, the employee shall be returned to the same or similar position within the employee's former classification. If the employee's former classification no longer exists, the employee shall be placed in a classification similar to that formerly occupied.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 9.09 (1)

New-Hire Administrative Leave of Absence

Timesheet Code: ALX

An administrative leave may be granted to a newly-appointed employee if they need to take time off from the job due to extenuating circumstances, subject to the following limitations:

- New-hire administrative leave is unpaid;
- New-hire administrative leave may only be granted during a newly-appointed employee's first year of employment with the County;
- Prior approval must be received from the newly-appointed employee's Department Director:
- New-hire administrative leave may only be granted when all appropriate accumulated paid leave time (e.g., sick leave, vacation, and compensatory/exchange time) leave has been exhausted: and
- An employee is limited to a total of 80 hours of new-hire administrative leave.

Personal Leave of Absence

Timesheet Code: LOA

A personal leave of absence involves a temporary separation from active pay status, authorized by the County. Such leave must be for a minimum of ten (10) working days, but may not exceed a maximum period of six (6) months, with no extension or renewal allowed.

A personal leave of absence may only be granted when all appropriate accumulated leave (vacation leave and compensatory/exchange time leave) has been exhausted.

A personal leave of absence must be requested by the employee in writing and must be approved by the employee's Department Director. The request shall state reasons for taking the leave and the dates for which such leave is being requested and shall be accompanied by any necessary supporting documentation. Every request for leave will be given fair consideration in accordance with the following policies and procedures and the staffing needs of the departmental unit.

Extended Unpaid Sick/Medical Leave

Timesheet Code: LOA

An employee who is unable to perform any of the essential functions of his or her position due to disabling illness, injury or condition, and the disability continues after the employee has exhausted his or her accumulated sick leave benefits, may be granted a leave of absence without pay for a period of up to six (6) months upon presentation of evidence as to the probable date for return to active work status.

The following additional criteria must be met in order to be granted leave under this section:

- The employee's absence will be at least five (5) consecutive workdays;
- The employee must demonstrate that the probable length of absence will not exceed six (6) months; and
- The employee must present to Human Resources, at the time the request is made, a licensed physician's report stating diagnosis, prognosis, and probable period for which the employee will be unable to perform the essential job duties of their position.

Cuyahoga County
Department of Human Resources



Section 9.09 (2)

Policies and Procedures Manual

uyahoga County epartment of Human Resources	Section 9.09 (<u>3</u>)
nployee may be given a disability say apply for <u>OPERS</u> disability reting formation for <u>OPERS</u>).	

9.10 Maternity/Paternity Leave Policy

Maternity leave permits parents who have care giving responsibilities to have time off to spend with a child newly added to the family. In the event of female employees' pregnancy and male employees' wives' pregnancy, they shall be granted leave of absence from work for maternity purposes.

Cuyahoga County is firmly committed to protecting the rights of expectant mothers and complying with Title VII of the 1964 Civil Rights Act as amended by the Pregnancy Discrimination Act of 1978. Cuyahoga County's policy is to treat women, and men whose wives are affected by pregnancy, childbirth or related medical conditions in the same manner as other employees unable to work because of their physical condition in all employment aspects, including recruitment, hiring, training, promotion and benefits.

Further, the County fully recognizes eligible employees' rights and responsibilities under the Family and Medical Leave Act (section 8.0 of the County Personnel Policies and Procedures Manual),

In an effort to treat all employees equally, the County will provide an maternity/paternity leave of absence for those employees who do not meet the requirements of the Family and Medical Leave Act (Section 8.0). The leave will not exceed eight (8) weeks. The employee may use any accrued sick or vacation time concurrent with the approved leave. Requests (with supporting medical certification) must be made to Human Resources thirty (30) days prior to the expected date of delivery or as soon as practical.

At the conclusion of the eight (8) weeks, additional leave request of a medical necessity will be reviewed in conjunction with current leave policies.

If both spouses are employed by the County while neither of them meets the eligibility of FMLA, the unpaid leave is limited to a combined eight (8) weeks.

Note: An employee on an unpaid leave of absence does not accrue sick or vacation leave. In addition, time spent on an unpaid leave of absence does not count towards an employee's new hire probationary period. See Section 5.07 for more detail.

Pregnant employees may continue to work until they are certified as unable to work by their physician. At that point, pregnant employees are entitled to receive benefits according to Section 11.3 (Benefits During Unpaid Leaves of Absence) of the County's Personnel Policies and Procedures Manual).

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 9.10 (1)

Last Revised: 4/5/11

Page 326 of 403

Comment [jjs8]: New Section

Note: An employee in an unpaid status on the first day of a month will not receive health care coverage and related benefits unless he or she elects to pay the entire expense. The employee may be required to pay up to one-hundred and two percent (102%) of the entire health insurance and related benefits premium costs to retain his or her benefits status. Contact the Benefits Department at (216) 443-3539 for more information.

The employee is entitled to be reinstated with full seniority to the same or equivalent job with no loss of service or other rights or privileges provided the employee contacts the County within thirty (30) days after the birth and indicates a return-to-work date with a certificate from the employee's physician as to fitness to perform the duties. Should the employee not return to work within the approved time frame the employee will be considered to have voluntarily terminated employment with the County

Employees will be able to reserve 40 hours of sick time to be utilized upon their return from an approved FMLA Maternity or Paternity Leave.

Cuyahoga County Department of Human Resources



Section 9.<u>10</u> (2)

Policies and Procedures Manual Last Revised: 4/5/11

Section 10: MILITARY LEAVE

NOTE: County Employees who are related to members of the Armed Forces, including members of the National Guard or Reserves, should be aware that they may be eligible for military-related FMLA leave. Please see <u>Section 8</u>. for more information regarding military-related FMLA leave.

10.01 Military Leave With Pay (Military Reserve Leave)

Timesheet Code: ML

Permanent County employees who are members of the Ohio National Guard, the Ohio defense corps, the Ohio naval militia, or any other reserve component of the United States armed forces, are entitled to a leave of absence from their duties without loss of pay for a period or periods not to exceed twenty-two (22) working days (176 hours) in any one calendar year, provided they are called for field training or active duty. This service does not need to be in one continuous period. The maximum number of hours for which payment under this provision will be made in any calendar year is one-hundred seventy-six (176).

Any military pay received for this twenty-two (22) or less day period of military service shall have no effect on pay from the County. This twenty-two (22) day period of military leave will not affect accrual of sick leave or vacation credits. The employee will continue to accrue sick leave and vacation credits at the normal rate.

An employee who is entitled to leave provided under this Section, and who is called or ordered to the uniformed services for longer than a month, for each calendar year in which the employee performed service in the uniformed services, because of an executive order issued by the governor pursuant to the Revised Code is entitled, during the period designated in the order or act, to a leave of absence and to be paid, during each monthly pay period of that leave of absence, the lesser of the following:

- The difference between the employee's regular gross monthly wage or salary and the sum of the employee's gross uniformed pay and allowances received that month; or
- Five hundred dollars (\$500).

No employee, however, shall receive payments pursuant to the above paragraph if the sum of the employee's gross uniformed pay and allowances received in a pay period exceeds the employee's regular gross wage or salary. In order to obtain any paid leave under this Section, the employee must submit to Human Resources a copy of military orders or other statement in writing from the appropriate military authority as evidence of a call to training or duty.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 10.01

10.02 Unpaid Military Leave

complete an initial period of obligated service.

Pursuant to the Uniformed Services Employment and Reemployment Rights Act (USERRA), an eligible County employee shall be granted, upon giving notice to his or her Department and Human Resources, a leave of absence to serve in the uniformed service, as defined in the Revised Code. This leave shall be without pay and shall be considered as a leave of absence from service with reinstatement rights. No single leave of absence or combination of uniformed service leaves of absence may exceed five (5) years or a single, longer period required to

Employees on military leave without pay for up to thirty (30) days must be given the option of making direct payments of the employee's share of the health insurance premium. Employees with longer periods of service will be given the option of continuing health care coverage and related benefits for up to twenty-four (24) months. Such continuation shall be at the employee's expense.

NOTE: the employee may be required to pay up to one-hundred and two percent (102%) of the entire health insurance and related benefits premium costs.

Employees on military leave who are reinstated to employment with the County pursuant to this Section are entitled to immediate reinstatement of health insurance for the employee and any other individual who is covered by such plan by reason of the reinstatement of the coverage for the employee, with no waiting period and no exclusion of preexisting conditions, except for service-connected illness or injury. Even if the employee does not elect to continue health coverage during his or her military service, he or she will have the same right to immediate reinstatement of health insurance.

Employees returning from military service will be reinstated in the job that they would have attained had they not been absent for military service and with the same seniority, job status and pay, as well as other rights and benefits determined by seniority. The County will make all reasonable efforts to enable returning employees to refresh or upgrade their skills to enable them to qualify for reemployment. If refresher training is not successful, the County will reinstate the employee in a position that most nearly approximates that position originally held.

An employee returning from military leave must apply for reinstatement to their Department or Human Resources within the following time frames:

- Leave of less than thirty (30) days the employee must report back to work at the next
 regularly scheduled shift on the day following release from the military, safe travel home,
 and eight (8) hours of rest.
- Leave of thirty-one (31) to one hundred eighty (180) days the employee must apply for reinstatement within fourteen (14) days following release.
- Leave of one hundred eighty-one (181) days or more the employee must apply for reinstatement within ninety (90) days after release.

Cuyahoga County

Department of Human Resources

(4)

Section 10.02 (1)

Timesheet Code: MLX

Policies and Procedures Manual Last Revised: 4/5/11

Failure to return to work or apply for reinstatement in accordance with these time limits does not necessarily forfeit the employee's reinstatement rights, however, the employee will be subject to the County Attendance Control Policy (Section 14).

Following a period of service of thirty-one (31) days or more, employees will be required to provide documentation to the County which establishes the following:

- The timeliness of the employee's application for reinstatement;
- That the employee has not exceeded their cumulative 5-year limit of military leave; and
- That the character of their discharge or separation from uniformed service does not meet one of the types set forth in 20 CFR 1002.135.

If the documentation is not readily available, or does not exist, the County will not deny reinstatement, however, if the documentation later becomes available and it shows that the employee did not qualify for reinstatement, the employee will be immediately removed from employment.

For employees who were disabled while on military duty, or if a pre-existing disability is aggravated by military service, the County will make reasonable efforts to accommodate the disability (*See* Section 4.07 for more information on disability accommodations.)

If the employee's service was one-hundred eighty-one (181) days or more, they are protected from discharge, except for cause, for one (1) year. If the employee's service was for thirty-one (31) to one-hundred eighty (180) days, the period of protection from discharge, except for cause, is one-hundred eighty (180) days.

Employees may elect to use any vacation leave they have accrued for a period of unpaid military leave. Use of vacation is strictly optional. Employees on unpaid military leave do not accrue sick or vacation leave.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 10.02 (2)

Section 11: **BENEFITS**

11.01 Health Care Benefits

The County's comprehensive benefits program, consists of group medical, prescription, dental, vision, life insurance and flexible spending accounts.

NOTE: Part-time employees hired on or after January 1, 2008 are not eligible for County provided health care benefits. Part-time employees who were receiving County health care benefits prior to January 1, 2008 will continue to remain eligible to receive health care benefits.

New employees have thirty (30) days from their hire date to make their benefit elections and to submit documentation regarding dependents. Under most circumstances, if an employee fails to make benefit elections within that time period, the employee will have to wait until open enrollment to elect benefits.

New employees shall be eligible to receive health care benefits on the first day of the month following their date of hire. The costs of the employee's selected benefits plans are shared between the employee and the County. The County will make its first contribution on the first payroll deduction for the first month following the employee's hire date. Employees, at their own expense, can elect to begin medical, dental, vision and flexible spending account coverage on the first day of hire. Employees who choose this option are responsible for 100% of the cost of their benefits.

Employees who take a leave of absence during any benefits waiting period may incur a delay in the effective date of County paid benefits. The new effective date will be calculated from the day the waiting period is completed.

The County's health benefits may cover an employee's spouse and dependent children. To be covered, children must be under age twenty-eight (28), unmarried, and legal dependents of the employee. Health benefits may also cover certain other dependent children if the employee has legal guardianship, as outlined in each of the County's group insurance carrier contracts. No dependent can be covered by more than one County sponsored benefit plan. Documentation is required before dependents can be added.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 11.01 (1)

Last Revised: 1/24/12

NOTE: Effective January 1, 2011, employees may be able to add dependents whose age is greater than 23 to their benefits coverage. Federal rules will allow employees to add dependents who will be covered through the month of their 26th birthday (with proper documentation). The State of Ohio allows dependents to be added following the month of their 26th birthday through the month of their 28th birthday if they meet all the State of Ohio criteria.

If an employee is covered by medical coverage elsewhere, the employee may be entitled to waive County coverage and receive a taxable wage incentive for maintaining non-County coverage. Employees should contact the County Benefits Department at (216) 443-3539 or benefits@cuyahogacounty.us for more information on this incentive.

Employee contributions for health benefits are paid through a pre-tax payroll deduction, both during and after the waiting period.

<u>Fraud Warning</u>: Any person who, with intent to defraud or knowing that he or she is facilitating a fraud, submits an application or files a claim containing a false or deceptive statement may be guilty of insurance fraud. Any employee found making false or deceptive statements may be subject to disciplinary action, up to and including removal.

By enrolling in a County benefit plan, an employee agrees to comply with eligibility rules for themselves and for all of their dependents in these plans. The enrolling of ineligible dependents may be considered fraud. An employee may be subject to an eligibility audit during the benefit year. An employee selected for an eligibility audit may be required to submit copies of documentation such as certified birth certificate(s), marriage certificate(s), income tax return(s) and/or other related documentation including affidavits.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 11.01 (2)

Last Revised: 1/24/12

11.02 Open Enrollment

Open enrollment generally occurs between October and December. Employees must go on line annually to make their elections. Employees who do not make elections on line by the open enrollment deadline will default to the medical plan that the County designates. Once an employee is defaulted to this designated plan, he or she cannot appeal or change this plan unless a qualifying event occurs. Employees are also bound by the deadlines to submit documentation during open enrollment for addition and/or deletion of dependents.

11.03 Benefits During Unpaid Leaves of Absence

If an employee is on an unpaid leave of absence, benefits may be maintained. To continue benefits, employees must make required monthly contribution payments which may be greater than those paid by employees in active pay status. Employees should contact the County Benefits Department at (216) 443-3539 or benefits@cuyahogacounty.us for more information on continuation of benefits during a leave of absence.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 11.02 & 11.03

11.04 Changes in Medical Benefits Coverage

Under most circumstances, employees can only make changes to benefit elections during the annual open enrollment period. After enrollment, revision to an employee's elections is only permitted due to an eligible change in status. Employees must notify Human Resources and complete a <u>Benefits Office Change Form</u>, including attachment of required documentation, within thirty (30) days of an eligible change of status.

Examples of changes of status include:

- Marriage;
- Divorce, legal separation, or annulment;
- Birth, adoption, placement for adoption, or legal guardianship;
- Change of status in child custody;
- Death of dependent;
- A child who is no longer a legal dependent or a dependent reaching age twenty-three (23) (please see <u>note box in</u> Section 11.01 regarding changes to this requirement on or after January 1, 2011);
- Loss of alternate coverage; and/or
- Certain reductions/increases in work hours and or work status (i.e., from part-time to full-time).

NOTE: Employees are responsible for notifying the Benefits Division if their legal guardianship ends or if their dependent children get married, leave home with their own job and/or are no longer shown by the employee as a dependent on their income tax returns.

Changes in status can affect the employee's single/family coverage entitlement for medical, prescription drugs, dental and vision, as well as impacting flexible spending accounts and life insurance. In the case of divorce, legal separation and annulment, employees must adjust their covered dependents by removing former spouses within thirty (30) days of divorce, legal separation or annulment. Copies of the first and last pages of the court document must be included. Failure to adhere to this requirement may subject the employee to discipline, up to and including removal.

NOTE: Divorced employees who are required by court order to pay benefits for their ex-spouse cannot cover the ex-spouse under County benefits. A separate policy must be purchased outside of County benefits. It is illegal to cover an ex-spouse under County benefits.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 11.04

11.05 Continuation of Coverage

In compliance with Federal COBRA laws, all new employees receive a comprehensive letter concerning their rights with respect to continuing group health care benefits. Generally, a former employee can continue benefits for up to eighteen (18) months, providing that they pay the monthly group rate in addition to a small administrative fee. Some unique situations, as defined by law, may warrant a longer period of up to twenty-nine (29) months. Upon severing employment with the County, eligible employees will receive notification of their COBRA entitlement, duration and payment instruction.

In addition, Federal HIPAA laws require that the County (or its designated agent) provide eligible employees a certificate of group health care coverage. This document may be used as evidence of their insurability to another employer or organization whose group health carrier may otherwise impose an exclusion for certain pre-existing conditions. Employees needing a copy of their certificate of group health care coverage should contact the County Benefits Department at (216) 443-3539 or benefits@cuyahogacounty.us.

The County's group term life insurance has two features allowing employees to take a portion or all of their life insurance with them under an individual arrangement with the County's insurance carrier:

- The "conversion provision" allows an employee to convert the basic insured amount, if applicable to their group, and the supplemental coverage into a whole life policy.
- The "portability provision" allows an employee to take the insured amount that is in excess of the basic coverage and create an individual term life policy at the same prevailing group premium. Changes in premium still occur at the same five-year (5) age intervals as when employed.

NOTE: The "conversion" and "portability" provisions both involve direct payment to the insurance carrier.

There are deadlines associated with both the "conversion" and "portability" provisions of the County's sponsored life insurance program. The County routinely sends an informative notification concerning preservation of life insurance when employment ends. However, employees should contact the County Benefits Department at (216) 443-3539 or benefits@cuyahogacounty.us prior to separation from employment with the County.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 11.05

11.06 Retirement – Ohio Public Employee Retirement System (OPERS)

All County employees are required by state law to participate in **OPERS** through payroll deduction.

Information regarding <u>OPERS</u> is available by contacting Human Resources or by directly contacting <u>OPERS</u>:

Ohio Public Employee Retirement System 277 East Town Street Columbus, Ohio 43215-4642 1-800-222-PERS (7377)

www.opers.org

Field Code Changed

11.07 Death Benefit

Employees who elect life insurance coverage must complete a <u>Beneficiary Form</u>. Upon the death of an employee or their covered dependents, death benefits will be payable. Contact the County Benefits Department at (216) 443-3539 or benefits@cuyahogacounty.us for more information regarding death benefits.

11.08 Optional Additional Employee Benefits

At the sole discretion of the County, certain additional optional employee benefits may be offered to County employees, including:

- Voluntary benefits, such as accident insurance, whole life insurance with long term care rider, and/or short term disability;
- Discounted RTA bus/transit passes and parking;
- Health and wellness programs; and
- Tuition reimbursement.

County employees will be notified of these offers as they become available.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 11.06 & 11.08

11.09 Flexible Spending Accounts (FSAs)

Flexible Spending Accounts are offered by the County as an additional benefit. The County offers two types of FSAs: (1) Medical Payment Accounts; and (2) Dependent Care Payment Accounts. More information on both of these plans can be found in the open enrollment material distributed by the Benefits Division or by contacting the Benefits Division at (216) 443-3539 or benefits@cuyahogacounty.us.

Medical Payment Account (MPA)

The MPA allows employees to use pretax deductions to pay medical, prescription drug and dental expenses that are not covered by the employee's medical plan (e.g., co-pays, deductibles, orthodontia, etc.). Employees deposit these payroll deductions in a separate account with the County's third party provider. To draw on the account, employees must submit receipts for eligible expenses to the third party provider through use of <u>reimbursement form</u>. The employee will then be reimbursed by the third party provider.

Dependent Care Payment Account (DCPA)

The DCPA allows employees to use pretax deductions to pay childcare expenses. The DCPA works in the same manner as the MPA with regard to submission of receipts and <u>reimbursement form</u>. Childcare services can be provided in or out of the employee's home as long as the sitter is not a dependent that you claim on your income tax. The DCPA covers childcare for children less than thirteen (13) years of age, but also may be used to pay for an incapacitated adult who is dependent on the employee for 50% of their support and who needs care while the employee is at work.

NOTE: Employees should be aware that there are strict deadlines associated with the submittal of claims for withdrawal of funds from flexible spending accounts. Employees that fail to submit claims prior to the deadline will forfeit any remaining balances in their flexible spending accounts. More information regarding specific deadlines can be obtained by contacting Human Resources.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 11.09

Section 12: WORKERS' COMPENSATION

12.01 Workers' Compensation Policy

The State of Ohio Workers' Compensation Laws cover County employees. A County employee who sustains a job-related injury or occupational disease may be eligible to receive benefits from the Bureau of Workers' Compensation (BWC).

12.02 Employee Procedure

A County employee who is involved in a work-related accident/injury or is diagnosed with an occupational disease must adhere to the following procedure:

- 1. The employee must immediately notify their supervisor of the work-related accident/injury or occupational disease diagnosis. If the employee is incapacitated, the employee must notify their supervisor as soon as practicable.
- 2. The employee must complete a Cuyahoga County Accident Investigation Report within 24 hours of the occurrence, unless confined to the hospital. If confined to the hospital, the employee must complete the accident report within 24 hours of release from the hospital.
- 3. The employee may seek treatment at the medical provider of their choice so long as the provider is BWC certified. If medical care is needed, the employee should obtain a Managed Care Organization (MCO) card from Human Resources for submittal to their provider. If immediate medical care is needed, the employee does not need to obtain a MCO card. Rather, the employee should inform their provider that they have a work-related injury and refer them to the County's MCO, 1-888-OHIOCOMP. University Hospitals CompCare (www.UHCompCare.com) 1-800-818-7273.

NOTE: An employee who is involved in a work-related accident/injury may be required to submit to a drug and/or alcohol test pursuant to Revised Code <u>Section 4123.54</u>. The results of the drug and/or alcohol test, or the employee's refusal to submit to the test may affect the employee's eligibility for compensation and benefits pursuant to Revised Code <u>Section 4123.54</u> and <u>Chapter 4121</u> of the Revised Code and will subject the employee to the provisions of the County's Fitness for Duty Program (<u>Section 7.04</u>).

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 12.01 & 12.02 (1)

Employees who are off work due to a work-related accident/injury or occupational disease are responsible to keep their supervisors notified of their work status and to provide appropriate documentation from their physician. Appropriate documentation includes certification of total disability (BWC Form - C-84) and physician's reports of work ability (BWC Form - Medco-14).

If the employee is released to light or full duty, the employee must provide appropriate documentation from their physician to Human Resources and the County Workers' Compensation Office.

If an employee is required to attend an Industrial Commission hearing, he or she will be given up to two (2) hours of paid leave. The employee must present the Hearing Notice from the Industrial Commission to their supervisor prior to the hearing date in order to be granted the paid leave.

If an employee is required to attend a County or BWC mandated medical examination, he or she will be given up to four (4) hours of paid leave time to attend. The employee must present their supervisor with the written letter from the BWC, MCO or TPA in order to be granted the paid leave.

NOTE: Time off for attendance at any medical appointment not mandated by the County or BWC must be covered by the employee's own leave time.

Cuyahoga County Department of Human Resources

C

Policies and Procedures Manual

Section 12.02 (2)

12.03 Supervisor Procedure

Immediately upon receipt of notification from an employee of a job-related accident/injury or occupational disease diagnosis, the supervisor shall:

- Provide the employee with a "Workers' Compensation Injury Reporting
 PacketCuyahoga County Injury Care Packet." Reporting packets can be obtained from
 Human Resources. The reporting packets contain the Accident Investigation Report that
 must be filled out by the employee; and
- 2. Contact Human Resources to notify them of the incident.

NOTE: If the incident in question involves a vehicle accident, the supervisor is also required to contact Fitness for Duty and Protective Services in accordance with <u>Section 16.04</u> of this Manual.

Upon receipt of the completed Accident Report from the employee, the supervisor shall immediately give the form to Human Resources.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 12.03

12.04 Relationship to Leave Time

Employees who are off work for more than seven (7) calendar days due to a job-related accident/injury or occupational disease may be entitled to disability benefits through the BWC. It may be several weeks, however, before the BWC determines eligibility and issues the first disability check to an injured employee. County employees who are off work due to a job-related accident/injury or occupational disease have the following options while waiting for the BWC to determine eligibility and issue disability payments:

- Sick Leave An employee may elect to use their accrued sick leave. The employee will continue to receive their full wages and benefits while using sick leave. The BWC, however, will not reimburse an employee for the usage of their sick time (i.e., the BWC will not issue retroactive payment of disability payments to cover the period of time where sick leave was used). In addition, the County will not reimburse or recredit the employee for the sick time used.
- Vacation/Compensatory/Exchange/Personal Leave An employee may elect to use their accrued vacation, compensatory, exchange or personal leave. The employee will continue to receive their full wages and benefits while using vacation, compensatory, exchange or personal leave. Should the BWC determine that the employee is eligible for disability payments, the BWC will issue retroactive disability payments to cover eligible periods where vacation, compensatory, exchange or personal leave were used. The County, however, will not reimburse or re-credit the employee for the vacation, compensatory, exchange or personal leave time used.
- Unpaid Workers' Compensation Leave Timesheet Code: WCX Injured employees also have the option of not receiving a paycheck from the county and waiting on BWC wage benefits. However, the decision to go into a no-pay status may affect health care benefits. Should the BWC determine that the employee is eligible for disability payments, the BWC will issue retroactive disability payments to cover eligible periods where unpaid workers' compensation leave was used. Should the BWC determine that the employee is ineligible to receive disability, the employee will not receive any retroactive payment or reimbursement from the BWC or the County.

In addition, in the event that an FMLA eligible employee suffers lost time of three (3) days or more, the employee will be required to use their available FMLA time to cover up to the first twelve (12) weeks of lost time.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 12.04

12.05 Alternative Work Program

Policy

On July 1, 2003, the County passed a Resolution adopting the Cuyahoga County Alternative Work Policy. The passage of this Resolution indicates that the County fully supports the provision of alternative work opportunities for employees who have been injured at work and are eligible for Workers' Compensation Benefits. Employees who have been injured may be able to work for a specific period of time in a modified form of their present job or at a different job assignment.

This policy only pertains to employees with County work related injuries who have been released by their physician for alternative work/light duty. The policy may also be limited based on individual departmental work needs at a particular time.

<u>Procedure</u>

- When an employee is injured, the County's MCO, <u>University Hospitals CompCare</u> (<u>www.UHCompCare.com</u>) 1-800-818-7273. 1-888 OHIOCOMP, will review the employee's medical records and discuss restrictions with the employee's physician (or review Medco-14 form from physician). The MCO will also make available Functional Job Analyses for certain positions for review by the physician. The MCO will then make recommendations on appropriate candidates. A Department may also make recommendations for appropriate candidates. Appropriate candidates will likely miss more than 7 calendar days of work.
- The employee must file a workers' compensation claim with the BWC and complete the County's accident report before being considered for alternative work. However, the County's initial allowance order does not have to be issued prior to participation in the program.
- The County Workers' Compensation Office will work in conjunction with Department management to determine whether there is an appropriate alternative position within the employee's department.
- The Department may object to the assignment if there are already two employees from the Department participating in this program.
- If a bargaining employee is involved, the union will be notified of the assignment. The
 employee will be notified of their right to union representation regarding the assignment.
- The employee will maintain their current rate of pay and their bargaining status will remain in force. Where possible, all efforts will be made to assign the employee to their regular shift.

Cuyahoga County
Department of Human Resources

Section 12.05 (1)Last Revised: 4/5/11

Policies and Procedures Manual

- The County Workers' Compensation Office will inform the employee by certified mail in writing of the new assignment, reporting time and work hours. A copy of this letter will be sent to the union (if applicable), Human Resources, Department Director(s) and supervisor(s).
- An employee will be eligible for alternative work for up to sixty (60) calendar days. This initial sixty-day period may be extended by agreement between the Department and employee (or union if a bargaining employee). At no time may an alternative work assignment exceed one hundred twenty (120) days.
- For an employee to be allowed to return to their prior position, the employee must provide a full release from their physician, indicating that the employee can perform the essential functions of the position, with or without accommodation, or have been granted a reasonable accommodation under the ADA. (See Section 4.07 of this Manual for more information on making a request for an ADA Accommodation.)

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 12.05 (2)

<u>12.06</u> <u>Wage Continuation Program</u>

The County Wage Continuation Program is an optional program available for employees who are injured at work and meet specific eligibility requirements. Under the Program, the employee will continue to receive their regular County wages as opposed to receiving workers' compensation disability benefits for up to sixty (60) calendar days while they are recovering from their injuries. The employee will also continue to receive their applicable benefits, including accrual of sick, vacation and retirement credit.

The program is entirely voluntary and the employee may opt out of the program at any time. The employee must meet all eligibility requirements and follow all mandated procedures to participate in the program. The County reserves the right to make determinations on eligibility on a case by case basis.

Eligibility

- Injury or occupational disease must be compensable under Workers' Compensation Laws
 of the State of Ohio;
- Employee must miss more than seven (7) calendar days of work;
- Employee must fill out a county accident report in accordance with Section 12.02 of this Manual:
- Employee must use a medical provider from either <u>MetroHealth</u> or <u>OccuCenters</u> facilities.
- Employee must fill out a Wage Continuation Form (available from Human Resources), and return it to the County Workers' Compensation Office within 72 hours of the injury/diagnosis. If employee is hospitalized, the employee must return the form within 72 hours of their discharge from the hospital. There will be no retroactive application to this program;
- Employee must submit FMLA paperwork to Human Resources and are required to use their FMLA leave. Employees who are not eligible for FMLA leave are not eligible to participate in this program; and
- Employee must agree to participate in alternative work if eligible. The employee's participation in the alternative work program shall be governed by the rules set forth in Section 12.05 of this Manual.

Procedure

- Upon receipt of the required forms, the Workers' Compensation Office will consult with the employee's supervisor and make a decision to either certify or reject the injury/occupational disease claim.
 - o **Rejected** the employee will not be eligible to participate in the program.
 - Certified the Workers' Compensation Office will enroll the employee in the program.
 - Certified, but later disallowed by the BWC the employee's participation in the program will be suspended until a final decision is issued on the claim. The employee will be required to use accumulated paid leave or FMLA leave during

Cuyahoga County
Department of Human Resources



Section 12.06 (1)

Policies and Procedures Manual

this period of time if the employee has not been released by their doctor. If the employee does not have sufficient leave time to cover the absence, the employee may be placed on an extended unpaid sick/medical leave.

- If, after the appeals are exhausted, the claim is allowed, leave time used will be refunded.
- The employee's physician will be required to submit both a C-84 and Medco-14 along with an estimated return to work date to the Workers' Compensation Office. The Workers' Compensation Office will use these forms to determine the appropriateness of an alternative work assignment.
- Wage continuation may continue for a maximum of sixty (60) calendar days or until the
 employee's physician releases the employee to either full duty or alternative work,
 whichever is earlier.
 - o The sixty-day limit applies to each injury (i.e., if an employee returns to work after forty-five (45) days of wage continuation, but later needs to be off work again for the same injury the employee will only be authorized for fifteen (15) additional days of wage continuation. After a total of sixty (60) calendar days of wage continuation, the employee will not be eligible to participate in wage continuation for the same injury for the duration of the participant's service with the County.)
- If the employee is unable to return to work at the end of the sixty-day time period, he or she will be directed to seek benefits from the BWC. The employee will be required to continue on FMLA leave until exhausted.
- An employee may discontinue the program at any time. If the employee decides to opt
 out of the program, the employee may use his or her accumulated leave time and apply to
 use the standard Workers' Compensation Program. An employee who has opted out of
 the program may not opt back in at a later date.

Medical Treatment

- Employees in the program must use a medical provider from either MetroHealth or OccuCenters facilities. It is recommended that employees with less serious injuries use OccuCenter, and employees with more serious injuries use MetroHealth.
- Any employee who fails to fully comply with his or her treating physicians recommendations, including treatment, medication, therapy, vocational rehabilitation, and/or alternative/transitional work assignment, will be immediately removed from the program and will forfeit any future claim to program benefits. Missing appointments will be considered non-cooperation.
- Employees in the program who are working in an alternative work position will be granted paid leave time for medical or physical therapy appointments with their approved physician. The employee will be required to submit documentation of the appointment to their supervisor, including the time he or she arrived and left the appointment. Failure to submit appropriate documentation will result in the employee being charged with sick leave for the time away from work
- Employees in the wage continuation program who have returned to full duty will be granted up to two (2) hours of leave time per day for medical appointments <u>related to</u>

Cuyahoga County
Department of Human Resources



Section 12.06 (2)

Policies and Procedures Manual Last Revised: 4/5/11

their work-related injury for up to one (1) year following their date of injury. The appointment must be with their approved physician under this program and the employee will be required to submit documentation in the manner described in the prior paragraph.

Prohibitions

- An employee is not eligible to use leave donation while participating in this program.
- Employees in the program shall only be paid for their regularly scheduled hours during the period of disability.
- Employees may not maintain any other employment while collecting wage continuation.
- In the event that the County finds that an employee who has received wage continuation has filed a fraudulent claim, the County reserves the right to discipline the employee pursuant to Section 13 of this Manual, up to and including removal, and to recoup the full worth of all wage and benefit payments rendered to the employee.

NOTE: The County reserves the right to modify the wage continuation program at any time.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 12.06 (3)

Section 13: EMPLOYEE CONDUCT POLICY

13.01 Guidelines for Appropriate Conduct

All County Employees are responsible for performing their jobs in a competent, professional manner and for conducting themselves, at all times, in a way that advances the goals of the County and increases public confidence in Cuyahoga County government. This requires County employees to exercise due care and regard for the safety and security of persons and property, and to refrain from behavior which might be harmful to the county's interests, or which violates or conflicts with the county polices and procedures.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 13.01

13.02 Employee Dress/Appearance

County employees are required to dress in a manner appropriate for their position. Unless otherwise required by their position, employees are generally required to dress in a manner that is "business appropriate."

Although it is not possible to compile a complete list of acceptable and unacceptable attire, employees should always strive to dress in a way that maintains a business appearance and promotes a positive image and level of professionalism.

Employees should be aware that their respective Departments may have additional rules and regulations or more specific guidelines regarding employee dress/appearance. In determining the appropriate dress standards for individual units, Department management should consider the following factors:

- The nature of the work performed by the unit;
- The likelihood of interaction with the public; and
- Consistency amongst similarly situated units.

Department management has the right to take corrective action when an employee is improperly dressed. An employee who is dressed inappropriately may be sent home to change his or her clothes and may be subject to further discipline under this Section. An employee who is sent home shall be required to use his or her own accumulated vacation, compensatory or exchange leave to cover the time that he or she is away from the worksite to change clothes. In the event the employee has no vacation, compensatory or exchange time, the employee will be placed in AWOL status for the time that he or she is away from the worksite.

Nothing in this policy is intended to abridge an employee's rights relating to non-discrimination or to hinder the advancement of diversity at the County. The County will reasonably accommodate those employees whose bona fide religious belief or disability requires special attire (*see* Section 4.08 for information regarding requests for religious accommodation; *see* Section 4.07 for information regarding disability accommodation requests).

Cuyahoga County Department of Human Resources



Section 13.02

13.03 Employee Responsibility for County Property

Employees shall not abuse, neglect, waste or misappropriate County property. All employees are responsible for the proper care of any tools, materials, equipment or vehicles assigned for the performance of their jobs. No County tools, equipment, materials or vehicles shall be used for any purpose other than authorized work-related activities. No County tools, equipment or materials shall be taken from the worksite for any purpose unless specifically authorized by the employee's supervisor and/or management. Any violation of these requirements may subject an employee to discipline, up to and including removal.

Upon separation from County employment, employees must return to their Department Management all County-issued cell phones, pagers, computers, keys, uniforms, and other such equipment. In addition, employees must return their ID badges to Human Resources.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 13.03

13.04 Recording of Conversations

The County has established the following guidelines for the recording of telephone conversations, office meetings and office conversations concerning County business. Any violation of this Section could result in disciplinary action, up to and including removal.

Telephone Conversations

County employees shall not record any telephone conversation involving the discussion of County business unless at least one of the following applies:

- 1. All parties first consent either verbally or in writing;
- 2. The party intending to record the conversation notified the other party both at the beginning and at the end of the conversation that the call is being recorded;
- 3. The Law Director or designee authorizes in writing an employee to record a telephone conversation without the other party's consent (e.g., in furtherance of a sexual harassment investigation); or
- 4. County management authorizes the recording of a telephone conversation as a standard course of business (e.g., all telephone calls to the 696-KIDS hotline are recorded).

Office Meetings and Conversations

County employees shall not record any internal office meetings, conversations, and/or any oral communication involving the discussion of matters pertaining to County business unless at least one of the following applies:

- 1. All parties first consent either verbally or in writing; or
- 2. The Law Director or designee authorizes in writing an employee to record a meeting/conversation without the other party's consent (e.g., in furtherance of a sexual harassment investigation).

An internal office meeting is defined as an assembly of two or more County employees for the purpose of discussing matters relating to the conduct of County business.

NOTE: This policy shall not apply to public meetings or any other meeting or forum involving the discussion of County business that is open to members of the general public, including, but not limited to, public hearings.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 13.04

13.05 Workplace Search/Privacy Policy

NOTE: *See* <u>Section 15.02</u> for information regarding the County's electronic equipment/e-mail privacy policies.

County management may question employees and/or inspect any personal property or any area from which the County conducts business, including any spaces, facilities and/or vehicles leased by the county, when there are **reasonable grounds** for suspecting that the search will reveal evidence of work-related misconduct, including, but not limited to:

- When management has a reasonable suspicion that a search/questioning is necessary to safeguard another employee's safety or property;
- When management has a reasonable suspicion that a search/questioning is necessary to safeguard a County customer or their property;
- To protect County property from destruction and/or theft;
- To investigate possible violations of County policy; or
- In furtherance of an internal workplace harassment/discrimination investigation.

County management will obtain authorization from either Human Resources or the Protective Services Division prior to conducting a workplace search.

NOTE: Employees who are without their County issued ID Badge may be required to submit to a search through the use of a metal detector. In addition, all County employees may periodically be required to submit to a search through the use of a metal detector.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 13.05

13.06 Professional Licenses

Any County employee who is required as a condition of his or her employment to possess and maintain in good standing, a professional license (e.g., social worker license, license to practice law, etc.), shall immediately report any change in his or her licensure status to their Department Management (i.e., license revoked due to bad behavior or failure to renew). An employee who fails to promptly report any change in their licensure status may be subject to disciplinary action, up to and including removal.

13.07 Criminal Conviction

Pursuant to Section 5.05 of this Manual, all applicants for appointment to positions under the County are required to declare whether they have been previously convicted of certain crimes. Any employee that fails to disclose a previous criminal conviction or current arrest for one of the listed crimes on their application is subject to disciplinary action, up to and including removal for falsification of their application.

Any current County employee that is arrested <u>and/or charged</u> for one of the following crimes must immediately report said arrest to Human Resources:

- Felonies (or under arrest for crime punishable as a felony);
- A crime involving dishonesty, fraud, or theft (e.g., forgery, burglary, robbery, credit card fraud, perjury, bribery, tax evasion);
- Crimes of moral turpitude (e.g., sex offenses, pandering, prostitution, importuning, public indecency);
- Crimes involving physical violence (e.g., assault, patient abuse or neglect); and
- Drug-related crimes (e.g., trafficking offenses, drug possession).

Any employee who fails to disclose the fact that he or she has been arrested <u>and/or charged</u> for one of the above-described crimes is subject to disciplinary action, up to and including removal.

Conviction of a crime or a history of criminal convictions may constitute cause for disciplinary action. Determination of such action will be based on an analysis of the responsibilities of the position and the nature and time frame of the conviction.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 13.06 & 13.07

13.08 **Inappropriate Conduct / Grounds for Discipline**

Inappropriate Conduct

In accordance with Ohio civil service laws, the County considers the following conduct to be inappropriate and grounds for disciplinary action:

Incompetency Violation of County Policy and/or Procedure Inefficiency Violation of any Ohio Civil Service Law

Dishonesty Failure of Good Behavior

Drunkenness / Substance Abuse Misfeasance Immoral Conduct Malfeasance Insubordination Nonfeasance Discourteous Treatment of the Public Felony Conviction

Neglect of Duty

 $\underline{\textit{Examples of Inappropriate Conduct}}$ The following examples of inappropriate conduct are provided to assist County management with administration of the Progressive Discipline Program. The examples, however, are provided solely for illustration, and are NOT intended to be exhaustive or exclusive.

MINOR INFRACTIONS:

The following is a non-exclusive list of examples of minor infractions:

Type of Conduct	Examples:
Neglect of Duty /	• Failing to follow call-in procedures for a single day of absence without
Inefficiency /	approved leave.
Incompetency	Taking unauthorized or extended lunch or other breaks.
	 Being away from the worksite without permission or leaving work prior to the end of the shift without authorization.
	• Loafing, loitering, or failing to perform work in a prompt and efficient manner.
	 Reading material during regular work hours that is not job-related or authorized by management.
	 Conducting non-work related business while on County time.
Violation of County	Abuse of County telephones for personal, local calls.
Policy and/or	Using County equipment or property for other than its intended
Procedure	purpose (including, but not limited to, e-mail and Internet).
	• Failing to sign time sheets, properly register time or to punch time card.
	• Smoking on County owned or leased properties (1 ticket)-
Insubordination	• Failing to follow supervisor's instructions or Departmental procedures.
<u>Failure of Good</u>	Behavior unbecoming toward the public or fellow employee
Behavior	

Cuyahoga County Section 13.08 (1) Department of Human Resources Policies and Procedures Manual Last Revised: 4/5/11

MAJOR INFRACTIONS:

The following is a non-exclusive list of examples of major infractions:

Type of Conduct	Examples:
Neglect of Duty / Inefficiency / Incompetency	 Sleeping while on duty. Leaving a post of continuous operation without authorization prior to the end of the shift or prior to proper relief. Failing to follow call-in procedures for two consecutive days of absence without approved leave. Failing to report for overtime after being scheduled to work overtime or refusing to work mandated (mandatory) overtime when assigned. Failure to complete a legitimate job assignment. Failure of a supervisor to properly supervise or enforce work rules or
Violation of County Policy and/or Procedure Discourteous Treatment of the Public	 County policies and procedures. Abuse of County telephone for personal long distance calls. Violation of County Policy and/or Procedures, "Social Media Policy." Smoking on County owned or leased properties (2 tickets) Failure to provide prompt, courteous service to the public.
Failure of Good Behavior	 Verbal abuse or profanity directed at a co-worker, supervisor, or the public. Unauthorized sharing of confidential information about clients and/or employees. Using another employee's confidential computer identification code without authorization or giving another's code to anyone to use. Any act that would bring discredit to the employer Failure to cooperate with a County investigation
Insubordination	 Refusing a legitimate job assignment or failing to perform a directive from a supervisor or management representative.

REMOVABLE INFRACTIONS:

The following is a non-exclusive list of examples of removable infractions:

Type of Conduct	Examples:
Neglect of Duty/	Egregious, flagrant or willful neglect in the performance of assigned
Inefficiency /	duties.
Incompetency	Disregarding safety or security regulations that results or would likely
	in serious physical harm or major property loss or damage.
	• Failing to follow call in procedures for three consecutive days of

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual

Section 13.08 (2)

Last Revised: 4/5/11

Egregious, flagrant, or willful failure of a supervisor to properly supervise and enforce work rules or County policies and procedures.		-t
Supervise and enforce work rules or County policies and procedures.		absence without approved leave.
Intentionally destroying County property or records without prior authorization.		
## authorization. ## Procedure Violation of County Policy and/or Procedures. "Social Media Policy." Smoking on County owned or leased properties (3 or more tickets) Palsification of employment records or other County records in manual or automated systems, including falsification of stated reason for use of leave (e.g., abuse of sick leave or using sick leave for unauthorized purposes.) Unauthorized punching, signing or altering other employee's time cards or timesheets Unauthorized altering of one's own time card or sheet. Making false claims or providing false information in investigations of workplace misconduct or accidents. Making false claims or misrepresentations in an attempt to obtain any County benefit (e.g., health benefits, promotion, paid leave of absence, etc.). Felony Conviction Conviction of a felony. Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer.	T. 1	
Procedure Violation of County Policy and/or Procedures, "Social Media Policy." Smoking on County owned or leased properties (3 or more tickets) Falsification of employment records or other County records in manual or automated systems, including falsification of stated reason for use of leave (e.g., abuse of sick leave or using sick leave for unauthorized purposes.) Unauthorized punching, signing or altering other employee's time cards or timesheets Unauthorized altering of one's own time card or sheet. Making false claims or providing false information in investigations of workplace misconduct or accidents. Making false claims or misrepresentations in an attempt to obtain any County benefit (e.g., health benefits, promotion, paid leave of absence, etc.). Felony Conviction Failure of Good Behavior Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business.		
Smoking on County owned or leased properties (3 or more tickets) Palsification of employment records or other County records in manual or automated systems, including falsification of stated reason for use of leave (e.g., abuse of sick leave or using sick leave for unauthorized purposes.) Unauthorized punching, signing or altering other employee's time cards or timesheets Unauthorized altering of one's own time card or sheet. Making false claims or providing false information in investigations of workplace misconduct or accidents. Making false claims or misrepresentations in an attempt to obtain any County benefit (e.g., health benefits, promotion, paid leave of absence, etc.). Felony Conviction Failure of Good Behavior Conviction of a felony. Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Drunkenness / Substance Abuse Possessing an open container of alcohol on County property or while engaged in County business.	•	
Falsification of employment records or other County records in manual or automated systems, including falsification of stated reason for use of leave (e.g., abuse of sick leave or using sick leave for unauthorized purposes.) Unauthorized purposes.) Unauthorized purposes.) Unauthorized purching, signing or altering other employee's time cards or timesheets Unauthorized altering of one's own time card or sheet. Making false claims or providing false information in investigations of workplace misconduct or accidents. Making false claims or misrepresentations in an attempt to obtain any County benefit (e.g., health benefits, promotion, paid leave of absence, etc.). Felony Conviction Conviction of a felony. Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer.	Proceaure	
manual or automated systems, including falsification of stated reason for use of leave (e.g., abuse of sick leave or using sick leave for unauthorized purposes.) • Unauthorized purching, signing or altering other employee's time cards or timesheets • Unauthorized altering of one's own time card or sheet. • Making false claims or providing false information in investigations of workplace misconduct or accidents. • Making false claims or misrepresentations in an attempt to obtain any County benefit (e.g., health benefits, promotion, paid leave of absence, etc.). Felony Conviction Failure of Good Behavior • Conviction of a felony. • Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. • Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. • Striking or physically assaulting another person. • Fighting, except for self-defense. • Theft of property from co-workers, the County or others. • Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. • Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. • Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. • Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. • Any act that would bring discredit to the employer. • Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business.		
cards or timesheets Unauthorized altering of one's own time card or sheet. Making false claims or providing false information in investigations of workplace misconduct or accidents. Making false claims or misrepresentations in an attempt to obtain any County benefit (e.g., health benefits, promotion, paid leave of absence, etc.). Felony Conviction Failure of Good Behavior Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Drunkenness / Substance Abuse Possessing an open container of alcohol on County property or while engaged in County business.	Dishonesty	manual or automated systems, including falsification of stated reason for use of leave (e.g., abuse of sick leave or using sick leave for unauthorized purposes.)
 Making false claims or providing false information in investigations of workplace misconduct or accidents. Making false claims or misrepresentations in an attempt to obtain any County benefit (e.g., health benefits, promotion, paid leave of absence, etc.). Felony Conviction Conviction of a felony. Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 		cards or timesheets
workplace misconduct or accidents. Making false claims or misrepresentations in an attempt to obtain any County benefit (e.g., health benefits, promotion, paid leave of absence, etc.). Felony Conviction Failure of Good Behavior Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Making an a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business.		
County benefit (e.g., health benefits, promotion, paid leave of absence, etc.). Felony Conviction Failure of Good Behavior Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business.		
 Failure of Good Behavior Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 		County benefit (e.g., health benefits, promotion, paid leave of absence,
 Soliciting or accepting a gratuity or bribe from anyone conducting business with the County or seeking to do business with the County. Using obscene, abusive, or threatening language or gestures, or performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 	Felony Conviction	Conviction of a felony.
performing an act with intent to cause injury to another. Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Punkenness / Substance Abuse Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business.	Failure of Good	Soliciting or accepting a gratuity or bribe from anyone conducting
 Striking or physically assaulting another person. Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 		
 Fighting, except for self-defense. Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 		, ,
 Theft of property from co-workers, the County or others. Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 		
 Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was committed. Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 		
 Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer, and/or touching someone else if it is unwelcome and uninvited. Engaging in a pattern of behavior which creates a hostile, intimidating, or offensive work environment based on a protected characteristic. Any act that would bring discredit to the employer. Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 		 Unauthorized possession of a firearm, explosives, or other dangerous weapons on County property or while engaged in County business. Performing an act which constitutes a felony under the laws of the United States, the State of Ohio or the jurisdiction in which the act was
 Any act that would bring discredit to the employer. Drunkenness / Substance Abuse Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 		Making an unwelcome, uninvited sexual advance or request for a sexual favor, especially of a subordinate, a vendor or a customer,
 Drunkenness / Substance Abuse Using alcoholic beverages on County property, or using alcoholic beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 		or offensive work environment based on a protected characteristic.
 Substance Abuse beverages while engaged in County business. Possessing an open container of alcohol on County property or while engaged in County business. 	Drunkenness /	
 Possessing an open container of alcohol on County property or while engaged in County business. 	27.00.00000	
		Possessing an open container of alcohol on County property or while
• Manufacturing, using, sching, of distributing account of linest drugs on		
		wandacturing, using, sening, or distributing account of fincit drugs of

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 13.08 (3)

	•	County time or County property. Refusal or failure to submit to a drug test as required. Failing an alcohol or drug test in violation of the County's Fitness for
	•	Duty Policy. Tampering with a specimen or drug test including but not limited to the introduction of any foreign substance or specimen from another individual into or in place of the employee's specimen.
Job Abandonment	•	Failing to follow call-in procedures for three (3) consecutive work days of absence without approved leave.

Evaluation of Inappropriate Conduct

All inappropriate conduct shall be evaluated on a case by case basis, considering the following factors:

- 1) Level of disruption to County business.
- 2) Level of harm to the County's interests.
- 3) Level of damage to the public's trust and confidence in Cuyahoga County government.
- 4) The employee's position and the degree of responsibility inherent in that position (i.e., supervisory vs. non-supervisory positions). Employees who occupy a supervisory or management position are held to a higher standard of conduct commensurate with the level of leadership required of them.
- 5) Whether the employee's conduct is part of a continuing problem.
- Whether the employee's conduct put a co-worker, vendor or customer's safety and/or security at risk.
- 7) Whether the employee's honesty and veracity is brought into question because of his or her conduct.
- 8) Whether there are extenuating or mitigating circumstances.
- 9) Whether the inappropriate factor involves a violation of the County's Code of Ethics (*See* Section 3.).

Cuyahoga County Department of Human Resources



Section 13.08 (4)

13.09 Progressive Discipline Program

Purpose

The County subscribes to a policy of progressive discipline. Progressive discipline is not intended to be punitive. Rather, the goal of this policy is to help the employee recognize and correct certain unacceptable behavior before it becomes serious enough, or frequent enough, to warrant termination of employment.

Application

The Progressive Discipline Program applies to:

- Regular full-time and part-time employees (bargaining and non-bargaining); and
- Promotion or transfer probationary employees.

The Progressive Discipline Program may not apply to:

- New-hire probationary employees;
- Interns; and
- Temporary or seasonal employees
- Any employee who commits an offense deemed to be "removable" pursuant to <u>Section</u> 13.08.

In the event of a conflict, Federal, State and Local laws, rules, and regulations take precedence over Progressive Discipline policies and procedures. In addition, in the event of a conflict, collective bargaining agreements take precedence over Progressive Discipline policies and procedures.

Relationship to Attendance Control Program

All attendance related discipline shall be administered in accordance with the Attendance Control Program (Section 14). Attendance related discipline, however, may be relevant under this Section as evidence of a continuing performance problem. In addition, failure to follow call in procedures is considered a Progressive Discipline offense and may be addressed separately from the related attendance control issue.

Procedure

When a management representative determines that an employee has conducted himself or herself in a manner that is deemed "inappropriate" under <u>Section 13.08</u> of the Policies and Procedures Manual, the management representative shall consult with Human Resources to determine the appropriate action to take. If an investigation is necessary, Human Resources will work with the management representative to gather the necessary information regarding the inappropriate conduct.

If Human Resources determines that the facts of the case may justify the imposition of a suspension, demotion, or removal, Human Resources will have the management representative complete a request for Pre-Disciplinary Conference in accordance with <u>Section 13.11</u> of this Manual.

Cuyahoga County Department of Human Resources



Section 13.09 (1)

Policies and Procedures Manual

Depending on the nature of the discipline that is ultimately issued, the employee will either be notified of the discipline by his or her supervisor or Human Resources.

NOTE: All finalized disciplinary actions are permanently placed in the employee's personnel file. The disciplinary action, however, will only remain active for two (2) years for purposes of progressive discipline. The two-year period shall be measured backward from the date of the subsequent offense to the date the prior discipline was imposed.

Types of Disciplinary Action

- Verbal Reprimand- A verbal reprimand is an articulation of the problem by the supervisor to the employee. This serves as a warning and may be administered before a supervisor applies stronger action. A verbal reprimand shall note the date and nature of the reprimand as well as specifically state that the employee is receiving a formal warning. A written record of the reprimands will be made and such will be made a part of the employee's personnel file. Such records should include the time, date, a brief description of the problem and direction given to the employee.
- Written Reprimand The purpose of a written reprimand is to give an employee formal written notice that his or her conduct has been deemed to be inappropriate and of any action that may be necessary to correct said conduct. The written reprimand should also specify consequences if the employee does not correct his or her inappropriate conduct. If Human Resources determines that a written reprimand is appropriate pursuant to the Progressive Discipline Program, the supervisor or management representative shall prepare the written reprimand and submit it to Human Resources for review. Once approved by Human Resources, the supervisor or management representative will give the written reprimand to the employee. The written reprimand shall be signed by the supervisor or management representative, the employee and any third-party who witnessed the issuance of the written reprimand. The employee's signature is an acknowledgement of receipt of the form, not agreement with its contents. The original written reprimand shall be sent to Human Resources to be placed in the employee's personnel file, with copies to the employee and the employee's direct supervisor.
- Working Suspension A working suspension results when the County requires an
 employee who is suspended to report to work to serve the suspension. An employee
 serving a working suspension shall continue to be compensated at the employee's
 regular rate of pay for hours worked. The disciplinary action shall be recorded in the
 employee's personnel file and has the same effect as a suspension without pay for the
 purpose of recording disciplinary actions. Working suspensions may be applied to
 situations other than attendance control issues at management's discretion with
 consultation from HR.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 13.09 (2)

- Suspension A suspension is a forced unpaid leave of absence from employment
 with the County. The County may require an employee who is suspended to report to
 work to serve the suspension. An employee serving a suspension in this manner shall
 continue to be compensated at the employee's regular rate of pay for hours worked.
 The disciplinary action shall be recorded in the employee's personnel file and has the
 same effect as a suspension without pay for the purpose of recording disciplinary
 actions.
- Reduction See Section 5.15.
- Demotion See Section 5.14.
- Removal See Section 5.17.

NOTE: At the sole discretion of the Director of Human Resources or designee, an employee may be temporarily placed on paid administrative leave when the employee's inappropriate behavior is so serious that immediate removal from the workplace is necessary. This may be necessary because the employee's actions indicate that remaining on the job or returning to the job may be detrimental to the employee, co-workers, customers or the County. The employee will be required to leave County property pending an investigation into the employee's behavior. Examples of behavior that may require a placement on paid administrative leave include, but are not limited to, the "Removable Offenses" set forth in <u>Section 13.08</u> of this Manual. Employees who are placed on paid administrative leave shall be prepared to return to work each day on leave and are required to call Human Resources prior to their regular starting time each day to determine whether they should return to work that day.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 13.09 (32)

13.10 Conduct/Performance Improvement Measures

The utilization of the measures described below can oftentimes help an employee improve his or her performance or conduct without having to initiate formal disciplinary procedures. Their purpose is to establish an understanding of the issues, exchange information, and establish job expectations. If the employee's performance or conduct does not improve, however, the supervisor shall initiate disciplinary proceedings in accordance with <u>Section 13.09</u> of this Manual.

Management is strongly encouraged to consult with Human Resources prior to making use of the following measures:

NOTE: The following measures are NOT considered disciplinary action for purposes of the Progressive Discipline Program. These measures, however, may be mandated in conjunction with disciplinary action under the Progressive Discipline Program. An employee's failure to fulfill a mandate pursuant to this section constitutes insubordination and may subject the employee to disciplinary action. For example, an employee that is being suspended for inappropriate conduct may also receive a mandatory referral to the Employee Assistance Program as part of the disciplinary order. The employee's failure to fulfill the mandated referral would subject them to further disciplinary action for insubordination.

Verbal Counseling

Verbal counseling involves a private meeting between supervisor and employee, whereby the employee is informed in detail of his or her conduct that the supervisor has deemed inappropriate and of any action that may be necessary to correct said conduct. The supervisor shall create a document detailing the reasons for and outcomes of the counseling. The supervisor and the employee shall sign the document. The supervisor shall retain the original and the employee shall receive a copy of the document. The employee's signature is an acknowledgement of receipt of the form, not agreement with its contents.

NOTE: "Verbal counseling" may also be referred to as "oral counseling" or "coaching".

Performance Improvement Plan ("PIP")

A performance improvement plan is a formal process used by supervisors to help employees improve performance or modify behavior. The performance improvement plan, or PIP, as it is often called, identifies performance and/or behavioral issues that need to be corrected and creates a written plan of action to guide the improvement and/or corrective action. A PIP may be used in conjunction with a verbal counseling, a disciplinary action or a poor performance review.

Fundamentally, a PIP is a structured communication tool designed to facilitate constructive discussion between the employee and the supervisor. An effective PIP will:

Cuyahoga County Department of Human Resources



Section 13.10 (1)

Policies and Procedures Manual

- Consider the employee's input;
- Specifically identify the performance to be improved or behavior to be corrected;
- Provide **reasonable and clear expectations** about the work to be performed or behavior that must change;
- Identify the support and resources available to help the employee make the required improvements;
- Establish a plan for reviewing the employee's progress and providing feedback to the employee for the duration of the PIP; and
- Specify consequences if performance standards as identified in the PIP are not met.

NOTE: A "performance improvement plan" may also be referred to as a "corrective action plan or "CAP".

Training

In an effort to further the professional development of its employees, the County will frequently offer a wide variety of training opportunities for its employees. Examples of the types of training that may be offered includes:

- Computer skills;
- Management/leadership training;
- Professionalism and business etiquette;
- Conflict management/resolution;
- Problem solving;
- Effective communication; and
- Project management.

Attendance at training sessions can be mandated by County management. An employee that fails to attend a mandated training session will be subject to disciplinary action, up to and including termination of employment.

Please contact Human Resources or refer to the County Intranet for more information regarding training opportunities.

Management Referral to Employee Assistance Program

Under certain circumstances, the County may deem it necessary to request that an employee seek assistance through the County's Employee Assistance Program (EAP) (See Section 7.03). An employee who receives a management EAP referral will be treated in the same manner as an employee who voluntarily seeks assistance through the EAP except that the employee's attendance, motivation level, and willingness to follow recommendations will be reported back to County management by the EAP administrator.

Cuyahoga County Department of Human Resources



Section 13.10 (2)

Policies and Procedures Manual

13.11 Pre-Disciplinary Conference (PDC)

If Human Resources determines that the severity of the employee's conduct may justify imposition of a suspension, demotion or removal, Human Resources will prepare a Notice of Pre-Disciplinary Conference for issuance to the employee.

At any point prior to the PDC, the employee may choose to waive his or her opportunity to have a PDC. Attendance at the PDC is mandatory unless waived. Upon written request, the presiding Human Resources' official may grant the employee a reasonable continuance, if necessary.

The following rules of procedure shall apply to the PDC:

- a) The employee may present oral and/or written statements, testimony, documents etc., in response to each allegation of inappropriate conduct.
- b) The employee may choose to have one additional person (other than a subordinate employee) accompany them to the PDC as an observer. The observer shall not be permitted to speak on the employee's behalf. It is the employee's responsibility to notify their chosen observer that their attendance is desired.
- c) If the employee is a member of a union, he or she may be accompanied by a union representative in accordance with the applicable <u>collective bargaining agreement</u>. It is the employee's responsibility to notify the union representative that their attendance is desired.
- d) The County reserves the right to record the PDC. The employee has no right to record the PDC, unless otherwise authorized in their <u>collective bargaining agreement</u>.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual

(4)

Sections 13.11

Last Revised: 4/5/11

Page 362 of 403

13.12 Appeals

Comment [c9]: Incorporated new appeals language

The Cuyahoga County Charter provides non-bargaining employees an appeals process for certain disciplinary action through the HRC. Revised Code <u>Section 124.34</u> sets forth the specific types of discipline that may be appealed and outlines the procedures for filing an appeal with the HRC.

Should an employee elect to appeal a disciplinary action, they must file their written appeal with the Cuyahoga County Human Resource Commission (HRC). The appeal must actually be received and date stamped by the HRC by the tenth calendar day from the date the disciplinary letter is delivered to the employee by hand or to their last known mailing address, as maintained by Cuyahoga County, whichever occurs first. Employees may file with the HRC by email at humanresourcecommission@cuyahogacounty.us, by facsimile at (216) 443-3694, in person by obtaining a date stamp, or by mail postmarked no later than the tenth calendar day from the date the disciplinary letter is delivered. It is advised that employees retain records of proof of service upon the HRC.

NOTE: Certain progressive discipline decisions may be appealed in accordance with the complaint process set forth in Section 17.03. Certain claims, such as claims of discrimination or harassment, do not fall under the jurisdiction of the HRC. If an action is not subject to HRC appeal, employees may elect to appeal these actions in accordance with the terms set forth in sections 4 or 17 of this policy manual.

Cuyahoga County Department of Human Resources

(4)

Policies and Procedures Manual

Section 14: ATTENDANCE CONTROL PLAN

<u>14.01</u> <u>Purpose</u>

Absenteeism and tardiness represent two of the most serious problems in county government. Absenteeism and tardiness disrupt work schedules, cause unnecessary overtime, and place an unfair burden on responsible employees and supervisors. The purpose of the Attendance Control Plan is to increase productivity and employee morale through the systematic reduction of employee absenteeism and tardiness.

14.02 **AWOL (Absent Without Official Leave)**

AWOL occurs when an employee is away from his or her scheduled place of work or is tardy for reasons other than authorized leave. AWOL and tardiness are work rule violations.

The mere procurement of a physician's statement will not prevent the accumulation of AWOL hours, unless the absence is covered by approved leave.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 14.01 & 14.02

14.03 Attendance Control Plan

Attendance control is administered separately from the Progressive Discipline Program in Section 13.09. Failure to follow departmental call-in procedures, however, is considered a separate offense which may be addressed under the Progressive Discipline Program in Section 13.09.

NOTE: An employee may receive separate disciplinary action for a failure to follow call-in procedures and for AWOL arising out of the same incident. For example, if an employee fails to follow call in procedures for a single day of absence without approved leave, the employee would be subject to discipline for a "minor infraction" under <u>Section 13.08</u> and would also receive eight (8) hours of AWOL time pursuant to the Attendance Control Plan.

The Attendance Control Plan shall be administered as follows:

Stage 1

If the employee accumulates ten (10) or more hours of AWOL in a two-year period, the employee shall receive an **AWOL Written Reprimand**. The supervisor and employee shall sign the AWOL Written Reprimand and the original shall be sent to Human Resources to be placed in the employee's personnel file, with a copy to the employee.

Stage 2

If the employee accumulates ten-twenty (2010) or more hours of AWOL in the-a two-year period after the date of the issuance of their Stage 1 AWOL Written Reprimand, the employee shall be subject to a **three-day working suspension**.

Stage 3

If the employee accumulates sixteen thirty-six (3616) or more hours of AWOL in the-a two-year period after the third day of their Stage 2 three day suspension, the employee shall be subject to removal.

Attendance Control Plan Summary Chart

	AWOL ACCUMULATION	DISCIPLINE
STAGE 1	10 hours in a 2-year period	Written Reprimand
STAGE 2	10-20 additional hours in the a two-year period after the date of the issuance of their Stage 1 AWOL Written Reprimand	3-day Working Suspension
STAGE 3	16-36 additional hours in the a_two-year period after the third day of their Stage 2 three day working suspension	Removal
Cuyahoga County Department of Hur Policies and Proce		Last Revised: 4/5/11

NOTE: Non bargaining staff who have accumulated over 36 hours of AWOL as a first or second offense shall be subject to removal.

Section 14.03

14.04 Appeals

Comment [c10]: Incorporated new appeals language

The Cuyahoga County Charter provides non-bargaining employees an appeals process for certain disciplinary action through the HRC. Revised Code <u>Section 124.34</u> sets forth the specific types of discipline that may be appealed.

Should an employee elect to appeal a disciplinary action, they must file their written appeal with the Cuyahoga County Human Resource Commission (HRC). The appeal must actually be received and date stamped by the HRC by the tenth calendar day from the date the disciplinary letter is delivered to the employee by hand or to their last known mailing address, as maintained by Cuyahoga County, whichever occurs first. Employees may file with the HRC by email at humanresourcecommission@cuyahogacounty.us, by facsimile at (216) 443-3694, in person by obtaining a date stamp, or by mail postmarked no later than the tenth calendar day from the date the disciplinary letter is delivered. It is advised that employees retain records of proof of service upon the HRC.

NOTE: Certain progressive discipline decisions may be appealed in accordance with the complaint process set forth in Section 17.03. Certain claims, such as claims of discrimination or harassment, do not fall under the jurisdiction of the HRC. If an action is not subject to HRC appeal, employees may elect to appeal these actions in accordance with the terms set forth in sections 4 or 17 of this policy manual.

Cuyahoga County Department of Human Resources



Policies and Procedures Manual

Section 14.04 Last Revised: 4/5/11

Section 15: ELECTRONIC EQUIPMENT AND COMMUNICATIONS

15.01 Electronic Equipment and Communications Policy

Internet/Intranet/Extranet-related systems, including but not limited to computer equipment, software, operating systems, storage media, network accounts providing electronic mail, World Wide Web browsing, FTP, cellular telephones, and "smart phones" are the property of Cuyahoga County. These systems are to be used for business purposes in serving the interests of the County, the public and agency customers in the course of normal operations. Access and use of County provided communication equipment and services are provided at the discretion of the County and may be revoked at will.

Effective security is a team effort involving the participation and support of every Cuyahoga County employee and affiliate who deals with information and/or information systems. It is the responsibility of every computer user to know these guidelines, and to conduct their activities accordingly.

The purpose of this policy is to outline the acceptable use of electronic equipment at Cuyahoga County. These rules are in place to protect the employee and Cuyahoga County. Inappropriate use exposes Cuyahoga County to risks including virus attacks, compromise of network systems, data, services, and legal liability issues.

This policy applies to employees, contractors, consultants, temporaries, and other workers at Cuyahoga County, including all personnel affiliated with third parties. This policy applies to all equipment that is owned or leased by Cuyahoga County.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 15.01

15.02 Privacy Expectations

County employees do not have a right, nor should they have an expectation, of privacy while using any County electronic equipment at any time, including accessing the Internet and/or using County owned/provided e-mail. By using County electronic equipment, County employees make express agreement to consent to disclose the contents of any type of information maintained on or passed through County electronic equipment. In addition, any record created by an employee when using County electronic equipment (e.g., e-mail record, internet usage history), is generally considered a public record subject to disclosure upon request.

By using County electronic equipment, consent to monitoring and recording is implied with or without cause, including, but not limited to, accessing the Internet and using County owned/provided e-mail. Any use of County communication resources is made with the understanding that such use is generally not secure, is not private, and is not anonymous.

All County provided electronic equipment, and its contents, may be monitored and inspected at any time without prior notice. Electronic communications may be disclosed within a Department to those who have a need to know in the performance of their duties. Department Directors, system managers, and supervisors may access any electronic communications at any time.

Cuyahoga County Department of Human Resources



Section 15.02

Policies and Procedures Manual Last Revised: 4/5/11

15.03 Security and Proprietary Information

The following guidelines are designed to protect County employees, partners and the public from illegal or damaging actions by individuals, either knowingly or unknowingly:

- 1. Keep passwords secure and do not share accounts. Authorized users are responsible for the security of their passwords and accounts. System level passwords should be changed quarterly, user level passwords should be changed every six months.
- 2. All PCs, laptops and workstations should be secured with a password-protected screensaver with the automatic activation feature set at 10 minutes or less, or by logging-off (control-alt-delete for Win2K users) when the host will be unattended.
- 3. Use encryption of information in compliance with Information Services Center's Acceptable Encryption Use policy. This policy may be found at: http://10.4.1.23/ccisc/pdf/policies/ISCSecurityPolicyProcedures 0309.pdf
- 4. Because information contained on portable computers is especially vulnerable, special care should be exercised. Protect laptops in accordance with good judgment and best practices in protecting county owned equipment.
- 5. Postings by employees from a Cuyahoga County email address to newsgroups should contain a disclaimer stating that the opinions expressed are strictly their own and not necessarily those of Cuyahoga County, unless posting is in the course of business duties.
- 6. All hosts used by the employee that are connected to the Cuyahoga County Internet/Intranet/Extranet, whether owned by the employee or Cuyahoga County, shall be continually executing approved virus-scanning software with a current virus database unless overridden by departmental or group policy.
- 7. Employees must use extreme caution when opening e-mail attachments received from unknown senders, which may contain viruses, e-mail bombs, or Trojan horse code.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 15.03

15.04 Prohibited Uses of Electronic Equipment and Communications

Prohibited use of County equipment and/or electronic communications may subject the violator to disciplinary action, up to and including removal. Prohibited usage may also expose the violator to criminal prosecution. Examples of prohibited uses of electronic equipment and communication are:

NOTE: Employees may be exempted from these restrictions during the course of their legitimate job responsibilities (e.g., systems administration staff may have a need to disable the network access of a host if that host is disrupting production services.)

System and Network Activities

The following activities are strictly prohibited, with no exceptions:

- 1. Violations of the rights of any person or company protected by copyright, trade secret, patent or other intellectual property, or similar laws or regulations, including, but not limited to, the installation or distribution of "pirated" or other software products that are not appropriately licensed for use by Cuyahoga County.
- 2. Unauthorized copying of copyrighted material including, but not limited to, digitization and distribution of photographs from magazines, books or other copyrighted sources, copyrighted music, and the installation of any copyrighted software for which Cuyahoga County or the end user does not have an active license is strictly prohibited.
- 3. Exporting software, technical information, encryption software or technology, in violation of international or regional export control laws, is illegal. The appropriate management should be consulted prior to export of any material that is in question.
- 4. Introduction of malicious programs into the network or server (e.g., viruses, worms, Trojan horses, e-mail bombs, etc.).
- 5. Sharing or revealing your account password to others or allowing use of your account by others. This includes friends, family and other household members when work is being done at home.
- 6. Using a Cuyahoga County computing asset to actively engage in procuring or transmitting material that is in violation of sexual harassment or hostile workplace laws in the user's local jurisdiction.
- 7. Making fraudulent offers of products, items, or services originating from any Cuyahoga County account.
- 8. Making statements about warranty, expressly or implied, unless it is a part of normal job duties.
- 9. Effecting security breaches or disruptions of network communication. Security breaches include, but are not limited to, accessing data of which the employee is not an intended recipient or logging into a server or account that the employee is not expressly authorized to access, unless these duties are within the scope of regular duties. For purposes of this section, "disruption" includes, but is not limited to, network sniffing, pinged floods, packet spoofing, denial of service, and forged routing information for malicious purposes.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 15.04 (1)

- 10. Port scanning or security scanning is expressly prohibited unless prior notification to the Research and Security Department is completed.
- 11. Executing any form of network monitoring which will intercept data not intended for the employee's host, unless this activity is a part of the employee's normal job/duty.
- 12. Circumventing user authentication or security of any host, network or account.
- 13. Interfering with or denying service to any user other than the employee's host (for example, denial of service attack).
- 14. Using any program/script/command, or sending messages of any kind, with the intent to interfere with, or disable, a user's terminal session, via any means, locally or via the Internet/Intranet/Extranet.
- Providing information about, or lists of, Cuyahoga County employees to parties outside Cuyahoga County.
- 16. Accessing inappropriate websites (e.g., pornography, gambling, etc.).

Email and Communications Activities

- 1. Sending unsolicited email messages, including the sending of "junk mail" or other advertising material to individuals who did not specifically request such material (email spam).
- 2. Any form of harassment via email, telephone or paging, whether through language, frequency, or size of messages.
- 3. Unauthorized use, or forging, of email header information.
- 4. Solicitation of email for any other email address, other than that of the poster's account, with the intent to harass or to collect replies.
- 5. Creating or forwarding "chain letters", "Ponzi" or other "pyramid" schemes of any type.
- 6. Use of unsolicited email originating from within Cuyahoga County's networks of other Internet/Intranet/Extranet service providers on behalf of, or to advertise, any service hosted by Cuyahoga County or connected via Cuyahoga County's network.
- 7. Posting the same or similar non-business-related messages to large numbers of Usenet newsgroups (newsgroup spam).

Weblogs ("Blogging")

- Blogging by employees using Cuyahoga County's equipment or systems is subject to the
 terms and restrictions set forth in this Policy. Use of Cuyahoga County's systems to
 engage in blogging is acceptable, provided that it is done in a professional and
 responsible manner, does not otherwise violate Cuyahoga County's policy, is not
 detrimental to Cuyahoga County's best interests, and does not interfere with an
 employee's regular work duties. Blogging from Cuyahoga County's systems is subject to
 monitoring.
- Cuyahoga County's Information Sensitivity policy also applies to blogging. As such, employees are prohibited from revealing any Cuyahoga County confidential or proprietary information, trade secrets or any other material covered by Cuyahoga County's Information Sensitivity policy when engaged in blogging.
- 3. Employees are prohibited from making any discriminatory, disparaging, defamatory or harassing comments when blogging or otherwise engaging in any conduct prohibited by Cuyahoga County's Non-Discrimination and Anti-Harassment policy.

Cuyahoga County Department of Human Resources



Section 15.04 (2)

Policies and Procedures Manual

- 4. Employees may also not attribute personal statements, opinions or beliefs to Cuyahoga County when engaged in blogging from Cuyahoga County systems. If an employee is expressing his or her beliefs and/or opinions in blogs, the employee may not, expressly or implicitly, represent themselves as an employee or representative of Cuyahoga County. Employees assume any and all risk associated with blogging, to include legal liability. Employees may blog freely from their home personal computers or other access outlets assuming legal liability at their own risk.
- Apart from following all laws pertaining to the handling and disclosure of copyrighted or export controlled materials, Cuyahoga County's trademarks, logos and any other Cuyahoga County intellectual property may not be used in connection with any blogging activity

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 15.04 (3)

Comment [jjs11]: New Section

The same standards, principles and guidelines that apply to Cuyahoga County employees in the performance of their assigned duties apply to online activities. The County expects that all employees will exercise personal responsibility whenever they participate in social media and recognizes that the proper use of social media is another important tool available to improve communications with the public.

Employees are not permitted to use personal computers for social media networking on behalf of the County unless they have been designated to do so by their Agency Director. Employees with such designations should remember County Electronic Equipment and Communications Policy in Section 15.01 of this manual, County Acceptable Use of Government Office Equipment, County Internet Policy and County policies regarding the use of communication tools when using social media in the performance of their assigned job duties.

Employees should be aware that what they post online (comments, images, videos) is in a domain no longer controlled by them, and that their shared experiences, thoughts, and comments are no longer solely accessed by the employee alone. There is no such thing as a "private" social media site. For example, while personal Facebook privacy settings may be adjusted so only friends can view content, those same friends can forward comments to anyone or take screenshots of posts. When in doubt about the appropriateness of content, it is best to refrain. Employees should think of social media in the same manner as writing on a wall in public with the addition of their name and address attached to everything written. There are no truly anonymous postings to social media.

The County has the right to monitor employees' social media use on County and personal equipment and will exercise its right as necessary. Users do not and should have no expectation of privacy. Social media is not a secure means of communication.

Employees are allowed to have personal social networking sites. These sites must remain personal in nature and be used to share personal opinions or non-work related information. This helps ensure a distinction between sharing personal and County views. In addition, *employees should never use their County e-mail account or password in conjunction with a personal social networking site* and employees should never refer or link back to their own personal site from the Cuyahoga County web site. Personnel should also remember Cuyahoga County Employee Conduct Policy in Section 13.01 and Cuyahoga County Ethics Policy in Section 3.01 of this Manual when using social media.

Cuyahoga County Department of Human Resources

(4)

Section 15.055

Policies and Procedures Manual

Section 16: TRAVEL / DRIVING ON COUNTY BUSINESS

16.01 Traveling on County Business

Employees may be required to travel on County business in conjunction with their normal job assignments or in order to fulfill a special departmental need. Any decision to travel must be made in accordance with the policies and procedures set forth in the Cuyahoga County Travel Policy. An employee may be denied reimbursement for their travel expenses and may be subject to discipline, up to and including removal, if they fail to follow the policies and procedures set forth in the Cuyahoga County Travel Policy. A copy of the Cuyahoga County Travel Policy is available on the County intranet. A copy may also be obtained by contacting Human Resources.

County employees who are required to travel are responsible for conducting themselves, at all times, in a manner that advances the goals of the County and increases public confidence in County government. This requires County employees to refrain from behavior that might be harmful to the County's interests, or which violates or conflicts with County policies, practices and/or procedures. County employees shall exercise good judgment while traveling on County business and shall observe all of the County's rules while traveling, including, but not limited to rules regarding:

- Personal appearance (<u>Section 13.02</u>);
- Alcohol and drug use (Section 7.01); and
- Sexual or other workplace harassment (Section 4.03).

Any failure to adhere to these requirements constitutes an infraction of the County Discipline Policy (Section 13), which may result in disciplinary action, up to and including removal from County employment.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 16.01

Last Revised: 1/24/12

Safety Belt Usage

The County recognizes that safety belts are an important item of personal protective equipment and that safety belts save lives and reduce the severity of injuries to those who wear them. It is the County's commitment to do everything reasonable to prevent injuries to employees and damage to property and to protect the County, its employees and the general public from the results of vehicle accidents.

This policy applies to all County employees and to all occupants of vehicles driven by employees on County business. Occupants shall use safety belts in all vehicles driven on County business (whether County owned or privately owned vehicles). It is especially important that all employees demonstrate their commitment to and support of this policy by their strict adherence to it.

Any employee who is cited by a law enforcement agency for not wearing a safety belt will be responsible for any fines or other actions that may result as part of the citation. Employees who violate this policy may be subject to disciplinary action, up to and including removal.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 16.02

16.03 Driving on County Business / Use of County Fleet Vehicles

Every employee who drives or operates a County fleet vehicle at any time, or who operates <u>any</u> motor vehicle (i.e., employee's personal vehicle, rental vehicle) on, or in the course of County business must strictly adhere to the following requirements:

- The employee must have and maintain at all times, without interruption, a <u>valid</u> <u>driver's license</u> and the minimum <u>automobile insurance coverage</u> required by Ohio law. Employees are solely responsible to make sure that their licenses and automobile insurance are properly renewed/maintained.
- Employees are responsible to make sure that the <u>Ohio Bureau of Motor Vehicles</u> (<u>BMV</u>) has the employee's correct mailing address. Employees can notify the BMV of an address change by visiting the BMV website (http://www.ohiobmv.com/). If the BMV sends notification of a license suspension or other mailing to the address in its records, the employee is deemed to be properly notified and held responsible for having knowledge of the suspension or other matter.
- If the employee's <u>driver's license and/or insurance is expired, suspended, revoked, or otherwise invalid</u>, the employee shall <u>immediately report</u> this fact to their immediate supervisor, and <u>immediately stop driving</u> on, or in the course of County business. The employee's Department Director, with the concurrence of the Director of Human Resources, may allow the employee to resume driving on, or in the course of County business upon the employee providing acceptable proof of insurance coverage and driver's license (or appropriate occupational driving privileges granted by a court of competent jurisdiction). Other conditions may be imposed as appropriate in light of the circumstances of each individual case.

NOTE: Driving records are public information. County management has the authority and does conduct **routine inspections** of the driving records of its employees. Employees who are required to drive on, or in the course of County business are encouraged to periodically visit the Ohio BMV website (http://www.ohiobmv.com) to monitor the status of their license.

- All occupants (including non-employees) of any motor vehicle being used on, or in the course of County business are required to wear a **safety belt** at all times.
- Employees who are required to transport children on, or in the course of County business must follow all applicable Ohio laws regarding the use of <u>car seats</u> or other restraints.
- Employees must follow all <u>traffic laws and parking regulations</u>. Employees are solely responsible for the cost of any driving/moving infraction/violation, parking

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 16.03 (1)

tickets, impound charges, towing charges, and/or storage charges incurred while driving a County fleet vehicle or any vehicle on, or in the course of County business.

- The use of <u>alcohol</u> and/or other <u>controlled substances</u> including a prescription or over the counter medication, which may temporarily render an employee unable to operate a vehicle safely is strictly prohibited.
- An employee involved in a <u>motor vehicle accident</u> must strictly adhere to the vehicle accident reporting requirements set forth in <u>Section 16.04</u> of this Manual.

NOTE: It is strongly recommended that an employee operating a vehicle pull to a safe location and park the vehicle **before** using a cell phone. It should also be noted that several municipalities strictly prohibit the use of a cell phone while driving.

Use of County Fleet Vehicle

Immediately upon becoming aware of the need for the use of a County fleet vehicle, a County employee should contact the Fleet Services Department at (216) 443-8952 to determine if a vehicle is available on their travel date. If a vehicle is available, the employee should submit a Vehicle Request Form to the Fleet Services Garage via facsimile at (216) 443-7806.

A County employee that operates a County fleet vehicle must adhere to the following <u>additional</u> requirements:

- Personal use of a fleet/pool vehicle is strictly prohibited.
- Fleet/pool vehicles may only be used during the **employee's regular working hours** or as approved by the employee's department management.
- Fleet/pool vehicles shall only be **operated by County employees**.
- <u>Non-County employees</u> shall not travel in a fleet/pool vehicle unless specifically authorized by the employee's department management.
- **Smoking** is strictly prohibited inside a fleet/pool vehicle.
- Employees must comply with the Cuyahoga County <u>Vehicle Idling</u> Policy. A copy of the Idling Policy can be obtained by contacting the Fleet Services Department or by logging onto the County intranet.
- Fleet/pool vehicles may not be driven <u>out-of-state</u> without prior approval by the County.
- Fleet/pool vehicles may never be driven <u>out of the country</u>.
- Fleet/pool vehicles and keys must be <u>returned</u> by the specific date and time given and in the condition that they were received.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 16.03 (2)

Use of Personal Vehicle

A County employee who is required to drive or operate their personal vehicle on, or in the course of County business must adhere to the following additional requirements:

- The employee's personal vehicle's <u>automobile registration</u> must be current. If the employee's personal vehicle's automobile registration expires or is otherwise invalid, the employee must immediately notify his or her supervisor and immediately stop driving in the course of County business. The employee's Department Director, with the concurrence of the Director of Human Resources, may allow the employee to resume driving their personal vehicle on, or in the course of County business upon the employee providing acceptable proof of valid automobile registration.
- An employee on, or in the course of County business <u>may not be transported by a</u> <u>non-County employee unless approved by the employee's supervisor.</u>
- Every employee who submits a <u>Travel Expense Report</u> for operating a personal vehicle on, or in the course of County business, certifies that he or she has a valid driver's license, vehicle registration, financial responsibility (insurance), and personally operated the vehicle for which the travel expense is claimed.
- The employee must maintain the vehicle in a good and safe operating condition.
- Smoking is strictly prohibited within an employee's personal vehicle when transporting other employees and/or the public throughout the course of county business.

Employees who are required to drive their personal vehicle on, or in the course of County business should refer to the policies and procedures set forth in the Cuyahoga County Travel Policy for information regarding mileage and parking reimbursement. A copy of the Cuyahoga County Travel Policy is available on the County intranet. A copy may also be obtained by contacting Human Resources. Any failure to conform to the requirement set forth in this Section constitutes an infraction of the County Discipline Policy (Section 13), which may result in disciplinary action, up to and including removal from County employment.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 16.03 (3)

16.04 Vehicle Accident Reporting Procedure

For purposes of this Section, a "vehicle accident" is an accident that results in property damage and/or physical injures to any parties involved in the accident.

Employee Duties

An employee involved in a vehicle accident either, (1) occurring in the course of County business in a private/rental vehicle, or (2) occurring in a County vehicle at any time, <u>must</u> follow the following steps:

- 1) Remain at the accident site unless it is necessary to immediately go to hospital.
- 2) Contact the appropriate authorities (i.e., dial 911).
- 3) Contact County Protective Services at 216-443-2141 or 1-888-814-3578.
- 4) Contact their immediate supervisor.

After such accidents, the County employee may be required to submit to an alcohol and drug test depending on the circumstances. (*See* Section 7.04 for more information on County drug/alcohol testing policies and procedures.) Therefore, the employee shall remain at the accident site until they receive further instruction from either their supervisor, Protective Services or the Fitness for Duty Coordinator.

NOTE: An employee who is incapacitated and/or transported by EMS to the hospital as a result of a vehicle accident shall contact their immediate supervisor as soon as practicable.

Supervisor Duties

Upon receipt of notification of a vehicle accident involving a County employee, the supervisor shall immediately notify County Protective Services (216-443-2141).

Any failure to conform to the requirement set forth in this Section constitutes an infraction of the County Discipline Policy (Section 13), which may result in disciplinary action, up to and including removal from County employment.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Section 16.04

Section 17: EMPLOYEE COMPLAINT PROCEDURES

17.01 Equal Employment Opportunity Complaints

See Section 4.05 for information regarding the Equal Employment Opportunity Complaint procedure.

17.02 Fair Labor Standards Act (FLSA) Complaints

Non-bargaining employees shall use the following complaint procedure for disputes regarding alleged violations of the FLSA (*see* Section 6.01). Employees covered by a collective bargaining agreement shall use the grievance procedure outlined in their agreement.

An employee who has been designated by the County as being exempt from receiving overtime and/or compensatory time may file a written complaint if he or she believe that:

- He or she has been improperly designated as an exempt employee; or
- His or her paycheck has been improperly docked.

The written complaint should outline why the employee should not be considered exempt or the specific dates, amounts and circumstance regarding when the alleged improper docking occurred. This written complaint should be submitted to the County Human Resources Director for review.

Upon receipt of the complaint, the Human Resources Director or designee will investigate the allegation. After review, if the employee has been improperly designated or docked, the County will re-designate, reimburse and/or correct the wages of the employee. The decision issued pursuant to this section will be the final decision of the County.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 17.01 & 17.02

17.03 General Complaints

Policy

For all complaints not involving equal employment or FLSA issues, County employees are encouraged to make every reasonable effort to resolve such complaints with his or her immediate supervisor. If the complaint cannot be resolved informally, employees may file a County General Complaint pursuant to the procedure outlined in this Section.

Limitations

Certain issues are not subject to the County General Complaint Procedure. Examples include, but are not limited to:

- Administration of disciplinary action that entitles the employee to a right of appeal to the HRC;
- Removals, reductions or demotions during the new-hire probationary period;
- Collective bargaining issues (<u>bargaining unit employees should refer to the grievance</u> process set forth in their collective bargaining agreement);
- Position audit results;
- Temporary transfers of thirty (30) days or less;
- Layoffs;
- Disability separation decisions
- Performance evaluations; and
- Decisions based on fitness for duty examinations.

Procedure

STEP 1: WRITTEN COMPLAINT TO DIRECTOR

- Employees wishing to file a complaint under this Section shall document his or her complaint in writing and submit it to his or her Department Director within ten (10) working days of the event upon which the complaint is based. The complaint must specifically state the remedy the employee is seeking.
- The Department Director and/or designee will investigate the complaint and/or discuss the issue with the employee.
- The Department Director will issue a written response within a reasonable period of time, generally not to exceed thirty (30) calendar days.

STEP 2: APPEAL TO DIRECTOR OF HUMAN RESOURCES

- If the employee is not satisfied with his or her Department Director's response, the employee may request review by the Director of Human Resources.
- The employee shall send his or her original complaint and the Department Director's response to the Director of Human Resources within ten (10) working days of receipt of the Department Director's response.
- The Director of Human Resources and/or designee will investigate the complaint and/or discuss the issue with the employee.

Cuyahoga County
Department of Human Resources



Section 17.03 (1)

Policies and Procedures Manual

• The Director of Human Resources and/or designee will issue a written response to the complaint within a reasonable period of time, generally not to exceed thirty (30) calendar days.

STEP 3: APPEAL TO THE COUNTY EXECUTIVE/DESIGNEE

- If the employee is not satisfied with the response submitted by the Director of Human Resources, the employee may request review by the County Executive/Designee.
- The employee shall send their original complaint and both prior responses to the County Administrator within ten (10) working days of receipt of the Director of Human Resources' response.
- The County Executive and/or designee will investigate the complaint and/or discuss the issue with the employee.
- The County Executive or designee will render a final decision on the appeal within a reasonable period of time, generally not to exceed forty-five (45) calendar days.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual

(\$

Section 17.03 (2)

Section 18: RECORDS MANAGEMENT

18.01 Confidential Information

The County prohibits the release of confidential information, which consists of any information that is required by state or federal law or state or federal administrative rule to be kept confidential.

In addition, some information is specifically exempt from public inspection and copying under state public records law but is not required to be kept confidential. Release of such information is permitted only by any of the County's designated public records managers, and release of such information by anyone else is a violation of this section.

An employee who violates this section may be subject to disciplinary action, up to and including removal.

18.02 Compliance with Records Retention Schedule

Each Department has established a policy to ensure that all County records are retained in compliance with all known local, state, and federal laws and regulations, all Ohio Historical Society requirements, and the internal operating needs of the County. Before disposing of any County records, employees must determine whether the disposal is in accordance with the applicable Records Retention Policy adopted by the employee's Department.

An employee should contact his or her Department's designated public records manager with any questions regarding records retention.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 18.01 & 18.02

18.03 Personnel Records

A personnel file on each employee is kept in the Human Resources Department's office. It contains information regarding training, experience, and employment as well as the work history of the employee with the County. An employee shall have the right to inspect his or her personnel file provided reasonable notification is given to the Human Resources Department.

It is the responsibility of the employee to immediately notify the Human Resources Department of any changes in address, telephone number, emergency information, name change and marital status. Employees are required to have a current permanent residence address and contact number on file at all times with the Human Resources Department. Post Office boxes are not sufficient for this purpose. Name and marital status changes require legal documentation (e.g., marriage certificate, divorce decree, etc.), which will also be provided to the Personnel and Benefits Divisions.

18.04 Electronic Mail Records

Employees using electronic mail in the normal course of the County's business are responsible for adhering to the rules established for electronic mail (see <u>Section 15</u>). Electronic mail is to be used judiciously as such for conducting the business of the County. The use of e-mail for improper purposes, such as intimidation, harassment, non-County business, etc., may subject the employee to disciplinary action, up to and including removal.

There is no expectation of privacy for County owned/provided e-mail accounts. The County, without notice to employees, reserves the right to routinely and randomly monitor and/or access any employee's County owned/provided e-mail account. In addition, any record created or received by an employee when using County owned/provided e-mail accounts is generally considered a public record subject to disclosure upon request.

NOTE: See Section 15 for more information on the County's Electronic Equipment and Communications Policy.

All e-mail sent or received through County owned/provided e-mail accounts is the property of Cuyahoga County.

The County has established a policy that electronic mail messages are records of the County and that they are subject to all established rules concerning records retention and management. The deletion of e-mail messages shall be done in accordance with the applicable Records Retention Policy adopted by the County.

Cuyahoga County Department of Human Resources

Policies and Procedures Manual



Sections 18.03 & 18.04

18.05 Complying with Requests to Inspect and Copy County Records

In cooperation with the County's designated public records managers, all County employees shall maintain public records so that they are readily available for inspection and copying and shall make public records available for inspection and provide copies of public records in accordance with the County's Public Records Policy, which is as follows:

Public Records Policy of Cuyahoga County, Ohio

Chapter 1: Rationale

Believing that open government leads to a better-informed citizenry, greater public participation in government, better government, better public policy, and more effective use of public resources, the County Council establishes this Public Records Policy to insure the preservation and public accessibility of records relating to all functions of Cuyahoga County government. Cuyahoga County's policy in all of its functions is to strictly adhere to all of its obligations under Ohio's Public Records Law and to exceed those obligations whenever it is practical and makes sense to do so.

Chapter 2: Definitions

Section 2.1: "Committee" shall include subcommittees.

Section 2.2: "Public office" includes the following:

- (A) the office of the County Executive
 - (B) the Cuyahoga County Council, including all of its committees
 - (C) all departments, divisions, offices, or other organized bodies operating under the administration of the County Executive,
- (D) the Cuyahoga County Prosecuting Attorney,
- (E) all Boards, Commissions, and Advisory Councils to which the County Executive and/or the County Council appoint at least a majority of its members,
- (F) all Boards, Commissions, Advisory Councils and any similar body created by the Cuyahoga County Charter, the County Council, and/or the County Executive.

Section 2.3: As used in Section 2.2, a "similar body" must be formally organized, be on-going, and be involved in making or advising on public policy decisions.

Section 2.4: "Public record" includes any document, device, or item, regardless of physical form or characteristic, including electronic records, created or received by or coming under the jurisdiction of any public office, which serves to document the organization, functions, policies,

Cuyahoga County
Department of Human Resources



Section 18.05 (1)

Policies and Procedures Manual

decisions, procedures, operation, or other activities of the office. All records which meet this definition are public records, unless exempted under section 149.43 of the Ohio Revised Code.

Section 2.5: "Electronic record" includes prepared documents such as word processing documents, spreadsheets, and graphic presentations as well as written electronic communications, including but not limited to electronic mail and text messages.

Chapter 3: Scope of Ordinance

Section 3.1: Applicability

This ordinance is hereby adopted as the public records policy, required under Section 149.43 of the Ohio Revised Code, for every public office in Cuyahoga County government, as public office is defined in Section 2.2 of this ordinance, over which the Cuyahoga County Council has legislative authority.

Chapter 4: County Records Commission

Section 4.1: Creation of County Records Commission

There is hereby created, the County Records Commission, which shall succeed and replace the current County Records Commission. The County Records Commission shall consist of the County Executive, who shall serve as chairperson, the President of County Council, the Prosecuting Attorney, the Fiscal Officer, and the Clerk of Court of Common Pleas.

Section 4.2: Appointment of Designees

Each member of the County Records Commission shall appoint one designee, an employee of his/her office who is knowledgeable about the maintenance of public records, who shall attend meetings of the County Records Commission whenever the member is unable to do so.

Section 4.3: Powers and Duties of County Records Commission

The County Records Commission shall have all the powers and perform all the duties of County Records Commissions provided for in Section 149.38 of the Ohio Revised Code, including, but not limited to, the power to recommend applications for one-time disposal of records or proposed records retention schedules to the Ohio Historical Society.

Section 4.4: Rules and Procedures of the County Records Commission

The County Records Commission shall adopt its own rules and procedures, which shall be consistent with Section 149.38 of the Ohio Revised Code. Until such time as the County Records Commission adopts such rules, it shall operate under the rules of the prior County Records Commission that existed prior to the enactment of this ordinance.

Section 4.5: Meetings of the County Records Commission

The County Records Commission shall meet at the call of the chair as often as needed to respond to proposed records retention schedules and proposed one-time disposals of records, but shall meet a minimum of once every six (6) months.

Cuyahoga County
Department of Human Resources

Policies and Procedures Manual



Section 18.05 (2)

Section 4.6: Within sixty (60) days after receiving a request for one-time disposal of records or a proposed records retention schedule from any office, the County Records Commission shall either approve the request and send it to the Ohio Historical Society for its consideration or return the request disapproved to the office that submitted it with a letter stating the reasons for disapproval.

Chapter 5: Public Records

Section 5.1: Maintenance of Public Records

All public offices within the scope of this ordinance shall organize and maintain all their public records so that they are readily available for inspection and copying in accordance with the Ohio Public Records Law and the Public Records Policy of Cuyahoga County.

Section 5.2: Maintenance of Electronic Mail

All electronic mail sent or received through the Cuyahoga County Information Services Center shall be the property of Cuyahoga County. The necessity to maintain electronic mail as public records shall depend on the content of the records, not on the medium in which it is kept. It shall be the responsibility of the public records managers and each individual user to insure that electronic mail is maintained in accordance with the records retention schedule for each office, and that records which must be kept for an extended length of time will not be placed in files where they will be automatically deleted.

Each person covered by this ordinance shall also insure that all public record electronic mail sent or received outside of the County Information Service Center system are maintained so that they are readily available for inspection and copying in accordance with the records retention schedule for each office.

Section 5.3: Designation of Countywide Public Records Manager

The County Archivist is hereby designated as the Countywide Public Records Manager and shall do the following:

- (A) Manage the public records of Cuyahoga County to insure that they are organized so as to be readily available to the public for inspection and copying and are maintained and disposed of in accordance with the records retention schedules of the various offices within Cuyahoga County government.
- (B) Assist the public records managers of the various public offices in Cuyahoga County in implementing a sound and consistent countywide public records process in accordance with this ordinance.
- (C) Assist the public offices within the scope of this ordinance in preparing and updating public records retention schedules.

Section 5.4: Designation of Deputy Countywide Public Records Manager

The County Executive shall designate one employee of the executive office staff as Deputy County Public Records Manager, who shall do the following:

- (A) Assist the Countywide Public Records Manager in the performance of his/her duties;
- (B) Maintain the public records request log provided for in Section 6.5 of this ordinance;

Cuyahoga County
Department of Human Resources



Section 18.05 (3)

Policies and Procedures Manual

- (C) Work with each public office to determine what kinds of public records requests are received by that office that relate to personal or business matters, rather than governmental operations, and are not required to be included in the public records request log;
- (D) Provide a copy of Cuyahoga County's Public Records policy to each public records manager and obtain a written acknowledgement from each records manager that the policy was received; and
- (E) Serve as the public records manager for the County Executive's office.

Section 5.5: Designation of Public Records Managers

- (A) Each public office shall designate a public records manager who shall be responsible for the maintenance of the public records for that office and for handling public records requests directed to that office.
- (B) For the following offices, the person designated as public records manager shall be an employee of the office who works at the principal place at which that office does business:
 - (1) the County Executive, including all executive office staff
 - (2) the County Council
 - (3) the Law Department
 - (4) the Sheriff
 - (5) the Medical Examiner
 - (6) the Clerk of Courts
- (7) the Department of Economic Development, which shall include the Office of Regional Collaboration
 - (8) the Information Officer
 - (9) the Department of Public Works
 - (10) the Department of Purchasing
 - (11) the Department of Human Resources
 - (12) the Fiscal Office
 - (13) the Child Support Enforcement Agency
 - (14) the Division of Children and Family Services
 - (15) the Division of Employment and Family Services
 - (16) the Division of Senior and Adult Services
- (17) the Director of Human Services, which shall include the Director's Office and all other offices in the Department of Human Services not covered by items twelve (12) through fifteen (15) above.
 - (18) the Inspector General
- (C) Each office not listed in Subsection B of this section shall designate a public records manager who shall be one of the following:
- (1) an employee of that office or an officer of a Board, Commission, or Advisory Council,
 - (2) the Deputy County Public Records Manager, or
- (3) an employee of Cuyahoga County, approved by the Deputy County Public Records Manager, who is the public records manager for another office with responsibilities related to those of the designating office.

Cuyahoga County Department of Human Resources



Section 18.05 (4)

Policies and Procedures Manual

Section 5.6: Public Records Manager for County Council

The Clerk of Council is hereby designated as public records manager for Cuyahoga County Council.

Section 5.7: Records Retention Schedules

Each public office shall have a records retention schedule in place, which shall specify, consistent with state law, the methods by which and the length of time that records shall be kept. For any office that has a records retention schedule in place at the time that this ordinance becomes effective, that records retention schedule shall remain in effect until it is amended according to the procedure set forth in Section 149.38 of the Ohio Revised Code. Each public office that does not have a records retention schedule in place at the time that this ordinance becomes effective shall propose a public records retention schedule to the County Records Commission, in accordance with the procedure set forth in Section 149.38 of the Ohio Revised Code, not later than June 30, 2011.

Section 5.8: Interim Transient Records Retention Schedule

Except to the extent that a different records retention schedule on transient communications is required for an office by state law, each public office that does not have a records retention schedule on transient communications in place at the time that this ordinance becomes effective shall use the transient records retention policy and schedule adopted by the County Commissioners on January 12, 2009, until such time as the office's records retention schedule on transient communications is updated, according to the procedure set forth in Section 149.38 o the Ohio Revised Code.

Section 5.9: Publication of Public Records Policy

- (A) Each public office having public office space shall prepare a poster which shall describe the public records policy of that office, explain how to obtain public records, and name the public records manager for that office. The poster shall be displayed in a conspicuous place at the office and at any branch office where the office conducts business. Each office shall post the same information and its public records retention schedule on its web-page on the county's website.
- (B) The County Executive and the Clerk of Council shall each post on their respective webpages the full Public Records Policy of Cuyahoga County, a summary of that policy, instructions on how to obtain public records, and a list of all of the public records managers for Cuyahoga County government and their contact information, and the public records retention schedule for each office.
- (C) The manual of general policies and procedures issued to all employees shall include the county's public records policy.

Section 5.10: County Website

The County shall maintain a readily accessible website, which shall include separate pages for the County Executive, the County Council, and each department in County government. The County Executive and the Clerk of County Council shall insure that the website is regularly updated to provide current information, including the notice, agenda, minutes, and reports of all

Cuyahoga County
Department of Human Resources



Section 18.05 (5)

Last Revised: 4/5/11

Policies and Procedures Manual

public meetings conducted by offices within the scope of this ordinance and instructions on how to obtain public records.

Chapter 6: Public Records Requests

Section 6.1: Form of Records Request

Any person requesting public records shall identify the records requested with sufficient clarity to allow the public office to identify, retrieve, and review the records requested. No specific language or form is required to make the request. If the records request is not sufficiently clear, the public records manager must contact the requestor for clarification, and should assist the requestor by providing information about the manner in which the office keeps its records.

Section 6.2: Identity of Requestor Not Required

The requestor is not required to put a public records request in writing and does not need to provide his/her identity or the intended use of the records requested. The public office may request this information, particularly to aid in complying with the request, but must clearly state that providing this information is voluntary.

Section 6.3: Availability of Records

Records shall be made available promptly for inspection or copying. Public records requests shall be given priority attention in any office receiving them, but reasonable time shall be allowed to comply with requests that are large, involve records stored other than at the site where the request was made, or involving records that must be inspected for possible redaction of information exempt from the public records law. Whenever a request is received that cannot be complied with immediately, the public office shall provide the requestor a receipt acknowledging and describing the public records request and may provide an estimate as to when a response can be provided.

Section 6.4: Denial of Request and Redaction of Records

Any denial of records requested must include a written explanation, including legal authority. If part of a record requested is exempt from public records law, that part shall be redacted, and the remainder provided. Each redaction shall be accompanied by a written explanation, including legal authority.

Section 6.5: Public Records Request Log

Each office shall maintain a log of all public records requests received that relate to governmental operations and shall forward a copy of the log at the end of each week to the Deputy Countywide Records Manager, who shall maintain a countywide public records log. Each office shall, with the approval of the Deputy Countywide Public Records Manager, determine what kinds of public records requests received by that office relate to personal or business matters, rather than governmental operations; and these requests shall not be required to be included in the log. For each public records request required to be included in the public records request log, the following information shall be provided:

- (A) the office that received the request
- (B) the date that the request was received,

Cuyahoga County Department of Human Resources



Section 18.05 (6)

Policies and Procedures Manual

- (C) the name of the requestor, if known
- (D) a brief description of what records were requested,
- (E) the date that response to the request was completed,
- (F) a brief description of any denials or redactions required.

Chapter 7: Costs of Public Records

Section 7.1: Persons requesting copies of public records shall be required to pay for the cost of making copies, at a rate not to exceed the actual cost of making copies. Payment in advance may be required.

Starting in calendar year 2011, the County Council shall biannually determine and establish the copying costs for public records.

Section 7.2: Except as otherwise provided by court order, the following copying costs shall apply until the County Council first determines and establishes copying costs for public records:

- (A) The charge for paper copies shall be three cents (\$.03) per page. The charge shall be waived when less than one dollar (\$1.00).
- (B) The charge for computer files downloaded to a compact disc shall be the actual cost, not to exceed \$1.26 per disc. The charge shall be waived when only one (1) disc is required to fulfill the request.
 - (C) There shall be no charge for copies provided by email.
- (D) Each public office shall have the discretion to waive copying costs of ten dollars (\$10.00) or less when it determines that it is practical and cost-effective to do so, provided that the public office follows a consistent policy on waiver of copying fees for all requestors.

Cuyahoga County Department of Human Resources



Section 18.05 (7)

County Council of Cuyahoga County, Ohio

Ordinance No. O2013-0004

Sponsored by: County Executive	An Ordinance amending Ordinance No.		
FitzGerald/Department of Health	O2011-0045, as amended, which		
and Human Services	established divisions and sections within		
	the Department of Health and Human		
	Services, to create the Office of the		
	Director and to delineate displacement,		
	reinstatement and reemployment		
	procedures; and declaring the necessity		
	that this Ordinance become immediately		
	effective.		

WHEREAS, Article VIII, Section 8.01 of the Charter of Cuyahoga County created the Department of Health and Human Services, and provides that the Department of Health and Human Services "shall administer all programs and activities for which the County has or has assumed responsibility for the protection and enhancement of the health, education and well-being of County residents; and

WHEREAS, under the prior form of government these services were performed by four separate agencies, namely the Department of Employment and Family Services, the Cuyahoga Support Enforcement Agency; the Department of Children and Family Services; and the Department of Senior and Adult Services; and

WHEREAS, the work performed by the four above-named agencies required specialized knowledge and skills; and

WHEREAS, Article III, Section 3.09(2), of the Charter grants Council the power to establish departments, and divisions and sections within departments, under the supervision of the County Executive as the Council determines to be necessary for the efficient administration of the County; and

WHEREAS, Ordinance No. O2011-0045 established as separate divisions within the Department of Health and Human Services the following: the Division of Employment and Family Services/Cuyahoga Support Enforcement Agency; the Division of Children and Family Services; the Division of Senior and Adult Services; and the Division of Community Initiatives; and subsequent Ordinance No. O2012-0029 renamed the Division of Employment and Family Services/Cuyahoga Support Enforcement Agency as Cuyahoga Job and Family Services; and

WHEREAS, Ordinance No. O2011-0045 authorizes the Director of Health and Human Services to employ additional deputies, assistants and employees as shall be conducive to the efficient performance of the duties of the Department of Health and Human Services; and

WHEREAS, the establishment of an Office of the Director, within the Department of Health and Human Services, to employ additional employees, officers, and assistants, will benefit the operations of the Department; and

WHEREAS, the diversity of programming needs and issues under the new Department of Health and Human Services has not changed since the Charter consolidated the previously-separate departments, and continues to require specialized knowledge and skills; and

WHEREAS, the equalization and delineation of displacement rights within the Department of Health and Human Services is necessary to maximize the specialized knowledge and skills, minimize service disruption, and make better use of County resources; and

WHEREAS, Article XVIII, Section 3 of the Ohio Constitution grants Cuyahoga County broad home rule powers to adopt rules, as designated by ordinance, as the Council determines to be necessary for the efficient administration of the County; including ordinances at variance with state statute; and

WHEREAS, it is necessary that this Ordinance become immediately effective in order that critical services provided by Cuyahoga County can continue and to provide for the usual, daily operation of a County department.

NOW, THEREFORE, BE IT ENACTED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. The Office of the Director is hereby established within the Department of Health and Human Services.

SECTION 2. Displacement rights following the abolishment of a position within the Department of Health and Human Services are hereby delineated and wholly defined as displacement within the Division to which each employee is assigned (i.e., the Division of Children and Family Services; the Division of Senior and Adult Services; Cuyahoga Job and Family Services; or, the Division of Community Initiatives Services). In the case of the employees of the Office of the Director, displacement rights shall be limited to that office.

SECTION 3. Displacement, reinstatement, and reemployment procedures shall apply separately within each Division and/or the Office of the Director, as specified in Section 2 of this ordinance.

SECTION 4. This Ordinance specifically overrides any and all conflicting ordinances, as well as the provision set forth in OAC 123:1-41-12(E) that creates an additional displacement category for employees of county departments of job and family services defined as displacement in classifications previously held. Unless provided for otherwise in a collective bargaining agreement, in which case the terms

of the agreement shall apply, employees in the Department of Health and Human Services shall have only those displacement rights contained herein.

SECTION 5. Nothing in this Ordinance is intended to limit in any way the ability of the County Executive to hire employees, abolish positions, or lay off employees for the purpose of enhancing efficiency of operations or for any other reason permitted by general law.

SECTION 6. It is necessary that this Ordinance become immediately effective in order that critical services provided by Cuyahoga County can continue, and to continue the usual and daily operation of a County department. Provided that this Ordinance receives the affirmative vote of eight members of Council, this Ordinance shall become immediately effective upon the signature of the County Executive.

SECTION 7. It is found and determined that all formal actions of this Council relating to the adoption of this Ordinance were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

enacted.	_, seconded by	, the foregoing Ordinance was duly
Yeas:		
Nays:		
	County Council Preside	nt Date
	County Executive	 Date
	Clerk of Council	Date
First Reading/Referred t Committee(s) Assigned:		
Journal		

County Council of Cuyahoga County, Ohio

Ordinance No. O2012-0030

Sponsored by: County Executive	An Ordinance establishing t	the	
FitzGerald/Department of	organizational structure of the Departme	ent	
Development	of Development and creating divisions		
_	within the department, and declaring the		
	necessity that this Ordinance become		
	immediately effective.		

WHEREAS, Article VII, Section 7.02 of the Charter created the Department of Development, which states in pertinent part, "There shall be a Department of Development, under the direction of the Director of Development, which shall develop, direct and implement programs and activities for carrying out the purposes of this Article"; and

WHEREAS, Section 7.01 of the Charter, states that the County shall have as a primary responsibility the promotion and enhancement of the economic well-being and prosperity of the County and all of its residents; and

WHEREAS, in order to carry out the purpose of the Economic Development Department, it is necessary to create divisions within the Department of Development for the efficient administration of the County, and

WHEREAS, Article III, Section 3.09(2), of the Charter grants Council the power to establish departments, and divisions and sections within departments, under the supervision of the County Executive ... as the Council determines to be necessary for the efficient administration of the County; and

WHEREAS, Section 13.04 of the Charter provides for the transfer of powers and duties, as designated by ordinance, of departments, offices, and agencies existing at the time of the Charter's adoption; and

WHEREAS, the Council of Cuyahoga County has determined that establishment of divisions within the Department of Development is necessary for the efficient administration of the County; and

WHEREAS, The Department of Development has recommended the following divisions: (1) Division of Economic Development, (2) Division of Community Development, (3) Division of Regulatory Compliance & Monitoring, (4) Division of Fiscal Services, and (5) Division of Airport. The duties, functions and employees of the Department of Development shall be incorporated into these divisions.

WHEREAS, it is necessary that this Ordinance become immediately effective in order that critical services provide by Cuyahoga County can continue without interruption, and to continue the usual, daily operation of a County Department.

NOW, THEREFORE, BE IT ENACTED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. The Department of Development shall contain the following divisions: (1) Division of Economic Development, (2) Division of Community Development, (3) Division of Regulatory Compliance & Monitoring, (4) Division of Fiscal Services, and (5) Division of Airport. The duties, functions and employees of the Department of Development shall be incorporated into the divisions listed herein.

SECTION 2. Nothing in this Ordinance is intended to limit the ability of the County Executive and the Director of the Department of Development to abolish positions for the purpose of enhancing the efficiency of operations or for any other reason permitted by general law.

SECTION 3. It is necessary that this Ordinance become immediately effective in order that critical services provide by Cuyahoga County can continue without interruption, and to continue the usual, daily operation of a County Department. Provided that this Ordinance receives the affirmative vote of eight members of Council, this Ordinance shall become immediately effective upon the signature of the County Executive.

SECTION 4. It is found and determined that all formal actions of this Council relating to the adoption of this Ordinance were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion by was duly enacted.	, seconded by, the	foregoing Ordinance
Yeas:		
Nays:		
	County Council President	
	County Executive	 Date

Clerk of Council	Date

First Reading/Referred to Committee: October 9, 2012 Committee(s) Assigned: Economic Development & Planning

Committee Report/Second Reading: <u>January 8, 2013</u>

Journal CC009 January 22, 2013

County Council of Cuyahoga County, Ohio

Ordinance No. O2012-0035

Sponsored by: County Executive	An Ordinance amending Ordinance No.	
FitzGerald and Councilmembers	O2011-0027 dated 7/12/2011, which	
Germana and Conwell	established divisions and sections within the	
	Department of Public Works, to add the	
	operations of the Cuyahoga County Airport.	

WHEREAS, pursuant to Section 3.09(2) of the Charter Cuyahoga County, Ordinance No. O2011-0010 created the Department of Public Works as contemplated by Section 5.05 of the Charter; and,

WHEREAS, Cuyahoga County Ordinance No. O2011-0010 was subsequently amended by Ordinance No. O2011-0027, creating divisions and sections within the Department of Public Works; and

WHEREAS, it is necessary to amend Ordinance No. O2011-0027 in order to add the Cuyahoga County Airport under the Department of Public Works; and

WHEREAS, the County Airport is currently under the Department of Development, but the County has evaluated the efficiency of its operations and determined that it is in the best interests of the County to have the Cuyahoga County Airport under the Department of Public Works.

NOW, THEREFORE, BE IT ENACTED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. Ordinance No. O2011-0027 establishing divisions and sections within the Department of Public Works shall be amended to add and include the Cuyahoga County Airport as follows (language additions are boldfaced and underlined, deleted language is stricken out):

SECTION 1. The Department of Public Works is hereby established under the supervision of the County Executive and the Director of Public Works who, pursuant to Article V of the Charter, shall be appointed by the County Executive, subject to confirmation by Council, and shall serve at the pleasure of the County Executive.

SECTION 2. The duties and functions of the former Department of Central Services shall be incorporated into the Department of Public Works unless otherwise provided for in a future ordinance. The Department of Public Works shall contain the following divisions and sections: (a) Finance and Planning Division, containing a Finance and a Planning Section, (b)

Maintenance Division, (c) County Engineer Division, containing a Design Section and a Construction Section, and (d) Animal Shelter Division.

SECTION 3. The operations of the Cuyahoga County Airport, including, but not limited to, all functions, duties, responsibilities, employees and assets of the Airport are hereby transferred from the Department of Development to the Department of Public Works. The operations shall be allocated within the established divisions under the Department of Public Works as applicable.

SECTION 34. With the consent of the County Executive, the Director of Public Works is authorized to employ persons in the service of the Department within established budgetary parameters and in compliance with the Human Resources policies and procedures of the County. Nothing in this ordinance is intended to limit the ability of the County Executive and the Director of Department of Public Works to abolish positions for the purpose of enhancing the efficiency of operations or for any other reason permitted by general law.

SECTION 2. It is found and determined that all formal actions of this Council relating to the adoption of this Ordinance were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion byduly enacted.	, seconded by	, the forego	oing Ordinance was
Yeas:			
Nays:			
	County Council	President	Date
	County Executiv	ve	Date
	Clerk of Counci	 1	 Date

First Reading/Referred to Committee: <u>December 11, 2012</u>

Committee(s) Assigned: <u>Public Works, Procurement & Contracting</u>

Additional Sponsorship Requested: <u>January 3, 2013</u>

Committee Report/Second Reading: <u>January 8, 2013</u>

Journal CC009 January 22, 2013

[PROPOSED SUBSTITUTE ORDINANCE]

County Council of Cuyahoga County, Ohio

Ordinance No. O2012-0035

Sponsored by: County Executive	An Ordinance amending Ordinance No.		
FitzGerald and Councilmembers	O2011-0027 dated 7/12/2011, which		
Germana and Conwell	established divisions and sections within the		
	Department of Public Works, to add the		
	operations of the Cuyahoga County Airport;		
	and declaring the necessity that this		
	Ordinance become immediately effective.		

WHEREAS, pursuant to Section 3.09(2) of the Charter Cuyahoga County, Ordinance No. O2011-0010 created the Department of Public Works as contemplated by Section 5.05 of the Charter; and,

WHEREAS, Cuyahoga County Ordinance No. O2011-0010 was subsequently amended by Ordinance No. O2011-0027, creating divisions and sections within the Department of Public Works; and

WHEREAS, it is necessary to amend Ordinance No. O2011-0027 in order to add the Cuyahoga County Airport under the Department of Public Works; and

WHEREAS, the County Airport is currently under the Department of Development, but the County has evaluated the efficiency of its operations and determined that it is in the best interests of the County to have the Cuyahoga County Airport under the Department of Public Works.

WHEREAS, it is necessary that this Ordinance become immediately effective in order that critical services provide by Cuyahoga County can continue without interruption, and to continue the usual, daily operation of a County Department.

NOW, THEREFORE, BE IT ENACTED BY THE COUNTY COUNCIL OF CUYAHOGA COUNTY, OHIO:

SECTION 1. Ordinance No. O2011-0027 establishing divisions and sections within the Department of Public Works shall be amended to add and include the Cuyahoga County Airport as follows (language additions are boldfaced and underlined, deleted language is stricken out):

SECTION 1. The Department of Public Works is hereby established under the supervision of the County Executive and the Director of Public Works who, pursuant to Article V of the Charter, shall be appointed by the County Executive, subject to confirmation by Council, and shall serve at the pleasure of the County Executive.

SECTION 2. The duties and functions of the former Department of Central Services shall be incorporated into the Department of Public Works unless otherwise provided for in a future ordinance. The Department of Public Works shall contain the following divisions and sections: (a) Finance and Planning Division, containing a Finance and a Planning Section, (b) Maintenance Division, (c) County Engineer Division, containing a Design Section and a Construction Section, and (d) Animal Shelter Division.

SECTION 3. The operations of the Cuyahoga County Airport, including, but not limited to, all functions, duties, responsibilities, employees and assets of the Airport are hereby transferred from the Department of Development to the Department of Public Works. The operations shall be allocated within the established divisions under the Department of Public Works as applicable.

SECTION 34. With the consent of the County Executive, the Director of Public Works is authorized to employ persons in the service of the Department within established budgetary parameters and in compliance with the Human Resources policies and procedures of the County. Nothing in this ordinance is intended to limit the ability of the County Executive and the Director of Department of Public Works to abolish positions for the purpose of enhancing the efficiency of operations or for any other reason permitted by general law.

SECTION 2. It is necessary that this Ordinance become immediately effective in order that critical services provide by Cuyahoga County can continue without interruption, and to continue the usual, daily operation of a County Department. Provided that this Ordinance receives the affirmative vote of eight members of Council, this Ordinance shall become immediately effective upon the signature of the County Executive.

SECTION 23. It is found and determined that all formal actions of this Council relating to the adoption of this Ordinance were adopted in an open meeting of the Council, and that all deliberations of this Council and of any of its committees that resulted in such formal action were in meetings open to the public, in compliance with all legal requirements, including Section 121.22 of the Ohio Revised Code.

On a motion by duly enacted.	, seconded by	, the foregoing Ordinance was
Yeas:		
Nays:		

County Council President	Date
County Executive	Date
Clerk of Council	Date

First Reading/Referred to Committee: <u>December 11, 2012</u>

Committee(s) Assigned: Public Works, Procurement & Contracting

Additional Sponsorship Requested: January 3, 2013

Committee Report/Second Reading: <u>January 8, 2013</u>

Journal CC009 January 22, 2013